#### BEFORE THE METRO COUNCIL

)

) )

)

)

)

FOR THE PURPOSE OF ADOPTING THE 2004 REGIONAL TRANSPORTATION PLAN AS THE REGIONAL TRANSPORTATION SYSTEM PLAN AND THE REGIONAL FUNCTIONAL PLAN FOR TRANPORTATION TO MEET STATE PLANNING REQUIREMENTS ORDINANCE NO. 03-1024

Introduced by Rod Park

WHEREAS, federal law requires Metro to demonstrate every three years that its Regional Transportation Plan ("RTP") conforms to the Clean Air Act; and

WHEREAS, the U.S. Department of Transportation and the U.S. Environmental Protection Agency last found the RTP to conform to the requirements of the Clean Air Act on January 26, 2001; and

WHEREAS, cities and counties in the region have made amendments to their transportation system plans ("TSPs") in order to comply with Metro's 2000 RTP, and these TSP amendments have generated proposed amendments to the Regional Street Design and Freight System maps and minor revisions to other model system maps in the RTP; and

WHEREAS, cities and counties, in the course of amending their TSPs, identified new transportation projects and studies and changes in the location, description, cost or timing of previously approved projects; and

WHEREAS, Metro and cities and counties of the region have completed corridor studies, and concept plans pursuant to Title 11 of the Urban Growth Management Functional Plan, since adoption of the 2000 RTP, and these plans have generated proposed technical amendments to Chapter 6 (Implementation) of the RTP; and

WHEREAS, the Council directed that this update to the RTP be limited in scope to reflect changes in projects and programs since adoption of the 2000 RTP, in anticipation of a major review of RTP policies and projects in the next three-year cycle, due for completion by 2007; and

WHEREAS, the Metro Council has received and considered the advice of its Joint Policy Advisory Committee on Transportation and its Metro Policy Advisory Committee, and all proposed amendments identified in Exhibit "A" have been the subject of a 30-day public review period; and

WHEREAS, the Council held public hearings on the 2004 RTP on December 4 and December 11, 2003; now therefore,

#### THE METRO COUNCIL ORDAINS AS FOLLOWS:

- 1. Chapter 2 (Transportation) of the Regional Framework Plan ("RFP") and Chapter 1 (Regional Transportation Policy) of the RTP are hereby amended as set forth in Part 1 of Exhibit A, attached and incorporated into this ordinance.
- 2. Chapters 3 and 5 of the 2000 RTP are hereby amended as set forth in Part 2 (Project Update) of Exhibit A, attached and incorporated into this ordinance, to identify the scope and nature of the proposed transportation improvements that address the 20-year needs and a financial plan for implementing the recommended projects.
- 3. Chapter 6 (Implementation) of the RTP is hereby amended as set forth in Part 3 (Technical Update) of Exhibit A, attached and incorporated into this ordinance, to demonstrate regional compliance with state and federal planning requirements and establish regional TSP and functional plan requirements for city and county comprehensive plans and local TSPs.
- 4. Metro's 2000 RTP and these amendments to it, together with Titles 2 and 10 of the Urban Growth Management Functional Plan, comprise Metro's 2004 RTP, adopted as the regional functional plan for transportation under ORS 268.390, the regional "metropolitan transportation plan" required by federal transportation planning law, and the regional transportation system plan required by state planning law.
- 5. The Findings of Fact and Conclusions of Law in Exhibit B, attached and incorporated into this ordinance, explain how these amendments to the RTP conform to the requirements of the Clean Air Act and comply with state transportation and land use planning laws and the RFP.

ADOPTED by the Metro Council this \_\_ day of December, 2003.

Ę

David Bragdon, Council President

Attest:

Approved as to Form:

Christina Billington, Recording Secretary

Daniel B. Cooper, Metro Attorney



# Exhibit "A" to Ordinance No. 03-1024

Too Large to Copy Regional Transportation Plan may be found on the Metro Website or by contacting Kim Ellis at (503) 797-1617



PEOPLE PLACES OPEN SPACES





# 2004 Regional Transportation Plan **Policy Update**

# October 31, 2003



#### Metro People places • open spaces

Metro serves 1.3 million people who live in Clackamas, Multnomah and Washington counties and the 24 cities in the Portland metropolitan area. The regional government provides transportation and land-use planning services and oversees regional garbage disposal and recycling and waste reduction programs.

Metro manages regional parks and greenspaces and owns the Oregon Zoo. It also oversees operation of the Oregon Convention Center, the Portland Center for the Performing Arts and the Portland Metropolitan Exposition (Expo) Center, all managed by the Metropolitan Exposition Recreation Commission.

#### Your Metro representatives

Metro Council President – David Bragdon Metro Councilors – Rod Park, District 1; Brian Newman, District 2; Carl Hosticka, District 3; Susan McLain, District 4; Rex Burkholder, District 5; Rod Monroe, District 6. Auditor – Alexis Dow, CPA

Metro's web site: www.metro-region.org

Metro 600 NE Grand Ave. Portland, OR 97232-2736 (503) 797-1700 TDD (503) 797-1804

Printed on 100 percent recycled paper, 30 percent post-consumer fiber



# 2004 Regional Transportation Plan Policy Highlights

#### **Recent Policy Amendments**

Since the last update to the Regional Transportation Plan (RTP) in August 2000, a number of policy amendments have been adopted. These include:

- Oregon Land Conservation and Development Commission (LCDC) acknowledgement amendments (2001)
- TriMet's Elderly and Disabled Transit Study (2001)
- Regional Corridor Priorities project (2001)
- I-5 Partnership corridor study (2002)
- Metro's Green Streets project (2002)
- South Corridor Transit Study (2003).

These amendments to policies and policy maps have already been adopted by ordinance prior to this RTP update, and incorporated into the plan document.

#### **Proposed Policy Map Amendments**

The proposed policy amendments for the 2004 Regional Transportation Plan are limited to several transportation system map changes. No changes to policy text are proposed as part of this update.

This policy packet details a number of proposed amendments to the Regional Street Design and Regional Freight System maps that reflect the Oregon Transportation Commission's interest in creating "special transportation areas" where compact urban centers and main streets are planned along state-owned arterial streets. *These proposed map changes are shown in the table in Attachment 1.* 

The updated system maps also include a number of "housekeeping" amendments that reflect fine-tuning of the various model system maps, as recommended by local cities and counties through transportation plans adopted since the last RTP update in August 2000. *These changes are also summarized in Attachment 1.* 

Finally, a new map is proposed to be added to Chapter 1 of the RTP that identifies the Metropolitan Planning Organization (MPO) Planning Boundary. This boundary defines the area that the Regional Transportation Plan applies to for federal planning purposes. The boundary includes the area inside Metro's jurisdictional boundary, the 2003 urban growth boundary and the 2000 census defined urbanized area boundary for the Portland metropolitan region. This map is shown in Attachment 2 (note: a larger version of this map is available from Metro upon request).

# Attachment 1 Proposed Amendments to RTP System Maps

#### Figure 1.12 Motor Vehicle Functional Classification Map

Supele Nemes	Locadans	Guident Rite Seessinemen	Proposed RTP classified to to	Source of any source of any source of any source of a source of
Allen Boulevard	Hall Boulevard to Murray Boulevard	Collector of regional significance	Minor arterial	Beaverton TSP
Hart Road	Murray Boulevard to 170 <sup>th</sup> Avenue	Collector of regional significance	Minor arterial	Beaverton TSP
Murray Boulevard	Scholls Ferry Road to Barrows Road	Collector of regional significance	Minor arterial	Beaverton TSP
				A CARLES AND A
Sandy Boulevard	207 <sup>th</sup> Avenue to I-84	Collector of regional significance	Minor arterial	Fairview TSP
				<b>进行的时候,</b> 他们的时候
David Hill Road	Thatcher Road to Sunset Dr (Hwy 47)	No road	Planned minor arterial	Forest Grove TSP
'B' Street (Old Highway 47)	Hwy 47 to Pacific Avenue	Not classified	Minor arterial	Forest Grove
Sunset Drive	Main St. to Hwy 47/ NW Nehalem Highway	Not classified	Collector	Forest Grove TSP
Thatcher Road	David Hill Road to Gales Creek Road	Not classified	Minor arterial	Forest Grove TSP
	, et Maria e Cara e En 1999 - Cara e Car			
Riverside Drive Extension			Amend the dashed line to reflect alignment in TSP	Gresham TSP
Railroad Avenue	SE 37 <sup>th</sup> Avenue to Linwood Avenue	Not classified	Minor arterial	Milwaukie TSP
		and the difference of the second s		an a
Stark Street	Kane Road to UGB	Collector	Minor arterial	Multnomah County Functional Classification Study

#### Figure 1.12 Motor Vehicle Functional Classification Map (continued)

StreetName	Location 200	CurrenteRTP Classification	CPIODOSEOR PIER CALASSING ALCONT	Source of the change of the second se
SE Clatsop Extension	SE Mt. Scott	Future collector	Remove from	Portland TSP
	Boulevard to	of regional	the RTP motor	
	Deardon / 132nd	significance	venicle map or	
	Avenue		Willomette	
			National	
			Cemetery	
			boundaries	
SE Flavel Street / Mt.	SE 82 <sup>hd</sup> Avenue to	Minor arterial	Collector of	Portland TSP
Scott Boulevard	the city limits		regional	
			significance	
N Interstate Avenue	Fremont Bridge to N Denver Street	Major arterial	Minor arterial	Portland TSP
N Ivanhoe Street	N Philadelphia	Not classified	Minor arterial	Portland TSP
	Avenue to N		(should be	
	Lombard Street		Identined as the	
			Route)	
N Richmond Avenue	N Lombard Street to	Not classified	Minor arterial	Portland TSP
	N Ivanhoe Street		(should be	
			identified as the	
			US 30 Bypass	
			route)	
Water Avenue On-	Central Eastside	Principal	Delete from	Portland TSP
Ramp	Industrial District	arterial	Motor Vehicle	
		a and a second and a second and a second a se	System wap	
Boones Ferry Rd	SW Norwood Road to	Minor arterial	Major arterial	Tualatin TSP
	Nyberg Street		major artona.	
Lower Boones Ferry	Boones ferry Road to	Major arterial	Minor arterial	Tualatin TSP
Road	Bridgeport Street			
Martinazzi Avenue	Boones Ferry Road to Tualatin Sherwood	Not classified	Minor arterial	Tualatin TSP
Martinazzi Avenue	Tualatin Sherwood to	Not classified	Collector	Tualatin TSP
	Pinto Drive to			
	Vermillon Drive to	•		
	Driver to Boons Ferry			
	Road			
Nyberg Street	65 <sup>th</sup> Avenue to	Minor arterial	Major arterial	Tualatin TSP
	Tualatin-Sherwood			
	Road			
Tualatin Sherwood	Nyberg Street to	Minor arterial	Major arterial	Tualatin TSP
Road	Cipole Road			
			a second a second s	SRASS STATISTICS

.

.

#### Figure 1.12 Motor Vehicle Functional Classification Map (continued)

Street Name and the	Location	Current/RTP classification	Propose I RTP. Classification	Source of a set
Grant Street	Brookwood Parkway to 28th Avenue	No Designation	Collector of regional significance	Hillsboro TSP
A REPORT OF A R		A prist of the line of the	- A CONTRACTOR OF A	7. F
Beef Bend Road		Collector of regional significance	Minor arterial	Tigard TSP
Gaarde Street	:	Collector of regional significance	Minor arterial	Tigard TSP
Walnut Street	Gaarde Street to Scholls Ferry Road	Collector of regional significance	Minor arterial	Tigard TSP

#### Figure 1.4 Street Design Classification Map

Sugardane	Lecenter	Congnue 19	Promised RTPs	Source of
	같은 것이 있는 것이 같은 것이 있는 것이 있는 것이 있다. 같은 것이 있는 것이 같은 것이 있는 것이 있는 것이 같은 것이 있다.	noties the star	<b>Obsaulterion</b>	change and
Allen Boulevard	At Murray Boulevard	Possible	Delete	Beaverton
	intersection	boulevard	"Possible	Comprehensive
		intersection"	boulevard	Plan and
			intersection"	Development
<u> </u>			designation	Code
Hall Boulevard	Allen Boulevard to	Regional	Delete	Beaverton
1	Denney Road	boulevard	Regional	Comprehensive
			boulevard"	Plan and
			designation	Development
		-		Code
Murray Boulevard	At Farmington Road	"Possible	Delete	Beaverton
	intersection	boulevard	*Possible	Comprehensive
		intersection"	boulevard	Plan and
1			intersection"	Development
			designation	Code
				an a
McLoughlin Boulevard	Gloucester	Regional	Regional Street	Gladstone Town
(Highway 99E)	Avenuenue to	Boulevard		center moved to
	Arlington Street			Main Street
SE Railroad Avenue	SE 37 <sup>th</sup> Avenue to	Not classified	Community	Milwaukie TSP
	Linwood Avenue		Street	
				2010 Bill .
Broadway Bridge		Community	Regional Street	Portland TSP
		Boulevard		
E Burnside Street	108 <sup>th</sup> Avenue to	Regional	Regional Street	Portland TSP
	117 <sup>th</sup> Avenue	Boulevard		<u>.</u>
E Burnside Street	127 <sup>th</sup> Avenue to	Regional	Regional Street	Portland TSP
	143rd Avenue	Boulevard	-	

## Figure 1.4 Street Design Classification Map (continued)

StreetName	Location	Gurrent RTP	Proposed RIE	Source of 22
		selessification.	elassineation	change .
E Burnside Street	151 <sup>st</sup> Avenue to 162 <sup>nd</sup> Avenue	Regional Boulevard	Regional Street	Portland TSP
Burnside Bridge		Community Boulevard	Regional Boulevard	Portland TSP
SW Capitol Highway	SW Galeburn to SW Luradel	Community Street	Community Boulevard	Portland TSP
SW Capitol Highway	SW Brugger to SW Baird	Community Boulevard	Community Street	Portland TSP
SW Capitol Highway	SW Hume to SW Multnomah	Community Street	Community Boulevard	Portland TSP
SW Capitol Highway	SW 31 <sup>st</sup> to SW 33rd	Community Street	Community Boulevard	Portland TSP
SE Clatsop Extension	SE Mt. Scott Boulevard to Deardorf / 132nd	Future Community Corridor	Remove from the RTP street design map or realign south of Willamette National Cemetery boundaries	Portland TSP
NE Cully Boulevard	NE 57 <sup>th</sup> to NE Prescott Street	Community Street	Community Boulevard	Portland TSP
SE Division Street	SE 129 <sup>th</sup> to SE 130 <sup>th</sup>	Regional Street	Regional Boulevard	Portland TSP
SE Division Street	SE 117 <sup>th</sup> to SE 122nd	Regional Street	Regional Boulevard	Portland TSP
SE Division Street	SE 82 <sup>nd</sup> to SE 89 <sup>th</sup>	Regional Street	Community Boulevard	Portland TSP
SE Division Street	SE 75 <sup>th</sup> to SE 82 <sup>nd</sup>	Community Street	Community Boulevard	Portland TSP
SE Division Street	SE 33 <sup>rd</sup> to SE 50th	Community Street	Community Boulevard	Portland TSP
NE 82 <sup>nd</sup> Avenue	NE Sandy to NE Beech	Regional Street	Regional Boulevard	Portland TSP
NE 82 <sup>nd</sup> Avenue	NE Thompson to NE Halsey	Regional Street	Regional Boulevard	Portland TSP
SE 82 <sup>nd</sup> Avenue	SE Mill Street to SE Clinton Street	Regional Street	Regional Boulevard	Portland TSP
SE 82 <sup>hd</sup> Avenue	SE Raymond to SE Martins	Regional Street	Regional Boulevard	Portland TSP
Foster Road	SE 80 <sup>th</sup> to SE 82nd	Regional Street	Regional Boulevard	Portland TSP
Foster Road	SE Holgate to SE 75 <sup>th</sup>	Regional Street	Regional Boulevard	Portland TSP
Hawthorne Bridge		Regional Boulevard	Community Street	Portland TSP
St. Helens Road	NW Harbor through Linnton to north end of Kingsley park	Highway	Urban Road	Portland TSP

# Figure 1.4 Street Design Classification Map (continued)

Sile et Name a stal and	Location	CurrentRTP dessification	Proposal R P.	Sourceroft a
N Ivanhoe Street	N Richmond to N Philadelphia	Community Street	Community Street	Portland TSP and STA coordination meeting
NE Killingsworth Street	NE 35 <sup>th</sup> PL to NE 30 <sup>th</sup>	Community Street	Community Boulevard	Portland TSP
NE/N Killingsworth Street	NE MLK to N Interstate	Community Street	Community Boulevard	Portland TSP
N Killingsworth Street	N Interstate to N Greeley	Not Classified	Community Street	Portland TSP
N Lombard Street	N Woolsey to N Philadelphia	Community Street	Community Boulevard	Portland TSP
N Lombard Street	N Interstate to N Seward	Community Street	Community Boulevard	Portland TSP
N Lombard Street	At Philadelphia Street	Boulevard intersection	Delete	STA coordination meeting
N Lombard Street	At Ida Street	Boulevard intersection	Delete	STA coordination meeting
Macadam Avenue (Highway 43)	Bancroft to Taylor's Ferry Road	Regional Street	Regional Boulevard	STA coordination meeting
McLoughlin Boulevard	Grand/MLK Boulevard to SE Woodard (1 block north of Powell)	Highway	Regional Boulevard	Portland TSP
Mcloughlin Boulevard	SE 17 <sup>th</sup> Avenue to City Limits	Highway	Urban Road	Portland TSP
Morrison Bridge		Community Boulevard	Regional Street	Portland TSP
SW Multnomah Boulevard	SW 30 <sup>th</sup> Avenue to SW 35th Avenue	Community Street	Community Boulevard	Portland TSP
SE 92 <sup>™</sup> Avenue	SE Liebe to SE Harold Street	Regional Boulevard	Not classified	Portland TSP
SE 92 <sup>nd</sup> Avenue	SE Harold to SE Tolman Street	Regional Boulevard	Community Boulevard	Portland TSP
SE 92 <sup>no</sup> Avenue	SE Tolman to SE Duke	Community Street	Community Boulevard	Portland TSP
NE 122 <sup>nd</sup> Avenue	NE Multnomah to NE Oregon Street	Community Boulevard	Community Street	Portland TSP
SE 122 <sup>nd</sup> Avenue	SE Stark to SE Morrison Street	Community Street	Community Boulevard	Portland TSP
SE 122 <sup>nd</sup> Avenue	SE Clinton to SE Powell Boulevard	Community Street	Community Boulevard	Portland TSP
N Richmond	N Lombard to N Ivanhoe Street	Community Street	Community Boulevard	Portland TSP & STA coordination meeting
SE/NE Sandy	SE 12 <sup>th</sup> Avenue to	Community	Regional	Portland TSP

L Roulevard	L RAUIAVard	Roulovard	
			1

### Figure 1.4 Street Design Classification Map (continued)

Street Name	Location	Current RTP/	Proposed RUP	Source of
		classification	classification	change
			All and the second second	
NE Sandy Boulevard	NE 47 TO NE 82	Regional Street	Boulevard	Portiand ISP
NE Sandy Boulevard	NE 98 <sup>th</sup> to NE 122 <sup>nd</sup>	Community	Regional	Portland TSP
<u> </u>		Boulevard	Boulevard	
NE Sandy Boulevard	NE 122 <sup>nd</sup> to NE 163 <sup>rd</sup>	Urban Road	Regional Street	Portland TSP
Sellwood Bridge		Regional Street	Community Street	Portland TSP
SE 17 <sup>th</sup> Avenue	SE Nehalem to SE Tacoma	Unclassified	Community Boulevard	Portland TSP
SE 17 <sup>th</sup> Avenue	SE Tacoma to SE Andover	Community Street	Community Boulevard	Portland TSP
Steel Bridge		Regional Boulevard	Community Street	Portland TSP
NE/SE 39 <sup>th</sup> Avenue	NE Broadway to SE Powell	Community Street	Regional Street	Portland TSP
SE 39 <sup>th</sup> Avenue	SE Powell to SE	Unclassified	Community	Portland TSP
	Woodstock		Street	
	Woodstock		Street	
Macadam Avenuenue	Woodstock	Regional	Street Regional Street	STA
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination meeting; West
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination meeting; West Linn to focus
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination meeting; West Linn to focus boulevard
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination meeting; West Linn to focus boulevard improvements on interior town
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streats
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn	Regional Boulevard	Street Regional Street	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets
Macadam Avenuenue (Hwy 43)	Woodstock In West Linn Brookwood Parkway	Regional Boulevard	Street Regional Street	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets
Macadam Avenuenue (Hwy 43) Grant Street	Woodstock In West Linn Brockwood Parkway to 28th Avenue	Regional Boulevard	Street Regional Street Community boulevard	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets Hillsboro TSP
Macadam Avenuenue (Hwy 43) Grant Street	Woodstock In West Linn Brookwood Parkway to 28th Avenue	Regional Boulevard	Street Regional Street Community boulevard	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets Hillsboro TSP
Macadam Avenuenue (Hwy 43) Grant Street Beef Bend Road	Woodstock In West Linn Brookwood Parkway to 28th Avenue	Regional Boulevard No Designation	Street Regional Street Community boulevard Community street	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets Hillsboro TSP Tigard TSP
Macadam Avenuenue (Hwy 43) Grant Street Beef Bend Road Gaarde Street	Woodstock In West Linn Brookwood Parkway to 28th Avenue	Regional Boulevard No Designation No Designation	Street Regional Street Community boulevard Community street Community street	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets Hillsboro TSP Tigard TSP Tigard TSP
Macadam Avenuenue (Hwy 43) Grant Street Beef Bend Road Gaarde Street Walnut Street	Woodstock In West Linn Brookwood Parkway to 28th Avenue Gaarde Street to	Regional Boulevard No Designation No Designation No Designation	Street Regional Street Community boulevard Community street Community street Community	STA coordination meeting; West Linn to focus boulevard improvements on interior town center streets Hillsboro TSP Tigard TSP Tigard TSP

#### Figure 1.16 Regional Public Transportation System Map

StreetiName #*	Location Comments	Current RTR elassification	Proposed RTR. classification	Source of ser- change at
181 <sup>st</sup> Avenue	Gresham	Regional Bus	Frequent Bus	Gresham TSP
I-84 Corridor	Troutdale – Portland	Unclassified	Potential	Gresham TSP
		· ·	Commuter Rail	

#### Figure 1.17 Regional Freight System Map

Suaat Nema	Leenlen	Surrani STP BESEILIE IIION	27090540 3417 GESSING.(100)	Source of change seen
N Lombard Street	N St Louis to N Philadelphia	Road Connector	No designation	STA coordination meeting
McLoughlin Boulevard (Hwy 99E)	Hwy 224 to I-205 south ramps	Main roadway route	No designation	STA coordination meeting; Freight route provided by Highway 224 to I-205
N Ivanhoe Street	N St Louis to N Philadelphia	No designation	Road Connector	STA coordination meeting
N St Louis Street	N Lombard to N Ivanhoe	No designation	Road Connector	STA coordination meeting
Tualatin Valley Highway	Hwy 47 bypass to western Forest Grove city limits	Main roadway route	No designation	STA coordination meeting; Freight route provided by Highway 47 bypass

#### Figure 1.18 Regional Bicycle System Map

Street Name ************************************	erocationer in a series and a series of the	Current RTP Hassification	Processing Passifications	Source of Area change f
MAX Multi-Use Path	Gresham – Ruby Junction to Cleveland Avenue	None	Regional Corridor Off- street Bikeway	Gresham TSP
Tonquin Trail	Tualatin River to Willamette River	None	No change to classification; update off- street bikeway alignments to reflect regional greenspaces plan	Metro Parks and Greenspaces Master Plan
Lower Tualatin River Greenway Trail	Tualatin River to Willamette River	None	Same as above	Same as above
Washington Square Regional Center Trail	Washington Square	None	Same as above	Same as above
Oregon City Loop Trail	Willamette River to Clackamas River	None	Same as above	Same as above
Trolley Trail Connector	Springwater Trail to Trolley Trail in Milwaukie	None	Same as above	Same as above
East Buttes Power Line Corridor Trail	Springwater Trail to Clackamas River	None	Same as above	Same as above
East Buttes Loop Trail	Powell Butte to Gresham	None	Same as above	Same as above
Scouter Mountain Trail Extension	Scouter Mountain Trail to East Buttes Loop Trail	None	Same as above	Same as above

#### Figure 1.19 Regional Pedestrian System Map

SURGER MAINE	L DG: (10)	ourran)::गर्म ====================================	Proposed R 12 Passingelien	Source of changes
MAX Multi-Use Path	Gresham– Ruby Junction to Cleveland Avenue	None	Multi-use Facility	Gresham TSP
Tonquin Trail	Tualatin River to Willamette River	None	No change to classification; update off- street bikeway alignments to reflect regional greenspaces plan	Metro Parks and Greenspaces Master Plan
Lower Tualatin River Greenway Trail	Tualatin River to Willamette River	None	Same as above	Same as above
Washington Square Regional Center Trail	Washington Square	None	Same as above	Same as above

#### Figure 1.19 Regional Pedestrian System Map (continued)

SireetName and a		Current RTP "classification"	Broposed BIP classification	Source of Change
Oregon City Loop Trail	Willamette River to Clackamas River	None	Same as above	Same as above
Trolley Trail Connector	Springwater Trail to Trolley Trail in Milwaukie	None	Same as above	Same as above
East Buttes Power Line Corridor Trail	Springwater Trail to Clackamas River	None	Same as above	Same as above
East Buttes Loop Trail	Powell Butte to Gresham	None	Same as above	Same as above
Scouter Mountain Trail Extension	Scouter Mountain Trail to East Buttes Loop Trail	None	Same as above	Same as above
General	Region	None	Update pedestrian district boundaries to reflect updated 2040 center boundaries	Metro 2040 Growth Concept

#### Attachment 2



-----

# How to Comment on the update to the 2004 Regional Transportation Plan

The public comment period for the 2004 Regional Transportation Plan (RTP) begins on October 31, 2003 and concludes with a public hearing on December 4, 2003. You may submit comments online at Metro's website:

#### www.metro-region.org/rtp

Comments and questions may also be mailed using the form below, or left on Metro's Transportation hotline at (503) 797-1900, Option 2.

#### **Comments:**

	<u> </u>			
			·····	<b>_</b>
			<del>-`</del> -	
	······································			
<u>.                                    </u>	·······			
		<u>.</u>	<u></u>	

#### Submitted by:

Name			
Street Address	City/Zip		
Phone	E-Mail		
Send me more info:	·		
2000 RTP Document CD	Other RTP Info:		
Please add me to the RTP interested citizens mailing/e-mail lists			

## **Regional Transportation Plan Update Calendar**

- October 31 Public comment period begins; staff recommendation on draft 2004 RTP released for 30-day public comment period; draft RTP and conformity determination submitted to FHWA and FTA to begin review
- November 3 Air quality conformity analysis begins
- November 5 MTAC comments on draft 2004 RTP
- November 12 MPAC comments on draft 2004 RTP
- November 13 JPACT tentative action on draft 2004 RTP
- November 13 Metro Council first reading of Ordinance on draft 2004 RTP
- November 26 TPAC review and discussion of draft 2004 RTP and air quality conformity analysis
- **December 4** Public hearing on draft 2004 RTP; public comment period ends at 5 p.m.
- December 5 TPAC special meeting to comment on draft 2004 RTP
- December 10 Tentative final MPAC action on 2004 RTP
- December 11 Tentative final JPACT action on 2004 RTP
- **December 11** Metro Council second reading of Ordinance and consideration of adoption of 2004 Regional Transportation Plan

#### FOLD HERE



Place first class postage here.

Metro 600 NE Grand Avenue Portland, Oregon 97232 Attention: Marilyn Matteson





# 2004 Regional Transportation Plan **Project Update**

# October 31, 2003



PEOPLE PLACES OPEN SPACES

#### Metro People places • open spaces

Metro serves 1.3 million people who live in Clackamas, Multnomah and Washington counties and the 24 cities in the Portland metropolitan area. The regional government provides transportation and land-use planning services and oversees regional garbage disposal and recycling and waste reduction programs.

Metro manages regional parks and greenspaces and owns the Oregon Zoo. It also oversees operation of the Oregon Convention Center, the Portland Center for the Performing Arts and the Portland Metropolitan Exposition (Expo) Center, all managed by the Metropolitan Exposition Recreation Commission.

#### Your Metro representatives

Metro Council President – David Bragdon Metro Councilors – Rod Park, District 1; Brian Newman, District 2; Carl Hosticka, District 3; Susan McLain, District 4; Rex Burkholder, District 5; Rod Monroe, District 6. Auditor – Alexis Dow, CPA

Metro's web site: www.metro-region.org

Metro 600 NE Grand Ave. Portland, OR 97232-2736 (503) 797-1700 TDD (503) 797-1804

Printed on 100 percent recycled paper, 30 percent post-consumer fiber



# 2004 Regional Transportation Plan Project Highlights

### **Recent Project Amendments**

Since the last update to the Regional Transportation Plan (RTP) in August 2000, the Metro Council adopted a number of project amendments that stem from transportation corridor studies, including:

- the I-5 Partnership corridor study (2002)
- the South Corridor Transit Study (2003).

These amendments have already been adopted by ordinance prior to this RTP update, and are included in the published RTP project lists.

## **Proposed Project Amendments**

The proposed project changes in the draft 2004 RTP combine the "Preferred" and "Priority" systems contained in the 2000 RTP as a single Preferred system of projects needed to serve the region over the 20-year planning period, through 2025. This proposed \$9.9 billion preferred system establishes the universe of projects eligible for inclusion in the \$4.2 billion subset of "Financially Constrained" projects that are eligible for federal funding.

The Financially Constrained system is also the source of transportation projects that may be funded through the Metropolitan Transportation Improvement Program (MTIP) and Metro's Transportation Priorities process. The MTIP allocates federal funds in the region, and is updated every two years, and includes a rolling, four-year program of transportation improvements. The 2003 Regional Transportation Plan will provide an updated set of financially constrained projects and programs for future MTIP funding allocations.

Metro worked with local cities and counties to develop a comprehensive inventory of regional transportation projects identified in local plans and special studies adopted since the 2000 RTP was completed. This inventory includes:

- new projects or studies that are not currently in the 2000 Regional Transportation Plan, but that have been adopted in local transportation system plans (TSPs) and regional corridor studies through a public process
- updates to existing 2000 RTP projects or studies to reflect changes in project location, description, cost and recommended timing

Nearly all city and county transportation plans in the Metro region have been updated during the past three years to be consistent with the 2000 RTP. In the process of completing these updates, many local plans identified new transportation projects of regional significance that are proposed as part of the draft 2004 RTP as amendments.

Some corridor studies that have been completed (or are nearing completion) since the last RTP update in August 2000 have been endorsed by resolution with the expectation that the new projects generated by these studies would be incorporated into the current RTP update. This includes the Powell/Foster Corridor Study, Phase 1.

Finally, the Pleasant Valley Concept Plan, Powell Boulevard Streetscape Study and the McLoughlin Boulevard Enhancement Plan were completed in 2003 with the expectation that new projects generated by these local planning efforts would be incorporated into the 2004 RTP. The recommendations endorsed in each of these efforts are also reflected in the enclosed draft amendments.

### **How Projects Were Prioritized**

in October, Metro staff worked with members of the Transportation Policy Alternatives Committee and other interested parties to update the RTP project lists. In a series of four half-day workshops, this effort focused on incorporating all "housekeeping" amendments generated by local plans that have been adopted since the RTP was approved in August 2000. Since Metro commented separately on all of these local plans during their respective adoption activities, friendly amendments that were consistent with RTP policies, had already been identified for most projects.

The principal focus of the TPAC workshops was to define an updated Financially Constrained system of improvements. This exercise is a federal requirement, and defines a subset of roughly half of the Preferred system projects that are demonstrated to confirm to the federal Clean Air Act, and subsequently eligible for federal funds. The purpose of the exercise is to demonstrate that those projects most likely to be funded over the 20-year planning period will not result in a lapse in conforming to federal Clean Air Act standards for auto emissions.

Some notable differences in the 2004 RTP constraint exercise include a somewhat larger revenue projection for the constrained system through the new plan horizon year of 2025. Coupled with the fact that projects from the current plan have been built since it was adopt, this revenue increase results in a net gain in projects than can be included under the constraint ceiling. The expanded constrained revenue is largely the result of modest increases in local revenue sources devoted to regional transportation improvements, or revenues that reduce the backlog of maintenance obligations, which in turn expands the budget for capital projects.

There has also been an extensive discussion of factoring future Oregon Transportation Investment Act (OTIA) revenue into the forecast, but due to the limited timeframe for completing the RTP update, this assumption was not possible. Future OTIA revenues are expected to be incorporated into future state forecasts, and will be reflected in the next update to the RTP. However, the first three OTIAs are included in the forecast, and are part of the increased state revenue stream shown in the 2004 forecast amount.

The TPAC exercise followed the basic principles of (a) maintaining the Region 2040 Plan policy emphasis of the current RTP by focusing improvements in areas that serve as the economic engines for the region, including centers, ports and industrial areas, and (b) **2004 Regional Transportation Plan** Packet 2 – Project Amendments Page 2

maintaining a similar project balance among travel modes, including roads, transit, bikeways, pedestrian improvements and other project categories. Figure 1 is a summary of how the proposed 2004 RTP projects compare with the existing 2000 RTP according to these principles:

2040 Policy Emphasis (by number of projects)	2000 RTP	Draft 2004 RTP
Projects in Central City & Regional Centers	40%	60%
Projects in Industrial Areas and Ports	35%	17%
Projects in Town Centers & Main Streets	15%	17%
Projects in Other Areas	10%	7%
Balancing Modes of Transportation (by dollars)	2000 RTP	Draft 2004 RTP
Road & Bridge Projects	35%	46%
Bicycle & Pedestrian Projects	7%	9%
Transit Projects	55%	41%
Boulevard Projects	3%	4%

#### Figure 1 Distribution of Financially Constrained System Projects

The shift in projects from industrial areas and ports to the central city and regional centers is partly due to a number of changes to the proposed transit improvements in the constrained system. While number of major transit projects have been completed since the 2000 RTP was adopted, such as the Central City Streetcar, Interstate MAX and Airport MAX projects, the major rail improvements planned for the south corridor to Clackamas and extensions of the Central City Streetcar will increase the emphasis of major transit service on serving regional centers and the central city.

Though the share of dollars devoted to transit projects appears to decline, the actual amount is similar to the 2000 RTP, and the change is instead due to growth in the road revenues. As the lower part of Figure 1 shows, road revenues are expected to increase beyond the 2000 projections at both the local and state level, boosting the share of road and bridge projects, relative to transit projects. These most expensive road improvements are concentrated in major corridors and centers that are traditional hubs of the transportation system, thus adding to the increase in share of projects serving the central city and regional centers.

The slight increase in bicycle, pedestrian and boulevard projects shown in Figure 1 reflect a continued emphasis on many specific projects carried over from the 2000 RTP system, as well as new revenues for such projects proposed by ODOT and several local jurisdictions. While the percentage devoted to these projects is comparatively low, the cost of bicycle and pedestrian projects, in particular, tend to be modest since they can often be constructed without purchasing right-of-way.

Table 1 of this packet provides a more detailed summary of the proposed project changes to the RTP Financially Constrained System, as developed by Metro and TPAC members. Table 2 is a comprehensive list of RTP projects that includes all Financially Constrained and Preferred system improvements.

### Timing of the RTP Update

This RTP update comes at a critical turning point on a number of technical fronts. First, the current plan is due to lapse in late January 2004 under federal planning regulations, and must be updated in order to ensure the continued flow of federal funds for RTP projects. Second, the air quality analysis tool used in the region will soon be replaced with a new "Mobile 6" model that still requires testing to determine whether the current mix of RTP projects could conform to the Clean Air Act.

Compounding the transition to a new air quality tool is the fact that the Oregon Department of Environmental Quality (DEQ) is embarking on an update to their Air Quality Maintenance Plan, a governing document for RTP air quality assessments. This effort is expect to take as much as two years, counting federal approval of the updated air quality plan. During this period, it could be difficult to add or change projects in the RTP, which underscores the importance of including critical projects in this RTP update, and completing the update well in advance of the January 2004 lapse date.

# Table 1 Summary of 2004 RTP Financially Constrained System Project List Changes October 31, 2003

RTP #	Projects Added	Projects Dropped	Summary of Change	Est. Project Cost in 2003 dollars
1000		Interstate MAX LRT	Deleted (under construction)	
			Moved to Preferred System pending approval of LRT strategy in Clark County.	Washington State
1002		Vancouver Light Rail Loop	Wa.	Project
1008	I-5 South Corridor Study			\$ 1,732,500
1010	Morrison Bridge Deck Replacement			\$ 10,000,000
1012	Sellwood Bridge Replacement			\$ 90,000,000
1014		Central City Street Car	Deleted (Construction completed)	
1015	Central City Street Car - Phase 2a			\$ 15,350,000
<u>1016</u>	· · · · ·	Central City Street Car	Deleted (under construction)	
1021	·	Peninsula Crossing Trail	Deleted ( constructed)	
1024	I-5/McLoughlin Ramps			\$ _ 23,100,000
1025	I-5/North Macadam Access Improvements			\$ _ 20,000,000
1027	South Portland Improvements			\$ <u>_28,293,000</u>
1030	Ross Island Bridge Interchange			\$5,082,000
1033		Lovejoy Ramp Removal	Deleted (Construction completed)	
1034		Lower Albina RR Crossing	Deleted (Construction completed)	
1039	SE Belmont Ramp			\$ 1,732,500
1056		Lloyd District TMA Startup	Deleted (project completed)	
1057	Eastbank-Springwater Trail Connector (Three Bridges) Improvement			\$ 4,700,000
1058		SW Moody Bikeway	Deleted (Construction completed)	
1063		SE Morrison / Belmont Bikeway	Deleted (local level improvement)	
1064		N Interstate Bikeway	Deleted (under construction)	
1065		SE 17th Avenue Bikeway	Deleted (Included in project 1066)	
1066		SE Milwaukie Bikeway	Deleted (local level improvement)	
1069		East Burnside Bikeway	Deleted (kycal level (mprovement)	
1079		Steel Bridge Pedestrian Way (RATS Phase	Deleted (Construction completed)	、
1081		Eastbank Esplanade	Deleted (Construction completed)	
1082	SE Grand Avenue Bridgehead Improvements			\$ 1,600.000
1086	Central City Street Car - Phase 2b			\$ 20,000,000
1087	Central City Street Car - Phase 2c			\$ 12.000.000
1089	East Burnside/NE Couch Couplet and Street Improvements			\$ 7.500.000
1090	W Burnside/NW Couch Couplet and Street Improvements			\$ 7.500.000
1097	Nalto Parkway Street and Pedestrian Improvements			\$ 3,250,000
1098	Aerial Tram			\$ 15,000,000
1106	Eastside Streetcar - Phase 1			\$ 36,900,000
1107	Eastside Streetcar - Phase 2			\$ 44,000,000
1118	Sandy Boulevard Frequent Bus			\$ 1.760.000

# Table 1 Summary of 2004 RTP Financially Constrained System Project List Changes October 31, 2003

RTP #	Projects Added	Projects Dropped	Summary of Change	Est. I	Project Cost
1119	Sandy Boulevard/Burnside/12th Avenue Intersection			\$	4 620 000
1135	MLK/Lombard Frequent Bus			\$	2,100,000
_ 1138	Lombard/39th Frequent Bus			\$	2,700,000
1143	N / NE Lombard Bikeway			\$	1.155.000
1144		N Portland Road Bikeway	Deleted (Construction completed)		
1145		N St. Louis/Fessenden Bikeway	Deleted (Construction completed)		
1146		N Greeley/Interstate Bikeway	Deleted (Construction completed)		
1163	I-205 Ramps Construction			\$	12,000,000
1164	I-205 Ramp Study - PE/EA			\$	1,000,000
1165	I-205 Ramp Right-of-way Acquisition			\$	2,000,000
1177	SW Sunset Pedestrian and Bicycle Improvements			\$	1.386.000
1195		Barbur Boulevard Multi-modal	Moved to Preferred System	\$	15 000 000
1198	· · · · · · · · · · · · · · · · · · ·	SW Taylors Ferry Bikeway	Moved to Preferred System	s	2 079 000
1199	Barbur Boulevard Pedestrian Access to Transit Improvements			e	4 620 000
1207	· · · · · · · · · · · · · · · · · · ·	Barbur Boulevard ITS	Deleted (Construction completed)	<u> </u>	4,020,000
1209	NW 23rd Avenue Reconstruction				4 040 000
4040		NE/SE 122nd Avenue Bikeway		3	1,810,000
1213		Multnomah Pedestrian District	Deleted (under construction)		
1217			Deleted (Construction completed)		
1222		SE Milwaukie Pedestrian Improvements	Moved to Preferred System	\$	993,300
_ 1225	Lower Albina Area Improvements			\$	5,000,000
1226	Killingsworth Bridge Improvements	· · · · · · · · · · · · · · · · · · ·	·	\$	2,700,000
1229		Woodstock Mainstreet	Deleted (Construction completed)	├───	
1232	NW 23rd/Belmont Frequent Bus			\$	2, <u>490,</u> 000
1233	Hawthome Boulevard Frequent Bus			\$	2,460,000
1234	Lombard Street Improvements		· · · ·	\$	2,800,000
1235	Prescott Station Area Street Improvements			\$	3,400,000
1236	Improvements			\$	930,000
1237	Fessenden Frequent Bus Improvements		· · ·	\$	1,485,000
1252	Inner Powell Streetscape Plan			n/a	
1257		NE Russell Bikeway	Deleted (Construction completed)		
_ 1271	Linnton Community Bike and Pedestrian Improvements			\$	550,000
1277	NW Champlain Viaduct Reconstruction			\$	283,000
1278	SE 39th Avenue Reconstruction, Safety and Pedestrian Improvements			\$	2,200.000
1279	Holgate Street Improvements			\$	797.000
2000	Hogan Corridor Improvements			s	13,860.000
2001		Hogan Corridor Improvements	Moved to Preferred System	s	27.720.000
2010	Halsey/Weidler Boulevard and ITS			s	12,127,500

# Table 1Summary of 2004 RTP Financially Constrained SystemProject List Changes<br/>October 31, 2003

RTP #	Projects Added_	Projects Dropped	Summary of Change	Est. Project Cost in 2003 dollars
2013		NE Halsev Bikeway	Moved to Preferred System	\$ 1.420.000
2028	Powell Boulevard Improvements - East County			\$ 21,000,000
2029	242nd Avenue Reconstruction			\$ 2,400,000
2032	Bumside/Hogan Intersection Improvement			\$ <u>546,0</u> 00
2035	Cleveland Street Reconstruction	· · · · · · · · · · · · · · · · · · ·		\$ 1,732,500
2036	Wallula Street Reconstruction			\$ 1,732,500
2038	Walters Road Reconstruction			\$ 1,155,000
2039	Regner Road Reconstruction			\$ <u>14,200,0</u> 00
2042	257th Avenue Intersection Improvements	······································		\$ 4,899,510
2044	Orient Drive Improvements			\$ 4,158,000
2045	190th Avenue Improvements			\$ <u>1</u> 2,500,000
2051	Improvement			\$25,000,000
2055	SW Walters Road/Springwater Trail Access		····	\$346,500
2062		Gresham Regional Center TMA	Deleted (Project completed)	
2068		I-205 Ramps	Deleted (Construction completed)	
2069	I-205 Interchange Improvement			\$23,100,000
2070	I-205 Interchange Improvement			\$ 650,000
2074	Sandy Boulevard Widening			\$ 11,800,000
2076	181st Avenue Frequent bus		· · · · · · · · · · · · · · · · · · ·	\$ 1,350,000
<u>2077</u>	181st Avenue Widening	·		\$1,097,500
2079		185th Avenue Railroad Crossing	Deleted (Construction completed)	
2080	202nd Railroad Crossing Improvement			\$ 4,042,500
2086		NE 138th Avenue Improvements	Deleted (Construction completed)	
2087	· · · · · · · · · · · · · · · · · · ·	NE 158th Avenue Improvements	Deleted (Construction completed)	
2099	201st/202nd Avenue Corridor Improvements			\$ 9,909,900
2103	181st Avenue Improvements			\$3,326,400
2104	Bumside Road Boulevard Improvements			\$4,200,000
2109	Glisan Street Improvements			\$ 1,800,000
2110	MKC Collector	· · · · · · · · · · · · · · · · · · ·	,,,,,,	\$ 1,100,000
2111		207th Avenue Connector	Deleted (Construction completed)	
2115	Pairview-wood Village IC Pedestrian			\$ 1,386,000
2120	Sandy Boulevard Bicycle and Pedestnan			\$ 8,3 <u>1</u> 6,000
2124	Halsey Street Improvements - Troutdale	· · · · · · · · · · · · · · · · · · ·		\$ 3,742,200
2125	Troutdale TC Pedestrian Improvements			\$115,500
3004	US 217 EIS Study		· · · · · · · · · · · · · · · · · · ·	\$ 6,000,000
3005	US 26 Refinement and EA Study			\$ 577,500
3006	US 26 Improvements			\$25,410,000

# Table 1 Summary of 2004 RTP Financially Constrained System Project List Changes October 31, 2003

RTP #	Projects Added	Projects Dropped	Summary of Change	Est. Project Cost in 2003 dollars
		Us 26 Improvements	Deleted (Construction completed)	
3008	US 26 Improvements			\$ 37,600,000
3011	US 26 Improvements			\$ 12,300,000
3017	Beaverton Hillsdale Highway- Frequent Bus	· · · · · · · · · · · · · · · · · · ·		\$ 3,300,000
3021	2040 Centers and Station Areas Pedestrian System Infill			\$5,000,000
3022	2040 Centers and Station Areas Bicycle System Infill			\$ 5,000,000
3026		Millikan Extension	Deleted (Construction completed)	
3027		Davis Improvements	Deleted (Construction completed)	
3028	·····	Hart Improvements	Deleted (under construction)	
3035	Hocken Avenue Improvements			\$1,300,000
3039	Hocken Avenue Improvements	· · · · · · · · · · · · · · · · · · ·		\$2,000,000
3055	and Bicycle Improvements			\$ 12,127,500
3057	Improvements			\$ 24 <u>2,5</u> 50
3076	Allen Boulevard Improvements			\$ 1,155,000
3085		170th Improvement	Deleted (Construction completed)	
3096		Pedestrian Access to MAX	Deleted (included in Project #3021)	
3099	1st Avenue/Glencoe Road		· · · · · · · · · · · · · · · · · · ·	\$4,467,000
3108		Baseline Road Improvements	Deleted (Construction completed)	
3110	Transladia Mallari Martina da	Jackson School Road Improvements	Deleted (Construction completed)	
3118	Intersection Alignment	· · · · · · · · · · · · · · · · · · ·		\$ 10,000,000
3130		Evergreen Road Improvements	Deleted (Construction completed)	
3132		Comeilus Pass Road Improvements	Deleted (Construction completed)	
3136		Brookwood/Parkway Avenue Improvements	Deleted (Construction completed)	
3138		Murray LRT Overcrossing and Pedestrian	Deleted (Construction completed)	
	US 26 Overcrossing - Sunset IA			\$ 6,633,74 <u>3</u>
3149	Shute Road Interchange Improvements			\$6,382,000
3152	······	Westside TMA	Deleted (Project completed)	
<u>315</u> 3	David Hill Road Connector	·····		\$7,165,000
3154		Forest Grove Northern Arterial	Deleted (Construction completed)	
3159	Highway 8 Improvements - Forest Grove		· · · · · · · · · · · · · · · · · · ·	\$ 9,240,000
316 <u>2</u>		TV Highway (Pacific/19th) Blkeway	Deleted (included in Project #3159)	
3164	TV Highway Frequent Bus			\$ 1,575,000
3171	North Davis Street Reconstruction			\$ 1,600, <u>00</u> 0
3172	23rd/24th Avenue Extension			\$ 2,782,000
3175		Barnes Road Improvements	Moved to Preferred System	\$ 7,161,000
3182	Cornell Road Improvements - West Cedar Mill			\$ 6,930,000
3188	Saltzman Road Improvements			\$ 19,000,000

.

## Table 1 Summary of 2004 RTP Financially Constrained System Project List Changes

600,055	Ś			ssecon Pedestrian and Bicycle Access	4084
12,000,000	\$		· · · · · · · · · · · · · · · · · · ·	Ramsey Rall Complex	4082
		Deleted (Project completed)			1804
		Deleted (Project completed)			4080
300,000	\$	metsys benefered of bevom	Additional tracks - North Rivergate		6207
8'200'009	\$	moted to Preferred System	WHI Rait Yard	· · · · · · · · · · · · · · · · · · ·	8207
2,000,000	- \$	metay2 beneters of bevom	Penn Junction Realignment		1207
		Deleted (Included In Project #4073)		· · · · · · · · · · · · · · · · · · ·	\$20¥
53'100	\$		Rivergate Bicycle and Pedestrian Trail	И. Рогсе/Вгоздасте/Victory Вікемау	7207
009'261'9	\$	Moved to Preferred System	Bames Yard Expansion		L/07
000'009'71					0/05
000,000,6					6005
000 000 2		metav2 benefar of bevold			6908
000 000 21	•	metry2 benefet 9 to beyow	Rivergate Rail excession		8907
		Deleted (Construction completed)	r eserte chonovorani evino eninem		2907
000.018.78	\$ _	Moved to Preferred System	West Hayden Island Bridge and Acces Road	· · · · · · · · · · · · · · · · · · ·	1907
14'000'000	\$			Lightrall station/track realignment	0907
550,000	\$			Airtens/Comfoot Rd Intersection	9907
		Deleted (Construction completed)			4047
000'061	\$			Airport Way'1 22nd Avenue Improvements	4046
000'051'1	\$			Columbis/S2nd Avenue Improvements	4044
002,808	\$	Moved to Preferred System			1032
000,328,11	\$		Columbia and Lombaration	STI XO9	4059
		Deleted (Construction completed)	Airport Way/Cascades grade separation		2202
009'ZEL'L	\$			Cascades Parkway Connection	4056
		Deleted (Construction completed)	Cascades Parkway	· · · · · · · · · · · · · · · · · · ·	4052
					ļ .
	-	Deleted (Construction completed)	noisnetx3 bsoЯ boowreblA	······································	4024
363,825	\$	metays benefered to beyom	Marx Drive Extension		4053
		Deleted (Construction completed)	Airport Way Widening, East		4020
14,000,000	\$	Moved to Preferred System	Lightrail station/track realignment	· · · · · · · · · · · · · · · · · · ·	6104
15,000,000	\$			2130 1 1911 bir Study and Tler 1 DEIS	6007
000,000,15	\$			Sauvie Island Bridge Replacement	2007
000'000'99	\$				900*
000'0#5'#	\$				100+
	•	(pereiduos ແດແລກັນຣັບດຣ) ກອງອາອາອາດີ			000
		Diologia uno lo racio vi comminimo no potoci	A Proprieta Sector Annual Sect		0007
		(our on wold this period of aub bateled)	Cedar Mill Multi-Use Path		7615
4 1911 6 1		Deleted (Included in Proiect #3183)	Cornell Road Boulevard Improvement		5915
. Project Cost 2003 dollars	.}e∃ ∶ni	epnedO to Vismmu2	Projects Dropped	Projects Added	# 97.Я

7 to 2 eged

# Table 1 Summary of 2004 RTP Financially Constrained System Project List Changes October 31, 2003

RTP #	Projects Added	Projects Dropped	Summary of Change	Est. Project Cost in 2003 dollars
4085	Terminal area Bicycle and Pedestrian Improvements			\$ 750,000
4086	PIC Bike and Pedestrian Improvements			\$ 240,000
4087	Leadbetter Street Extension and Grade Separation			\$ 8,000,000
4088	Terminal 4 Driveway Consolidation			\$ 1,000,000
5013	I-205 Climbing Lanes			\$ 46,200,000
5018		Highway 213 Intersection Improvements	Deleted (Construction completed)	40,200,000
5020	Highway 213 Improvements			\$ 17 335 000
5022		Highway 213 Widening		\$ 17,323,000
5038		Johnson Creek Boulevard, Phase 2	Deleted (Construction to be completed in 20	
5041	37th Avenue Bike/Ped Improvement	Pointeri ereen bouleraid, rindse z	Deleted (Construction to be completed in 20	
5046		Railmad Crossing (monyements		\$ 410,000
5050		Namoad Crossing improvements	Deleted (Construction completed)	
5050	· · · · · · · · · · · · · · · · · · ·	Hamson Street Bikeway	Moved to Preferred System	\$560,000
<u>505</u> 1		Lake Road Bikeway	Deleted (Included In Project #5037)	
5065		Clackamas Regional Center TMA Startup	Deleted (TMA has been formed)	
5070	Otty Road Improvements	· · · · · · · · · · · · · · · · · · ·		\$1,848,000
5076	Fuller Road Improvements		· · · · · · · · · · · · · · · · · · ·	\$2,600,000
5087	West Sunnybrook Road Extension			\$2,310,000
5098	King Road Frequent Bus		·	\$1,236,000
5099	Webster Road Frequent Bus		·	\$1,510,000
<u>510</u> 8		Jennifer Street/135th Avenue Extension	Deleted (Construction completed)	<u>\$</u>
5126	South Amtrak Station Phase 2			\$ 1,500,000
5130		99E/2nd Avenue Realignment	Deleted (Construction completed)	
5142	Mollala Avenue Frequent Bus			\$ 1,085,000
<u>51</u> 52	Willamette River Shared-Use Path			\$ 500,000
5157	Mollala Avenue Streetscape Improvements			\$ 15.000.000
5163		*A* Avenue Reconstruction		
5171	Transit Station Relocation		Deneties (Construction completed)	¢ ( 400 000
5105		Highway 43 Improvements	Delated (Balantia ha executive data and b	\$ 4,190,000
6100		righway 43 improvements	Deleted (Project to be completed through Pro	oject #5196)
- 0188 - 6011	Highway 217 Overseasing Concerds Place			\$8,000,000
0011	righway 217 Overcrossing - Cascade Plaza			<u>\$</u> 26,000,000
6014	·		Deleted (Construction completed)	
6020	· · · · · · · · · · · · · · · · · · ·	Powenine Trail Comdor	Deleted (Project included in #3014 and #307	2)
6027		I-5/21/ Interchange Phase 2	Moved to Preferred System	\$ 45,045,000
6029	Hall/Kruse Frequent Bus	Walnut Street Improvements, Phase 1		\$ 275,000
6033			Deleted (Construction completed)	
6035	Gaarde Street Improvements			\$ 4,620,000

# Table 1 Summary of 2004 RTP Financially Constrained System Project List Changes October 31, 2003

				Est. Project Cost
<u>  RTP #</u>	Projects Added	Projects Dropped Walnut Street Improvements, Phase 2	Summary of Change	In 2003 dollars
6046	Washington Soure Regional Center		Deleted (Construction completed)	
6057	Greenbelt Shared Use Path			\$ 2,000,000
6059		Beef Bend Road Improvements	Deleted (Construction completed)	
6064	Hall Boulevard Frequent Bus			\$ 7,700,000
6065	Herman Road Improvements			\$ 12,000,000
6072		Tualatin Road Improvements	Deleted (Construction completed)	· · · · · · · · · · · · · · · · · · ·
6076	Myslony/112th Connection			\$ 1,500,000
6086	Kinsman Road Extension			\$ 7,620,000
6088	Elligsen Road Improvements			\$ 1,750,000
6111	· · · · · · · · · · · · · · · · · · ·	Beef Bend/Elsner Road Improvements	Deleted (Construction completed)	
6113		Oregon Street Improvements	Deleted (Construction completed)	
6119	Teal Boulevard Extension			\$ _4,000,000
6125		Bangy Road Improvements	Deleted (Construction completed)	
6128		Carmen Drive Intersection Improvements	Deleted (Construction completed)	
6138	Wilsonville Road/I-5 Interchange Improvements (Phase 1 and 2)			\$ 20,900,000
6141	I-5/99W Connector: Phase 1 Arterial			\$ 53,000,000
6142	Upper Boones Ferry Road Improvement			\$ 1,000,000
7008		147th Avenue Improvements	Deleted (under construction)	
7022	Sunovside Road Frequent hus			\$ 913,000
7034	Easter Road Extension	······································		\$ 1 700 000
7035	Glese Road Extension	•		\$ 2,900,000
7000				\$ 4 100 000
7030				4,100,000
7037	172nd Avenue Improvements		·····	\$1,900,000
7038	172nd Avenue Improvements			\$5,600,000
7039	Giese Road Improvements		· · · · · · · · · · · · · · · · · · ·	\$ 4,300,000
7040	Glese Road Improvements			\$ 3,000,000
7041	Foster Road bridge			<u>\$ 1,100,000</u>
7042	Giese Road Extension bridge		· · · · · · · · · · · · · · · · · · ·	\$ 1,100,000
7043	Butter Road Bridge			\$ 1,700,000
8007	Preservation/Maintenance Projects			\$ _10,000,000
8049	Priority Pedestrian Access to Transit Improvements			\$ 20,000,000
8050	SMART TDM Program			\$ 1,500,000
8057	LIFT Vehicle Purchases			\$ 16,890,000
8058	Ride Connection Vehicle Purchases			\$ 4,767,600

.

### **Public Comment Draft** 2004 RTP Project List October 31, 2003

						2025 RTP Preferred	2025 RTP Financially Constrained	2003 dollars ( *** indicates phasing in financially	RTP Program
RTP#	2040 Link Deleted (under con	Jurisdiction struction)	Project Name (Facility)	Project Location	Project Description	System	System	constrained	Years
11001				Current Market Contract		S. S.		\$-, 475,000,000	2004409
1002	Region	CTRAN	Vancouver Light Reil Loop	Expo Center to Vancouver, Washington	Construct LRT	x		Washington State Project	2016-25
1003		THE TIME	Milwanger moralt=\line		Constant	the t	- ·x	\$. 515,000,000	2010-15
1004	Region	ODOT	I-5 South Improvements	1-5 south of central city/1-405 to Charbonneau	Implement safety and modernization Improvements recommended by studies in Projects 1008 and 1098	x		\$ 57,750,000	2016-25
1005	Region	Multnomsh Co.	Rehabilitation of Wilamette River Bridges	Broadway, Burnside, Morrison, Sauvie Island Bridges	Provide for long-term rehabilitation and structural needs of bridges	x		\$93,334,395	2004-25
1006	Region	Multhomah Co.	(Painting)	Sumakte, Morrison, Sauvie Island Bridges	Provide for long-term painting preservation needs of bridges	x		\$ 37,338,840	2004-25
1007		allena nut Sa	Lite own of a string of the	Georgen Stellander Stellen	De las vientes pola se la transmite de vier adecardos na artículas entres caracter replacarte de constructiones de la constru- parecente de constructiones de la construction provincial construction de la construction de la constructiones de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la constructi				543954 543954 543954
1008 111	e a contra da entra da entra da entra da entra da entra entra entra entra entra entra entra entra entra entra Entra entra	seren norden. Se Classification	L'ASTRE TRANSPORT	ergines // in Wilson Deconformation	toos endtoinin rave in company and a second state of the second st		<u>x</u> -9		ineis⊇7⊂) 2
	r F		is (nationalis) and particular and an an	SETALCOCIONEL SEAS			×.	5 5. 123101000 14	
SERVIO	and the second second	20 Million and Cos	AND INCOMPANY AND	Transit center and park-and-ride upgrades	stepace decidation consideration of the set	· · · · ·	2002 - X.	\$	1200-202
1011	Region	TriMet	Transit center and park-and-ride upgrades	throughout subarea	Transit center and park-and-ride upgrades	X		see Tri-Met total	2004-25
1012	The Regional		MCRAP Subury Physics Project			×	X.	\$ 90,000,000	2004-05
1013	Region	Multnomah Co.	Implementation	Sellwood Bridge	Eastside Undercrossing; Light Pole Relocation	x		\$ 635,250	2016-25
1014	Deleted (Construct	ion completed)	Se Barrert an all folged states toget and in the sign	an street a province of the street of the Contractor and the street of t	Na 42 Merril Mark 1 (1991) 11. 1993) 11. 11. 1994 (1991)				
71015	Deleted (under con	struction)			Consucciatives cars		× * *	\$ 4 06,350,000	200.00.
1017	Region	ODOT/Metro	Macadam/Highway 43 Transit/TDM Study	Portland central city to Lake Oswego	Study to define additional transil and demand management improvements in corridor	x		\$ 1,155,000	2004-09
1018	Region	Portland	Willamette Greenway Trail extension	St. Johns Bridge to Pier Park and connect to Smith and Bybee Lakes and to Kelly Point Park	Study feasibility of shared-use path			n/a	2016-25
1019	Central City	TriMet	Barbur Bouleverd Rapid Bus	PC80 to King City Without Budgets States and	Construct improvements that enhance Repid Bus service	x		see Tri-Met total	2004-09
1021	Deleted ( construct	ed)				<u> </u>		S	20044092
1022	Region	Portland	I-84/Banfield Trail	Willamette River/Eastbank Esplanade to I-205 bike lanee	Study feasibility of shared-use path	X		n/a	2016-25
1023	Region	ODOT/Metro	Banfield (I-84) Transit/TSM Study	I-205 to Portland central city	Study to define additional transit and system management improvements in corridor	x		\$1,155,000	2010-15
1024	Comma Cay	CODEUS	EdMid Odhini Renge	Anessinallitie arane in stations in the second state	Densition new (-SB) off-ramp and I/o NE off-ramp at	. *X*	X	se	2018-25
1025	2.Contral City	ODOT	15/North Macadam Accession Investmental		Construct Instantian and a second second second	X.	<b>X</b>	\$ 20,000,000	2016-25
1026	Deleted (alternative	improvements prov	(ded)			<u>ನಿವೇಶದ ಕಾರ್ಯಗಳು ಎಂದು ಬಿನ</u>			
		and a second			recomplet each was the neighborhood collector and of recomplet each was focul stream. Rebuild Ross Island Ridge Romps to separate regional traffic from 2 regional traffic to a constrained regional traffic from 2 regional stream and improve accession 1-405 and to				
1027	Control City	Portand/ODOF	Source Portland Improvements	Soun Portanti Starates :	S Part of the State	X .	🤆 X	\$ 28,293,000	2010-15-

# Public Comment Draft

2004 RTP Project List

RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Professed System	2025 RTP Financially Constrained System	2003 dollars ( *** Indicates phasing in financially constrained	RTP Program Years
		U.S	สาราชรรษ สมาริการสาร	an a	Improve Stanterey Stead Interesting Stread Interesting			C. 10	
							in the second		
-H(1029							1. <b>.</b>	5) 288,760.	2004-09
1031	Central City	ODOT	1-405/US 26 Connector	Ross Island Bridge to 1-405 to US 26	Construct new freeway access	<u>ү</u>		\$ 67 750 000	2018.25
					ปกฎษณะได้ (เมื่อมีการสายสายๆและประการสายสา		West 2	2 31,750,000	2010-23
1032	Senrel Chile	inta <u>i la</u>	in novements	lan suprivillament schere set service danses		O X	n 7 ix 1	\$2.887.500	2010/25
1033	Deleted (Construct	on completed)	<u> </u>						
1034	Deleted (Construct	on completed)	en ander som Rent ander an	A STATE AND A STATE AN				the second second	
1036	Contraction		The many state of the second		Incon-Concellancent	and Antonia Sheet Xarrow	X	\$	2004004
1037	- chatty,		Report Scholler and Grant Compare	Brits and wards in the full of the new of the			主教 "	\$ 4042,500	101200305
1038	Central City	Portland	SE 11th/12th Rall Crossing	Western edge of SE Division Street		x		\$ 98,175	2016-25
1039	Contra chy	المنابعة المنافعة		HEROMANNE AND THE BURNESS BURNESS BURNESS	IN COMPACT A STATE OF A STAT	х Х	X.	\$ 1,732,500	- 2010-15
		<b>-</b>			Geometric, signalization and channelization improvements to allow transit and general traffic access				
1040	Central City	Portland	SE Clay/MLK Intersection Improvementa	SE Clay and MLK	to westbound Clay street from southbound MLK	X		\$323,400	2016-25
1041	Central City	Portand	Interstate Avenue Selemic Reform	Interstate Avenue bridge at Lamabe Avenue		×		\$1,455,300	2016-25
1042	Central City	Portland	NE 12th Avenue Seismic Recom	NE 12th Avenue/Lloyd Boulevard	Seismic revoin project Major bridge maintenance, including painting,	X		\$415,800	2016-25
1043	Central City	Portland	Steel Bridge Rehabilitation	Steel Bridge	mechanical maintenance and structural improvements	X		\$ 30,000,000	2004-09
1044	Central City	Portland	Retrofit	Kittridge Street bridge at Yeon Avenue	Selsmic retrofit project	x		\$ 623,700	2016-25
1045	Central City	Portland	Steel Bridge East Ramps	Selamic retrofit project		<u> </u>	and the second second second second	\$ 831,600	2016-25
1048				<ul> <li>Scotling (1997) And the second se Second second se Second second se</li></ul>	Reality and an an arrival and a second	a X	X	\$3 2852850	ં સામ્યે
-1047			tt chills Schenstein Linger	Composition and States	And the Manual States of the second states of the second			5 577500	
			nt - H. Ale 2048 www.ale 2049 and 2019		hopeon as a filling and also here as the second			Ren 🕈	
					าะสถาสพรณะ คาะการสะโหญ่เหตุ จากสัตร์ แต่สาวสตร์จาก สามารถหนายาก ราการสาวสุดที่มีสุราวิทยาก การสาวสตร์จาก				
11048			Sara Selution and the second		HALL MANAGEMENT REPORT OF A STRATE AND A STRATE	and the same	i XX Ο	SE CALL SOIL	
			Section and the section of the secti		A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACT OF				
1213					For a construction of the second s				
1050	Calification -		Noine Sale in Jun 2022	Stein Altheorie (Stiffing edget e annelle hy	Rechtschult all mit führe sholore Asies	arex	317 p.,	Site 200000	
240	a de la caración de l				ីនេះនៅស្ថាននៃនៅភ្លើកហើសចំណើរ សារាជនៅតែមានសំពេងស្ថិន និង ស្ថិននាយនៃសារាជាស្ថិនដែលសំនើនដែរ សារាស់ស្ថារជាអំពីស្ថានាណ៍ អំពី				
1051	Central City	E. Ponto Con	UV - LUIS do Stor A linphove in sets	W HULL WYZDU	crossings; namesigneds to W2005Pietra W22nd, and wither the negotient to limit moderate delayers as a second		X	\$*0000000	
	1.	1			Inclement according overheads Mentiod Inche Southers Wetermonit terretwork Plans to bong periods Band A			A . M. 16	
3069	Central City	Poten	Note: Macadem Street Inscove rolling	Sedia vytera na slove or out slove zav	Convertever Parkway, Plants Connector, Key access Interactions and other ensat volvovements		x	s 20.501.250	
	10 2 2 3 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4				Complete Solid and the standard standard bike				
1053	Central City	Portlered SC	in new service improvements	NW COMPLEXIVITY AND SAME THE COMPLEXION	laries, penesinan crossings and pavement reconstruction	X	X	5 7.400,000	2004-09

## **Public Comment Draft** 2004 RTP Project List October 31, 2003

DTD #		Installation	Dmiest Name /Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	(*** indicates phasing in financially constrained	RTP Program Years
	ZUHU LINK		Sectors and sector on the sector sector is a sector of the	A State of the second state of the second	eres and the second state of the second second state of the second second second second second second second s		Strate State		Sugar Carl
1055			and a state of the second s	Constant Const	Consider the distance residual and report to the		tink t	5 3,455,000	2016/25
1058	Deleted (project co	mpteted)							
38					4 Optimization of the state				
			Francial Spin grades to historicates	SI HWARDING AN SINAL	<ul> <li>Million to a publication part of the total data of public boundary age and compared states.</li> </ul>	16 X 1	The Art	5 .24.700.000	2002:055
1058	Deleted (Construct	ion completed)							
1059	Deleted (alternative	route provided)							
1060	Deleted (local level	Improvement)			· · · · · · · · · · · · · · · · · · ·	<u> </u>	·		
1061	Deleted (local level	improvement)	e i sudde formingen geren skine i i i i i			1000	A		
1082	Centra City 41	en der einen som s	Antibar Profit Faile, Profit Alaptan, al	iscalutEuplo Cost		5767 X	2 X	\$ 1,468,050	1300 × 20
1063	Deleted (local level	Improvement)			· · · · · · · · · · · · · · · · · · ·				+
1064	Deleted (under con	struction)		· · · · · · · · · · · · · · · · · · ·					+
1065	Deleted (Included I	n project 1086)							
1066	Deleted (local level	Improvement}			· · · · · · · · · · · · · · · · · · ·				┟╌┥───┥
1067	Central City	TODO	SE McLoughlin Boulevard Bikeway	SE 17th Avenue to SE Clatsop Street	Retroft bike lanes to existing street	X		\$ 577,500	2016-25
1088		and a protocol and a set		ST WE ANALL SCHEMENTED STOLEN.		A CARGE	- X I.	19,635	2010-251
1069	Deleted (local level	improvement)	·						┢╼┽╸────┩
1074	Deleted (Construct	ion completed)			<u> </u>				
1075	1075 Deleted (Construction completed)							<u> </u>	┼-┼───┦
1076	Deleted (Included I	n project 1027)	West Burnakle Pedestrian and Bicycle		Retrofit bikeway to existing street, improve sidewalks,				┟┼╌──┘
107B	Central City	Portland	Improvemente	Tichner to Skyline	lighting and crossings		ļ	\$ 317,625	2016-25
1079	Deleted (Construct	ion completed}	ne unternet andret debekraut des sites andre		Manuta industria in the state of the state o	sker i strange	5-5-5-5		
1080	les i sources du	a start ti de		Chick Company and Chick Street	holenopolis vite biel bedan a contact a second		Alta X 内容	5 868,250	CT 200-200
1081	Deleted (Construct	fon completed)					in the second second second	anna wara sumati tayat	
1	N.4.4 <sup>N</sup> 9		"是你,你说,你说我们你的是你,你就是你。" 我们,我们们们我们的我们的我们的?" 我们就能能能能能能能能能能能能。		ให้แล้มกับที่มีสร้างสร้างการทำให้ สมกับที่สามันกับไม่ไม่มีที่มี #1 ให้พร้างสามสาม และกลังการการการที่มีสมกับที่ไม่มีสามาณีที่ #1				
1082	THE CONTRACTOR	a and the second second	In novie hit way to planacion	stan. Easteinin an Rhiss.	Volume sources and the second personal procession of the second person o	X	X	1,600,000	2004-09
1083	Central City	Portland	Improvementa	SE Powell Boulevard at Milwaukie Avenue	on the east leg of the intersection.	x		\$ 288,750	2004-09
2/1084	Central City	· · · Portland	Cievizind Pedectian Vanicle Skinal	OVE LIEV STORE BIRD SWY THE AVERUSE SWY AVER	and we all a leader interest and a second	X • X • • •		a 23175,500	C
1085	Deleted (Included i	n project 1119)							
1085	Connectives	the ImMetric dated	Constant State Constant			5. X 44	x 2.	20,000,000	200109
1087	· Cantral City	Trimet/Portanda	Contracting Subtream Production and	EIST STORADIELINGO STORADER, STORADER	Control and Carl Control Contr	¥X	X	5 12,000,000	2004-09
1088	Deleted (Study cor	npleted}	an a						1977 105 Milian
ŤČ89	Control City	Portfand -	Envernment DNE course courses and a Street Increasements	Energy 2015 Avoince on Burnelds Bridge	Rinals Maria Silevaka chi extension Die lares, In Sinei pa king and steel fields (* 1997).	×	×	5 7 500,000	12010 IS
.

2004 RTP Project List

								2003 dollars	[	
						2025 RTP	2025 RTP Financially	obasing in		RTP
			1			Preferred	Constrained	financially	P	rogram
RTP#	2040 Unk	Jurisdiction	Project Name (Facility)	Project Location	Project Description	System	System	constrained		Yeara
				한 모양 한 것 같은 것	ស្រុកក្នុងនេះ សម្តែនសម្តាំងក្លុងសម្តាំងសម្តែងស្រុកក្តីដែរ សម្តាំងដែរ			21 17 ST.		
			ala selene al della serie della predicta della statica della della della della della della della della della de	anningandersaliti Alexander (1. 19	orginen, winderen undevenderbeitenderbeitenen oder einer,		1	5 7 500,000	- 13	
		October 1	Orachart Franklide Tarrak Assess Obuda	Control Control of Industrial Olderial	Complete truck occore study			-1.		
1091	Central City	Portand	Central Eastikes Truck Access Study		Complete buck access study	X		nva		2016-25
1092	Central City	Portland	NW 14th/16th Study	Burnside to Vaughn	Signalization and improved access to I-405	<u>x</u>		n/a	2	2016-25
1093	Central City	Portland	Study	Central City	Study pedestrian enhancemente	x		n/a	2	2004-09
	0	Portland	SE Sandy Boyleword Shudy	Stark Street to Burnelde	Realize blocks to improve circutation in the area	· ·		n/a		2016 25
1094	Central City		Se sandy coulevare cluby			<u> </u>		Tira I		2010-20
1095	Central City	Portland	Union Station Multi-modal Center Study	North transit mail in Central City	Identify Improvements to meet additional transportation services to Union Station.	x		\$ 115,500	2	2016-25
1096	Central City	Portland	Barbur/I-5 Corridor Study	I-405 to Highway 217	Assess corridor Improvement options	x		\$ 1,732,500	2	2010-15
			Reinergeliker, The Stand Reininger and	ร้าน ที่ระบบสายแสดน มีแต่สร้างกายอาการแบบสายแบบจ	Cudies - a Calendra (Calendra) - Cale		Sec. Sector -			1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -
rencer maleto		e vicensie		The main second s	Participation of the second second second		Steering .	Test Concernence		
			[]2:		Manufacture in the second of the					
				a an	Charlen and a calebra statistical and the calebra statistics of the second statistics of the sec					400.00
				alle a standard a standard standard betaken i saya saya saya saya saya saya saya sa	Land and the second		100 C			
S161160	مى ئىلىكى <u>ئىل لىمى</u> ا		a a ser a la ser a l La ser a la s	al an an an ann an an an an an an an an an	MUTURE TRANSPORTED AND A STREET AND A STREE		and the second			105222
				a da anti-anti-anti-anti-anti-anti-anti-anti-	enterna valdaktentaan oo kutoka serina daga ahaa ahaa ahaa ahaa ahaa ahaa aha					
3 31UO				APP CARACTER AND A CONTRACT OF A CONTRACT	(sourcesting) description of the provide states of the second states of					
	ra constant da 1911 - La constant									
3 (110Z		Ch diana	alan (n. 1994). Maria (n. 1997). Maria (n. 1997).		A manager of the state of the second state of the second state of the second state of the second state of the s					0.000
initalUS TS	الكوريا المنتق	19. Sec (19. <u>111</u> 1)			Summer and the second				1993 - S	4.610 ° 12. 3
14 - A				and dente consider a star star star star and the	and a stand with a stand and a stand and a stand a stan					an a
1-191-191			ilen er en in 1995 i finn seren i som er Byre også finner		kontentinski man og som					
			abanara, Malaki aw Anganasanan.	ราวอาการแก่ และว่าวัดที่เป็น อย่าง กร้างกิจ อาร์ว (Without Barris	ในสาวไปหมือสาวที่ได้ได้จะเป็นชิงได้จะไปประวัติสาวได้ จะเหตุสมได้มีผู้ได้ สาวมีร้องสาวที่สาวที่สาวได้มีสาวการเรื่องจะเป็นประวัติสาวได้ จะเหตุสนได้มีผู้ได้					
್ಷಚ್ಚಿತ್ರ	ر لارت المحمد الم		<ul> <li>A statistical statist statistical statistical statis</li></ul>	a na statu na	Constitute the destinant of the state of the state of the	Suchassicher Stationer (1997) Stationer (1997)				
فياله الم	(r. ú. íð rið 👘	- steeling	hand shall sais a					2516-+ X86 900 0000		ain an t
	Print and all the second		IS MARKEN STATISTICS	ponessio di simple sintende di di di se	ana man'ny faritr'i Andrea ana amin'ny faritr'oran'i Angrika. I Sana manina dia kaominina dia kaominin'i Angrika amin'ny faritr'oran'i Angrika.			4,000,000		<u>mes</u>
1108	Deleted (included	n project 1108}			รัฐปลุดที่เป็นสุดที่สี่มีสี่รัฐสี่ประสูญสี่ รัฐการไปกาะพูเอสสี่ได้รับและสี่งเรื่องเป็น		<b>1</b>	an a	88 I I I	
					នៅដែលលោកដែនត្រូវបានស្រុកដល់នៅដែរ ហៅលាអ្នរនាំដែនដែ		Sec. 20			
	orstant nurolis i			active constantial for a sense of the sense						40.02.383
1-0-18		e de política	isteine Shushi Harver	Lagara Hand	annentes kontranse statuten na s		Steel of the	en • and house		QE
				the latter we want to see the second s	เชื้อแต่อนใจไม่มีการและ เมื่อไม่เหมินในการสมัยงแปลไม้ได้ ไปประก	2. 2. 1993			- 	
1.110	Controllywood (Control					Ave. an officer	44 A A A A A A A A A A A A A A A A A A			010.300
	TROWCOUT OF	and out here.		ssaiplean an teanni séará an Anteinia	Kulusphant a seen	1278 X. 34	<u>x .</u>	\$ +1 24,820,000	52	2004-09
	10. X						<b>A</b>			
	1. S. S. S.				minuacione and an analysis of the second second					
32		Dellan	Standy Boneyard Mine Modal Base		crossings, bike and she street participation and safety			A . 17 325 0m		
F7-3720	SCHORYWOOD IG	AND A DECIDENCE	Hunk-Alottoniche tiono harteste a serve al			RANK CAL	A STATE OF STATE OF STATE	A DEALER IN THE REAL PROPERTY OF	× - 17	

RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financfally Constrained System	2003 dollars ( "** Indicates phasing in financially constrained	RTP Program Years
		1			Contraction of the second with many second second			1.2.	Mark Barry
n fa Cari					to store ments of classing recessing of selected.	the second			
+	State to Gra	1. A.C.R	معار البناية المقال مدينة إلى المالية المنابعة المنابعة المنابعة المنابعة المنابعة المنابعة المنابعة والمنابعة		ក្នុមជំនាំអនុទាស់ទៅក្នុងស្រុកស្រុកស្រុកស្រុកស្រុកសារ				
	STHOSYWOOD ITC:	and a subsection of the subsec		CANTAVERIES OF CONTRACT.	Juidene numera an	1.X W	. X. J	\$ 4,820,000	Se suit Se
1126	Section version		111251 (di 1111)	di membri lege vicentes.	Reneed meast low later the of more	L*AXIS 9	2 - <b>X</b>	577,500	er 20200
				Manager and Annual States and Annual States and Annual States of the	Additional and a physical static strain a second state of the seco				
1130	E	enchi	In scale the		2007-Bertrio Handrochter	Strewest		\$	111-111
e1135			MANAMAMINA			and the second of	The second second		
							t de la sur a		
: <u>5</u> 1138	TO SCOOME TO US		a na kuna na kuna na kuna na kuna na kuna ku	CHARACTER AND CHARACTER AND		2 #X 1		\$ 2,700,000	2004-092
1139	SL Johns TC	Portland/ODOT	SL Johns Bridge Restoration	St. Johns Bridge	Complete restoration improvements	X		\$ 71,263,500	2010-15
1140	S1. Johns TC	TODO	WRBAP Future Phase Project Implement.	St. Johns Bridge	Bridge Avenue trell	x		\$ 346,500	2016-25
	1.1.1			สายและ (1996) สายและ (1996) สายและ (1996)		ан (° )			
1143	SC Johns TC	CODINE .	Contraction of the second		Renditions limite to opporting supports and the second	± x ···	9 - X + -	\$ 1,165,000	1 2010-15
1144	Deleted (Constructi	on completed)							
1145	Deleted (Constructi	on completed)		· · · · · · · · · · · · · · · · · · ·					
1146	Deleted (Constructi	on completed)							
1147	Scaenne TC +	Portion and	Will Chitalte ( State State In a state	Willing HCCenter Security អ្នកប្រក	Subvioring of the data of the	· 📈 .	A A A	n/at.	2004-00
1148	St. Johns TC	Portland	North Willsmette Greenway	Steel Bridge to Willamette Cove	Study feasbillty of shared-use path	x		nie	2016-25
. Alexand	CALLS IN THE					in the		Carl State	
	SE Some TC and				antipolitical experiment of Pechanical Clause States 2				
1150	DOMOBILI MIS	ere on anotorior of the	St. Some distributed planter the	ne mbina Sueen avia corregioni da sue da a corregionalitatione da a corregione da a corregionalitatione da a co	umproved igning and crossings	<u>м</u> т. Х	X	\$ 2 000,000	2004092
1151	Deleted (Study com	pleted; pending ado	ption)						
1152	Deneted (Study com	pratea)				the second second	N. HWE ALL . S		
1158		2 20 (1 20 1 20 Hz			ACCENTER INTERNET INTERNET INTERNET			462,000	-2016:25
1157		Sec. In	Un carried and			X		Real TROPOD	
v 9158			a sin 145 (Association Skiller)	which who wanted the solar shifts i	Folds and adding the former of the second state of the second stat	建铁公司			
vie de	- 我们的"你们会会		ada wanada waa waanad		individual for the second second states and the second second second second second second second second second	5		1 - A - A - A - A - A - A - A - A - A -	
71159	SE LONG TO SEE	in a second second					······································	23310,000	6. 02004005
	. 8 . 34				alar generality of status seconds and statistical second second second second second second second second second				
	the second state			124 No. 1 (No. 1) (No.	l der Walter, enderte in erverstimen alle der teinten Song der				
			<ul> <li>Address of the second seco</li></ul>			a de la comp	2.00.2	46 YE 10,000,000,000	000005
					Information Instant Centre Building District Plan with		g		
1161	LenteTC	Portlend	Friday School States	Constitution of the Second S	new terro storale pacetrane minutes, where	· 2 *	4		
									C States
					rew transform is present an an intervence.				1. 保護
1162	Lents TC	Portand	Poeter Roed Improvements	ANTO A UNAVANUES CONTRACTOR OF A	Increased on-Erdel parking, as appropriate;	15 X 1	x .	\$ 2.310,000	1 2010 24
1163	Reatin	- CEON -	-ZUDP-owen exclever/UDV000ber255 and Interchanges	2010 INTERNAL INTERNAL DE LA COMPANY	Consult: The overnence to show but to make movements	The states	ez-ye z	S. S. S. Million	

RTP #	2040 Link	Juriadiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( "" indicates phasing in financtally constrained	RTP Program Years
					in della poversia e a si prima contra a della si a si prima Decidada de contra a manera della contra della si prima della si prima della si prima della si prima della si p				
一種			araha masang sing sing sing sing sing sing sing si		ીતિ શાકભાજી અને આંગણવાડી પ્રાપ્ય છે. આ ગામ પ્રાપ્ય છે. આ ગામ આ ગામ કે આ ગામ આ ગામ આ ગામ આ ગામ આ ગામ આ ગામ આ ગામ આ ગામ આ ગા આ ગામ આ ગ				
1. A			and the second	uzumantanangan	A COLORED TO A COLOR		. <u>.</u> .		
<b>1</b> 1768	il	RED	and the and the second second second	นสมบัติสาราช (100 ระโ. การ. 2016) สาราช 1975 (1976)	LCSCOMMENT MUCH CONTRACTOR CONTRACTOR		et x	5	
	- دو ميند.	isterije.	Etter state and state state		Convint that as a the first			5 Lat 3,465,000	1262
	di <u>natio</u> n til States		ISW Shirwood, Elkoward () Stream of the stream of the	The adaption of the Contract Single				Service and sole a	- 1.:: <b>!</b> :0:::4
1173	Hiladale TC	Portland/ODOT	Hillsdale TC Pedestrian Improvements	Capitol, BH Highway, Berlha, and neighborhood streets		x	98 <b>9</b> - A	\$ 3,465,000	2016-25
1176		in the second		Ruelli Marageneusti (Wante	structure al dollar costando provincio acosta a la costa a la contra de la contra de la contra de la contra del			5 - 26 TOD	an an training an
	ورون <u>مح</u> د ال		line and the state of the second s	st pite sha witte at the second se	Stun Dons Denois Developmente a la contra de la contra de En la contra de la c Esta de la contra de		X	Sector Contract	000 (E
<b>Fist</b>		in the second	u	ana ang 19 19 pang 19 ang 19 pang 19 pa Pang 19 pang 19	องกันสารรถและไม่การระบุไม่มีที่ได้สารการกำรังได้ได้เร็ด กลางรุ่มแปล configuration			122 - 100	
					Nucleurin des Lieden extreme versieler and bliefen 2000 Seine eine Stiller ist er Bekter en stellen ist et statigt Versienen versiert ander statigte stellen stellen ist et statigte				
		(a)# s) fa (c) (a m	uth algorith 20 Cantes and 20 Cast	HENDING CONTRACTORING COLOR	in our sile second allowers and the second sile of the second second second second second second second second Second second				TABÛRE
<b>新加</b>			Si da Ro- In civin di	Solution Constant Constant States	մանություններին ու որ ընդեր էլ, դեռի, երություն դե Կինենը, չես Անգեյին, եր Անգեյինը Շորու լիրնել բան				
1186	Raleigh Hills TC	Washington Co.	Scholls Ferry Bikeway	Multhometh County line to BH Highway	Retrofit street to add bike lanes	x		\$ 548,625	2016-25
	i <u>alan in</u> in the				thin handle states to meets of paints so they are			3 2 5 115 50	2 205 25
1195	an a			an in the second state of the s	<ul> <li>Any supervised as a supervised and supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as a supervised as</li></ul>		<b>.</b>	30 <sup>-2</sup> -704551	
1794	West Portland TC	Portland	Capitol Highway Selemic Retrofit	Capitol Highway bridge at Barbur Boulevard	Selamic retrofit project	x		\$ 1,039,500	2016-25
		Rediend/ODOT	Barbur Soulevard Multi-modal	Tanulitaan Deviloymed to poutb Destinand site Resite	sidewalke and street trees, safe pedestrian crossings, enhance transit access and stop locations, traffic signal				
1195	West Porband TC	Formandrobot	Barbur Boulevard Multi-modal		Construct Improvements for transit, bikes and pedestrians. Transit improvements include preferential	<u>x</u>		\$15,000,000	2004-09
1196	West Portland TC	Portland/ODOT	Improvements, Phase 2	Terwilliger Boulevard to 3rd Avenue	signals, pullouts, shelters, left turn lenes and sidewalks Retrofft bike lenes to existing street; shoulder widening,	x		\$ 4,000,000	2010-15
1198	West Portland TC	Portiend	SW Taylors Ferry Bikeway	SW Capitol Highway to Portland City Limits	arainage Indoeventeerenkustatingis versiteren beverstatioeren	X The Street	<b>.</b>	\$ 2,079,000	2004-09
1200	West Portland TC	Portland/ODO7	Pedesirlan Overpass near Markham School	SW Barbur and I-5; connects SW Alfred Street and SW 52nd Avenue	Construct pedestrian crossing over I-5	× × ·	× X	\$ 3,465,000	2016-25
1201	West Portland TC	Portiand/ODOT	West Portland TC Pedestrian District	Barbur, Capitol and neighborhood streets	Improve sidewalks, lighting, crossings, bus shelters & benches	×		\$ 5,775,000	2016-25
1202	West Portland, TC	er Pontand 3	SW Copie) Highway Pedestran and BISTCID Hophovement	Understanding and the reader of the second	Construct sidewaites (improve crossing and bike	X	× 1	s 1,365,000	2004-09
1205	West Portland TC	ODOT	West Portland I-5 Access Study	Taylora Ferry and Barbur remps to I-5	Identify possible new connections over I-5 to serve motor vehicles, pedestrians, and bicycle travel	X		n/a	2004-09

					<del></del>			2002 dollars	
RTP#	2040 10-5	Jurisdiction	Project Name (Fecility)	Project Location	Protect Description	2025 RTP Preferred System	2025 RTP Financially Constrained	{ "" Indicates phasing in financially	RTP Program
	2040 Link								
1207	Deleted (Construct	on completed)							
1210									enes acar concertar
*~1209	Portene Memorea	an ann an an an	Santy/Parries Connectivity	i anna sian ir arigratia shiri 's an a saa	Complete bike and pedering connections between the	<u>⊜</u> 75	çî. X⊄	1.610,000	2006-09
1210	Portland Mainstreet	Portland/ODOT	Improvements	Killingsworth/102nd to 109th, 1-205 to 101st	205 and Parkrose neighborhoods.	x		\$ 578,524	2016-25
			Screensensensensensensen In seconde	and a straight and a	teren Bestellin <del>stein</del> n demikisteristiks				all son the
<b>R</b> Èire			H-eleitin Einen	e se	Childle or shut the second state				
1211	Deleted funder con					and the particular is the		A	
					Inducessory and selenting second second building to a second	*			
1212	POCODO MANAGAS		Ovision Street Transit Improvements,	A STATISTIC STATEMENT AND A	in house sidewalks, lighting, crossings, bus shellers &		x	S 5 6 814 500	2004.00
1215	Portland Mainstreet	Portland	Phase II	SE 136th Avenue to 174th Avenue	benches	X		\$ 1,270,500	2016-25
1216	Portland Mainstreet	Portland/ODOT	82nd Ped Access to Transit Improvements	NE Killingsworth to SE Clatsop	benches	x		\$1,732,500	2016-25
1217	Deleted (Constructi	on completed)							-
1218	Portland Mainstreat	Portland	SE Foster Road/82nd Avenue Intersection	SE Foster Road/82nd Avenue	Pedestrien improvements			£ 040.500	
		an a			handalay high said family and a state of the			346,500	2010-25
			같은 것은 가 가 가 있는 것이 있다. 이가 가 가 가 같은 것은 것은 것은 것은 것은 것이 있다. 것은 것은 것은 것은 것이 있는 것이 있다. 것은 것은 것은 것이 있는 것이 있는 것이 있는 것이 있다. 것은 것은 것은 것이 있는 것이 있는 것이 있는 것 같은 것은 것은 것은 것은 것은 것은 것은 것이 있는 것이 있는 것이 있는 것이 있는 것이 있는 것이 없다. 것은 것이 없는 것이 있는 것		pisilation assesse handle opped anter site 25. Gulight Christian provide a duration (Sarahan (Sarah				
S Pab	nee a Manazak	ra seodor		When winds the Bargarchite	hindille, beskiets one is estimated to starting		XX	99	2. Souse
3.27	Paramet Millionset	and Kalini	ใสมาตร์กู้ ได้ระไม่ไม่ที่ เป็นได้เป็นแม้ได้ระ	st Physical and Philipping			a and the	S	1 200
			an teoris de la constante de la Adria. En la ferma en la constante de		a senten ser and an and a series of the seri			2.232-28	
		C. C. Hearing	สียโตโลกาก ก็เม่าให้กละได้มีสีไปได้	a an isan a tan isan isan isan isan isan isan isan is					
1222	Partian d Malastrat	Portland	SE Milvarda Padastrian Immovaments	SE Milwaukie and Yukon to Tecome	Plan and develop streetscape and transportation			• • • • • • • • • • • • • • • • • • •	
	- or barrel international							\$ 993,300	2016-25
			ton whe the standard duration of the	ter and the method shows a start of the	Nonetwards and state and a property of an interaction of the second state of the secon				
			And Alexandra Management		ugarishi dagaraan di dikarda dhe guide Na Amerikan Maraha dadaraa				
1000		a an	fiel and the	ST Tacara of Streampile Ming	and you'd a lange that and				- Zaciel
			2 3 4 1 <u>1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 </u>		nandolskingdovskillander (* 1994) 1994 - Jahren Alder (* 1994) (* 1995) (* 1995) (* 1995)				
			AND	THE AND A MURICIPAL AND A DESCRIPTION OF A DESCRIPTION OF A DESCRIPTION OF A DESCRIPANTINON OF A DESCRIPTION OF A DESCRIPTION OF A DESCRIPTION	entingale da a scanastinati neno elle mandriktini estanti et. Seconante utili estiluti estisaruniten				er ser contra
					ງກາຍແບບການເປັນໃຫ້ເປັນການເປັນເປັນເປັນແຮງ ແລະ ແລະ ກາຍເປັນເປັນໃນ		(H. 17		AUTO IN
1226	د <u>بودیند را م</u> از ک	and the second	Kangeren bielen neuen au				Ω:γ	3,700,000	
			e ega e Maria de Sel a Santa de		Spokeney (medite bikey biney are to contrate 1 admina-		50 <b>30</b> 30		
02244	Pottend Mainauset	Car Provident Cargo	Sporane of Omedia One Construction	Angewennes wat acommercy and ossing the second second	Conduct the next phase of a condor study that develops	C-4-X-	<u> </u>	\$1 197250,000	22004-03
					multi-modal transportation strategies and specific roadway, bicycle and pedestrian projects that provide		-		
122R	Region	Portland/Metro/ ODOT	Powell Boutevard/Foster Road Corridor Study - Phase 2	I-205 to Damascus	access to Pleasant Valley, Damascus, and the urban prowth boundary expansion areas	x		\$ 1 200 000	2004-00
1229	Deleted (Constructi	on completed)				^		· 1,200,000	2004-09
12.44									18 17 3 1
(230	Portland Memorie	Constant of the second	NEISE /Zhenkenterite	Statistics and the second statistics and	and come summeries a series as a series	ľs⊾tX.	х <u>,</u>	231,000	2010-15
				For Spirit Provide Solverd Spirit Uses	companyations and the state signs for random and incoments				1. 1. D. 4
1231	Portland Mainstreet	Portland	SELECTION IS	45U/JUDINERITC / Fellevine	and control of inable now	Max 21	<u>x+</u>	5 TH5,500	2010-15

2004 RTP Project List October 31, 2003

			1		<u> 2003</u>		r	2003 dollars	-
RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	(*** indicates phasing in financially constrained	RTP Program Years
				new whether states of the stat					
					en anderen generalisten sinderen anderen generalisten.			State Mage	
01233	Portient: Malmetreet	e - Uluu:					111.74	\$30-2.400.000	200409
					<ul> <li>Contract of the state of the st</li></ul>				
				an airean ille airean a' chuir an				14 2800.000	5 - 10 - 10 - 10 - 10 - 10 - 10 - 10 - 1
1. X.) 				initianing a state and a second s In the second	Sadaray (Berry and Street and Street)			1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	
	2.0000000		Hugersteller	Andrew Stantagen and Antonia	<ul> <li>Manufactures and a second se</li></ul>			5 10 100.000	and the second
		and a state of the second	THE CHARTER AND	n Benne in de la constante en la constante en la constante en la constante de la constante de la constante de La constante en la constante en La constante en la constante en	ให้มีเป็นประทันสุดขึ้นสี่เม็นที่ให้เราะที่มีการราชกระบบที่สี่มีเราะ การสร้าง				
		a san ang ang ang ang ang ang ang ang ang a	[1] A. M.		ระบทแก้สระบุญจากแหน่งกฎก เกมเกม ระบุการกระทั่งระ				
<u>1</u>	Contented Automatical	ing tal <del>ula</del> Tagangan	And a second				α + Χ	en 486 0002	n: #2004-09)
T				Allefficiants of Allefer and the second s Allefficiations and the second seco	and the second se				
			HURSLAU HALLANDURS		- สุโทษไทยได้เหตุ (การสารสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการส สารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสารการสา				
	2453 1.2.3			a a serie de la companya de la comp A companya de la comp		- e <b>n</b> a	<b>.</b>	ter figure	
			: 1976년 1977년 1978년 1971년 - 1971년 1971년 1971년 1971년 1971		<ul> <li>รายสมัยชาติสาราชไปที่มีสาราชสาราชอาสาราชอาสาราชสาราชสาราชสาราชส</li></ul>		<u> </u>		
14.626	in the state of the second		Bann Miller (Belgenale	aring weather and same with the shall with	Colleging Regionitation		2 <u></u>	500-00 - eo4,250	dipe.co
	4 4								
					and have a straight and the set of the set o			24 - A - A - A - A - A - A - A - A - A -	
	Containing and an	i i i i i i i i i i i i i i i i i i i	and a state of the second s Second second		and an			<b>1</b> 2010 2010	
1.126	Series designed							760.00	
والكلام والم			hte in something of the care.	Wir Chu waana maaan kang kang kang	the application of the second s	- (f) 2018.		0.000	1.
			the activity land, a	Selfi Andres Millionadan.	ALLEGY ALL HE COLLEGE ALLER		1. х		
0.4 <u>72</u> .0	Formand Como	in the second	el mandelican an all	TAMES APRICES STORAGE STORE STORAGE				C 1.0 000000	6. Zuitze
1249	Portland Corridor	Portland	SW Boones Ferry Bikeway	SW Terwilliger to Portland city limits	Retrofit bike lanes to existing street	x	-	\$ 5,775,000	2016-25
1250	Portland Corridor	ODOT	SW Mecedam Corridor	SW Front Avenue to Multnometh County line	Bikeway design to be determined	x		\$ 577,500	2016-25
1251	Portland Corridor	Орот	SE Powell Bikeway	SE 71st Street to I-205 Multi-use Path	Retrofit bike lanes to existing street	x		\$ 5,197,500	2018-25
		and the second	And a star strange and	same blemmetrefferingste auftragerunge	[3] A. S. Marketter and A S. Marketter and A. S. Marketter and A. S Marketter and A. S. Marketter and A				and a straight the second s
			NETHER CONTRACTOR OF A CONTRACTOR STORES		Bor COMPACTOR CONTRACTOR CONTRACTOR INTERVAL AND INCOME AND INCOME INCOME AND INCOME AND IN INTROVINCIÓN AND INCOME AN				
	T Utilini COMO	Parland	136th Avenue Bike and Pedestrian	Castas Read to Division Street			Apple Apple		CHE YOU WOULD BE
1254	Portland Corridor	Portand	Obdelae Chart Billion and Billion	OF Frank Augure to 75thb Augure	Peterst bile lange to existing street	<u> </u>			2016-25
1255	Portland Corridor		Division Street Dikeway Improvementa			X			2016-25
1257	Deleted (Constructi	on completed)					<u> </u>		
1258	Deleted (local level	Improvement)							
h 7259	4. South North SC	- Ponteino -	Whitestkinner del Konstate in L		Plan and develop improvements to the redection	×		\$ 76,078	2004-09
		Dedicad	Millione under Bedeutiden Dietrict	East of LSI ampoand CBI ( DT station	environment; improve sidewaiks, lighting, crossings, bus				
1260 44.9.05	South/North SC	Portend		Cast of t-o, proposed SAV LK1 station area		<u>х</u> Парадолі		\$ 773,850	2016-25
- 1263	Banfield SC	Portant/ODO	Connect SC Posteriar In province State		Dentification and states with the states	\$ <b>`_</b> X	<b>X</b>	2,598,750,	*E2010-151

# 2004 RTP Project List

NTP         3840 Lak         Jurtafiction         Project Name (Pacify)         Project Leastion         Project Description         2005 FPT Project Description        <									2003 dollars	· · · · · · · · ·
Number of the second	RTP#	2040 Link	Jurtadiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	(*** indicates phasing in financially constrained	RTP Program Years
Automatic         Automatic <t< td=""><td></td><td></td><td></td><td></td><td></td><td>linder (</td><td></td><td></td><td>- Arr</td><td></td></t<>						linder (			- Arr	
Market Structure         Market Structure         Market Structure         Conduct a project development study to determine right Market Structure	1264			venun sein autoren darine		North 19 Stor Improve State (Northeam) and Instal Store - St And Store - State - Store		e i		
1262         Portiand Condex         Portiand Project Development Statistic         Access of the second statistic of the Condex statistic of the Co	2.40				NEXTRACTION AND A CONTRACTOR OF A CONTRACT OF A	Iterrations and social polytometric and second as	202	*8	5	200403
1287         Portiand Corridor         Portiand Boulerand Project Development         205 to 174th Avenue         Conduct a project development study to determine high construction media and planned later uses.         X         \$         400,000,000         2016-25           1286         Portiand Corridor         ODOT/Protiand         Weed Boulward - Portiand         LOS is 174th Avenue         Wide street for unlines with aldewards and planned later uses.         X         \$         400,000,000         2016-25           1289         Portiand Corridor         ODOT/Protinn         US 30 ktW112th Avenue         Add signal at interaction         X         \$         400,000,000         2016-25           1289         Portiand Corridor         COOT         US 30 ktW112th Avenue         Add signal at interaction         X         \$         900,000         2016-25           1289         Portiand Corridor         TAML         US 30 in Limiton         Conduct a potentiant ADM patient MV 201V/con         X         \$         900,000         2016-25           1280         Portiand Corridor         TAML         US 30 in Limiton         Conduct a potentian for another with all diverses and the street balance MV 201V/con         X         \$         900,000         2016-25           1281         Portiand Corridor         DOOT         US 30 in Limiton         Conduct a potentian for a	-4200	CALIBRE WEY PLC.		NETOE 0900 AVERUE VIEICO XILORDIU	AVERSING CONTRACTOR OF STREET, STRE	re-protection and the second state of the seco	- X *	X	4,042,500	2010-15
1288         Portial Control         ODOTPortial         Portial South 125 in 174B Avenue         Widen street to four lates with addwalks and bits lates         X         \$         45,000,000         2016-25           1289         Portial Controls         ODOT         US 30 Perfecting Access to Transit         US 30 at NW 112b Avenue         Add spall at biselection         X         \$         45,000,000         2016-25           1270         Portial Controls         US 30 Perfecting Access to Transit         US 30 at NW 112b Avenue         Add spall at biselection         X         \$         900,000         2016-25           1270         Portial Controls         US 30 Perfecting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b Interesting Access to Transit         US 30 at NW 102b	1267	Portland Corridor	Portland	Powell Boulevard Project Development Study	1-205 to 174th Avenue	Conduct a project development study to determine right- of-way needs and achematic designs to support identified transportation needs and planned land uses	<b>x</b> ·		n/a	2004-09,
129         Portiand Contridy         US 30/W112b Interaction         US 30 at WW 112b Avenue         Add signal at Interaction         X         \$         135.000         2010-15           1270         Portiand Contridy         TriMes         US 30 in Liniton         Devices transmitties within Liniton area and construct ADA point at Use to positive within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton area and construct ADA point at Use to positive Within Liniton         X         \$         900,000         2016-25           1277         Portiand Contridy ODOT         US 30 Podestrian Networkshill Add States and Bate States States and Bate States St	1268	Portland Corridor	ODOT/Portland	Poweil Boulevard - Portland	I-205 to 174th Avenue	Widen street to four lanes with eldewalks and bike lanes	x		\$ 48,000,000	2016-25
Device frame         US 30 Padestrian Access to Transit Improvementa         US 30 In Linkon         Device frame and construct ADA pade to bus stops between NV 28h7/cen and Sauvis fame and Sauvis fame and Sauvis fame and Sauvis fame and Sauvis fame an	1269	Portiand Corridor	ODOT	US 30/NW 112th Intersection	US 30 at NW 112th Avenue	Add signal at intersection	x		\$ 135,000	2010-15
Automatication       Description       Description       Description       Section	1270	Portland Corridor	TriMet	US 30 Pedestrian Access to Transit Improvements	US 30 In Liniton	Develop transit amenities within Linnton area and construct ADA pade at bus stops between NW 29th/Yeon and Sauvie Island Bridge	x		\$ 900,000	2016-25
Izra       Purified Condor       ODOT       US 00 pedestine Overcosing       X       \$       350,000       2018-25         Izra       Perified Condor       ODOT       US 30 hitered/on improvements       VV 108/h Avenue       Construit a pedestine overcosing       X       \$       350,000       2018-25         Izra       Perified Condor       ODOT       US 30 hitered/on improvements       VV 108/h Avenue       Construit a pedestine overcosing       X       \$       350,000       2018-25         Izra       Perified Condor       ODOT       US 30 hitered/on improvements       VV 108/h Avenue       Construit affertinered/on filterinered/on filtered/on improvements       X       \$       1076,000       2018-25         Izra       Perified Condor       ODOT       US 30 Streetscape Improvements       VV 108/h Avenue       Construit affertinered/on filterinered/on filterinered/on filtered/on exceedsing to correct of filter interescipe Improvements       X       \$       1076,000       2018-25         1277       Perified Condor       ODOT       US 30 Streetscape Improvements       US 30 in Linkon       Construct affertine filtered/on exceedsing to correct of filter interescipe Improvements       X       \$       1076,000       2018-25         1278       Perified Condor       ODOT       US 30 Streetscape Improvements       US 30 in Unitor	新規の			[25] A. A. Martin, A. M. Martin, A. M. Martin, M. M. Martin, "A strain and the strain of the stra		Right Andre galleen with Andrew Speed	1	Sie bie "		
1272       Portland Contdor       ODOT       US 30 Padaethien Overcrossing       NV 108th Avenue       Construct a padestrian overcrossing       X       \$       350,000       2018-25         1273       Portland Contdor       ODOT       US 30 Intersection Improvemente       US 30 at NVS Bitzman and Balboa streets       Realign intersections       X       \$       600,000       2018-25         1274       Portland Contdor       ODOT       US 30 Bits and Padestrian Improvements       NV 106th KNthröge Avenues       Construct streetscape improvements to Visually name       X       \$       1,746,000       2016-15         1275       Portland Contdor       ODOT       US 30 Streetscape improvements       US 30 in Linnton       Construct streetscape improvements to Visually name       X       \$       400,000       2016-25         1276       Portland Contdor       ODOT       US 30 Streetscape improvements       US 30 in Wilbridge Improvements       US 30 in Wilbridge       Install center turn lane to Front Avenue       X       \$       400,000       2016-25         1277       Portland Contdor       ODOT       US 30 - Wilbridge Improvements       US 30 in Wilbridge       Install center turn lane to Front Avenue       X       \$       400,000       2016-25         1277       Portland Contdor       ODOT       US 30 - Wilbridge Impro	1271		e wan	and the contraction of the second	Some winder State worden	Active references an approximate a contract of the second strategy and the sec	14 х х		\$ 50000	2016-251
1273       Portland Contidor       ODD       US 30 Intersection Improvements       US 30 at MW Seitzman and Balboa streets       Realign intersections to correct offset intersections       X       \$       600,000       2016-25         1274       Portland Contidor       ODD       US 30 Bitse and Pedestrian Improvements       NW 105th to Kithridge Avenues       Construct allowalisa and bits facilities       X       \$       1.746,000       2016-25         1275       Portland Contidor       ODD       US 30 Streetacape Improvements       US 30 in Linnton       Construct allowalisa and bits facilities       X       \$       400,000       2006-09         1275       Portland Contidor       ODD       US 30 Wilbridge Improvements       US 30 in Wilbridge       Install center turn lane to Front Avenue       X       \$       135.000       2016-25         1277       Portland Contidor       ODD       US 30 Wilbridge       Install center turn lane to Front Avenue       X       \$       135.000       2016-25         1277       Portland Contidor       ODD       US 30 Wilbridge Improvements       US 30 in Wilbridge Improvements       US 30 in Wilbridge       Install center center of Front Avenue       X       \$       135.000       2016-25         1277       Portland Contidor       ODDOT       US 30 Wilbridge Improvements       US 30 in	1272	Portland Corridor	0001	US 30 Pedestrian Overcrossing	NW 108th Avenue	Construct a padestrian overcrossing	×		\$ 350,000	2016-25
1274       Portland Corridor       ODOT       US 30 Bike and Pedestrian Improvements       NW 105th to Kithidge Avenues       Construct streatescape improvements to Vieually narrow radway, including indicating to Vieually narrow radway, vieually vieualy to Vieually narrow radway, vieually to Vieua	1273	Portland Corridor	ODOT	US 30 Intersection Improvements	US 30 at NW Seltzman and Belboa streets	Realign intersections to correct offset intersections	x		\$ 600,000	2016-25
1275       Portland Corridor       ODOT       US 30 Streetlacape Improvementa       US 30 in Linnton       Construct streetlacape Improvementa fol Veuelly narrow rodwey, including landscaping, pedestrian builto uts and median       X       \$       400,000       2004-09         1275       Portland Corridor       ODOT       US 30 Streetlacape Improvementa       US 30 in Linnton       Install center tum late to Front Avenue       X       \$       135,000       2016-25         1277       Portland Corridor       ODOT       US 30 - Willbridge Improvements       US 30 in Willbridge       Install center tum late to Front Avenue       X       \$       135,000       2016-25         1277       Portland Corridor       WUCH: NURLING Constructions       NUCH: NURLING Constructions       NURLING Construc	1274	Portland Corridor	ODOT	US 30 Bike and Pedestrian Improvements	NW 105th to Kittridge Avenues	Construct eldewelks and bike facilities	x		\$ 1.746.000	2010-15
1275       Portland Contidor       ODOT       US 30 Streetscape Improvements       US 30 in Linnton       and median       and median       x       \$       400,000       2004-09         1276       Portland Contidor       ODOT       US 30 - Willbridge Improvements       US 30 in Willbridge       Install center turn lane to Front Avenue       X       \$       135,000       2016-25         1277       Portland Contidor       ODOT       US 30 - Willbridge Improvements       US 30 in Willbridge       Install center turn lane to Front Avenue       X       \$       135,000       2016-25         1277       Portland Contidor       Mol StatingUp Variation Variat						Construct streetscape improvements to Visually narrow				
1278       Portland Contdor       ODOT       US 30 - Willbridge Improvements       US 30 in Willbridge       Install center lum lane to Front Avenue       X       \$       135,000       2016-25         1277       Portland Contdor       WVChumpUl/-XConton Disconstruction       WVChu	1275	Portland Corridor	ODOT	US 30 Streetscape Improvements	US 30 in Linnton	and median	x		\$ 400,000	2004-09
1277       Person Summer       Over Example, Vertice Receiver and Participation States       Sta	1276	Portland Corridor	ODOT	US 30 - Willbridge Improvements	US 30 in Willbridge	Install center turn lane to Front Avenue	<u>x</u>		\$ 135,000	2016-25
201       Region       Multinomah Co.       Hogan Confridence       S. Conference       <	41277	م <u>ورد المح</u> طينية الم		Nurthenneth Scientific States and a	Supervision of the supervision o	Televise and solved and explored and a strategy of the second second second second second second second second s				
Also						Kiegranous Inter (Elimetras (Septic)) Semiorer Sam		- 1 <u>5</u> 5		
2270       Percent Control       1 - Exercise Sector Secto						Elektrallesinse estinya eta anta (Sine - Senie) Manali, lakan seni tura Jasharan II. angerana eta				
State       State <th< td=""><td></td><td></td><td></td><td>- A STREAM AND A ST</td><td>리는 가지 않는 것을 가셨다.</td><td>West Field Infinited and Report And Andreas Strengther</td><td></td><td></td><td>9.e</td><td></td></th<>				- A STREAM AND A ST	리는 가지 않는 것을 가셨다.	West Field Infinited and Report And Andreas Strengther			9.e	
2001       Region       Multinoma Co.       Hogan Confrónt Improvemental       I-94 to Glisan Street       Construct new I-94 interchange       X       \$       \$       277,720,000       2010-15	1276	and the second	and the second	n sui fais a chuir fu dean an dha	enangi bulanyah di Wasangan anahising 🦿 🕺	and a state of the	2 1		Sec. in the second	- 20.00
2000       Higher       Automatics       Higher and discounce       Higher and discounce       Automatics       Aut	和时					วรุงการเห็นระว่า อาจุลาวก่องได้มีการเกิดเห็นกันได้ -			ST. Barris	
Interview       Description       Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>						and the manager of the state of the state of the states				
2000 F H Sglow P S - Swumon and Construct new Heat in the State of the	5 102/9 P			1. OL STILL STILL FUR ATTACAL AT		1111	2014 X 22	240 - X - X - X - X - X - X - X - X - X -	19	<u> </u>
2001 Region Multinoman Co. Hogan Comporting P.O.W Presenting P.O.W	2000	E TROUGH IS	COMUNICATION		State Strate Congress State and Soveral State	nounin capacity in provinants and increase controls (BEA	(4) X	Xi	\$ 23. 13,860,000.	
2000 Declar DUULI (LKUIN XK CORPORT V. U.W WEREBORIER POWER A NICTING) / K	2001	Region	Multhomah Co.	Hogan Corridor Improvements	1-84 to Glisen Street	Construct new I-84 interchange	X		\$ 27,720,000	2010-15
2002 Kegron 2007 Preserve ruture right-of-way X \$ 17,556,000 2004-09	2002	Region	0001	1-84/US 26 Connector R-O-W Preservation	Paimquist to Highway 26	Preserve luture right-of-way	X		\$ 17,556,000	2004-09
2003 Region Muthomath Co. Hogan Control Improvements Palimquiet to Highway 26 in UGB Construct new principal arterial connection X \$ 9,471,000 2016-25	2003	Region	Multionah Co.	Hogan Corridor Improvementa	Paimquiet to Highway 26 In UGB	Construct new principal arterial connection	X		\$ 9,471,000	2016-25
2004 Region ODOT I-84 Wildening 238th Avenue to Sandy River Bridge Wilden 1-84 X \$ 9,471,000 2016-25	2004	Region	ОРОТ	I-84 Wildening	236th Avenue to Sandy River Bridge	Widen 1-84	<u> </u>	·	\$9,471,000	2016-25
2005 Region ODOT II-84 Troutdale Interchange Improvement Troutdale Interchange (exit 17) Improve Troutdale Interchange (exit 17) 2016-25	2005	Region	ODOT	t-84 Troutdale Interchange Improvement	Troutdale Interchange (exit 17)	Improve Troutdale Interchange		Sincle in the	\$ 17,325,000	2016-25
2006 - Honor - Annoional Cont. Ingertendet inprovimente - naces allen Special Strengenet - 45, comentin anemodale.	2006	Henor	Multionan Cols-	income control support in all sectors	ellen Shreimsbingshon sin an	center tim alternedian.	ં દે પ્રેસર્ટ	X	\$ 1,155,000	2004-09
Construct expand and/at unacade transit stations and	2007	Region	TriMet	Transit center and park-and-ride upgrades	Various locations in subarea	park-and-rides throughout subarea	X			2004-25
2007         Region         TriMet         Transit center and park-and-ride upgrades         Various locations in subarea         Construct, expand and/or upgrade transit stations and park-and-ride upgrades         2004-25		5 <b>3 4 1</b> 1				the and the provide the second s	-			
2007         Region         TriMet         Transit center and park-and-ride upgrades         Various locations in subarea         Construct, expand and/or upgrade transit stations and park-and-rides throughout subarea         X         2004-25           2	2008	Gelewaytic	Porteline: A. P.	Improvements, Phase 1. 1. Scalety and Information of the second state of the second st	NEWARAN CIRCLES ST. S. S.	monities and chosings street lighting broade lanes and a multimodel Barsty imployements	(°~-x 3*	a k	\$ 3,234,000	2004-098

.

#### Public Comment Draft 2004 RTP Project List

. f	Actober	31	ົງກຸດຈ	
	THE 100 00 THE 1 THE 180 TH			

RTPS	2040 Lin <del>a</del>	.Furtheliction	Project Name (Facility)	Project Location	Project Description	2125 RTP Preferred Syntam	2025 RTP Financially Constrained System	constrained	RTP Program Yearn
2360	Geterney RC	Patiend	Haiony Street Bridge Seiami: Resolt	Phalacy Street at I-54	Balantic retrolit project	X		<b>3</b> 12,4	2016-25
			na Barlandia ang kanalang kan Kanalang kanalang kan Kanalang kanalang kan	an (fui fusio) - Basta di Sanagaran (fui fusio) Anna an		i.		s interes	77 Jane 21
				Service and the content of the service backward in the	In the first Multimetry and construction of the second of the second sec			<b>*</b> 2.710	NG. 2010-15
		-		Stratural of the Lord	In the most of the second seco				
2013	Carlorna y MC	Multremak Ca.	HE Hairoy Dismay	162nd Avenue to 201st Avenue	Widen to retrait tike larges to existing street	X		\$ 1,420,1	X01 3304-03
	Common Pro-	Gabonia C.C.	The set of	- State Riverson in 2010 Roomsof NE Differs Officer of Stational Disord	ninger ist for die besterne in worden is einer Aufgeberechtigt des verfassenses interseet in besterne das des eine Trade verfassenses om die sonder interseet interseet interseet onterseet interseet interseet interseeting affectiere onterseet	×		1 (1040) 1 7051	ni 2006-09 Dž. 2016-15
	Constants Constants		n ann domh. Si Infrich condoms	(d) Separate and Andreas and separate distribution of second second second second for 2008 and second second second second second distribution.	Sales Constants de la constant de la constant Autorit basalements activationes :	× . × .			07 - 200409 09 - 200409
	n Carrier Pictor Contenent Pictor		i de la constante de la constan La constante de la constante de	All Statistics of All Statistics and an and a second state of the	na contra a more francisco prove Rendi fuera interación prese Manante interación de la contracta de la contracta de la contrac	X			
			Charles of the second s	Channel ( Lagran - Lagran	High prints incluits are periodian intervenients in regional come				270-05 270-75
- 2022	<b>Series</b> 201	A Martin		COMPLEX CONTRACTOR	- Construction of the second secon	*	÷	<b>.</b>	(0
			and the second se		ang an witherpower parallelise two		X		
			in the second state of the	Chiefen of PCRD NE Strawn is Chiefen Chiefe Street and NO	Alexandra and a second se			<b>.</b>	9
2017 2017			America Charles and the strong states States for the strong strong states	Parte Men e him 975 an Dro Avelue MOTore vez of German Childer	inguised owner: URT setting and react sets supported breaktions strang helicet on Cristian stray.				
				in the second			<b>.</b>		
2030	dresham PC	Ganationa Ganation	Petropisi Post ingenerate	242ad Avenue to LIS 24	White is fee lands	uutuun Aprilia X		1 2,656.5	00 2016-25
2631	Onatan HC	otor	Hogen Cortáor improvemente	Hoganifiumatice from 1-64 to US 26	None freight from existing 181st/Summite marie	X		<b>s</b> 27,7	50 2016-25
	: interio			and a second stratight of the second strategy					olizeri
2034	Greehem RC	Mulmometi Co.	Children Bland Improvements International Specific and a state	2016 Avenue to 2026 Avenue ISBN creative Pointe exclusive and a second	ingenee Cheleise Street Management Steet State Street in Street Doorsevert	X A		13.349,5 2	x 2016.25

.

2004 RTP Project List October 31, 2003

								20	003 dollars	
						2025 RTP	Financially	F	ohasing in	RTP
RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	Preferred System	Constrained System	f	inancially onstrained	Program Years
2036	Gresham RC	Gresham	Wallula Street Reconstruction	Division Street to Stark Street	Reconstruct street from Division Street to Stark Street	x	x	s	1.732.500	2016-25
2037	Gresham RC	Gresham	Bull Run Road Reconstruction	242nd Avenue to 257th Avenue	Reconstruct street from 242nd Avenue to 257th Avenue	x		s	1,155,000	2016-25
2038	Grasham RC	Gresham	Walters Road Reconstruction	Powell Boulevard to 7th Street	Reconstruct to improve access to Springwater Trail	×	Y Y	5	1 155 000	2018-25
2039	Gradado BC	Greshern	Regner Road Reconstruction	Creveland Street to city limits	Reconstruct Regner Road from Cleveland to dry Imilia	Y	- <u>v</u>	e	14 200 000	2016.25
	0t. 20	Greeham	Greebern BC Collector Improvements	Barnes Road, Williams Street, Chase Road, Welch			~		14,200,000	201022
2040	Gresnam RC	Greatian		Road, Famblad Road, Salquist Road, Hillyald Road	Reconstruct street to arterials standards, including blke	<u> </u>		\$	5,775,000	2016-25
. 2041	Gresham RC	Mutnomah Co.	257th Avenue Compor Improvements	Division Street to Powell Valley Road	Tanes, sidewalks, drainage, lighting and traffic signals Realign intersection to provide for safety, capacity, bike	<u> </u>	x	5	4,800,000	2004-09
2042	Glesham RC	Multhomah Co.	257th Avenue Intersection Improvements	Intersection of 257th/Patriquist Road/US 28	and pedestrian movements	X	x	\$	4,899,510	2004-09
2043	Gresham RC	Multnomah Co.	Powell Valley Road Improvements	242nd Avenue to 282nd Avenue	facilities	X		\$	4,712,400	2016-25
2044	Gresham RC	Muthomah Co.	Orient Drive Improvements	282nd Avenue to 257th Avenue	Improve Orient Drive	x	x	\$	4,158,000	2018-25
			The second s	Set as a set of the se	and bike lanes. Widen and determine the appropriate	Sec. 2				
					cross-section for Highland Drive and Pleasent View Drive from Powell Bodevard to 190th Avenue based on the					
2045	Gresham RC	Muthometr Co	190th Avenue Improvements	Buter Road to Highland Drive and Powell Boulevard to 190th Avenue	recommendations from Phase 2 of the Powell Boulevard/Foster Road Contdor Study	x	x	5	12,500,000	2010-15
	0	Multaomah Co	Obdeion Street Improvements	Pinfadole Augurus to Malhula Augurus						
2040	Gresnam RC	- Multionali Co.				<u> </u>		\$	4,620,000	2016-25
2047	Gresham RC	Gresham.	Division Street Improvements	NE Welfula Street to Birdadale Road	Complete boulevard design improvements	x	X	5	4,620,000	2004-09
2048	Gresham RC	Muitnomah Co.	Burnside Street improvements	NE Wallula Street to Hogan Road	Complete boulevard design improvements	x		\$	7,484,400	2004-09
2049	Greshern RC	ODQT/Gresham	Powell Boulevard Improvements - Gresham RC	Eastman Parkway to Hogan	Complete boulevart design improvements	×	×	s	4.620.000	2004-09
					Study to identify additional access management strategies, define long-term freight route in corridor and					
2050	Perior	ODOT/Gresham/Mult	I-84 to US 26 Corridor Study (ROW and arterials)	1-84 to US 26	evaluate potential new alignment south Powell Boulevard to US 26	~			1 155 000	0040.45
2050		opor	US 26/Springwater Interchange			^		2	1,155,000	2010-15
2051	Springwater (A	Caselland	Contract Cold out Tank	Converting to the test of the	The man charge on OS 25 th serve in Listian erea	X	<u>X and a</u>	5	25,000,000	2004-09
2053	Greenam RC	Greenantica		Spingwater Trail at 182nd Avenue and Pleasant	Springwater traincomector	X	X	5	1,963,500	2004-09
2054	Greeham RC	Gresham	Springwater Trail Connections SW Walters Road/Springwater Trail	View/190th Ave	Provide bike access to regional trail Upgrade pedestrian signal to full traffic signal and	x	X	\$	1,039,500	2016-25
2055	Gresham RG	Gresham ,	Access	SW7th to Powell Boulevard	provide bike access to regional trail	- X	X	<b>\$</b>	346,500	2016-25
2056	Gresham RC	Mutnomah Co.	Division Street Bikeway	174th Avenue to Walfuls Avenue	Retroft street to add blke lanes	X	x	5	460,000	2010-15
2057	Grandware BC	Gleeban/ODOT	Gresham RC Pedestrian and Ped-to-MAX.	Pkwy, Main Street, Cleveland and Intersecting	Improve sidewalks, lighting, crossings, bus shelters and hanches		contain a start			
2007	Centhon BC	Gresham	Stringenter Trail Parlastrian Across	Fairmen Trivia Roberts Regimen Horme	Improve and designed the set the set		A 152	à	7,045,500	2004-09
6002		er uch raite	Division Street Pedestrian to Transit		Improve sidewalks, lighting, crossings, bus shelters and	X	X	3	2,000,000	2016-25
2059	Gresham RC	Gresnam	Access improvements	T7401 to vvalible Avenue	Delicites	X	X	\$	1,155,000	2016-25
2062	Deleted (Project co	mpleted)			Study LRT to Mt. Hood Community College; a			· 		+
2063	Gresham RC	TrlMet/Metro	Study LRT extension to Mt. Hood Community Col.	тво	preliminary study was done between 1993-95 as part of the East Multhomah County Long-Range Transit Plan.	x		l	n/a	2016-25
2065	Greeham RC	Gresham	Phase 3 Signal Optimization	Systemewide	Optimize signals	x	x	s	2,310,000	2004-09
2068	Deleted (Constructi	on completed)								2016-25

RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( *** Indicates phasing in financially constrained	RTP Program Years
		anot	and a second state of the		(Phase 1 in FC: modify signing, striping channelization				and a subsection of
2069	PDXIA	ODOT 21	I-203 Interchange Improvement	1-205 NB/Airport Way Interchange	and signal timing for NB on-ramp) Widen I-205 SB on-ramp at Aroort Ways modify simpler	X	× .	\$ 23,100,000	2004-09
2070	PDXIA	ODOT	1-205 Interchange Improvement	1-205 SB/Airport Way Interchange	striping channelization and/or signal timing for the I-205 NB on-ramp at Airport Way	~	×	* ese poo	2004.00
2071	PDX IA	ODOT	I-205 Auxillary Lane	Airport Way to Columbia Boulevard	New I-205 auxiliary lane from Airport Way to Columbia Boulevard	~	<u>^</u>	a 050,000	2004-09
2072	PDX IA	ODÓT	I-205 Auxiliary Lane	I-84 to Columbia Boulevard	New auxiliary lane from I-84 to Columbia Boulevard	×		\$ 23,100,000 \$ 5,775,000	2016-25
2073	South Shore IA	Multnomah Co.	I-84/I-205/Tillamook Shared-Use Connector Study	I-84/122nd Avenue to I-205	Study feasibility of corridor	~ ~ ~		s 5,775,000	2016-25
2074	South Shore IA	Multromati Co.	Sandy Boulevard Widening	122nd Avenue to 238th Avenue	Widens street to five James with streetelike and bits inner	<u>^</u>	-	n/a	2016-25
2075	South Shore IA	Multhomeh Co.	207th North Extension	Sandy Boulevard to Almost Way	New street connection between 207th Avenue and Arrort Wey	<u>, x</u>	*X	\$ 11,800,000	2016-25
2076	South Shore 18	TriMer	181st Avenue Frequent free	Grastian to Columbia South Stores	Construct Improvements that enhance Frequent Bus	X		\$ 6,699,000	2016-25
2077	South Shore IA	- Multhomati Co,	181st Avenue Widening	Hillsey Street to EB opcomp to L841	Wildow stream to three lenses on this and		X	\$ 1,350,000	2010-15
2078	South Shore IA	Multhomah Co.	162nd Reilroad Crossing Improvements	162nd Avenue/railroad bridge	Replacing railroad bridge to allow for med widening	x	X	5 1,097,500	2004-09
2079	Deleted (Constructi	on completed)						\$ 6,006,000	2016-25
2080	South Shore (A	Multinomati Co.	202nd Railroad Crossing Improvement	202nd Avenue/railroad bridge	Replacing relirced bridge to allow for road widening	x	Y	s /0/2 900	2016-25
2081	South Shore IA	Matriomati Co	223rd Railroad Crossing Improvement	223rd Avenue/railroad bridge	Replacing railroad bridge to allow for road widening and two crossings, one north of Sandy and one south of 1-84			6.002,000 8.000,000	2004-03
2082	South Shore IA	Multnomah Co.	Columbia River Highway Railroad Crossing Improvement	Columbia River Highway east of I-84	Replacing railroad bridge to allow for road widening	v		<b>s 5,240,000</b>	2004-09
2083	South Shore IA	Multnomah Co.	Sandy Boulevard Overpass	Sandy Boulevart at I-84	Construct overpass to reconnect Sandy Boulevard over I-			\$ 1,386,000	2016-25
2084	South Shore IA	Multrometr Co.	161st Avenue Intersection Improvement	181st Avenue/Gisarr Street Intersection	Improve Intersection	X		\$ 27,720,000	2016-25
2085	South Shore IA	Multhomah Co.	181st Avenue Intersection Improvement	181st Avenue/Bumside Road Intersection	Improve Intersection	<u>,                                    </u>	<u> </u>	\$ 623:700 5	2016-25
2086	Deleted (Construction	on completed)		· · · ·			~ ^	<b>3</b> 346,300	2010-25
2087	Deleted (Construction	on completed)			· · · · · · · · · · · · · · · · · · ·				2016-25
2088	South Shore IA	Portland	NE Marine Drive/122nd Avenue Improvements	NE Marine Drive/122nd Avenue Intersection	Signalization, widen dike to install left turn lane on Marine Drive	×	×	\$ 1,943,865	2004-09
2091	South Shore IA	Portland	NE/SE 148th Avenue Bikeway	Division	Retrofit bike lanes to existing street	x	X	\$ 35,805	2010-15
2093	South Shore IA	Multhomah Co.	Marine Drive Safety Corridor Plan	Marine Drive from Troutdale to Rivergate	Long-term traffic management plan	x		n/a	2016-25
2098	Rockwood TC	Multhoman Co.	201st/202nd Avenue Corridor	Glisan Street to Halsey Street	Reconstruct and widen to five lanes Reconstruct and widen to three lanes (Sandy to Halesey)	X	- Contraction of the Contraction	\$ 2,356,200	2016-25
2099	Rockwood TC	Multhomah Co.	Improvements	Sandy Boulevard-Powell Boulevard	In FC System)	×	X	\$ 9,909,900	* 2004-09
2101	Rockwood TG :	Greenam	Stark Street Improvements	1900 ib 1970	Complete boulevard design improvements	<u>x</u>	X	\$ 3,465,000	2010-15
2102	Contract TC	Mitthomah Co. 4	151al Avenue improvemente	Cilleon to Yourble	Complete Boulevard design improvements	X	X	\$ 3,465,000	2004-09
2104	Rockwood TC	Multriomati Co.	Burnalde Road Boulevard Improvements	181st Avecua in 197th Avecas	Complete bollevard design improvements	×	<u> </u>	\$ 3,326,400	2010-15
2105	Rectanged TC	Grestern	Rockwood TC Pedestrian and Ped-to-MAX	181st. 188th, Stark and Intersecting streets and LRT	Improve sidewalks, lighting, crossings, bus shelters and	<u> </u>	× ×	\$ 4,200,000	2004-09
2108	Deleted (Construction	on completed)		A LANGE AND A L	venulos	X	X	\$ 3,465,000	2016-25
2109	Fairwaw/WV TC	Multhomah Co.	Glisan Street Improvements	202hd Avenue to 207th Avenue	Complete reconstruiding of Gilson Smart & Sig land				
2110	Fairview/WV TC	Multhomah Co.	MKC Collector	Haisey Street to Arata Road	Construct new collector of regional stanificance	x	X	5 1,800,000 5 0,400,000	2004-09

				00000000			2025 RTP	2	003 dollars *** Indicates	
						2025 RTP Preferred	Financially Constrained		ohasing in financially	RTP Program
RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	System	System		onstrained	Years
2111	Deleted (Construct	ion completed)		· · · · · · · · · · · · · · · · · · ·	Improve sidewalks lighting crossings bus shelters and					
2112	Fairview/WV TC	Multnomah Co.	223rd Avenue Improvements	Glisan to Stark	benches	x		\$	1,155,000	2016-25
2113	Fairview/WV TC	Multnomati Co.	Halsey Street Improvements	190th Avenue to 207th Avenue	Widen to three lanes with sidewalks and bike lanes	x		<b>\$</b> .	2,772,000	2004-09
2115	Farview/WV TC	MultCo/FV/ WV	Improvements	Failvlew, Halsey, Glisan and neighborhood streets	Denches	×	x	\$	1,386,000	2016-25
2116	Fairview/WV TC	Multnomah Co.	NE 223rd Avenue Bikeway and Pedestrian Improvements	NE Halsey Street to Marine Brive	Retroft bike lanes and sidewalks on existing street	×	x	\$	577,731	2010-15
2117	Fairview/WV TC	Multnomah Co.	207th/223rd Access Management Plan	207th/Gilsan/223rd from i-84 to Burnside	Traffic Management Plan to protect mobility on 207th/223rd to Gresham	x		l	n/a	2016-25
2118	Fairview/WV TC	MultCo/FV/ WV	Arata Road Improvement	Wood Village Boulevard to 238th Drive	Upgrade street with center turn lane/median, sidewalks and bicycle lanes	×		s	1.000.000	2010-15
2120	Troutdale TC	Multriomah Co.	Sandy Boulevard Bicycle and Pedestrian	162hd to Troutdale	Retrofit bike lanes and sidewalks on existing street	¥	Ŷ		8 316 000	2016-25
2121	Troutdale TC	ODOT/MultCo	Columbia River Highway Improvements	Kibling Avenue to Sandy River	Upgrade to include bicycle and pedestrian facilities	x		s	1.386.000	2016-25
2122	Troutdale TC	Multnomah Co.	Troutdale Road Improvements	Cherry Park Road to Strebin Road	Upgrade to include bicycle and pedestrian facilities	x		\$	2,217,600	2016-25
2123	Troutdate TC	Multinomah Co.	Stark Street Improvements	257th Avenue to Troutcale Road	Widens street to five lanes	x	x	\$	3,465,000	2004-09
2124	Trouidale TC	Multhomah Co.	Halsey Street Improvements - Troutdale	238ih ta 257m	Improve Halkey Street to 3 lanes and complete boulevard design improvements	x	x	s	3,742,200	2010-15
2125	Troutdale TC	Mult. Co./Troutdale	Troutdale TC Pedestrian Improvements	Old Col: River Highway, 257th/Graham, Buxton	Improve sidewalks, lighting, crossings, bus shelters and benches	×	Y	5	115 500	2016-25
-2126	Troutdate TC	Troutdale	257th Avenue Pedestrian Improvements	Cherry Park Road to Stark Street	Improve sidewalks, lighting, crossings, bus shellers and benches	•		•	3 155 000	2204.00
2127	Troutdale TC	MultCo/Troutdale	Edgefield Station Recreational Intermodal	249th and Halsey	Develop Edgefield Station as a recreational intermodal fadility		<u> </u>		E 77E 000	2004-05
2128	Troutdale TC	Multhomah Co.	40-mile Loop Trail	223rd Avenue/Marine Drive to Troutdale town center	Study feesibility of corridor	 		3		2016-25
2131	Burnside SC	Gresham	SE 174th Avenue Bikeway	Springwater Trall to SE Stark Street	Retrofit bike lanes to existing street	×		s	23.100	2016-25
2132	Burnside SC	Gresham	Burnside SC Pedestrian Improvements	172nd, 197th, Glisan, Stark and intersecting streets	Improve sidewalks, lighting, crossings, bus shelters and benches	Y		e	7 103 250	2016-25
2133	Portland Corridor	ODOT	I-205 Shared-Use Path Crossing Improvements	Several locations	Improve access to I-205 shared-use path	^ X		s	317.625	2004-09
3000	Region	ODOT	Highway 217 Improvements	1-5 to US 26	Add capacity to existing highway	x			\$115,500,000	2016-25
3001	Region	ODOT	Highway 217 Improvements	NB - TV Highway/Canyon Road to US 26	Widen NB to three lanes; camp improvements	x	x	s	31.000.000	2010-15
3002	Region	ODOT	US 26/217 Interchange Improvement	EB US 26/SB Highway 217 Interchange	Braided ramps	x		\$	57,750,000	2010-15
3003	Region	ODOT	US 26/Jackson School Road Interchange	Jackson School Road at US 26	Construct new interchange	x	×	\$	18,480,000	2004-09
3004	Region	0007	US 217 EIS Study	1-5 to US 26	Complete planning and environmental works for Improvements in comdor	x	x.	s	5,000,000	2010-15
3005	Region	брот	US 26 Refinement and EA Study	Sylvan interchange to 185th Avenue	Complete planning and environmental work for Improvements in corridor	x	x	\$	577,500	2004-09
3006	Region	ODOT	US 26 Improvements	US 26 between Sylvan and Highway 217	Complete interchange improvements by adding third through-lane and collector distributor system from Camelot Court to Sylvan Road (Phase 3)	x	×	¢	25 410 000	2004-09
3007	Deleted (Construct	on completed)					<u> </u>	· ·	20,410,000	
3008	Region	ODOT	US 26 Improvements	Highway 217 to Murray Boulevard	Widen US 26 to six lands	- X-	x	\$	37,600,000	2004-09
3009	Region	OUOT MultCo/WashCo	US 20 Improvements Cornelius Pass Road	Murray Boulevard to Cornell Road	Widen US 25 to six lanes	<u> </u>	X .	\$	8,780,000	2004-09
3011	Region	0001	US 26 Improvements	Murray Boulevard to 185th Avenue	Widen US 26 to six lanes	x	×	> \$	28,875,000	2016-25
3012	Region	Hillisboro	Rock Creek Greenway Shared Use Path	TV Highway to Evergreen Parkway	Completes shared-use path elong Rock Creek from Tuelatin Vatley Highway to Everance Parkway	Ŷ		5	14 212 000	*portion
3013	Recipo	Various	Bronson Creek Greenway Shared-Use	Beaverion Creek to Powerline Trail	Study feasibility of corridor	×	v	e	874:000	2004.00
provinging and the second			T	Provide the second s		ACTING TO A CONTRACT OF A C	1	4	COLUMN A	COTFOR A GALAGE

Shaded projects are included in Financially Constrained System

Table 2

-

								2003 dollars	
RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	(*** Indicates phasing in financially constrained	RTP Program Years
10.55		A Contract of	Baundas Businetes Tail Complex Tail	Benning Const Constitution Family State	Plan decisioned exects of shared use path			and the second se	
3014	Region	VIEROUS -	Powerline Beaverion (Tail Contdo) (Tail	Bionson Creek Greenway to Panisnyion Noad	rian, uesigniano constructanareo-use pani	. <u>x</u>	<u>×</u>	\$ 3,118,500	2004-09
3015	Region	Various,	Beaverion Creek Greenway Confidor Study	Rock Creek to Fanho Creek Greenway	Sludy feasibility of contdor	<u> </u>	. x	5 1,500,000	2004-09
Sines	Rector	Weshington Co.	Westington County ATMS	Washington County	Acquire hardware for new traffic operations center and conduct needs energies	x y	×	\$ 12155,000	2004-09
			Beaverton Hillsdale Highway- Frequent		And the second				
3017	Region	TriMet	Bue	Beaverton-Hillsdale Highway	Improvements to enhance Frequent bus service	X	X	\$ 3,300,000	2004-09
3018	Region	TriMet	Transit center and park-and-ride upgrades	Various locations in subarea	park-and-rides throughout subarea	x	(	See Tri-Met Total	2004-25
10000				(1) Center: Cedar Hills to Hocken via	and the second				
1.1.1				(3) Millikan Way, Watson/Hall to 114th; (4)					
		Banardan	Beaverton Connectivity Improvements I:	Broadway to 115th connection; (5) Electric to	Complete protect Resources the edge of the				
3019	Bervenon RC	CORVERION	CASE TOSI	(6) Rose Blool: Westgate to Broadway: (7) 120th	Complete central perverion super connectoris	<u> </u>	<u>^</u>	\$ 19,100,000	2004-09
55 S.S.			Contraction of the second states of	Ave.: Center to Canyon; (8) 114th/115th. LRT to					
3020	Bernetinn RC	Beaverton	North/South	Tuslaway Ave;; Electric to Millikan	Complete central Beaverton street connections	y .	×	\$ 15,000,000	2004-09
			4						
		and states	2040 Centers and Station Areas				a de la parte de la companya de la c		A
3021	Region	Washington Co.	Pedestrian System Infil	Regional pedestrian system in Washington County	Fill in missing gaps in regional pedesition system	x	х	\$ . 5,000,000	2004-09
41.0	an a								
1.14		A., 1997	2040 Centers and Station Areas Bicycle						
3022	Region	Washington Co.	System Infil	Regional Doycle system in Washington County	Fill in missing gaps in regional bloycle system	<u> </u>	X	\$ 5,000,000	2004-09
		WashCo/Beaverton/		NB/SB at Walker Road, SB at TV Highway, NB/SB	Capacity increase and/or braided ramp between the highest priority interchanges identified through the				
3023	Beaverton RC	ODOT	Highway 217 Interchange Improvements	at BH Highway and at Allen Boulevard	Highway 217 Corridor study (#8009)	x		\$ 4,158,000	2004-09
3024	Region	ODOT	US 26 Improvements	Cornell Road to 185th Avenue	Widen US 26 to six lanes	<u> </u>		\$ 19,920,000	2010-15
					limited access from Murray to Brookwood and five lanes				
3025	Beaverton RC	ODOT/WashCo	TV Highway Improvements	Cedar Hills Boulevard to 10th Avenue	from Brookwood to 10th	X		\$ 38,346,000	2016-25
3026	Deleted (Construct	ion completed)							] [
3027	Deleted (Construct	ion completed)							
3028	Deleted (under con	struction)					-		
3029	Beaverton RC	Beaverton	Lomberd Improvements	Broadway to Farmington	the north with pedestrian facilities	x	Y	\$ 1848,000	2004-09
No ce de	200 <b>- 1</b>	i de la composición d	and the state and the state of the		Widen to five lanes; intersections improvements, add		and dependent		
3030	Beaverton RC	Beaverton	ramington Road Improvements	PROCKER Avenue to Multay Bouleverd	tum tanes, orke tanes and sidewalks	<u>×</u>	X	\$ 14,000,000	2004-09
3031	Beaverton RC	Beaverton	Allen Boulevard Improvements	Highway 217 to Murray Boulavard	Widen to fiva lanes	X		\$ 10,800,000	2016-25
3032	Beaverton RC	Beeventon	Cedar Hills Boulevard Improvements	Farmington Road to Walker Road	Widen to five lanes with sidewalks and bike lanes	- X	x	\$ 4,600,000	2010-15
-		Remerton	125th Avenue Extension	Brockman Stran)/Grannak to Hall Brockman	Construct two/three-tane extension with Intersection				0000
3033	Deaveron.HC	Cograting		And a second s	Construct three-lane extension with bikeways and	·····	X	3. TU,ZUU,000	2004-09.
3034	Beaverton RC	Beaverton	Hall Boulevard Extension	Cedar Hills Boulevard to Hocken	sidewalks	×	X*	\$ 5,700,000	2010-15
	a state of the second	Benjarden	Hereben Aussilia Inner Statistica Statistica	I PT in Restanton Creek	Widen to 3 lense with bike lanes and sidewalks and				artes
3035	Berverton KC	Destantion	Liverent cleaned with cleanance		LOWING MOUNTUP	<u>x</u>	X	000,000, F	2004-09
3036	Beaverton RC	Washington Co.	158th/Merio Road Improvements	170th Avenue to Walker Road	Widen to five lanes with sidewalks and bike lanes	X		\$ 4,620,000	2016-25
3037	Beaverton RC	Beaverton	Nimbus Road Extension	Hall Boulevard to Denney Roed	Extend two-lane roadway	x		\$ 10,300,000	2016-25
3038	Besvertmi RC	Beaverton	Center Street Improvementa	Hall Boulevard to 113th Avenue	Widen to three lanes with bikeways and sidewalks			\$ 3,696,000	2015-25

2004 RTP Project List October 31, 2003

						9134 BTP	2003 dollars		
				r.		2025 MTP	Phancially	phening in	RTP
RTP#	20142) i iste	Jurfadiction	Project Name (Facility)	Project Location	Project Concreption	Preferrez System	Constraines System	financially constrained	Yeare
				The second s	<ul> <li>Service Tublic Length Lengther and Family Inform Print.</li> </ul>	¥ -		s	
			n i programme de la companya de la Esta de la companya d Esta de la companya d						
i mati			n ar an an an an ann an ann an ann an ann an a	An Converte Collige Interaction		X	- X	1. <u>51000</u>	m
						x			
****	Basasser, Gr	Successive Washing	Winker Frant Inschmentents	Cector Hills Boulevard to Mumay Boulevard	Withen to anywri hannes with externation and brine income	"E		* 31 #75 nos	3146.95
	Hill advantation of the "The		influence in the second se						22 ND 12-
<u>. 3046</u>					er gescheiden gehannte des Einder einen der eine eine eine eine eine eine eine ei	2,	X		
. 9947								s	2765.03
ange	- D <del>arat</del> in PE		i ya da da dagi da angi sa ang Nga ang da da mangi sa angi sa a	A CONTRACTOR OF A CONT	obelanian berches	2	X .	1 12062	<b></b>
3056	<b>Benefiting RC</b>	Essweiten WestvCor Triblet	Weiker Road Pedesitian Improvements	Polishy/1020 to Highway 217	improve sidesces, agrong, ordesings, but sherers and benches	X		1 115.550	2516-25
		Wallio Careford	Real and state of the state of the second state of						i i kan is
					ti din kanan ing kanan manan kanan kan Manan kanan kana			•	20472
					The second second has been been a second side.		X	*	200-22
3954	<b>Demotion</b> RC	Weddington Co.	lingations marits	Scholis Ferry Road to TV Highway	internections, fill in Lénycle autourk gaps	N		\$ \$77,500	2016-25
3055			inen Arhunder, en der nährte Artung Artensen singer Ander State von State der State	a na antara ang ang ang ang ang ang ang ang ang an		y.	K	1 12.127,500	1 351a.25
		rener	Carryon Rick/TV Fighway Bits and References managements	5W Gist Average to Matemay 217	Rike Jones, sideopõios profinationation cryssings	v		4 + 660, 576	-144-4-1.44
					Ingran Stevens, crossing and it in boards where the	.е. С		12 1.0002/0110	20-0-20 1
				i a mila mante di content a tri cont	FITTE THE REPORT OF THE AND A DESCRIPTION OF THE ADDRESS OF THE ADDRES		( ), Y.		
9556			ne neunan sa mang senang mang senang sa			, x	<b>X</b>	\$ 200,000	
3960	<b>Berrenton</b> RC	OBOINMentCo	TV Highway Azzasa Maragament	1175h Avenue to Hilstory	Access (Tatagathan)	ж		\$ 17,325,000	2010-15
				The Highware from Highware 217 to 2007h	in direction and a sign of a provident of the second second second second second second second second second s	×	E E		* 201-10
								•	
		i and and a state of the second s	n and an	an ann an 1995. 'S minne fra 1995 i Reserves	ander and an and a state of the second			a ann ann	
2.75	Edeversit Corner	n halatina ngasa satas Litat kan ngasa satas							Endered I
201			i hining di Matania ang Makadi tang di tang di kaling di Kaling di Kaling di Kaling di Kaling di Kaling di Kali		in philippi destations in the second state in the second state of the second state in the second state of the s		, A		205-15
3068	Bennetten Contder	Washington Co.	Gantier: Honselit2nd Aserase improvements	Allen Sculewrd is Cleven Road	Widen to itoma larens with teknowyn and aldzwellus	Ц		\$ 5,1197,600	2018-25
3060	Beswetten Comfor	Washington Co.	Schola Ferry Read Insprovements	Garden Home Road to Hamilton Street	Whiten to three larges with aktreality and blue targes	X		5 9,240,002	2016-25
<b>1</b> 171	i Berter			Chainsteat Inn Is Scholt Fairs Folis	- Serients form Cast States in the state of the	Y.	[] . X	£	0.000
3677				Fernington Road to Constant way road	To end ford mail war are the reader the selection of	x	<b>y</b> .	s parter	
3573	Basvetter Contine	Washington Co.	Barres Road Bitemey	Gunside to Lewity Road	Retrofit to include tike lanos	x		s srr.sm	2014-24
					Statel and she saw about the same at				
					- The second is a start construction and the				, mant
3675	Carlena Contra					x	<b>.</b>		. i meto
<b>1</b> 35 6	Ömyrigi Canar	tomata 🗍					L., K.	1 nd55.07a	l yyncas

			1				·····	2002 dellare	
PTP #	204014-6	Indediction	Project Name (Facility)	Brolect Location	Brolest Description	2025 RTP Preferred	2025 RTP Financially Constrained	( "*" indicates phasing in financially	RTP Program
	2040 Link	Bequerton	Western Avenue Redestrian Improvements	Eth Street to 800 feet could of Eth Street	Improve sidewalks, lighting, crossings, bus shelters and	System	System	Constrained	Tears
3077	Beaverton Comdor	Deaventon	Canyon Road Bicycle and Pedestrian			X		\$ 55,440	2016-25
3078	Beaverton Corridor	0001		US 26 to 110th Avenue	Retrofit to include bike lanes/sidewalks	×		\$ 15,592,500	2010-15
3079	Beaverton Corridor	Beaverton	Allen Boulevard Bike/Ped Improvements	Western Avenue to Scholls Ferry Road	Retrofit to include bike lanes and fill in missing sidewalks	X	X	\$ 320,000	2010-15
3082	Beaverton IA	Beaverton	Western Avenue Bike Lanes	B-H Highway to Allen Boulevard	Retrofit to include bike lanes	x		\$ 360,000	2016-25
3083	Westside SC	Washington Co.	170th Improvement	Blanton Street to Farmington Road	Widen to five lanes with sidewalks and bike lanes	x		\$ 9,240,000	2016-25
3084	Westside SC	Washington Co.	170th Improvement	Alexander Road to Merlo Road	Widen to five lanes with sidewalks and bike lanes	x		\$ 9,240,000	2016-25
3085	Deleted (Construct	ion completed)							
3086	Westside SC	Washington Co.	158th Avenue Improvements	Walker to Jenkins Road	Widen to include bike lanes	x		\$	2016-25
3087	Westside SC	Beaverton	Millikan Way Improvements	TV Highway to 141st Avenue	Widen to five lanes with sidewalks and bike lanes	x		\$ 5,000,000	2016-25
3088	Westside SC	Beaverton	Millikan Way Improvements	141st Avenue to Hocken Road	Widen to three lanes with sidewalks and bike lanes	x		\$ 3,700,000	2016-25
3089	Westside SC	Washington Co.	160th Avenue Improvements	Tualatin Valley Highway to Farmington Road	Widen to five lanes with sidewalks and bike lanes	x		\$ 2,310,000	2016-25
3090	Westside SC	Washington Co.	Walker Road Improvements	173rd to Stucki Boulevard	Widen to include bike lanes	x		\$ 866,250	2016-25
3091	Westside SC	Hillsboro	Quatama Street Improvements	205th Avenue to 227th Avenue, 227th at Baseline	sidewalks and bike lanes	x	×	<b>\$</b> 9,436,350	2010-15
3092	Westside SC	Washington Co.	Powerline/Rock Creek Trail	Bethany/Kaiser Road to Evergreen Road/Rock Creek Greenway	Construct shared-use path for blcyclists and pedestrians just north of US 26	×	x	\$ 1,155,000	2004-09
3093	Westside SC	Washington Co.	Murray Boulevard Bikeway	Farmington Road to S of TV Highway	Retrofit to include bike lanes	x		\$ 231,000	2016-25
3094	Westside SC	Hillsboro	Cornell Road Bikeway	Etem Young Parkway (W) to Ray Circle	Retrofit to include bike lanes	x	x	\$ 884,730	2004-09
3095	Westside SC	Washington Co.	170th Avenue Pedestrian Improvements	Merio Orive to Elmonica light rail station	Fill in sidewalk gaps and extend to light rail eastside only	×	×	5 311.850	2004-09
3096	Deleted (included in	n Project #3021)							
3097	Westside SC	Washington Co.	Baseline Road Pedestrian Improvements	158th Avenue to 166th Avenue	Improve sidewalks and pedestrian crossings	×		\$ 110.880	2016-25
3098	Westside SC	Washington Co.	Walker Road Bike/Ped Improvements	Canyon Road to Cedar Hills Boulevard	Retrofit to include bike tanes and sidewalks	X	x	\$ 866,250	2016-25
3099	Hillsboro RC	Hillsboro	1st Avenue/Glencoe Road	Lincoln Street to Evergreen Road	Widen to three lanes with skiewalks and bike lanes	x	x	\$ 4,467,000	2016-25
3101	Hillsboro RC	Hillsboro	Jackson School Road Improvements	Evergreen Road to Grant Street	Widen to three lanes with sidewalks and bike lanes	x		\$ 5,162,850	2016-25
3102	Hillsboro RC	Washington Co.	Baseline Road Improvements	201st to 231st Avenue	Widen to three lanes with bike lanes and sidewalks	x	x	\$ 24,255,000	2004-09
3103	Hillsboro RC	Washington Co.	Baseline Road Improvements	Murray Boulevard to Brookwood Parkway	Widen to five lanes with bike lanes and sidewalks	x		\$ 6,930,000	2016-25
3104	Hillsboro RC	Hillsboro	NW Aloctek Drive Extension	NW Amberwood Drive to Comellus Pass Road	New three-lane facility with sidewalks and bike lanes	x	x	\$ 2.948.715	2004-09
3105	Hilistoro RC	Hillsboro	E/W Collector	185th Avenue to west of Cornelius Pass Road	New 3-lane facility	Y.	y	\$ 6,781,005	2004-09
	111 111		and the state of the		and the second			<u> </u>	200400
3106	Hillsboro RC	Washington Co.	229th/231st/234th Connector	Lois Street to Dogwood Street	New 3-lane fadility and bridge	x	×	\$ 24,300,000	2004-09
					Widen to five tanes, including bridge, sidewalks and bike lanes (sidewalk on easistice and bike tanes only in				100 m 50 m 50 m
3107	Westside SC	misboro/WashCo.	SW 200th Avenue Improvements	LRT to Baseline Road	financially constrained system)	X	X	\$ 7,078,685	2010-15
3108	Deleted (Constructi	on completed) ODOT/WashCo/		······································	Improve primary access route from regional center to US	-			
3109	Hillsboro RC	Hillsboro	Hillsboro to US 26 Improvements	Shute Road/Cornell Corridor	26	x		n/a	2016-25
3110	Deleted (Constructi	on completed)							

RTP #	2040 Unk	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP 2025 RTP Preferred System System		2003 dollars ( """ Indicates phasing in financially constrained	RTP Program Years
3111	Hillsboro RC	Washington Co.	First Avenue Improvements	Grant Street to Glencoe High School	Improve sidewalks and pedestrian crossings and make transit improvements	×	x	\$ 808,500	2004-09
3112	Hilisboro RC	ODOT: N	First Avenue Improvements	Oak Street to Baseline Street	Rechannelize NB and SB to provide protected left turn lanes and signel phasing at 1st/Oak and 1st/Baseline	x	x	\$ 190,575	2004-09
3113	Hillsboro RC	Hillsboro	10th Avenue Improvements	Main Street to Baseline Road	Add right turn lane and widen sidewalk	x	x	\$ 1,915,000	2004-09
3114	Hillsboro RC	Hillsboro	NE 28th Avenue Improvements	Grant Street to East Main Street	Widen to three lanes with sidewalks, bike lanes, street, lighting and landscaping	x	x	\$ 3,191,000	2004-09
3115	Hillsboro RC	Hillsboro	10th Avenue improvements	Washington Street to Main Street	Widen to provide third NB through lane	<b>x</b> .		\$ 734,000	2010-15
					Construct one additional NB turn lane and rechannelize WB Baseline Street approach to 10th Avenue to provide				
3116	Hillsboro RC	Hillsboro	10th Avenue Improvements	Walnut Street to Baseline Street	two approach lanes	x		\$ 2,255,715	2010-15
3117	Hillsboro RC	Hillsboro	East-West Connector	Brookwood Parkway to 28th Avenue	Extend Grant Street beyond 28th Avenue with a new 3- lane facility	x		\$ 9,061,600	2016-25
2110	Wileborn PC	Hilsborn	Tualatin Valley Highway/Brookwood	Tuelatin Vallay Highway at Brookwood Avenue	Hazel intersection and roadway improvements to Alexander Street	<b>,</b>		E 40.000.000	2010.05
3119	Hillsborn RC	ODOT	TV Highway Improvements - Hillsboro	Shute Park to Baseline/Oak Street to Tenth	Complete boulevard design improvements	× ×	<u>^</u>	\$ 2,310,000	2010-23
3120	Hillsborn RC	ODOT/Wash, Co.	TV Highway Pedestrian Improvements	10th to Cornelius Pass Road	Improve sidewalks, lighting, crossings, bus shelters and benches	×		\$ 0.5% 500	2016.25
5120	This boro TKG				Study to define access management strategy and define needed improvements for motor vehicle, truck transit	^		\$ 9,586,500	2010-23
3121	Region	ОДОТ	TV Highway Corridor Study	Highway 217 to downtown Hillsboro	bike and pedestrian travel in the corridor	X		\$ 1,732,500	2004-09
3123	Hillsboro RC	TriMet/Hillsborg	Hillsboro Regional Center TMA Startup	Hillsboro Regional Center	program with employers	x	×	\$ 200,000	2004-09
3124	Hillsboro RC	ODOT	TV Highway System Management	209th Avenue to 10th Avenue	Interconnect signals	x		\$ 1,732,500	2004-09
3126	Sunset IA	Washington Co.	Cornelius Pass Road Improvements	TV Highway to Baseline Road	Widen to five lanes including sidewalks and bike lanes	<u> </u>	- X	\$ 5,775,000	2010-15
3127	Hillsboro Corridor	WashCo	Hilfsboro RC Pedestrian Improvements	18th, 21st, Oak, Maple and Walnut streets	benches	х	x	\$ 1,914,500	2004-09
3128	Hillsboro RC	Washington Co.	Cornell Road Improvements	Arrington Road to Main Street	Widen to five lanes	X	x	\$ 6,930,000	2016-25
3129	Deleted (Outside N	letro Planning Area E	Boundary)						
3130	Deleted (Construct	ton completed)							
3131	Sunset IA	Washington Co.	Evergreen Road Improvements	25th Avenue to 253rd Avenue	Widen to five lanes including sidewalks and bike lanes	X	x	\$ 4,679,500	2004-09
3132	Deleted (Construct	tion completed) Washington Co./	Cornellus Pass Road Interchange		Construct full diamond Interchange and southbound	-			
3133	Sunset IA	ODOT	Improvement.	US 25/Comelius Pass Road	auxiliary lane to facilitate traffic flows on and off US 26 Widen to three lanes including sidewalks, bike lanes and	X	x	\$ 5,775,000	2004-09
3134	Sunset IA	Washington Co.	Cornellus Pass Road Improvements	TV Highway to Baseline Road	signals at Johnson and Francis	X	x	<b>\$</b> 10,395,000	2004-09
3135	Sunset IA Deleted (Construct	ilon completed)	Comellus Pass Koed Improvements	Baseline Road to Alociek Drive	widen to five lanes including sidewalks and bike lanes	X	x	\$ 17,325,000	2004-09
3137	Sunset IA	Washington Co.	Brookwood Avenue Improvements	TV Highway to Baseline Road	Widen to three lanes including sidewalks and bike lanes	×	x	\$ 8,662,500	2004-09
3138	Deleted (Construct	ion completed)							
					Construct two-tane new overcrossing with sidewalks and bike lanes to better connect areas north and south of US				1.26.6
3139	Sunset IA	Hillsboro	US 26 Overcrossing - Sunset IA	NW Bennett Avenue to NW Wagon Way	26	×.	X	\$ 6,633,743	2016-25
3140	Sumset IA	Hillsboro	229th Avenue Extension	NW Wagon Way to West Union Road	New three-fane facility with sidewalks and bike lanes	x	X	\$ 2,867,800	2010-15
3141	Sunset IA	washington Co.	T/Utiv1/3rd Improvements	Baseline to Walker	Improve to 3 lanes	X	X	\$ 6,352,500	2010-15

RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	200 ("*" ph fin con	3 doliars indicates asing in ancially instrained	P	RTP Program Years
					Three lane extension (two lanes west bound and one lane eastbound with tum lanes), including bike lanes and						
3142	Sunset IA	Washington Co.	Johnson Street Extension	170th Avenue to 209th Avenue	sidewalks	X		\$	1,155,000	2	2004-09
3143	Sunset IA	Washington Co.	Walker Road Improvements	Cedar Hills to 158th Avenue	Widen to five lanes including sidewalks and bike lanes	X	X	\$	23,100,000	2	2010-15
3144	Sunset IA	Washington Co.	Walker Road Improvements	158In Avenue to Ambergien Parkway	Widen to five lanes including sidewalks and bike lanes	X	x	\$	11,550,000	2	2010-15
3145	Sunset IA	Washington Co.	Walker Road Improvements	Highway 217 to Cedar Hills Boulevard	Widen to five lanes including sidewalks and bike lanes	x		\$	30,607,500	2	2016-25
3146	Sunset IA	WashCo/Hillsboro	Cornelius Pass Intersection Improvements	Intersection at Quatama	Improve Quatama/Cornelius Pass Road intersection	x		\$	577,500	2	2016-25
3147	Sunset IA	Hillsboro	25th Avenue Improvements	Cornell Road to Evergreen	Widen street to three lanes with bike lanes	х	x	\$	2,553,000	2	2010-15
3148	Beaverton RC	Washington Co.	Walker Road Improvements	Highway 217 to Cedar Hills Boulevard	Widen to three lanes including sidewalks and bike lanes.	x	x	\$	9,240,000	2	2010-15
3149	Sunset IA	ODOT/Washington Co.	Shule Road Interchange Improvements	Shute Road and US 26	Construct westbound to southbound loop and diagonal ramps each direction	x	x	s	6,382,000	2	2004-09
3150	Sungat (A	Washington Co.	Cornell Road System Management	10th Avenue to Multhomah County line	Upgrade traffic controllers and install CCTV cameras and monitoring stations	x	×	5	800.000	1,	2004-09
5150		TdNot	US 26 Corridor TDN Program	Sunsat Industrial Area	Implements a transportation management association	~		c	1 501 500		2016 25
3151	Sunset IA					<b>^</b>		\$	1,501,500		2010-25
3152	Deleted (Project co	mpleted)			Extend easterly from Thatcher Road to Sunset Drive			Sec. Subarts		t	
					lanes at major intersections, traffic signal at 47 and bike						
3153	Forest Grove TC	Forest Grove	Uavid Hill Road Connector	Thatcher Road to Highway 47 (Sunset Drive)	Iaries	Х	X	5	7,165,000	2	2004-09
3154	Deleted (Construct	ion completed)		Highway 47/Elm Street and Highway 47/Maple						+	
3155	Forest Grove TC	ODOT Forest Grove/	Highwy 47 Traffic Signals Forest Grove-Cornelius Industrial	Street	Add traffic signals at Elm and Maple streets	X		\$	500,000	2	2004-09
3156	Forest Grove TC	WashCo.	Connector	Yew to Holladay	Two-lane improvements parallel to TV Highway	X		\$	1,440,000	2	2010-15
3157	Forest Grove TC	Washington Co.	Sunset Drive Improvements	University Avenue to Beat Road	sidewalks	×	×	S	6,954,000	2	2004-09
3158	Forest Grove TC	Washington Co.	Martin Road/Cornelius-Schettilin Road	Forest Grove northern UGB to Roy Road	Realign with widened paved shoulders Martin Road and Comellus Schefflin Road	x	x	\$	14,206,500	2	2004-09
3159	Forest Grove TC	ODOT/Forest Grove	Highway 8 Improvements - Forest Grove	B <sup>1</sup> Street to Cornelius city limits	Complete boulevard design improvements (OTIA project in FC)	x	x	s	9,240,000	* ;	2010-15
3160	Forest Grove TC	Washington Co.	Verboort Road Intersection Improvement	at Highway 47	Intersection safety improvement	x	x	S	231,000	1;	2010-15
-		Formet Groue	Gales Creek Road Intersection	et Thatcher Road	Realign intersection to increase capacity	v			4 400 650	-	2016 25
3161	Deleted (Included I	n Project #3159)				<b>^</b>		\$	1,420,650		2010-25
2482	Formet Comun TC	ODOT/Forest Grove	Errest Grove TC Padestrian Improvements	TV Highway, Pacific, 19th, College, Sunset, "B" and Intersection streets.	Improve sidewalks, lighting, crossings, bus shelters and banches	×	v	r	3 463 334		2054.00
3103	FOIBSLOIDVE TO	Titten	Tuttilehumu Emmunet Dur	Forest Grove to Hillsdale via TV Highway and B-H		~			2,403,234		2004-03
3164	Forest Grove (C		Highway A7/Outince Street	Tuelatin Valley Highway/Ouince St. Interrection	Modify traffic signal and add turn lange at Outpoe Street	<b>X</b>	A	5	1,575,000	<u> </u>	2004-25
3105	Forest Grove TC	0001	Michael Bluther ation Description		Increase turning radii, and protected turn lanes, and	^		ې ۲	1,000,000		2016-25
3168	Cornellus	Comellus/ODOT	10th Avenue	at Baseline and Adair	and improve pedestrian and vehicle safety	x	x	\$	879,000	2	2004-09
			energian de College et		Avenue/20th Avenue at Highway 8, Improve S, 20th	- 10 A					
3167	Comellus	Comelius/ODOT	nignway 6 intersection Realignment - 19th/20th Avenue	Intersection of 19072000 Avenue and Highway B at Initiation of couplet	to RR crossing north of N. Davis)	×	×	5	3,100,000	3	2004-09
					Intersection geometry improvements and conversion of pedestrian signal to full mode signalization for improved			1.0			
3168	Comelius	Comellus/ODOT	Highway 8/14th Avenue Intersection	Intersection of 14th Avenue at Highway 8 couplet (Adair and Baseline)	Main Street District circulation and improved pedestrian safety on Adair and Baseline streets	` x	- x	\$	450,000		2004-09

			· · · · · · · · · · · · · · · · · · ·							
RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( *** indicates phasing in financially constrained		RTP Program Years
	a de la construcción	and the second second	and the second		Complete Soulevard design improvements to Baseline,	141.4 (M)		and the second		
		al an in the			pedestrian alley within the Adair/Baseline couplet in Main	and the second second				
3169	Comellus	Comellus/ODOT	Main Street Couplet Improvements	Highway 8 couplet from 10th to 19th Avenue	Street District	x	×	\$ 6,930,00		2004-09
3170	Cornelius	Comellus/ODOT	West Couplet Enhancement	1st Avenue to 10th Avenue	Complète boulevard design improvements	X	x	\$ 3,465,00	2	2010-15
3171	Comellus	Comellus/Wash Co.	North Devis Street Reconstruction	19th Avenue to 10th Avenue	Reconstruct street to urban standards	X	x	\$ 1,600,00		2010-15
3172	Forest Grove TC	Forest Grove	23rd/24th Avenue Extension	Hawthome Ave. to Quince St. (Hwy: 47)	Hawthome	×	×	\$ 2,782,00	5	2004-09
3173	Sunset TC	Washington Co.	US 26 Undercrossing - Sunset TC	Barnes to Butner west of Highway 217	Construct new underpass to better connect areas north and south of US 26	×		\$ 11,550,00		2016-25
3174	Sunset TC	Washington Co.	Barnes Road Improvements	Miller Road to 84th Avenue	Widen to three lanes with bike lanes and sidewalks	x		\$ 4,966,50		2016-25
3175	Sunset TC	Washington Co.	Barnes Road Improvements	Highway 217 to 119th Avenue	Widen to five lanes with blke lanes and sidewalks	x		\$ 7,161,00		2010-15
3176	Sunset TC	Washington Co.	90th/98th Avenue Extension	Leahy Road to Barnes Road	Construct new two-lane road connection with bike and pedestrian facilities	x		\$ 1,732.50		2016-25
3177	Sunset TC	Washington Co.	Cedar Hills Boulevard/Barnes Road	Cedar Hills at Barnes Road	Add through and turn lanes, new traffic signal and signal at US 26 EB off-ramp	Y		\$ 2,079,00		2004.00
		Werbington Co	Wasthman Road Cathurals	Marilana In Soringmenet	Constructs off-road pathway to improve bloyde and			\$ 2,075,00		2004-05
31/8	Sumset IC	Tradiningion Cos	Woodiaver, (Cood ) ad maya	Workson an Openguase	benearier, errear in coultar nation could	X	X	<b>\$</b> 577,50		2010-15
		Machington Co.		Remos Read to Comell Read	14/1 dam to the state of the st					
3180	Sunset IC	wasington co.	Cornell Road Improvements - West Cedar		Widen to three/five lanes with sidewalks and bike lanes	X		\$ 3,003,00	)	2010-15
3181	Cedar Mill TC	Washington Co.	Mill Comell Road Improvements - West Cedar	US 26 to 143rd Avenue	Widen to five lanes with bike lanes and sidewalks	X		\$ 3,465,00		2016-25
3182	Cedar Mill TC	Washington Co.	Mill	143rd Avenue to Murray Boulevard	Widen to five lanes with boulevard design treatment	X	X	\$6,930,00	>	2016-25
3183	Cedar Mill TC	Washington Co.	Cornell Road improvements	Murray Boulevard to Saltzman Road	Widen to three lanes with bikeways and sidewalks	X	x	\$ 9,200,00	<b>y</b>	2004-09
3184	Cedar Mill TC	Washington Co.	Mill	Saltzman to Miller Road	shelters	x		\$ 12,705,00		2016-25
3185	Cedar Mill TC	Washington Co.	Barnes Road Improvement	Saltzman Road to 119th Avenue	Widen to five fanes with intersection improvement at Saltzman	×	×	\$ 6,121,50	<b>)</b>	2004-09
3186	Cedar Mill TC	Washington Co.	Murray Boulevard Improvements - Cedar Mill	Science Park Drive to Comell	Widen Murray Boulevard to five lanes and improve Cornell/Murray Intersection	x	×	\$ 12,000,00		2004-09
3188	Cedar Mill TC	Washington Co.	Seltzman Road Improvements	Cornell Road to Thompson Road	Widen to three lanes with sidewalks and blke lanes	Y	v v	\$ 19,000,00	,	2004-00
3189	Deleted (included in	Project #3188)						* (o)000100		2001.00
3190	Cedar Mill TC	Washington Co.	143rd Avenue Improvements	Cornell Road to West Union Road	Widen to three lanes with sidewalks and bike lanes	¥		\$ 5,775,00		2010-15
3191	Deleted (Project in	cluded in other proje	ects on list)					• 0,,,,0,		2010 10
2402	Codes Mill TO	Weshington Co.	Cedar Mill Town Center Local Connectivity, Phase 1	Marious locations in the town carter	Construct additional local road connections to improve					
2102	Deleted (included in	- Project #2182\			ability on deside of the	λ		\$ 1,155,00		2004-09.
3193	Deleted (mciuded i	n Project #3 1833						·····		
3194		Westington Co	A design of the second s							
્યાઝ	Cedar Mill 1C	· wasnington Co.	Saizman Pedestrammprovements	Wateries Koso to Doffwood Hoad	Construct sidewarks on west side of road	· X	X	\$ 560,17	5	2004-09
			Bethany Boulevard Improvements, Phase				-		+	
3197	Bethany TC	Washington Co.	1 Bethany Boulevard Improvements, Phase	Bronson Road to West Union Road	Widen to three lanes with bike lanes and sidewalks	X	X	\$ 5,775,00	<u>)</u>	2004+09
3198	Bethany TC	Washington Co.	2	Bronson Road to West Union Road	Widen to five lanes with bike lanes and sidewalks	x		\$2,310,00		2016-25
3199	Bethany TC	Washington Co.	West Union Road Improvements	143rd Avenue to Cornelius Pass Road	Widen to three lanes, including sidewalks and bike lanes	x		\$ 17,325,000		2016-25
3200	Bethany TC	Washington Co.	Kaiser Bikeway	West Union to Springville Road	Widen to include bike lanes	x		\$ 739,20		2016-25

### Public Comment Draft 2004 RTP Project List

g				UCIODER 31	2003				
RTP #	2040 Unk	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( "*" indicates phasing in financially constrained	RTP Program Years
3201	Bethany TC	Washington Co.	Kaiser Road Pedestrian Improvements	Bronson Creek to Springville Road	Improve sidewalks, lighting, crossings, bus shelters and benches	x	-	\$ 577,500	2016-25
3202	Bethany TC	Washington Co.	West Union Road Improvements	185th Avenue to Cornellus Pass Road	Widen to five lanes including sidewalks and bike lanes	x			2016-25
3204	Tanasbourne TC	Washington Co.	Tenasbourne	179th Avenue to Bethany Boulevard	Widen to five lanes with sidewalks and bike lanes	x	x	\$ 6,600,000	2010-15
3205	Tanasbourne TC	Washington Co.	173rd/174th Undercrossing	Cornell Road to Bronson Road	Construct new two lane undercrossing with sidewalks and bike lanes	x		\$ 17.094.000	2016-25
3206	Tanasbourne TC	Washington Co.	Thompson Road Improvements	Bronson Creek Drive to Saltzman Road	Widen to three lanes with sidewalks and bike lanes	x		\$ 2,310,000	2016-25
3207	Tanasbourne TC	Washington Co.	185th Avenue Improvements	Improve 185th Avenue and Cornell Road with "boulevard" design treatment, including improved sidewaiks and bus stops, curb extensions, street trees, lichting, etc., within the town center.	Complete boulevard desirin improvements	v		¢ (200.000	
2208	Tonnahauma TC	Washington Co	Teneshourne TC Padastrian Impressmente	Compl. Electron State and International	Improve sidewalks, lighting, crossings, bus shelters and	×		\$ 4,620,000	2016-25
3200		Westington Co.	Carlastic Deed Padation Inc.	contain every source way and an executing streets	Improve sidewalks, lighting, crossings, bus shelters and	X	X	\$ 231,000	2016-25
3209	Tanasbourne TC	wasnington Co.	Springville Road Pedestrian Improvaments	Kalser to 185th	benches Improve sidewalks, lighting, crossings, bus shelters and	X		\$ 577,500	2016-25
3210	Tanasbourne TC	Washington Co.	185th Avenue Pedestrian Improvements	Westview HS to West Union Road	benches Widen to five lanes: complete boulevard design	x		\$ 51,975	2016-25
3214	Farmington TC	Washington Co.	Farmington Road Improvements	172nd Avenue to 185th Avenue	improvements	x		\$ 11,550,000	2016-25
3215	Farmington TC	Washington Co.	Kinnaman Road Improvements	Farmington to 209th Avenue	bikeways and sidewalks	x		\$ 6,006,000	2016-25
3216	Familington TC	Washington Co.	185th Avenue Improvements	TV Highway to Bany Road	Widen to three lanes	x	x	\$ 9,240,000	2010-15
3217	Farmington TC	Washington Co.	Farmington Road Improvements	185th Avenue to 209th Avenue	Widen to three lanes	x	x	\$ 10,000,000	2010-15
3220	Aloha TC	WashCo/ODOT	Aloha TC Pedestrian Improvements	streets	benches	x		\$ 1,155,000	2016-25
3221	Beaverton Corridor	Washington Co.	Kinnaman Road Pedestrian Improvements	Farmington to 198th	benches	x		\$ 231,000	2016-25
3223	Beaverton Corridor	Washington Co.	185th Avenue Improvements	Tualatin Valley Highway to Kinnamon Road	Widen to five lanes with sidewalks and bike lanes	×		\$ 8,085,000	2016-25
3224	Deleted	l							
4000	Deleted (Construct	ion completed)	i			x			
4001	Region	TriMet	Killingsworth Frequent Bus	Swan Island to Clackamas TC	service.	×	x	\$ 4,540,000	2010-15
4002	Region	ODOT	I-5 Interstate Bridge and I-5 Widening - RO	I-5/Columbia River to Columbia Boulevard	Acquire right-of-way	x		\$ 20,000,000	2004-09
4003	Region	ODOT	I-5 Interstate Bridge and I-5 Widening	I-5/Columbia River to Columbia Boulevard	Improve I-5/Columbia River bridge (local share of joint project) based on recommendations in I-5 Trade Comdor Study	×		\$ 231,000,000	2004-09
4004	Region	ODOT	15 Reconstruction and Widening	Greeiey Street to 1-84	Loyd District and Rose Quarter (Groeley ramp improvements in financially constrained system)	· •	~	£ 106 260 000	* 2004.00
4005	Region	ODOT	125 North Improvements	Lombard Street to Expo Center/Delta Park	Widen to six lanes	×		\$ 100,200,000	2004-09
4006	Region	ODOT	1-5/Columbia Boulevard Improvement	1-5/Columbia Boulevard interchange	Construct full direction access interchange based on recommendations from I-5 North Trade Corridor Study		×	<ul> <li>C (1,000,000)</li> </ul>	2004-05
4007	Region	Multriomah Co.	Sauvie Island Bridge Replacement	Sauvie Island Bridge	Replace substandard bridge		Ŷ	\$ 31,000,000	2010-15
4008	Region	Metro/ODOT	I-205 North Corridor Study	Highway 224 to Vancouver, Wa.	Develop traffic management plan	×		\$ 1 155 000	2010-15
4009	Region	ODOT	I-5 Trade Corridor Study and Tier 1 DEIS	1-405 (OR) to I-205 (WA)	Plan Improvements to I-5 to benefit freight traffic	x	x	\$ 15,000,000	2010-15
4010	Columbia Corridor	Portland	Columbia Boulevard Seismic Retrofit	Columbía Boulevard bridge at Taft Avenue	Seismic retrofit project	x		\$ 415.800	2016-25
4011	Columbia Corridor	Portland	NE Marine Drive Bikeway	NE 6th to 33rd Avenue and Gantenbein to Vancouver Way	Retrofit blke lanes to existing street, off-street paths in missing locations	X	×	5 519 750	2004.09

.

·····		T						2003 dollars	T
RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained Sy <del>s</del> tem	(""" indicates phasing in financially constrained	RTP Program Years
		1		Six signals: at junction, MLK: interstate, Greeley,	communications intrastructure, closed circuit 1v cameras, variable message signs for remote monitoring				
4012	Columbia Corridor	Portland	N/NE Lombard/Killingsworth ITS	Portsmouth and Philadelphia/Ivanhoe	and control of traffic flow	X	x	\$ 242,550	2010-15
					Refine long-term improvements as defined in the Columbia Corridor Study to consider additional TSM and				
4013	Columbia Corridor	ODOT/Portland	US 30 Bypass Phase i Refinement Study	1-5 to 1-84	access management	x		n/a	2004-09
	O-humble Oranda	ODOT/Portland	Northeast Portland Highway Study	Columbia/Lombard - 1-5 to 115-30	Define long-term improvements and primary freight strategy in contider	v		¢ 577.600	2016.05
4014	Columbia Comdor	ODO 1/1 Ortiana	Hourieast Portante Highway Olday	Columbia Blvd. to US and Lombard/MLK and	Improve transition of freight movement from Lombard to	<u> </u>		\$ 577,500	2010-25
4015	Columbia Corridor	ODOT/Portiand	US-30 Bypass Improvements Study	Columbia/MLK intersections	Columbia and from Columbia to US 30	Х		\$ 1,155,000	2004-09
4016	Columbia Corridor	ODOT/Metro	North Willamette Crossing Study	US 30 to Rivergate north of St. Johns	Study the need for a new bridge from US-30 to Rivergate	x		\$ 1,155,000	2016-25
4017	POXIA	Port	SW Quad Access	33rd Avenue	Provide street access from 33rd Avenue into SW Quad		x	\$ 1,732,500	2004-09
					Improve access from Columbia Boulevard to 33rd				
4018	PDX IA Port/Portland		Columbia/Lombard Street Crossover	at 33rd Avenue	Avenue to the north for air cargo-related development	×	· · · · ·	\$ 8,778,000	2016-25
4019	PDX IA Port/Portland		Lightrail station/track realignment	Portland International Center	Construction of light rail station	x		\$ 14,000,000	2004-09
4020	Deleted (Construct	ion completed)							
	(	,							110000
	and the second second	and the state of the state of the	the second below of the second second second					Contractor (Second and Second	
4021	PDX IA	Port	Airport Way Improvements, West	82nd Avenue to PDX terminal	Widen to three lanes in both directions	X	X	\$ 11,550,000	2010-15
		1.	A set of a s		Provide free flow connection from Columbia				
4022	PDX IA	Portland/Port	East Columbia/Lombard Street Connector	Columbia/US 30 Bypass: NE 82nd Avenue to I-205	boulevard/sznd Avenue to US 30 Sypass/1205	x	×	\$ 28,865,250	2004-09
4023	PDX IA	Port	Marx Drive Extension	Marx Drive to 82nd Avenue	Extend Marx to 82nd Avenue	<u> </u>		\$363,825	2010-15
4024	Deleted (Construct	ion completed)							
4025	Deleted (Construct	ion completed)							
199							1.6		
4026	PDXIA	Port/Portland	Cascades Parkway Connection	Cascades Parkway to Alderwood Road	Construct two-lane extension	X-	×	\$ 1.732.500	2004-09
4027	Deleted (Construct	1							
	Deleted (Colistituci	(in completed)		and the second				- Indonesia	
4028	PDX IA	Port	Airport Way/82nd grade separation	82nd Avenue/Airport Way	Construct grade separated overcrossing	x	x	\$ 12,705,000	2010-15
					Communications infrastructure; closed circuit TV cameras, variable message signs for remote monitoring		S. 199		
4029	PDX IA	Portland	PDXITS	Traffic signalization	and control of traffic flow	X	X	\$ 11,895,000	2010-15
4030	PDX IA	Portland	NE 11-13th Avenue Connector	NE 10/13th Avenue at Columbia Boulevard	New Inree-lane roadway and bridge	<u>x</u>	X	\$ 9,326,625	2004-09
4031	PDX IA	Port	Arport Way return and Exit Roadways	Airport Way	relocate Airport Way exit roadway and construct new return roadway.	x	x	S 16.170.000	2010-15
	or operation of the second second second	Manufacture de la company	Airport Way terminal entrance roadway	Party Robert Contractory	Relocate and widen Airport Way northerly at terminal				
4032	PDX IA	Port	relocation	PDX terminal	entrance to maintain access and circulation	<u> </u>	X	\$ 4,620,000	2004-09
			The second s						
4033	PDX IA	Port	Airport Way east terminal access roadway	PDX east ferminal	Construct Airport Way east terminal access roadway	X	X ·	\$ 9,240,000	2010-15
4034	PDX IA	Portland	Retrofit	NE 33rd Avenue et Columbia Boulevard	Seismic retrofit project	x		\$ 1,039,500	2016-25
4035	Deleted (duplicated	in Project #4034)		· · · · · · · · · · · · · · · · · · ·					1
4025		Portland	42nd Avenue Bridge Seismic Retmit	NE 42nd Avenue at Lombard Street	Seismic retrofit project	~		¢ (70.550	
+030	FUXIA		Columbia and Lombard Intersection		Improve left turn/right turn capacity at MLK/Columbia and	^		φ <u>4/3,55</u> 0	2010-25
4037	PDX IA	Port	Improvements	Columbia Boulevard and Lombard Street at MLK	MLK/Lombard	x		\$ 808,500	2004-09

RTP #     2840 Link     Judicidition     Project Name (Facility)     Project Location     Project Description     2015 RTP 2015 RTP Project Description     (*** indicate System       RTP #     2840 Link     Judicidition     Project Name (Facility)     Project Location     Project Description     2015 RTP Project Description     Project Description     System     System     System     System       4015     Physics     Physics     Physics     Physics     Physics     System     System	RTP Program Teara
ADD PONIA Part Inconstruint 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	
ADD PORTA PART PART AVELONATION OF A	
ADE FORM FOR CONTRACTOR CONTRACTOR CONTRACTOR FOR STATE OF STATE O	
	x
Afth Assess instancias and Processy in the second	
ADNO PODE TOTAL TOTAL AND A CONTRACT ON BETWEEN X X X X 2 230/0	0   20100
Countries Documents Thermony	
ADMI POTER Portex Inconverses at Anderwood Face Inconverses Wilson and agrantice brannesses 2 X S 19801	
ADV2 POC A Part Control Russ Internetion Advanced Control bits sector Add taxed Promotion large discovered in a sector and the sector	6 2054-59
Signator 33mSMachad Drive Internetion	
4043 Protect Increment x 3 2 2047	a tenara
And and a series - Performant - Devolutional days a transmission destant a state events as a series of the series as a series of the series of	ود و مرب
Add NS without any weak much participation and	
ADME POR UN Performant Appendiant Toposenetis Appendiant Strat Appendiants	ol [ mets
Anas article Bytend INF Artestand Planetry - NF Collaboration Declaration Declaration Trail Description and the sector street	
4947 Deleted (Construction completed)	
4040 Deleted (alternative route provided on 37th)	
anno proxia Portes Nike Colorise Decement Disease . R. Lorbert in Mile Breart Perry's the breast or existing Street	A : 2015-15
4051 - Port is the owned in MC decompedite MC and the Arth Avenue	64 2016.25
4557 Talakted if Construction rearring	
Protection and Bioplances . POX annial towards 5. Province Post and the	
AD13 POX IX POX	in iren
and my is Perfered Parts in Constant Section Perford Frank Arries and in Solida Constant Second and consta	ا يونو زيون الماري
APURATE AND A TRANSPORT	
Antos Por la provisi desenvoltes exercises parte a service a service and Control Road	n maaaa
40% POW A Portend Counsis basevoir ITS Strained action in Superstand 1555 ents dents for inthis from x x S 31645	ol moro l
4057 Protect INNE Marra Dreams	يتحر ويتهزوا الج
Control control in the second shall TM	
in a state of the	
ADSD PETRIA POST INCOMMENTAL ADDRESS ROAD PETRIA PETRIA PETRIA ADDRESS X X X X X X X X X X X X X X X X X X	56 <u>2014</u> 20
Print of a set of the	
4050 PDR IA PORTAZINA DUBANA	n i pose-pe
TYPHI FillyDefi Hilling Dirich and Dirich an	en neven ser
	<u>u zurenia</u>
4522 Deleted (Canadravia)	
4052 Ebergate A CODITIVATiant N. Dented ingrovements and Colorities Starty billion and a star	مريدورام
4064 Rivergate IA Port Marrie Drive Trace 2 Fail overcrossing Contract rail overcrossing X 5 20 785.0	8
and the second at Prof. Invested North Invested Investigation and Investigation and Investigation and Investigation	n   mare

RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars 25 RTP ("*" indicates ancially phasing in strained financially ystem constrained		RTP Program Years
4066	Rivergate IA	Port	Columbia River Channel Deepening Study	Astoria to Portland	Conduct feasibility/environmental study	x	· · ·	n/a	$\square$	2004-09
4067	Rivergate IA	Port	Columbia River Channel Deepening	Despen Columbia River Channel from Astoria to Portland	State-wide Issue, project is outside Metto region	×	×	statewide project		2004-09
4068	Rivergate IA	Port/RR	Rivergate Rail expansion	Includes a series of improvements in Rivergate	Expand rail capacity in and to the Rivergate area	x	• • • •	\$ 17,000,000		2004-09
4069	Rivergate IA	Port/RR	Hayden Island rail access	Rail facilities from Rivergate to Hayden Island	Rail access to Hayden Island development	x		\$ 3,000,000		2010-15
4070	Rivergate IA	Port/RR	Additional tracks - Kenton Line	on Line North Portland to Fir Street Add track and sidings between Pen Junction and I-205 x		x		\$ 17,600,000		2010-15
4071	Rivergate IA	Port/RR	Barnes Yard Expansion	Bonneville Yard to Barnes Yard	Construct additional unit train trackage between Bonneville and Barnes Yard for storage	x		\$ 5,197,500		2004-09
4072	Columbia Corridor	Portland	Force/Broadacre/Victory Bikeway NI Marthe Drive to N. Deriver. Signed bikeway connection to I-5 river crossing		x	x	\$ 23,100		2016-25	
4073	Rivergate IA	Portland/Metro	Kelley Point Park Access (rail/40 Mile Loop Trail	Vicinity of Kelley Point Park	Construct shared-use path	x	×	\$ 132,825		2004-09
4074	Deleted (included i	n Project #4073)								
4075	Rivergate IA	ODOT/RR	3rd Track Connector Study	North Portland to Vancouver, WA	Study additional rail capacity to address growth in high speed rail and commuter rail	x		n/a		2004-09
4076	Rivergate IA	Various	Columbia Slough Greenway Trail Study	Kelly Point Park to Blue Lake Park	significance	x		n/a		2004-09
4077	Rivergate IA	Port/RR	Penn Junction Realignment	UP/BNSF Main line	Realign track configuration and signaling	x		\$ 5,000,000		2004-09
4078	Rivergate IA	Port/RR	WHI Rail Yard	West Hayden Island	Construct 7 track rall yard	x		\$ 9,500,000		2010-15
4079	Rivergate IA	Port/RR	Additional tracks - North Rivergate	Rivergate	Yard	x		\$ 300,000		2016-25
4080	Deleted (Project co	mpleted)	· · · · · · · · · · · · · · · · · · ·							
4081	Deleted (Project co	mpleted)								
4082	Rivergate IA	Pon/RR	Ramsey Rall Complex	South of Columbia Slough bridge	Construct six tracks and one mainline track and lead	x	<u>x</u>	\$ 12,000,000	$\square$	2004-09
4084	PDX IA	Port	Improvements	Mt. Hood Avanue to Marine Drive	Hood Avenue and Marine Drive	x	x	\$ 550,000		2004-09
4085	PDX IA	Port	Terminal area Bicycle and Pedestrian Improvements	Southside of PDX terminal to 82nd Avenue	Provide bicycle and pedestrian connection between terminal and 82nd Avenue south of Airport Way	x	×	\$ 750,000		2010-15
4086	PDX IA	Port	PIC Bike and Pedestrian Improvements	Portland International Center	Alderwood Road and Mt. Hood LRT station	x	x	\$ 240,000		2010-15
4087	Rivergate IA	Port	Leadbetter Street Extension and Grade Separation	to Marine Drive	Extend street and construct grade separation	x	x	\$ 8,000,000		2004-09
4088	Rivergate IA	Port/Portland	Terminal 4 Driveway Consolidation	Lombard Street at Terminal 4	Consolidate two signalized driveways at Terminal 4	x	×	\$ 1,000,000		2004-09
4089	Columbia Corridor	Port/Portland	Columbia Boulevard Improvements	60th Avenue to 82nd Avenue	Widen street to five lanes	x		\$ 15,000,000		2010-15
4090	Region	ODOT	I-5 Reconstruction and Widening - PE/EA	Greeley Street to I-84	to modernize reeway and ramps to improve access to the Lloyd District and Rose Quarter	x		\$ 15 000 000		2010-15
4091	Region	ODOT	I-5 Reconstruction and Widening - ROW Preservation	Greeley Street to I-84	Acquire R-O-W	Y		\$ 5,000,000		2010 15
4092	Region	Region	BNSF Rail Bridge	Columbia River	Construct improvements to increase track speeds on approaches too movable river spans	x		\$ 8,000,000	$\parallel$	2004-09
4093	Region	Region	North Portland Junction	North Portland	Install revised rail corssovers and higher turnout speeds	x		\$ 9,200,000		2004-09
4094	Region	Region	Graham Line Connection	South of Steel Bridge	Restabilish a connection in the southeast quadrant at East Portland between UP's Brooklyn and Graham rail lines	x		\$ 11,000,000		2010-15
4095	Region	Region	Albina to Willsburg Junction Improvements	Between Milwaukie and UPRR Albina Rail Yards	Implement track and signal improvements to allow for increased track	x		S 8 800 000	+	2004-09
4096	Region	Region	Willsburg Junction to Clackamas	Milwaukle to I-205	Extend two tracks from Willsburg Junction to Clackamas	x		\$ 19,000,000	$\square$	2004-09

F		1	T				2003 dollars					
RTP #	2040 Lint	Jurisdiction	Project Name (Facility)	Project Location	Project Description		2025 RTP Financially Constrained	25 RTP ("** indicates nancially phasing in nstrained financially System constrained			RTP Program Years	
	2040 Link				Upgrade river lead tracks between Albina and East		Cystem					
4097	Region	Region	Albina Yard Mainline Improvements	Near UPRR Albina Rail Yards	Protland, and a second track through the East Portland yard, interlocking the Seattle and Brooklyn subdivisions	x		\$	12,000,000		2004-09	
4098	Region	Region	Graham Line Skling	Graham rail line	Add controlled siding on the UP Graham line	х		s	12 000.000		2004-09	
4099	Region	Region	North Portland Rail Grade Separation	BNSF Rail Bridge and Columbia Slough and North Portland Junction	Grade separation rail/highway traffic on North Columbia Boulevard at Penn Junction	x		\$	75,000,000		2016-25	
5000	Region	TriMet	Oregon City LRT Extension	Oregon City to Milwaukie extension	New LRT Service	x		\$	577,500,000		2016-25	
5001	Region	TriMet	Transit center and park-and-ride upgrades	Various locations in subarea	park-and-rides throughout subarea	×	x	See	Td-Met Total		2004-25	
5002	Region	орот	I-205 Improvements	99E to Highway 213	General purpose, express, HOV or peak period pricing capacity improvements to be determined based on I-205 South Corridor Study	X		\$	86,625,000		2016-25	
5003	Region	ODOT	Sunrise Highway -Unit 1, Phase 2	122nd Avenue to Rock Creek	at 135th and Rock Creek junction	x		\$	104,550,000		2004-09	
5004	Region	ODOT	Sunrise Highway R-O-W Preservation	Rock Creek to 257th Avenue	Acquire right-of-way	x		s	46,200,000		2004-09	
5005	Region	ODOT	Sunrise Highway - Unit 2, Phase 1	Rock Creek to 257th Avenue	Construct new 4-lane facility	x		\$	184,800,000		2016-25	
5006	Region	ODOT	Sunrise Highway - Unit 2, Phase 2	257th Avenue to US 26	Construct new 4-lane facility	x		\$	177,000,000		2016-25	
5007	Region	ODOT	Highway 212	Rock Creek to Damasous	Construct climbing lanes to 172nd Avenue	Y	v		1 501 500		partone	
6009	Posias	ODOT	Highway 212/I-205 Interchange	Hinbway 212/L205	Increase ramp capacity from (-205 to Highway 212			φ			200-03	
5008	Region		induction and in		General purpose, express, HOV or peak period pricing	<u>x</u>		\$	17,325,000		2016-25	
5009	Region	ODOT	I-205 Improvements	West Linn to 1-5	capacity improvements to be determined based on I-205 South Corridor Study	x		\$	80,850 <u>,000</u>		2016-25	
5010	Region	орот	I-205 Express Lanes	Highway 213 to just north of I-84	capacity improvements to be determined based on I-205 South Corridor Study	x		\$	34,650,000		2016-25	
5011	Region	ODOT/ClackCo	I-205 North Auxiliary Lane Improvements	I-205 at Sunnybrook Road	Complete interchange	x		5	10 510 500		2004-09	
5012	Region	орот	I-205 Bridge Improvements	I-205 Bridge In Oregon City	General purpose, express, HOV or peak period pricing capacity improvements to be determined based on I-205 South Corridor Study	x		\$	86,625,000		2016-25	
	1.01	and the second			New SB Truck climbing lane at 1-205 bridge (between		1.00	1.2	1945 (P. 1946)			
5013	Region	ODOT	1-205 Climbing Lenes	Willamette River to West Linn in Clackamas County	financially constrained system	x	x	\$	46,200,000	*	2016-25	
5014	Region	ODOT	I-205 Auxiliary Lanes	82nd Drive to Highway 212/224	Add auxiliary lanes	х		\$	9,240,000		2016-25	
5015	Region	ODOT	Highway 99E/224 Improvements	Ross Island Bridge to I-205	Access management, reversible travel lane from Ross Island Bridge to Harold and widen to six lanes from Harold to I-205	x		\$	110,880,000		2016-25	
5016	Region	ODOT	Highway 213 Grade Separation	Washington Street at Highway 213	Grade separate southbound Highway 213 at Washington Street and add a northbound lane to Highway 213 from just south of Washington Street to the I-205 on-ramp.	x	x	\$	10,395,000		2010-15	
5017	Region	ODOT	Highway 213 Intersection Improvements	Abernethy at Highway 213	Intersection Improvements	x	x	\$	3,465,000		2010-15	
5018 [	Deleted (Construct	ion completed)										
5019	Region	ODOT	Highway 213 Interchange Improvements	Beavercreek/Highway 213	Grade separate existing intersections	x		s	20,790,000		2016-25	
				and a second second second second	Access management, sidewalks and capacity . Improvements including adding one lane in each							
5020	Region	ODOT	Highway 213 Improvements	Clacksmas CC to Leland Road	direction north of Canyon Ridge Drive	<u>x</u>	X	\$	17,325,000	*	2010-15	
5021	Region	ODOT	Highway 224 Extension	1-205 to Highway 212/122nd Avenue	Highway 212/122nd Avenue Interchange	x	X	\$	84,315,000		2010-15	
5022 0	Deleted (Construct	ion completed)										
5023	Région	ODÓT	I-205/Highway 213 Interchange Improvement	1-205 at Highway 213	Reconstruct I-205 southbound off-ramp to Highway 213 to provide more storage and enhance freeway operations and safety	×	x	s	1,155.000		2010-15	

### Public Comment Draft 2004 RTP Project List

- 6

~								-		-	-	•		~	1
7	c	te	a	b	e	r	3	1	_	2	O	۵	3		

RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	RTP ( <sup>***</sup> indicates clally phasing in ained financially em constrained		RTP Progra	
	Contraction and the				Corridor analysis from H205 to 172nd Avenue to develop						
		ODOT/Clackames	A CONTRACT OF		determine selected alternative and develop phasing						
5024	Region	County	Sunnise Contdor Unit 1 Supplemental EIS	1-205 to 172nd Avenue	acquisition	x	x	s :	2,736,195	200	14-09
5025	Region	County	Sunrise Corridor Unit 2 Locational EIS	172nd to US-28	Damascus/Boring Concept plan	x	x	s -	1,848,000	200	14-09
5026	Region	Metro	Portiand Traction Co. Shared-Use Trail	Milwaukie to Gladstone	Planning, PE and construction of multi-use trail Conduct EIS corridor analysis to study long-term transit	<u>X</u>	X	\$	1,386,000	200	4-09
5027	Region	Metro/ODOT	I-205 South Corridor Study- EIS Highway 224/McLoughlin Boulevard	I-5 to Highway 224	and road improvements	x	X	\$	5,000,000	201	0-15
5028	Region	ODOT/Metro	Corridor Study	Portland central city to Clackamas regional center	improvements	x		<b>\$</b>	1,155,000	201	6-25
5029	Region	ОДОТ	(McLoughlin/Highway 224) and EIS	Ross Island Bridge to I-205	complete EIS	x		\$ 9	9,240,000	200	4-09
5030	Region	ODOT	Highway 213 Green Corridor Plan	Highway 213 south of Leland Road	Develop Green Corridor plan	x		n/	a	201	0-15
5031	Region	ODOT	Highway 213 Corridor Study	Highway 213 south of I-205	Corridor analysis to study long-term transit and road improvements	x		s	577.500	201	6-25
5032	Region	Various	North Clackamas Greenway Corridor Study	Milwaukie to Clackamas RC	Study feasibility of corridor	Y			2	200	4 00
5033	Region	Various	Willamette River Greenway Study	Selwood Bridge to Lake Oswego	Study feesibility of contrider					200	4-05
		ODOT/Clackamas	Control Website Dig Composite of			<u> </u>	λ	n/:	8	200	4-09
5034	Region	County	Sunnse Highway R-O-W Preservation		Acquire right-of-way	X		\$ 40	0,000,000	200	4-09
5035	Milwaukle TC	TriMet	McLoughlin Bouleverd Rapid Bus	Milwaukie TC to Oregon City TC	Construct improvements that enhance Rapid Bus service	x	x	see Tri-Me	t lotal	201	0-15
5036	Deleted	A STATE OF STREET			Permitting the second transformed to the second second						-
5027	Mitham Adv. T.C.	Mitwanikia/GlackCo	I ske Road Improvements	2 st Automotic Michael (224	and add sidewarks, landscaped median, curbs, storm						
5038	Deleted (Construct			12 Jac Person De la Chigh May 224	uramage and rer tant religies at some mersectors	X	X	\$	5,500,000	201	0-15
5038	Deleted (Construct	ton to be completed	in 2003)								
5039	Deleted (Included I	h Project #5049)	Delivered Average Directory of the								
5040	Mirweukie IC	Milwaukis	Rain Jau Avenue Dike/Peo Improvement	S/IT Avenue to Linwood Road	Reirolit Dike lanes and sidewalks	×	X	\$ 7	7,000,000	201	0-15
5041		Miwaukie	Stut Avenue Bike/Peo Improvement	rignway 224 to Hamson Street	Retrofft Dike Janes and sidewalks	X	X	\$	410,000	201	6-25
5042	Deleted (Project to	be completed throug	jh redevelopment)		Extend sidewalk to Johnson Creek Boulevard and					_	
5043	Milwaukie TC	Clack. Co./Milwaukie	Stanley Avenue Multi-modal Improvements	Willow Street to Johnson Creek Boulevard	accommodate bloycles	X		\$	173,000	201	6-25
5044	Milwaukie TC	Milwaukie	Oatfield Road Improvement	Oatfield Road/Lake Road intersection	Intersection	x		\$	207,000	201	0-15
5045	Milwaukie TC	Clack, Co./Milwaukie	Improvements	Linwood/Harmony/Lake Road intersection	left turn lane and grade separate UPRR	x	x	\$ 28	000,000	201	0-15
5046	Deleted (Construct	ion completed)	Mal auchlin Haulavard Improvements								
5047	Milwaukie TC	ODOT	Milwaukie	Scott Street to Harrison Street	Complete boulevard design improvements	х		\$ 3	300,000	200	4-09
5048	Milwaukie TC	ODOT	Miccougnin Boulevard Improvements - Milwaukie	Harrison Street to Kellogg Creek	Complete boulevard design improvements	x	x	\$ 1	1900.000	200	4-09
5049	Milwaukie TC	ODOT	McLoughlin Boulevard Improvements - Mitwaukie	Kellogg Creek to River Road	Complete boulevard design improvements	×		s 2	000.000	200	4.00
5050	Milwaukle TC	Mitwaukie	Harrison Street Bikeway	Highway 99E to King Road via 42nd Avenue	Retrofit bike lanes to existing street	×	-	<u> </u>	560,000	200	4-09
5051	Deleted (included l	n Project #5037)		······································		~			200,000	200	
5052	Milwaukie TC	Milwaukie	17th Avenue Trolley Trail Connector	Springwater Corridor to Trolley Trail	Construct sidewalks on 17th Avenue to provide trail connection	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~					
5054	Milwaukie TC	Milwaukie/ODOT	Milwaukie Town Center Pedestrian	McLoughlin, Harrison, Monroe, Washington, Main	Improve sidewalks, lighting, crossings, bus shelters and benches	~				200	4-09
						х		\$ 2	400,000	201	6-25

		1	1				r	20	03 dollars	- 1	
RTP #	2040    - -	lutediction	Project Name (Escility)	Brolect Location	Brainet Description	2025 RTP Preferred	2025 RTP Financially Constrained	("" P fi	" indicates hasing in inancially	P	RTP rogram
	2040 LINK	Junadiction	Floject Hame (Facility)		Improve pedestrian access to Willamette River from	aystem	aystem	cc	Jostraineu		rears
5055	Milwaukie TC	Milwaukie/ODOT	Milwaukie TC River Access Improvements	McLoughlin Boulevard	Milwaukie	X		\$	10,000,000	2	016-25
5056	Milwaukie TC	Clackamas Co.	Lake Road Pedestrian Improvements	Harmony Road to Johnson Road	benches	x		\$	115,500	2	016-25
5057	Milwaukie TC	Clack. Co./Milwaukie	Linwood/Flavel Avenue Pedestrian improvements	Johnson Creek Boulevard to Harmony Road	Improve sidewalks, lighting, crossings, bus shelters and benches	x		\$	600,000	2	010-15
5058	Milwaukie TC	Milwaukie	17th Avenue Pedestrian Improvements	Lava Drive to Ochoco Street	Improve sidewalks, lighting, crossings, bus shelters and benches	x		s	920.000	2	016-25
5059	Milwaukle TC	Milwaukia	King Road Boulevard Improvements	42nd Avenue to Linwood Avenue	Boulevard design, including wider sidewalks, bikeway, median treatment and access management	×	x	s	5 000 000	2	010-15
5062	Milwaukie TC	TriMet/Miwaukle:	Milwaukie TMA Startup	Milwaukia town center area	Implements a transportation management association program with employers	x	×	5	200.000	2	016-25
5064	Clackamas RC	TriMet	I-205 Rapid Bus	Clackamas RC to Oregon City via I-205	Construct improvements that enhance Rapid Bus service	x		see Ti	ri-Met total	2	004-09
5065	Deleted (TMA has	been formed)	we also also also also also also also also								
						-					
5068	Clackamas RC	Clackamas Co.	East Sunnyside Road Improvements	122nd Avenue to 172nd Avenue	Widen to five lanes to improve safety and accessibility to Damascus	X	x	s	45 045 000	* 2	010-15
5067	Clackamas RC	Clackamas Co.	Johnson Creek Boulevard Interchange	Johnson Creek Boulevard at 1-205	Add loop ramp and NB on-ramp; realign SB off-ramp	v			0.000.000		
6000	<u> </u>	Claskamas Ca	Internet Creat Reviewerd Improvements		Widen to three lanes and widen bridge over Johnson	<u> </u>	^	4	0,000,000	2	310-23
5068		Ciackamas Co.			Creek to improve freight access to I-205	X		\$	8,085,000	2	016-25
5069	Clackamas RC	Clackames Co.	Harmony Road Improvements	Sunnyside Road to Highway 224	Widen to five fanes to improve safety and accessibility	<u>x</u>	X	5	7,392,000	2	010-15
5070	Clackamas RC	Clackamas Co. 🦡	Otty Road Improvements	82nd Avenue to 92nd Avenue	Widen and add turn fanes	x	x	s	1,848,000	2	004-09
			And a state of the		Extend William Otty Road as two-lane collector to						
5071	Clackamas RC	Clackamas Co.	William Otty Road Extension	I-205 frontage road to Valley View Terrace	improve east-west connectivity	x	×	\$	5,313,000	2	016-25
5072	Clackamas RC	Clackamas Co.	West Monterey Extension	82nd Avenue to Price Fuller Road	Two-lane extension to improve east-west connectivity	x	x	5	1 767 150		010-15
5073	Clackamas RC	Clackamas Co.	Monterey Improvements	82nd to new overcrossing of 1-205	Widen to five tanes from 82nd to 1-205	x	x	s	5,197,500	2	004-09
6074		Clackamas Co	Courses Assessie Extension	Callery plat 205 to pay dat fratting and	Extend new three-lane crossing over I-205 to improve			1.0			
- 30/4	CIBCRETTIES INC.	Cistona nas co.	Columb Preside Excension	Cadady - Oval - 20210 Hisw desi 10/1/200 Dec	BERLEWOLL CONTINUED VIEW	<u> </u>	X	5	6,294,750	2	116-25
5075	Clackamas RC	Clackamas Co.	79th Avenue Extension	King Road to Clatsop Street	Build N-S collector west of 82nd Avenue	x		e	5 775 000	2	016-25
5076	Clackamas RG	Clackamas Co.	Fuller Road Improvements	Johnson Creek Boulevard to Otty Road	Widen street and add tum lanes	^		•	0,770,000	~ ~	001.00
					the second s	<u> </u>	<u> </u>	φ	2,000,000	- 2	204-09
5077	Cleckames RC	Clackemas Co.	Summers Lane Extension	122nd Avenue to 142nd Avenue	New three-lane extension to provide alternative e/w route to Sunnyside	x	x	5	8 373 750	* 2	016-25
5078	Clackamas RC	Clackamas Co.	Mather Road Improvements	97th Avenue to 122nd Avenue	Connect to Summers Lane extension and widen	x		\$	3,465,000	2	016-25
5079	Clackamas RC	Clackamas Co.	122nd/Hubbard/135th Improvement	Sunnyside Road to Hubbard Road	Reconstruct and widen to three lanes	x		\$	7,276,500	2	016-25
5080	Clackamas RC	Clackamas Co.	Fuller Road Improvements	Harmony Road to Monroe Street	Widen to three lanes with sidewalks and bike lanes; includes disconnecting auto access to King Road	x	×	e	4 755 135		018.75
	the later the second of		the second second second second	and the second		- <u></u>	A	φ	4,733,135	- 4	5.10-25
5081	Clackamas RC	Clackamas Co.	Boyer Drive Extension	82nd Avenue to Fuller Road	New two-lane extension	x	x	\$	1,963,500	2	016-25
	14 (A)									1	
5082	Clackamas RC	Clackamas Co.	82nd Avenue Multi-Modal Improvements	Clatsop Road to Monterey Avenue	Widen to add sidewalks, lighting, crossings, bike lanes and traffic signals	x	x	s	11,550,000	* 2	010-15
5083	Clackamas RC	Cłackamas Co.	Causey Avenue Extension	1-205 frontage road to William Otty Road	Construct new two lane extension	×		\$	13,629,000	20	010-15
5084	Clackamas RC	Clackamas Co.	Fuller Road Extension	Otty Road to King Road	Construct new two lane extension	x		\$	4,620,000	2	016-25

RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( "*" indicates phasing in financially constrained	RTP Program Years
and a second second	1000 00 00 00 00 00 00 00 00 00 00 00 00	and a second state	And the second se	and the second					
5085	Clackamas RC	Clackamaa Co.	Clackamas RC Bike/Pedestrian Corridors	Clackamas RC existing and new developments	Provide bike and pedestifan connections in the RC	x	x	\$ 5,775,000	2016-25
and the second se		North Martin	82nd Avenus Boulevard Design	A CONTRACTOR DE LA PROPERTA DE		Concerning of the			
5086	Clackamas RC	Clackamás Co.	Improvements	Monterey Avenue to Surinybrook Street	Complete boulevard design improvements	x	×	\$ 4,620,000	2004-09
5087	Clackamas RC	Clackamas Co;	West Sunnybrook Road Extension	82nd Avenue to Harmony Roed	route to Sunnyside Road	x	x	\$ 2,310,000	2016-25
5089	Clackamas RC	Clackamas Co.	Sunnyside Road Bikeway	SE 82nd Avenue to I-205	Restripe to include bike tanes	and the second s	Y	\$ 231,000	2010 16
5090	Clackamas RC	Clackamas Co.	Lawnfield Road Bikeway	SE 82nd Dr. to SE 97th Avenue	Widen to lockide bike lanes			201,000	2010-13
1000 C		Clashiman Ca			Control Control of Station	X	X	\$ 115,500	2016-25
5091	Charckamies RC	Chackernes CO.	Causey Averaus Dikaway	1-205 pam to SE Fuller	Restripé to include bike tenes	x	x	\$ 23,100	2010-15
5092	Clackamas RC	Clackamas Co.	SE 90th Avenue Bikeway	SE Causey to SE Monterey	Construct bike lanes	x	x	\$ 92,400	2016-25
5093	Clackamas RC	Clackamas Co,	SE 97th Avenue Bikeway	SE Lawrifield to SE Mather	Construct bike lanes	X	and the second	\$ 23.100	2016-25
.5094	Clackamas RC	Clackamas Co.	CRC Trail	Clackamas Regional Park to Phillips Creek	N Clackamas sharad-use path	-			2010 20
5095	Clackamas RC	Clackamas Co.	Phillips Creek Greenway Trail	Causey Avenue to Mt. Scott Greenway	Construct trail	x x	X	\$ 358,050	2010-15
5096	Clackamas RC	Clackamas Co.	District Park Trail	Phillips Creek Trail to Mt. Scott Trail	Construct trail	x		\$ 202,910	2004-09
5097	Clackamas RC	Ciackamas Co.	Hill Road Bike Lanes	Oatfield Road to Thiessen Road	Construct bike lanes	x		\$ <u>433</u> 125	2004-09
5098	Clackamas RC	TriMet	King Road Frequent Bus	Clackamas Regional Center	Construct improvements that enhance Frequent Bus	× .	¥	5 1236.000	2010-15
5099	Clackamas RC	TriMet	Webster Road Frequent Bus	Clackamas Regional Center	Construct Improvements that enhance Frequent Bus Service	The second s	×	\$ 1,E10,000	2010-15
5100	Clackamas RC	Clackamas Co.	Fuller Road Pedestrian Improvements	Harmony Board to King Board		~	~		2010-13
					1.10 OVC SLOVANS	X	x	\$ 635,250	2004-09
5101	Clackamas RC	Clack. Co/ODOT	Clackamas RC Pedestrian Improvements	82nd Avenue, Sunnyside, Sunnybrook, Monterey and Intersecting streets.	Improve sidewalks, lighting, crossings, bus shelters and benches			and the second	
E102	Ciashamas DO	Clashamas Co	Claskemes BC Redevalar		Master plan and retrofit existing site to construct future	X	X	\$ 1,732,500	2016-25
5102	Clackamas RC	Clackamas Co.		Clackamas Regional Center	street grid Advanced transportation system management and	x		n/a	2016-25
5103	Clackamas HC	Clackarnas CO.	Clackamas County ITS Man	County-wide	intelligennt transportation system program	×	x	\$ 6,514,200	2004-09
5104	Clackamas RC	Clackamas Co.	Sunnybrook Extension - west	82nd Avenue to Harmony Road	Construct two-lane extension	<b>v</b>		• • • • • • • • • • • • • • • • • • •	
			102nd Avenue/Industrial Way		Extend Industrial Way from Mather Road to Lowefloid	^		३ 2,541,000	2004-09
5105	Clackamas IA	Clackamas Co.	Improvements	Highway 212 to Mather Road	Road	x		\$ 7,680,000	2004-09
5106	Clackamas (A	Clackamas Co.	SE 82nd Drive Improvements	Hindway 212 to Lawroled Road	Wildow to fing two as to account of the final				
					When it we range to accommodate truck movement	<u> </u>	x	\$ 6,930,000	2016-25
5107	Clackamas IA	Clackamas Co.	SE 82nd Drive Improvements	Gladstone to Highway 212, phase 2	Widen to five lanes	x		\$ 8,662,500	2016-25
5108	Deleted (Constructi	on completed)						s -	
5109	Clackamas IA	Clackames Co.	82nd Drive Bloycle Improvements	SE Jennifer Street to Fred Meyer	Widen to include bike lanes	×	- x-	\$ 138.600	2010-16
5110	Clackames IA	Clackamas Co.	Jennifer Street Bicycle Improvements	SE 105th to 120th Avenue	Widen to Include bike Image			000,001	2010-15
				Contracting and the second	Mustrio mostle bike lanas	X	X	\$ 288,750	2004-09

#### Public Comment Draft 2004 RTP Project List

2003	15 1e	odota	0
וברו דופו			-007

92-9102	000 005 1 8	X	×	Jowniown Oregon City	Disveluog niinguojow	ubboveweups	Oregon City/ODOT	Cregon City RC	2144
2016-25	000'591'1 3	i x	X	setones provestates, normality, crossings, pus shered		cinemovorgmi natrice Pedestrian improvements	ODOT/THMM.	Cregon City RC	2143
2010-15	000'590'1 5	X	x	service construct improvements that entrance Frequent Bus	Oragon City to Clackamas Community College	su8 meuper3 eunevA sisiloM	THMM:	Cregon City RC	2715
2016-25	Seo.e87.r 2	;	×	Retrofit to include bike lanes and infill sidewalks	High Street to urban growth boundary	шргочетель:	Oregon City	Oregon City RC	1413
S016-25	ذذ ا	,	×	Right of way acquisition	081	South End Road Bike/Pedestrian	Oregon City	Oregon City RC	0715
2016-25	2 3'000'000		×	Construct sidewalks	Warner Milne to Meyers Road	Leland Road Pedestrian Improvements	Oregon City	Oregon City RC	62139
2016-25	2 1'254'900	×	χ	sinemevorumi nekseb braveluod eleiqino.	Abematily to Highway 213	sinemevorqmi seet2 rolgninseW	Oregon City	Cregori City RC	86138
SL-DLOZ	571,220,1 3	×	×	ainemevoidini ngiseb biaveluod eleiqino.D	and the Street statement of the statement of the statement of the street of the statement o	sinemevorqmi teetič notgnintaeW	Cregon City	Dregon City RC	1618
5016-25	000'000's \$	X	_X	complete bovlevard design improvements	Jeeri2 noisivid of Jeeri2 right	sinemevorqmi iseriiči rtt	Clackamas Co.	OC Contdor	96136
S1-0102	000'058'5 \$	X	×	Complete boulevard design improvements	1-205 to 1001 Steel	Phase 1 - Oregon City	ODOT/ClackCo	Cregon City RC	2132
2010-15	000,228,8 2		x	stnemevorqmi ngiseb braveluod etelqmoO	In the rest of the cost of the point rest is the rest of the cost of the cost of the cost of the rest	Phase 2- Oregon City	ODOT/ClackCo	Oregon City RC	PE13
S1-0102	2 4'000'000	X	X		rear in the solution of the so	Weshingtor/Abemetiny Connection	AltO nogenO	Cregon City RC	6133
5004-03	11\$°89 \$	×	×	Witden to Include Dire Janes	199412 triệM ci ⊒ê9 v⊮wrgiH	Main Street Extension.	Oragon City	Oregon City RC	2135
2016-25	3,580,500		×	teet Abemetiny from Highway \$13 to Main Street	Highway 213 to Main Street	ztnemevorqmi bsoЯ v/themedA	Clackamas Co.	Oregon City RC	1512
							ion completed)	Deleted (Construct	2130
5016-25	Istot teM-hT ees		x	Construct Improvements that enhance Rapid Bus service	Vancouver Mall to Oregon City via I-205	sud bidapid bus	JeMhT	Oregon City RC	6215
5016-25	Istot teM-hT ees		×	Construct improvements that enhance Rapid Bus service	T vity nogen O of A&9 of the Dregen City TC	Oregon City Rapid Bus	7eMhT	Oregon City RC	2128
5004-09	000,008,01 \$		×	Anter an and the state of the series of the	teevic nt01 of teevic nt8	Water Street Viaduct Improvements	Oregon City	Oregon City RC	2725
5004-09	2 1'900'000	x	X	notista käitinä evoitini Reproduktion alla sitti kaitin alla serieteen	Oregon City Ambak Station	S easing notials vientmy rituos	Oregon City	Oregon City RC	2158
5016-25	009'229 \$		×	peucpes	Johnson Road to Ostheid Road	zinemevorqmi nshitsebeq baoR teisdeW	enotsbelð. Co./Gladstone	OT enotabelo	SSLS
5016-25	099'11 \$		×	Bikeway design to be determined	bsoR revix of beach	Gioucester Street Bikeway	enotebele	OT enotebelo	2154
5016-25	055,11 \$		×	Bikeway design to be determined	82nd Dr. to McLoughlin Boulevard	Clackamas Boulevard Bikeway	enotebele	OT enotabelo	2153
5016-25	922'S \$		×	Bikeway design to be determined	Clackamas Boulevard to Jersey Street	Portiand Avenue Bikeway	Gladstone	OT enotabelo	2215
5016-25	000,033,11 \$		x	Comprete Industrial Improvements, such as poulevaid treatment at intersections, and appropriate TSM strategiestech as signal intertie	River Road to Clackamas River	McLoughlin Boulevard Improvement	Clackamas Co.	OT enotebelo	1212
5016-25	000'219'1 \$		x	Widen to three lanes; fill in sidewalks and bike lanes	eunevA bn28 of bs0A refaceW	sinemevorqmi bsoЯ blefitsO	өпотеряЮ	OT enotsbelO	2150
2004-09	\$ 353'400	X	X	eenel alide build to heel with a needed	SE Monroe Street to SE Johnson Creek Boulevard	sens. svid Road Bike Lanes	Clackamas Co.	nobimo 3 ésmechaelO	2115
5016-25	\$ 462,000		×	sensi shide bike lanes	eunevA sliskoM of bsoA J9 isstered	Wamer Milne Bikeway	Oregon City	Cleckemes Comdor	9112
2004-09	\$ 346,500		×	eansi exide bike lanes	369 YawrgiH oi bao'A Tevi'A 32	Roethe Road Bicycle Improvements	Clackamas Co.	Clacksmas Corridor	5115
5016-25	\$ \$1045,500		×	Setroft to include bike lanes	Hamson Street (Milw) to Clackamas R (OC)	Нідтияу 99Е Вікемау	1000	Clackamas Corridor	2114
5016-25	\$ 231,000		×	Widen to include bike lanes	SE Idleman to Ckackemas Co. Line	Mt. Scott Boulevard Improvements	Clackamas Co.	Clackamas Corridor	2113
дтр Ргодгат Уеаг <i>s</i>	2003 dollars phasing in financially financially financially financially	System Constrained 2025 RTP 2025 RTP	System Preferred 2025 RTP	Project Description	Project Location	Project Name (Facility)	nobologanu	3040 FTUK	<b>*</b> чтя

2004 RTP Project List

RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( *** indicates phasing in financially constrained	RTP Program Years
5147	Oregon City RC	TriMet/Oregon City	Intercity passenger station	Oregon City TC	Intercity passenger connections with LRT/Bus	x		\$ 2,310,000	2016-25
5149	Oregon City RC	Öregon City	Oregon City Bridge Study	Highway 43/7th Street in Oregon City	Evaluate long-term capacity of Oregon City bridge	x	×	0/a	2018-25
5150	Oregon City RC	TriMet/Oregon City	Oregon City TMA Startup Program	Oregon City Regional Center	Implements a transportation management association program with employers	x	Y	\$ 200.000	2016-25
5151	Oregon City RC	Oregon City	Clackamas River Shared-Use Path	I-205 to Clackamette Park	Construct shared-use path	x		\$ 265.650	2004-09
5152	Onegon City RC	Oregon City	Willamette River Shared-Use Path	Clackametre Park and Smurfit	Construct shared-use path	x	×	\$ 500,000	2010-15
5153	OC Corridor	Clackamas Co.	Beavercreek Road Improvements Phase 2	Highway 213 to Clackamas Community College	Widen to 5 lanes with sidewalks and bike lanes	x		\$ 3,003,000	2010-15
5154	OC Contdor	Clackamas Co.	Beavercreek Road Improvements Phase 3	boundary services and s	Widen to 4 lanes with sidewalks and bike lanes	×	X	\$ 2,310,000	2016-25
					Improve access management, and provide sidewalks and bike lanes to connect multiviamity and compercial				
6156	OC Corridor	Clackamas Co.	Beavercreek Road Improvements, Phase 1	Highway 213 to Motalia Avenue	employment areas	x	x	\$ 4,500,000	2010-15
				de la constante	sidewalks, sidewalk infill, ADA accessibility, bike tares, rectifing a stawalk lange, add bit don any site.				
5157	OC Corridor	Oregon City	Mollala Avenue Streetscape Improvements	7th Street to Highway 213 (9 segments)	streetscape	×	x	\$ 15,000,000	* 2004-25
5161	Lake Oswego TC	TriMet	Macadam Frequent Bus	Laka Oswego to PCBD	service	x	x	\$ 2,015,000	2010-15
5163	Deleted (Construct	ion completed)							
5164	Lake Oswego TC	Lake Oswego	"A" Avenue Bikeway	Iron Mountain to State Street	as B Ave.; bikeway design to be determined	x		\$ 1,732,500	2010-15
5165	Lake Oswego TC	Lake Oswego	Willamette Greenway Path	Roehr Park to George Rogers Park	shared-use path	×	x	\$ 127,050	2010-15
5166	Lake Oswego TC	Lake Oswego/ODOT	Improvements	Highway 43, "A" and neighborhood streets	Improve sidewalks, lighting, crossings, bus shelters and benches	×		\$ 1,155,000	2016-25
5167	Lake Oswego TC	ODOT/LOAWL	Highway 43 Pedestrian Access to Transit Improvements	key locations along Highway 43 and Intersecting streets	Improve sidewalks, lighting, crossings, bus shelters and benches	x		\$ 1,155,000	2016-25
5168	Lake Oswego TC	Lake Oswego	Country Club Road Pedestrian Improvements	Boones Ferry to "A" Avenue	Improve sidewalks, lighting, crossings, bus shelters and benches	X		\$ 577,500	2016-25
5169	Lake Oswego TC	Lake Oswego	Trolley Trestle Repairs	Lake Oswego to Portland	Repair trestles along rail line	x	x	\$ 1,155,000	2004-09
5170	Lake Oswego TC	ODOT	Highway 43 Tráffic Management Plan	Highway 43 from McVey to I-205	Develop traffic management plan to address growing demand	×		n/a	2004-09
5171	Lake Oswego TC	Lake Oswego	Transit Station Relocation	from 4th Avanue to location TBD	Relocate transit station	e x	x	\$ 4,190,000	2016-25
5172	Lake Oswego TC	TBD	Lake Oswego Trolley Study	Study phasing of future trolley commuter service between Lake Oswego and Portland	Study phasing of future trolley commuter service between Lake Oswego and Portland	x	x	ŋ/e	2004.09
5192	West Linn TC	Clackamas Co.	Highway 43/Willamette Falls Intersection Imp.	Highway 43/Willamette Falls Intersection	Improve safety/capacity of Highway 43 Intersection at Willamette Falls Dr.	×	-`	¢ 4 070 500	2010.05
5193	West Linn TC	West Linn	Willamette Fails Drive Improvement	10th Street to Highway 43	Upgrade street to urban standards with sidewalks and bike lanes	^ 		· 1,2/0,500	2010-25
5194	West Linn TC	Clackamas Co.	Highway 43 Intersection Improvements	Intersection at Pimlico Drive	Improve intersection to be safer for all modes of travel	×		<ul> <li>4,937,625</li> <li>3,914,500</li> </ul>	2004-09
5195	Deleted (Project to	be completed throug	th Project #5196)					4 3,811,500	2016-25
5196	West Linn TC	West Linn/ODOT	West Linn TC Pedestrian Improvements	Highway 43, Willamette Falls Drive, and neighborhood streets	Improve sidewalks, lighting, crossings, bus shelters and benches	X		\$ 1 155 000	2016-25
5197	West Linn TC	Clackamas Co.	Rosemont Corridor Plan	West Linn to Stafford Road	Study Rosemont as alternate n/s route; Study connection to I-205 at Exit 6	X		n/a	2016-25
5198	West Linn TC	ODOT	Highway 43 Improvements	Shady Hollow Lane to Robinwood Main Street	Complete boulevard design Improvements	x		\$ 9,240.000	2016-25
5199	Region	ODOT	1-205 Auxiliary Lanes	1-5 to Stafford Road	Add auxiliary lanes as part of pavement preservation project	x	x	\$ 8,000.000	2004-09
5200	Stafford UR	Clackamas Co.	Rosemont Road Improvements	Stafford Road to Parker Road/Sunset	Reconstruct and widen to three lanes; add turn lanes	x		\$ 6,121,500	2016-25

			· · · · · · · · · · · · · · · · · · ·	┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉┉					8888 I II		
RTP	2040 i Ink	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System		2003 dollars """ indicates phasing in financially constrained		RTP Program Years
5201	Stafford UR	Clackamas Co.	Childs Road Improvements	Stafford Road to 65th Avenue	Widen to three lenes including bike lanes and sidewalks	Y		e	4 897 200		2016-25
5202	Stafford LIR	Ciackamas Co.	Stafford Road Improvements	I-205 to Rosemont Road	Widen to three lanes including hike lanes and sidewalks	~			4,007,200		2010-23
5202						^		\$	4,389,000		2016-25
5203	Deleted (Project to	be completed public	Aprivate partitership)								
5204	Stafford UR	Clackamas Co,	Stafford Road	Stafford Road/Rosemont intersection	Realign intersection, add signal and right turn lanes	x	×	\$	866,250		2004-09
5205	Stafford UR	Clackamas Co.	Stafford Basin Future Street Plan	Develop future street plan for Stafford Basin		x			n/a		2016-25
5207	Happy Valley TC	Clack. Co./Happy Valley/NCPRD	Mt. Scott Creek Trail	Sunnyside Road to Mt. Talbert	Feasibility study and construction of undercrossing of Sunnyside Road to Mt. Talbert	x		\$	100.000	$\square$	2016-25
								<u> </u>		$\vdash$	
5208	Happy Valley TC	Clackamas Co.	Idleman Road Improvements	Johnson Creek Boulevard to Mt. Scott Boulevard	Reconstruct and widen to three lanes	x		\$	4,389,000		2016-25
				and the second second second second second	at the first of the second	1					et igner
5209	Happy Valley TC	Clackamas Co.	122nd/129th Improvements	Summyside Road to King Road	Widen to three lanes, smooth curves	x	x	s	3,465,000		2015-25
			Mt. Scott Boulevard/King Road					1			
5210	Happy Valley TC	Clackamas Co.	Improvements	Happy Valley City limits to 145th Avenue	Widen to three lanes	X	1	\$	4,620,000	2000	2016-25
19-25		and succession	Scott Creek Lans Pedestrian			and the second se	Section 2.0				7. 54.3
5211	Happy Valley TC	Happy Valley	Improvements	SE 129th Avenue to Mountain Gata Road	Construct pedestrian path and bridge crossing	<u> </u>	×	\$	103,950		2004-09
5212	Region	ODOT/Clackamas County	Sunrise Highway Unit 1, Phase 2 PE	135th Avenue to 172nd Avenue	facility and construct interchanges at 135th and Rock Creek Junctions	x		s	18 450 000		2004-09
		ODOT/Clackamas	Sunrise Highway Unit 1, Phase 2 R-O-W					<u> </u>	10,100,000		200.00
5213	Region	County	Preservation	135th Avenue to 1/2nd Avenue	Acquire right-of-way Peak-hour service only with 30-minute frequency in	X		\$	7,986,000		2004-09
6000	Region	Metro/ODOT	Beaverton-Wilsonville Commuter Rail	Wilsonville to Beaverton	existing rail corridor	X	X	\$	82,582,500		2004-09
6001	Deleted (Project de	fined in Project #600	00)							Π	
6002	Region	Metro/ODOT	Wilsonville-Salem Commuter Rail Extension Study	Wilsonville to Salem	Peak-hour service on existing tracks	x			n/a		2016-25
		Mater/ODOT	Tualatin-Portland Commuter Rall Extension	Tualatin to Union Station via Lake Oswego and	Parts have and an arts of a station for the			1			
6003	Region	Marbrobot	Siddy	Milwaukie	Conduct study and complete environmental design work	X			n/a		2016-25
6004	Region	ODOT	1-5/99W Connector Corridor Study	1-5 to 99W	for I-5 to 99W Connector	x	x	\$	1,732,500		2004-09
6005	Region	ODOT	1-5/99W Connector: Phase 2 Freeway	1-5 to 99W	Construct four-lane tollway with access control on 99W in Sherwood area	v			288 750 000	Π	2016 25
	Dealer		I-5/99W Connector: Phase 2 Freeway		Complete preliminary engineering for four-lane tollsway	^^		Ð	288,750,000		2010-25
6006	Кедюл	0001	Finetiminary Engineering	1-5 to 99W	with access control on 99W in Sherwood area to I-5	x	ļ	\$	15,000,000		2010-15
6007	Region	Various	Planning Weatherton Equato Consecutivity	Tigard to Tualatin	Planning and PE to extend greenway	x			n/a		2004-09
6008	Washington Sq. RC	Beaverton	Improvements	Washington Square Regional Center	recommendations in regional center plan	x			n/a		2016-25
6009	Deleted (Study und	егway)									
6010	Washington Sq. RC	ODOT/WashCo	Highway 217 Interchange Imp Denney Road	Denney Road at the Highway 217 on and off-ramps	Improve Denney Road at the Highway 217 on and off- ramps, including lights and covered culverts	x		•	577 500		2016-25
8011	Washington Sa PC	ODOT/Tioard	Highway 217 Overcrossing - Cascade	Nimbus to Locust	Provide a new connection from Nimbus to Washington	^			077,000		2010-20
					Improve existing roadway and construct new connections	×	× ×	13	26,000,000		2010-25
					and intersection alignments to provide connectivity and capacity from Walker Road in Western Avenue Project						
		Washington Ca			includes sidewalks and bike lanes and should be built as						
6012	Washington Sq. RC	ODOT	Hall Boulevard Improvements	Scholls to Locust	development occurs, Widen to 5 lanes with boulevard design	X	L	\$	6,000,000		2016-25
0010	Delated (Open					X		\$	5,428,500		2010-15
1 8014	Ueleted (Constructi	on completed)	1	1	I		1	1	,		

2004 RTP Project List

							2025 PTP	20	003 dollars		
						2025 RTP	Financially	` F	hasing in		RTP
RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	Preferred System	Constrained System	f Ci	inancially onstrained		Program Years
6015	Washington Sq. RC	Tigard/WashCo	Greenburg Road Improvements, North	Hall Boulevard to Washington Square Road	Widen to five lanes with bikeways and sidewalks	v			7 997 500		2004-00
6016	Washington Sq. RC	Tigard/WashCo	Greenburg Road Improvements, South	Shedy Lane to North Dakota	Widen to five lanes with bikeways and sidewalks	Ŷ	Y	5	2,007,000		2004-09
6017	Washington Sq. RC	Washington Co.	Taylors Ferry Road Extension	Washington Drive to Oleson Road	Three lane extension with bikeway and sidewalks	x	, , , , , , , , , , , , , , , , , , ,	ç	2 194 500		2016-25
6018	Washington So. RC	Washington Co.	Scholls Ferry/Allen Intersection	Scholie Ferry Road/Allen Boulevard Intersection	Realign intersection	•	v	•	0.040.000		2010 20
6019	Washington Sq. RC	Washington Co.	Oak Street Improvements	Hall Boulevard to 80th Avenue	Signal Improvement, bikeway and sidewalks			\$	2,310,000		2010-15
6020	Deleted (Project inc	luded in #3014 and	#3072)			^		е <b>р</b>	924,000		2004-09
6021	Washington Sq. RC	Beaverton/WashCo	Schoils Ferry Road Improvements	Highway 217 to 125th Avenue	Widen to seven lanes with access management	Y		c	19 202 800		2016.25
				Palm Boulevard, Washington Square Road, Ellande Lane, Scholls Ferry, Hall, Greenburg, Oleson,	ł			÷	10,202,000		2010-20
6022	Washington Sq. RC	WashCo/Tigard/ ODOT	Washington Square RC Pedestrian Improvements	Cascade, and streets within and through the mall area	Improve sidewalks, lighting, crossings, bus shelters and benches	~		*	6 020 000		2046.25
6023	Washington Sq. RC	Washington Co.	Scholls Ferry Pedestrian Improvements	Beaverton-Hillsdele Highway to Hall Boulevard	Improve sidewalks, lighting, crossings, bus shelters and benches	^		\$	6,930,000		2010-25
		, , , , , , , , , , , , , , , , , , ,		,	Implement appropriate TSM strategies such as signal	<u> </u>		3	577,500		2016-25
6025	Washington Sq. RC	Washington Co.	Scholls Ferry Road TSM Improvements	Highway 217 to 125th Avenue	interconnects, signal re-timing and channelization to improve traffic flows	×	x	s	577,500		2004-09
6026	Washington Sq. RC	TriMet/WashCo	Washington Square Regional Center TMA Startup Program	Washington Square Regional Center	Implements a transportation management association program with employers	x	x	s	200,000		2004-09
6027	Tigard TC	орот	I-5/217 Interchange Phase 2	Highway 217 and I-5	Complete interchange reconstruction	x		\$	45,045,000		2010-15
6028	Tigard TC	орот	I-5/217 Interchange Phase 3	Highway 217 and I-5	Complete interchange reconstruction with new southbound Highway 217 to I-5 flyover ramp	×		s	17.325.000		2010-15
6029	Tigerd TC	TriMet.	Hall/Kruse Frequent Bus	Tigard-Lake Oswego-Kruse Way	Construct Improvements that enhance Frequent Bos	Y	Y		275.000		2010 15
6030	Tigard TC	ОРОТ	Hail Boulevard Improvements	Locust to Durham Road	Improve Hall Boulevard to 5 lanes	×	~	\$	5 428 500		2010-13
6031	Tigard TC	Tigard	Greenburg Road Improvements	Tiedeman Avenue to 99W	Widen to 5 lanes	×		\$	5 544 000	1	2016-25
					Realign Hunziker Road to meet Hampton Street at 72nd Avenue and removes existing 72nd/Hunziker Road			<u>.</u>	0,0 1,000		2010 20
6032	Tigard TC	ODOT	Highway 217 Overcrossing - Tigard	Hunziker Street to 72nd at Hampton	intersection	×		\$	10,000,000		2016-25
6033	Deleted (Construct	on completed)	10								
6034	Tigard TC	Tigard	Walnut Street Improvements, Phase 3	135th Avenue to 121st Avenue	Widen to three lanes with bikeways and sidewalks	x	<u> </u>	\$	6,601,356		2010-15
oppe		Theread	Connels Street Internet and and								
0035		Tigard	Panita Read Improvements	Hell Parlamente Denne Band	widen to three lanes with bikeways and sidewalks	X	X	5	4,620,000		2004-09
6036	Tigard TC	Tigard	Durban Road Improvements	Laner Reenes Fort: Reed to Hell Reviewed		X		\$	9,240,000	+	2010-15
0037	rigard TC	Tigato			Extend street east of 99W to connecto to Hall Boulevard	X		\$	4,042,500	+	2010-15
6038	Tigard TC	ligard	Walnut Street Extension	Hall Boulevard to Hunziker Street	and Hunziker Street	×		\$	19,000,000		2010-15
6039	Tigard TC	0001	99W Improvements	I-5 to Greenburg Road	Widen to seven lanes	<u>×</u>		\$	28,875,000		2016-25
6040	Tigard TC	rigard	7 2Rd Avenue Improvements	999V to Hunziker Road	Widen to five lanes	X	X	\$	3,465,000		2004-09
6041	Tigerd TC	rigard.	72nd Avenue Improvements	HUNZIKER Road to Bonita Road	Widen to five Janes	X	X	\$	5,775,000		2010-15
6042	Tigard TC	Mashiartas C-	/ 2nd Avenue Improvements	Isonia Hoad to Duffiam Road	Widen to five lanes with bikeways and sidewalks	X	X	\$	5,775,000		2010-15
6043	Tigard TC	wasnington Co.	Oper Boones Ferry Road	I-5 to Durham Road	Widen to five lanes	X		\$	3,465,000	_	2016-25
6044	Tigard TC	iigard		Darmouth Road to Hunziker Road	Three lane extension; new Highway 217 overcrossing	X		\$	32,340,000		2016-25
6045	Tigard TC	Tigard	Darmouth Street Improvements	72nd Avenue to 68th Avenue	Widen to four lanes with turn lanes	- X .	x	5	577,500		2010-15

2004 RTP Project List

						2025 RTP	2025 RTP Financially	20	003 dollars *** indicates phasing in		RTP
RTP	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	Preferred System	Constrained System	f	Inancially onstrained	Pr	'ogram Years
8046	Deleted (Construct	ion completed)		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·					+	
6047	Tigard TC	ODOT	Highway 217/72nd Avenue Interchange Improvements	Highway 217 and 72nd Avenue	Complete interchange reconstruction with additional ramps and overcrossings	×		\$	17,325,000	20	010-15
6048	Washington Sq. RC	Beaverton/WashCo	Scholls Ferry Road Intersection Improvement	At Hail Boulevard	Add SB right turn lane from SB Hall Boulevard	x		\$	577,500	20	016-25
6049	Tigard TC	ODOT	Highway 99W Bikeway	Hall Boulevard to Greenburg Road	Retrofit for bike lanes	x		\$	577,500	20	010-15
6050	Tigard TC	WashCo/Tigard/ ODOT	Tigard TC Pedestrian Improvements	Highway 99W, Hall Boulevard, Main Street, Hunziker, Walnut and neighborhood streets	Improve sidewalks, lighting, crossings, bus shelters and benches	v			2 465 000		016.05
6051	Tigard TC	ODOT	Hall Boulevard Bikeway and Pedestrian improvements	Oak Street to Highway 99W	Bike lanes, sidewalks & pedestrian. crossings	^ X		\$	1,155,000	20	004-09
6052	Washington Sq. RC	Tigard/Beaverton	Highway 217 Overcrossing	Nimbus Drive to northern mall area	Two-lane overcrossing with sidewalks and bike lanes	x		\$	30,000,000	20	016-25
6053	Washington Sq. RC	Tigard	Nimbus Avenue Extension	Nimbus Avenue to Greenburg Road	Two-lane extension with sidewalks and bike lanes	x		\$	38 000 000	20	016-25
6054	Tigard TC	ODOT	Highway 99W Access Management Plan - Tigard	Highway 99W from I-5 to Durham Road	Develop access control plan for Highway 99W	x		*	n/a	20	004-09
6055	Tigard TC	ODOT	Highway 99W System Management	99W from I-5 to Durham Road	Signal interconnect on 99W from I-5 to Durham Road	x		\$	2,310,000	20	010-15
6056	Tigard TC	ODOT **	Highway 99W/Hall Boulevard Intersection	99W/Hall Boulevard	Add turn signals and modify signal	×	Ŷ	¢	4 273 500	21	010-15
6057	Washington Sq. RC	Tigard	Washington Squre Regional Center Greenbelt Shared Use Path	Hall Boulevard to Highway 217	Complete shared-use path construction	x	x	\$	2,000,000	2(	010-15
6058	King City TC	Tigard	Durham Road Improvements	Hall Boulevard to 99W	Widen to five lanes with sidewalks and bike lanes	x		\$	5,890,500	20	016-25
6059	Deleted (Construct	ion completed)									
6060	King City TC	WashCo/KC/Tigard/ ODOT	King City TC Pedestrian Improvements	Highway 99W, 116th, and Durham Road	Improve sidewalks, lighting, crossings, bus shelters and benches	x		s	3,465,000	20	016-25
6062	King City TC	King City	King City TC Plan	King City TC	Determine long-term transportation needs	x		·	n/a	20	010-15
6063	Happy Valley TC	Various	Lower Tualatin River Greenway Trail	Powerline Trail to Willamette River	Feasibility study to construct a shared-use pther	x		\$	75,000	20	016-25
6064	Tualatin TC	TriMel	Hall Boulevard Frequent Bus	Tualatin-Hall-TV Highway	Construct improvements that enhance Frequent Bus service	x	×	\$	7 700 000	2	010-15
6065	Tuelatin Ind. Area	Tualetin	Herman Road Improvements	Tualatin Road to Cipole Road	Widen to three larres including blke tarres and sidewalks.	x	x	\$	12,000,000	20	004-09
6066	Tuelatin TC	ODOT/Tualatin	I-5 Interchange Improvement - Nyberg . Roed	Nyberg Road/I-5 Interchange.	Widen Nyberg Road/I-5 Interchange	x	×	5	4,600,000	2	004-09
6067	Tualatin TC	ODOT	Boones Ferry Road Improvements	Durham Road to Wilsonville TC	Three lane improvement to complete sidewalks and bike facilities	×		\$	27 027 000	21	010-15
6068	Tualatin TC	ODOT	Boones Ferry Road Improvements	Tualatin-Sherwood Road to Wilsonville	Widen to five lanes with bikeways and sidewalks	×		\$	11,550.000	20	016-25
6069	Tualatin TC	Tigard/Tualatin	Hall Boulevard Extension	Extension from Durham to Tualatin Road	Extend Hall Boulevard to connect across the Tualatin River	X		\$	28,875,000	21	016-25
6070	Tuelatin TC	ODOTWashCo	Lower Boones Ferry	Boones to Bridgeport	Sidewalk, bikeway, interconnect signals	¥	Y	¢	5 900 000	2	004.00
6071	Tuelatin TC	Washington Co.	Tualatin-Sherwood Road Improvements	99W to Teton Avenue	Widen to five lanes with bike lanes and sidewalks: Intertie signals at Oregon and Cipple streets	x	x	s	28,875,000	21	010-15
6072	Deleted (Construct	ion completed)									
6073	Tuelatin TC	Tualatin	124th Avenue Improvements	Myslony Street to Tualatin-Sherwood Road	Construct new 3 lane arterial with bikeways and sidewalks	x	x	\$	7,854,000	2(	010-15
6074	Tualatin TC	Tualatin	65th/Tualatin River Crossing and connections	65th and McEwan between Lower Boones Ferry Road and Meridian Park Hospital	Construct new crossing of Tualatin River and connections to 65th and Lower Boones Ferry Road	Y		S	19,750,500		016.25
6075	Region	Various	Tonguin Trail	Connecting Wilsonville, Sherwood, tualatin, Tigard and Durham	Feasibility study to construct a shared-use path	~		-	100,000		110-23
6076	Tualatin Ind, Area	Tualatin	Mysiony/112th Connection	Myslony to Tuatatin-Sharwood Rd. @ Avery	Extend 3 lane road with sidewalks and bike lanes	^ X	Y	3 \$	1 500 000	20	00-15
6077	Tualatin TC	Washington Co.	Tualatin-Sherwood Road Bikeway	1-5 to Boones Ferry Road	Retrofit for bike lanes	X	A	s	1,155,000	21	016-25

						1	1	2002 dollars	
RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 donars ( *** indicates phasing in financially constrained	RTP Program Yaars
6078	Tualatin TC	Tualatin	Boones Ferry Road-Martinazzi Bike/Ped Path	Between Boones Ferry Road and Martinazzi north of	Construct new bike/pedestrian path	X		\$ 375.375	2016.25
6079	Tueletin TC	WashCo/Tuelelin/ ODOT	Tualatin TC Pedestrian Improvements	Nyberg, Boones Ferry, Tualatin, Tualatin-Sherwood, Segert and neighborhood streets	Improve sidewalks, lighting, crossings, bus shelters and benches	x	x	s 577,500	2004-09
6080	Tuelatin TC	Tuelatin/Durham	Tualatin River Pedestrian Bridge	Durhem City Park to Tualatin Community Park	Construct cantilevered pedesitian/bike path on railroad trestle across Tualatin River to Tualatin town center	×	x	\$ 1,155,000	2004-09
6081	Tualatin TC	WashCo/Tualatin	Nyberg Road Pedestrian and Bike Improvements	65th Avenue to 1-5	Complete sidewalks and bike facilities	x	x	<b>\$</b> 1,155,000	2004-09
6082	Tualatin TC	Washington Co.	Tualatin Freight Access Plan	Tualatin-Sherwood Road Corridor	Develop Interim circulation/freight management plan	x		n/a	2004-09
6083	Tuelatin TC	TriMet /WashCo	Tualetin Town Center TMA Startup	Tualatin Town Center	Implements a transportation management association program with employers	x	x	\$ 103,950	2004-09
6084	Wilsonville TC	Wilsonville	Kinsman Road Extension - south	Willsonville Road to Brown Road (5th Street extension)	Two-lane extension	x		\$3,200,000	2010-15
6085	Wilsonville TC	Wilsonville/SMART	Wilsonville-PCBD Express	Express bus service from Wilsonville Road/Boones Ferry Road to Portland CBD	Express bus service connection to PCBD	x		see Project #8035- 8037 costs	2016-25
6086	Wilsonville TC	Wilsonville	Kinsman Road Extension	Kinsman Road to Boeckman Road	Two-lane extension	x	x	\$ 7,620,000	2004-09
6087	Wilsonville TC	Wilsonville	Kinsman Road Extension	Boeckman Road to Ridder Road	Two-lane extension	x		\$ 3,910,000	2004-09
6088	Wilsonville TC	Wilson:/WashCo	Elligsen Road Improvements	Canyon Creek to Parkway Center	Improve Elligsen Road to 5 lanes	x	x	\$ 1,750,000	2010-15
6089	Wilsonville TC	Ciackamas Co.	Stafford Road Improvements	I-205 to Boeckman Road	Reconstruct, widen and add turn lanes	x		\$ 3,300,000	2016-25
6090	Wilsonville TC	Wilsonville	Boeckman Road Extension - West	Boeckman Road to Tooze Road	Extend 3 lanes with sidewaiks and bike lanes			s 36 170 000	2010-15
6001	Witnesselle TC	Wilsonville	Boeckman Road 1-5 Overcrossing	Parkway Avenue to 100th Avenue	Improve existing overcrossing to 6 lanes with sidewalks		<u> </u>	• 10,170,000	2010-13
6092	Deleted			Service contraction and the service services	und piles aires	X	λ.	2 3,990,000	2010-15
			······································						
6093	Wilsonville TC	Wilsonville	Barber Street Extension	Barber Street at Kinsman Road	Extend Barber Street as 3 lanes to 110th	x		\$ 7,310,000	2016-25
6094	Deleted (Construct	ion completed)							
6095	Wilsonville TC	Wilsonville	5th Street Extension	5th Street to Brown Road/Wilsonville Road intersection	Three lane extension from 5th Street to Brown Road, turn lanes at major intersections	x		\$ 6,390,000	2016-25
6096	Deleted								
6097	Wilsonville TC	Clackamas Co.	Stafford Road Safety improvements	I-205 to Boeckman Road	Safety Improvements	x		\$ 2,310,000	2010-15
6098	Wilsonville TC	Wilsonville	Kinsman Road Extension	Ridder Road to Day Road	Two-lane extension	x		\$ 4,700,000	2004-09
6099	Wilsonville TC	Wilsonville	Elligsen Road Improvements	Canyon Creek to Stafford Road	Two-lane extension	×		\$ 5,000,000	2010-15
6100	Wilsonville TC	Wilsonville	Barber Street Bikeway	Kinsman Road to Boberg Road	Complete N/S bikeway corridor	x		\$ 1,340,000	2016-25
6101	Wilsonville TC	Wilsonville	Wilsonville Road Bikeway	Rose Lane to Willamette Way West	Retrofit street to add bike lanes	x		\$ 577,500	2010-15
6102	Wilsonville TC	Wilsonville	Parkway Avenue Bikeway Parkway Avenue Bikeway (N bi	Town Center Loop to Boeckman Road	Retrofit to wide outside lanes	x		\$ 2,470,000	2010-15
6103	Wilsonville TC	Wilsonville	Boeckman)	Boeckman Road to Parkway Center Drive Wilsonville Road, Parkway Avenue, Boones Ferry	Retrofit street to add bike lanes	x		\$ 3,610,000	2016-25
6104	Wilsonville TC	Wilsonville	Wilsonville TC Pedestrian Improvements	Town Center Loop and intersecting streets	benches	X		\$ 2,160,000	2016-25
6105	Wilsonville TC	Wilsonville	Improvements	Parkway to Wilsonville Road	Retrofit street to add blke lanes and sidewalks	x	y the	\$ 251,000	2010-15
6106	Deleted (Constructi	on completed)					· · · · · ·		
6107	Wilsonville TC	Wilsonville	Boeckman Road Extension - East	Canyon Creek to Wilsonville Road	Three-lane extension with sidewalks and blke lanes	x		\$ 4,400,000	2016-25

							1	2003 dollars	
						2025 RTP Preferred	2025 RTP Financially Constrained	( "*" indicates phasing in financially	RTP
RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	System	System	constrained	Years
6108	Wilsonville TC	Wilsonville	Brown Road Improvements	Wilsonville Road to Evergreen Avenue	Three-lane extension with sidewalks and bike lanes	x		\$ 1,800,000	2010-15
6109	Sherwood TC	Washington Co.	Beef Bend/175th Avenue Realignment	Beel Bend at 175th Avenue	road with 175th Avenue	x	x	\$ 924,000	2016-25
6110	Sherwood TC	Washington Co.	Highway 99W Circulation Improvements Study	99W corridor from Tualatin-Sherwood to Chapman	Study potential of frontage roads on both sides of 99W to manage access	x		n/a	2004-09
6111	Deleted (Construct	ion completed)							
6112	Sherwood TC	Washington Co.	Beef Bend Road Improvements	Bull Mountain Road to Scholls Ferry Road	Widen to four lanes with limited access	x		\$3 465 000	2016-25
6113	Deleted (Construct	lon completed)							
6114	Sherwood TC	Sherwood/WashCo	Edy Road/Sherwood Improvements	Borchers to Pine/3rd Street	Widen; install signals; add bike lanes	x		\$ 1,732,500	2016-25
6115	Sherwood TC	Sherwood/WashCo	Edy Road Improvements	North city limits to 99W	Widen to include sidewalks and bike lanes	x		\$ 1,155,000	2016-25
6116	Sherwood TC	Sherwood/WashCo	Sherwood TC Bicycle/Pedestrian Bridges	Sherwood/Edy/ 99W; Meineke/99W; Sunset/99W		x		\$ 11,550,000	2016-25
6117	Sherwood TC	Sherwood/WashCo	Sherwood TC Pedestrian Improvements	Sherwood Road, Oregon, Pacific and intersecting streets	Improve sidewalks, lighting, crossings, bus shelters and benches	×		\$ 1,732,500	2016-25
		and the second	and the second					φ (,152,300	2010-23
6119	Murray/Scholls TC	Washington Co./Beaverton	Teel Boulevard Extension	Barrows Road to Scholis Farry Road	Construct 2-lane extension with sidewalks and bike lanes to fown center loop and Berrows Spect		a and a framework of the		
6120	Murrav/Scholls TC	Washington Co.	Barrows Roed Improvements	Murray Boulevard to 175th Avenue	Widen to add bike lanes	~	<u>^</u>	\$ 4,000,000	2004-09
		Beaverton/WashCo/	And a second sec	Scholls Ferry Road to Barrows Road at Walnut	Construct 2-lane roadway and bridge, additional turn			\$ 577,500	2016-25
8422	Mumay/Scholls TC	Bequarton	Provider Body Connection	Scholle Free Read to Remove 12	laries et intersections, bike laries, and sidewalks	<u>x</u> –	X	\$ 1,900,000	2004-09
0122	Multay/Schols 1C	Converter		Sciolis Ferty Roboto Barrows Road	Three lane connection with bikeways and sidewarks	X	X	\$ 1,900,000	2010-15
					Reconstruct and widen to three lanes to include bike				
6124	LO Comdor	Clackamas Co.			lanes	X		\$ 3,811,500	2010-15
6125	Deleted (Construct	on completed)							
6126	Deleted (under con	struction)			Widen to five lanes with sidewarks and bike lanes:				
			and the second se	and the second se	Boones Fany Corridor Stugy completed in 2000 with Lake Grove Town Center study work continuing in	100 C			
6127	LO Corridor	Lake Oswego	Boones Ferry Road Improvements-	Kruse Way to Washington Court	2003/04 funded by City. Project will be broken into three phases: upper, middle and lower		×	<ul> <li>A 200 data</li> </ul>	0010.45
6128	Deleted (Constructi	on completed)				<u>~</u>	<u>^</u>	.\$ 5,200,000	2010-15
6129	LO Corridor	Clackames Co.	Bangy Road Intersection Improvements	Bangy Road/Bonita Road intersection	Add traffic signal and turn lanes.	Y	~~~~	F 075 975	2040.45
6130	LO Corridor	Clackamas Cd.	Bangy Road Intersection Improvements	Bangy Road/Meadows Road Intersection	Add traffic signel and jurn tanes	Ŷ		\$ 3/5,3/3 \$ 375,375	2010-15
6131	LO Corridor	Lake Oswego	Willamette River Greenway	Roehr Park to Tryon Creek	shared-use path	× .		\$ 3/5,3/5	2010-15
6122	Laka Grova TC	Clackamas Co	Bonita Road Improvements	SE Bonow Pood to SE Comos Drive				• 540,000	2010-13
6135	Lake Grove TC	Clackamas Co.	Bonnes Ferry Road Bike Lanes	Khine Way to Multinemeth County line	Reconstruct and widen to three lanes	<u> </u>	10 ·	\$ 3,811,500	2010-15
6400		Portland	Beener Same Dadastica Lance		Improve sidewalks, lighting, crossings, bus shelters and	X	X	\$ 635,250	2004-09
0130	Lake Grove IC		ounds reny recestran improvements	Ferwiniger to ICLOSE MBA	Denches	x		\$ 1,155,000	2016-25
0137	Deleted (Study Real	completed)	Wilsonville Road/I-5 Interchange		Construct ramp Improvements (PE and ROW only in	an constant of the			_
6138	Wilsonville TC	ODO //Wilsonville	Wilsonville Road/I-5 Interchange	Town Center Loop to Boones Ferry Road ramps	financially constrained system)	X	×	\$ 20,900,000	2004-09
6139	Wilsonville TC	ODOT/Wilsonville	Improvements (Phase 3)	I-5 in Wilsonville area	Construct auxiliary lanes	X		\$ 11,300,000	2016-25
6140	Wilsonville TC	Wilsonville	Miley Road Improvements	French Prairle to west of I-5	Widen street to four lanes	x		\$ 2,300,000	2010-15

2004 RTP Project List October 31, 2003

			1	<del></del>				2003 dollars	
RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	( <sup>***</sup> indicates phasing in financially constrained	RTP Program Years
A Secondaria	Constant and the second of		and the second state of the second state of the		Accusing right-of-way and construct new attended based on	and the second second		and the second states and	1000
12000				Design of the second	recommendations from 1-5/99W Arterial connection study				
0.0.0		ODOT/MachCo	AND Connector Photo 1 Arigin	1.5 16 8004/	that protects through traffic movements between these				1
6141	Region	ODOT/WashCo.	1-35517 Connector, Filase i Alterial	1010 394	ingrivitys.	<u>x</u> .	<u> </u>	\$ 53,000,000	2004-09
6142	Durham TC	Durham	Upper Boones Ferry Road Improvement	Durham Road to Tualatin River	Widen to 3 lanes with sidewalks and bike lanes	<u>x</u>		\$ 1,000,000	2004-09
S	1. C.							diameter (	
7000	Damascus TC	Clackamas Co.	172nd Avenue Improvements	Foster Road to Highway 212	Widen to five lanes	X	x	\$ 8.085.000	2016-25
					and the second				
7001	Demostrate TC	Clackamas Co	Sunnvalde Road Improvements	172nd Avenue to Highway 212	Viden to five lanes in preferred/3 lanes in strategic and constrained			E (150.000	0010 45
	Carnacco ro					<u> </u>	<u> </u>	a 4,100,000	201013
7002	Damascus TC	Clackamas Co.	Foster Road Improvements	Highway 212 to 172nd Avenue	Widen to five lanes in preferred/3 lanes in strategic	х		\$ 20,790,000	2016-25
7003	Damascus TC	Portland	Foster Road Improvements	172nd Avenua to Jenne Road	Widen to five lanes	v		¢ = 775.000	2016.25
7000	Pumusous ro	Multi- amath O-				^		3,173,000	2010-23
7005	Pleasant Valley TC	Multhoman Co.	190th Avenue Extension	Butter/190th to 1/2nd/Foster Road Intersection	Five lane extension	X		\$ 11,550,000	2010-15
( Constant					Barbara Welch Road, Widen and determine the				
				A second second states and second second second	appropriate cross section of Foster Road from SE Barbara Walch Road to Jenne Road by completing				
	1		the state of the s	Hart States and States and States and States and States and	Phase 2 of the Powell Boulevard/Foster Road Corridor		1.1.1		
7006	Pleasant Valley TC	Portland	SE Foster Improvements	SE 122nd Avenue to Jenne Road	Study in order to meet roadway, transit, pedestrian and blke needs	v	v	* ** 000.000	2040.46
	1 Iotouri Villoy 10				Based on the recommendations from the Powell	<u>^</u>	~	3 14,000,000	2010-10
			An	and the second second second second second	Boulevard/Foster Road Corridor Study (#1228); construct				
					vicinity of SE 174th Avenue/Jenne Road between SE		1.5. See		
				and the second se	Powell Boulevard and Glese Road in Pleasant Valley.				
					Read to three lanes from Powell Boulevard to Foster	1994 (A)			
7007	Pleasant Valley TC	Portland/Gresham	SE 174th North/South Improvements	SE Foster to Powell Boulevard	Road	X	x	\$ 13,000,000	2010-15
7008	Deleted (under con	struction)							
7009	Pleasant Valley TC	Clackamas Co.	SE 145th/147th Bike Lanes	SE Clatsop to SE Monner	Widen to construct blke lanes	X	x	\$ 1,039,500	2010-15
7010	Pleasant Valley TC	Clackamas Co.	SE 162nd Avenue Bike Lanes	SE Monner to SE Sunnyside	Widen to construct blke lanes	X	Х	\$ 392,700	2016-25
7011	Pleasent Valley TC	Clackamas Co.	SC MORHER DIKE LERIES	Sc 147 m to tozno Avenue	Widen to construct bike lanes	<u>x</u>	<u> </u>	\$ 392,700	2016-25
7012	Deleted (Project inc	luded in #2045)							
7013	Deleted (Project inc	luded in #1228)							
					Develop a corridor plan to address N/S access to urban				
7015	Pleasant Valley TC	Metro	Towle/Eastman Corridor Plan	Towle/Eastman from Powell to 190th	reserves	X		n/a	2010-15
					Study a new extension of SE 174th Avenue between				
					in an amendment to planning documents to call for a new				
		Dealler d'Orachard			extension of SE 174th Avenue in lieu of widening Jenne				
7016	Pleasant Valley TC	Metro	Development Study	Jenne Road/174th from Powell to Foster	Boulevard (former project 7007).	×		Dia	2010-16
1.57	,,		4			Series Street Street		1//4	2010-13
7048	Sunshing Valley-DO	Clackamas Co.	242nd Avenue Improvements	Multhometh County line to Hichway 212	Baconstruct and widen to hiras issues				
	Containing Valley KR	Charles and Cook		and a second the second information of	Develop traffic management plan to protect rural	X	X	➡ 4,620,000	2016-25
7020	Sunshine Valley RR	Metro	Regner/222nd Corridor Plan	Regner/222nd Ave from Roberts to Highway 212	character/uses	х		n/a	2016-25
7021	Sunshine Valley RR	Metro	Hogan/242nd Comidor Plan	Hogan/242nd from Paimouist to Highway 212	Develop traffic management plan in urban growth boundary	~		-1-	2004.00
	canonic valoy (((				,	X		n/a	2004-09
7022	Damascus TC	INNEL	Sunnyside Road Frequent bus	Clackamas TC to Damascus TC	Construct improvements that enhance Frequent bus servi	X	X	\$ 913,000	2010-15
7023	Damascus TC	TrlMet	Powell/Foster Rapid Bus	PCBD to Damascus TC	Construct improvements that enhance Rapid bus service	<u>x</u>		See Tri-Met Total	2016-25

RTP #	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	2025 RTP Preferred System	2025 RTP Financially Constrained System	2003 dollars ( "*" indicates phasing in financially constrained	RTP Program Years
7024	Region	TriMet	Transit center	Damascus	Construct transit station to serve Damascus	x		See Tri-Met Total	2016-25
					Initiate a feasibility study of the trail proposed in the Pleasant Valley concept plan to evaluate property				
7025	Region	Various Partners	East Buttes Poweriine Corridor Trail	SE 172nd Avenue to Gresham-Fairview Trail	ownership, alignment options, environmental issues	x		\$ 100,000	2016-25
7026	Pleasant Valley TC	Gresham	Towle Avenue Improvements	Butler Road to Eastman Parkway	Improvements	x		???	2016-25
7027	Pleasant Valley TC	Gresham	Butler Road Improvements	190th Avenue to Regner Road	Construct sidewalks and bike lanes	×		777	2016-25
7028	Pleasant Valley TC	Gresham	Butler Road Improvements	Regner Road to 242nd Avenue	Construct sidewalks and bike lanes	x		777	2016-25
7029	Pleasant Valley TC	Gresham	162nd Avenue improvements	Powell Boulevard to Division Street	sidewalks and bike lanes	x		777	2016-25
7030	Pleasant Valley TC	Gresham	Regner Road Improvements	Butler Road to Roberts Road	Construct sidewalks, bike lanes and intersection Improvements	x		???	2016-25
7031	Pleasant Valley TC	Portland	Clatsop Road Bike Improvements, 1	132nd Avenue to 145th Avenue	Retrofit blke lanes to existing street	x		???	2016-25
7032	Pleasant Valley TC	Portland	Clatsop Road Bike Improvements, 2	Butler Road to Roberts Road	Retrofit blke lanes to existing street	×		777	2016-25
7034	Pleasant Valley TC	Gresham/Mult. Co	Foster Road Extension		New north extension of Foster Road	×	x	\$ 1,700,000	2010-15
7035	Pleasant Valley TC	Greshern/Mult, Co	Glese Road Extension	Glese Road to Foster Road	New extension of Glese Road to Foster Road	x	X	\$ 2,900,000	2018-25
7038	Pleasant Valley TC	Gresham/Mult. Co	190th Avenue Improvements	Butler Road to city limits	Widen to five lanes with sidewalks and bike lanes	X	x	\$ 4,100,000	2016-25
7037	Pleasant Valley TC	Gresham/Mult. Co	172nd Avenue Improvements	Glese Road to Butler Road	bigging street to urban standards with sidewalks and bike lanes	x	x	\$ 1,900,000	2016-25
7038	Pleasant Valley TC	Gresham/Mult_Co	172nd Avenue Improvements	Buller Road to Cheldelin Road	Upgrade street to urban standards with sidewalks and bike lanes.	x	x	\$ 5,600,000	2016-25
-7039	Pleasant Valley TC	Gresham/Mult. Co	Glese Road Improvements	172nd Avenue to 182nd Avenue	Upgrade street to urban standards with sidewalks and bike lanes	X	x	\$ 4,300,000	2016-25
7040	Pleasant Valley TC	Gresham/Mult. Co	Glese Road Improvements	182nd Avenue to 199th Avenue	Upgrade street to urban standards with sidewalks and bike lanes	x	x	\$ 3,000,000	2016-25
7041	Pleasant Valley TC	Gresham/Mult_Co	Foster Road bridge	Foster Roed	Construct bridge crossing	x	×	\$ 1,100,000	2016-25
7042	Pleasant Valley TC	Gresham/Mult. Co	Glese Road Extension bridge	Glese Road	Construct bridge crossing	x	x	\$ 1,100,000	2016-25
7043	Pleasant Valley TC	Gresham/Mult. Co	Butler Road Bridge	Buller Road	Construct bridge crossing	x	x	\$ 1,700,000	2016-25
8000	Region	Metro	Bicycle Travel Demand Forecasting Model	Region-wide	Develop regional bicycle travel demand forecasting model	x	x	\$ 115,500	2004-09
8001	Region	Metro	Bike Safety, Educia Encouragement Pliot. Project	Region-wide	Encourage bicyclist, pedestrian and motorist safety	×	×	\$ 115,500	2004-09
8002	Region	Metro	Expand "Bike Central" Program	Selected Regional Centers and Town Centers	Provide shower, locker and storage facilities for bike commuters	x	×	\$ 346,500	2010-15
8003	Region	Metro	LRT Station Area "Free Bike" Pilot Project	LRT Station Areas throughout the region	Administer free bike program in station areas	x	x	\$ 57,750	2016-25
8004	Region	TriMet	LRT and Transit Station Bike Parking	Selected LRT Station Areas and transit centers	Administer and maintain bloycle lockers	x	x	\$ 57,750	2010-15
8005	Region	Metro	Regional TOD Projects	Region-wide	Flexible funding program to leverage transit-oriented development	x	×	\$ 43,000,000	2004-25
8006	Region	Metro	Alternative transportation strategles study	Region-wide		x		n/a	2016-25
8007	Region	ODOT	Pedestrian/Bicycle Improvements to ODOT Preservation/Maintenance Projects	Various locations in region	Implement Dicycle and pedestrian anhancements as part of preservation and maintenance projects on ODOT facilities	x	×	\$ 10,000,000	2004-25
8008	Region	ODOT	Interchange Access Management	Various interchanges in the region	Implement access management strategies	×		\$ 46,200,000	2004-09
8025	Region	TriMet/SMART	Transit Center Upgrades	Region-wide	New or Improved transit centers at various locations in the region		×	\$ 20,002,072	2004-75
8026	Deleted (Priority Sy	stem dropped)					<u> </u>	w rotovejeto	2004-23
8027	Region	TriMet/SMART	Transit Center Upgrades	Region-wide	New or improved transit centers at various locations in the region	x		\$ 104,702,638	2004-25

						2025 RTP	2025 RTP Financially	2003 dollars (""" Indicates phasing in financially	RTP
RTP#	2040 Link	Jurisdiction	Project Name (Facility)	Project Location	Project Description	System	System	constrained	Years
8028	Region	TriMet	Vehicle Purchases	1.5% per year expansion	Vehicle purchases to provide for expanded service		×	\$ 169,785,000	2004-25
8031	Region	TriMet	Vehicle Purchases	4.5% per year expansion	Vehicle purchases to provide for expanded service	x		\$ 802,725,000	2004-25
8032	Region	InMet/SMART	Bus Operating Facilities	Region-wide	Bus operating facilities	and the second	X	\$ 75,000,000	2004-25
8034	Region	INMEVSMART	Bus Operating Facilities	Region-wide	Bus operating facilities	×		\$ 213,835,281	2004-25
8035	Region	THMet/SMART	Frequent/Rapid Bus Improvements	Baseline Network	priority and reliability improvements		x	\$ 26 297 000	2016-25
8037	Region	TriMet/SMART	Frequent/Rapid Bus Improvements	Preferred Network	Transit stations, improved passenger amenities, bus priority and reliability improvements	x		\$ 152,337,945	2004-25
8038	Begion	TriMet	Tri-Met Park and Ride Lota	Baseline Network	Park-and-ride facilities to serve bus and light rail stops and stations		× 1	F	
8041	Region	TriMet	Tri-Met Park and Ride Lots	Preferred Network	Park-and-ride facilities to serve bus and light rail stops and stations	x		\$ 5,782,970 \$ 89,620,839	2004-25
8042	Backso	SMART	SMART Park and Ridal res	SMART dietect	Park-and-ride facilities to serve bus and commuter rail	and a second second			
8043	Region	TriMet/SMART	Bue Stop Improvements	Region-wide	Bus stop improvements regionande	<u> </u>	<u> </u>	\$ 3,927,000	2004-25
8045	Region	TriMet/SMART	Bus Stop Improvements	Region-wide	Bus stop improvements region-wide			a 7,939,101 \$ 12,011,755	2004-25
8046	Region	TriMet/SMART	Bus Priority Treatments	Region-wide	Bus Priority Treatments	<u> </u>	Ŷ	< 10.801.089	2004-23
8048	Region	TriMet/SMART	Bus Priority Treatments	Region-wide	Bus Priority Treatments	Y		\$ 92,746,462	2010-25
12.25			Priority Pedestrian Access to Transit		Construct improvements that enhance pedestrian access	~		4 03,740,703	2004-23
8049	Region	TriMet	Improvements	Region-wide	to transit - sidewalks, crosswalks, ADA improvements	×	x	\$ 20,000,000	2004-25
8050	Region	Metro/SMART	SMART TDM Program	SMART district	Regional employer outreach, transit marketing, vanpool and carpool, station cars and car sharing programs	x	x	\$ 1,500,000	2004-25
8051	Region	Metro/TriMet	Regional Travel Options TDM Program	Preferred Network	Regional employer outreach, transit marketing, vanpool and carpool, station cars and car sharing programs	×		\$ 47,124,000	2004-25
8052	Region	Metro/TriMet	Regionel Travel Options TDM Program	Financially Constrained	Regional employer outreach, transit marketing, vanpool and carpool, station cars and car sharing programs.		x	<b>\$</b> 16,978,500	2004-25
8053	Region	Metro/TriMet	Region 2040 Initiatives	Region-wide	focations with high regional significance	x	x	\$ 6,063,750	2004-25
8054	Region	Metro/DEQ	ECO Clearinghouse	Region-wide	services	x	x	\$ 1,212,750	2004-25
8055	Region	Metro/TriMet	Transportation Management Associations Innovative Programs	Region-wide	Implementation of innovative transponation solutions in locations with high regional significance	Y	×	\$ 3,000,000	2004.25
			Future Transportation Management		Future implementation and sustainability of TMA's with	~	<u>^</u>	0,000,000	2004-25
8055	Region	Metro/TriMet	Associations Start-Up and Sustainability	Region-wide	employers	x	x	\$ 4,000,000	2004-25
8057	Region	TriMet	LIFT Vehicle Purchases	Region-wide	4 percent per year expansion	×	x	\$ 16,890,000	2004-09
8058	Region	TriMet	Ride Connection Vehicle Purchases	Region-wide	Purchase five vehicles per year	x	x	\$ 4,767,600	2004-09
				Total Capital Costs for each Network in Billions of 2003 Dollars		\$9.499	\$4.239		

### **Regional Transportation Plan Update Calendar**

- October 31 Public comment period begins; staff recommendation on draft 2004 RTP released for 30-day public comment period; draft RTP and conformity determination submitted to FHWA and FTA to begin review
- **November 3** Air quality conformity analysis begins
- November 5 MTAC comments on draft 2004 RTP
- November 12 MPAC comments on draft 2004 RTP
- November 13 JPACT tentative action on draft 2004 RTP
- November 13 Metro Council first reading of Ordinance on draft 2004 RTP
- November 26 TPAC review and discussion of draft 2004 RTP and air quality conformity analysis
- **December 4** Public hearing on draft 2004 RTP; public comment period ends at 5 p.m.
- December 5 TPAC special meeting to comment on draft 2004 RTP
- December 10 Tentative final MPAC action on 2004 RTP
- December 11 Tentative final JPACT action on 2004 RTP
- **December 11** Metro Council second reading of Ordinance and consideration of adoption of 2004 Regional Transportation Plan

#### FOLD HERE



Place first class postage here.

Metro 600 NE Grand Avenue Portland, Oregon 97232 Attention: Marilyn Matteson

# How to Comment on the update to the 2004 Regional Transportation Plan

The public comment period for the 2004 Regional Transportation Plan (RTP) begins on October 31, 2003 and concludes with a public hearing on December 4, 2003. You may submit comments online at Metro's website:

#### www.metro-region.org/rtp

Comments and questions may also be mailed using the form below, or left on Metro's Transportation hotline at (503) 797-1900, Option 2.

#### **Comments:**

		i
······································		
	· · · · · · · · · · · · · · · · · · ·	

### Submitted by:

· ·					
Name	· · · · · · · · · · · · · · · · · · ·				
Street Address	City/Zip				
Phone	E-Mail				
Send me more info:					
2000 RTP Document CD	Other RTP Info:				
Please add me to the RTP interested citizens mailing/e-mail lists					




# 2004 Regional Transportation Plan **Technical Update**

# October 31, 2003



PEOPLE PLACES OPEN SPACES

# Metro People places • open spaces

Metro serves 1.3 million people who live in Clackamas, Multnomah and Washington counties and the 24 cities in the Portland metropolitan area. The regional government provides transportation and land-use planning services and oversees regional garbage disposal and recycling and waste reduction programs.

Metro manages regional parks and greenspaces and owns the Oregon Zoo. It also oversees operation of the Oregon Convention Center, the Portland Center for the Performing Arts and the Portland Metropolitan Exposition (Expo) Center, all managed by the Metropolitan Exposition Recreation Commission.

# Your Metro representatives

Metro Council President – David Bragdon Metro Councilors – Rod Park, District 1; Brian Newman, District 2; Carl Hosticka, District 3; Susan McLain, District 4; Rex Burkholder, District 5; Rod Monroe, District 6. Auditor – Alexis Dow, CPA

Metro's web site: www.metro-region.org

Metro 600 NE Grand Ave. Portland, OR 97232-2736 (503) 797-1700 TDD (503) 797-1804

Printed on 100 percent recycled paper, 30 percent post-consumer fiber



# 2004 Regional Transportation Plan Technical Update Highlights

# **Recent Technical Amendments**

Since the last update to the Regional Transportation Plan (RTP) in August 2000, the Metro Council adopted a number of technical amendments that were mandated by the Oregon Land Conservation and Development Commission (LCDC) as part of the RTP acknowledgement process. These amendments were adopted in 2002, and are reflected in the published version of the RTP.

# **Proposed Technical Amendments**

Since the last RTP update, a number of corridor studies and concept plans for new urban areas have been completed, and approved by local or regional officials, or are about to be completed. The results of these studies include a number of technical changes to the RTP implementation chapter that frame future work that must be still be completed, and delete technical requirements that have been addressed by these studies. The changes reflected in the proposed technical amendments include:

- Powell-Foster Corridor Study Phase I Recommendations
- I-5 South Wilsonville Area Study
- Regional Travel Option Strategic Planning
- RTP Modal Target Study
- Damascus/Boring Concept Plan
- Transportation Adequacy Policy Transportation Planning Rule Requirements
- National Highway System (NHS) Routes Update

The proposed amendments are detailed in the attached strikethrough/underscore version of Chapter 6 of the 2000 Regional Transportation Plan. A number of other minor "housekeeping" edits are also shown in the proposed amendments to this chapter.

# **CHAPTER 6**

# Implementation

# 6.0 Introduction

The policies and transportation strategy in this plan reflect federal, state and regional planning requirements, while balancing the need for transportation improvements with increasingly limited funding. As such, the plan serves as a 20-year blueprint for transportation improvements in the region. However, there is much work to be done. Implementing this plan will require a cooperative effort by all jurisdictions responsible for transportation planning in the region, and will involve the following:

- adoption of regional policies and transportation strategies in local plans
- a concerted regional effort to secure needed funding to build planned transportation facilities and maintain and operate an expanded transportation system
- construction of the transportation improvements needed to serve expected growth and address existing safety concerns
- focusing strategic improvements that leverage key 2040 Growth Concept components
- periodic updates of the plan to respond to development trends and the associated changes in travel demand
- incorporating transportation solutions from corridor-level or subarea refinement plans
- ongoing monitoring for consistency with the local TSP development and other implementing agency plans, including the Oregon Department of Transportation's Six-Year Program and Tri-Met's Transit Development Plan

The transportation strategy described in Chapter 5 of the plan will not meet all of the region's 20year transportation needs, but it is a significant first step towards achieving the preferred system. Instead, it represents a pragmatic balance between the need to maintain existing infrastructure and keep pace with expected growth in the region and the realities of limited transportation funding. As the region moves forward with implementation of this plan, a new paradigm for how we view the transportation system must evolve. Like other urban utilities, transportation infrastructure must increasingly be viewed as a scarce commodity that should be managed and allocated to reflect the growing cost and complexity of expanding the system.

This chapter describes the steps necessary to implement the plan, including:

- compliance with federal, state and regional planning requirements
- implementation of the plan through local TSPs

- relationship to the Metropolitan Transportation Improvement Plan
- process for updating and amending the plan
- process for completing refinement plans, and locations where refinement plans must be completed
- outstanding issues that cannot be addressed at this time, but must be considered in future updates to the plan

Following this chapter are other important resources for implementing the plan, including appendices that describe proposed transportation projects and strategies in more detail, and a separate background document that describes much of the methodology used to develop this plan.

# 6.1 Demonstration of Compliance with Federal Requirements

# 6.1.1 Metropolitan Planning Required by TEA-21

The metropolitan planning process outlined by Congress in the federal Transportation Equity Act for the 21st Century (TEA-21) establishes a cooperative, continuous and comprehensive framework for making transportation investment decisions in metropolitan areas throughout the United States. Program oversight is a joint FHWA/FTA responsibility. The federal planning requirements were originally promulgated as part of the 1992 federal Intermodal Surface Transportation Efficiency Act (ISTEA), and were substantially reaffirmed by TEA-21 in 1998.

Among the most significant continuing provisions of TEA-21 for the Metro region are the following planning requirements:

- Metro, in cooperation with the ODOT, Tri-Met and other transit operators, remain responsible for determining the best mix of transportation investments to meet metropolitan transportation needs.
- Metro is responsible for adopting the Regional Transportation Plan.
- Metro is responsible for adopting the MTIP. ODOT must include the MTIP without change in the STIP. The Governor is designated to resolve any disagreements between Metro's MTIP and ODOT's STIP.
- The RTP must provide a 20-year planning perspective, addressing air quality consistency, fiscal constraint and public involvement requirements established under the original ISTEA.
- The Oregon Department of Environmental Quality must adopt an Oregon State Implementation Plan (SIP). The SIP includes actions that must be adopted by Metro and results in an emissions budget for carbon monoxide and ozone. Metro must demonstrate

progress toward implementing the actions identified in the SIP and demonstrate conformity with the carbon monoxide and ozone emissions budget.

- A Congestion Management System (CMS) is required in larger metropolitan areas that are designated as air quality maintenance or non-attainment areas. The Portland metropolitan region was designated as a maintenance area in 1997. Highway projects that increase single-occupant vehicle capacity must be consistent with the CMS.
- The CMS continues the requirement that alternatives to motor vehicle capacity increases be evaluated prior to adding single-occupant vehicle projects.
- Federal Highway Administration and Federal Transit Administration certification of the planning process is required in larger metropolitan areas, including the Metro region.

TEA-21 consolidated the 16 planning factors from the original ISTEA into seven-broad areas to be considered in the planning process (contained in section 1203(f) of the federal act). These factors are advisory, and failure to consider any one of the factors is not reviewable in court. However, the seven factors seek to:

- Support the economic vitality of the metropolitan area, especially by enabling global competitiveness, productivity and efficiency
- Increase the safety and security of the transportation system for motorized and nonmotorized users
- Increase the accessibility and mobility options available to people and for freight
- Protect and enhance the environment, promote energy conservation and improve quality of life
- Enhance the integration and connectivity of the transportation system, across and between modes, for people and freight
- Promote efficient system management and operation
- Emphasize the preservation of the existing transportation system

Each of these factors has been addressed through RTP policies identified in Chapter 1 of this plan and selection of the proposed transportation projects and programs identified in Chapter 3 of this plan. Specific sections that address the seven federal planning factors are detailed in the RTP Background Document.

In addition to changes to the ISTEA planning factors and scope of regional transportation planning, TEA-21 also modified several other elements of the federal ISTEA. Under the revised provisions, the Regional Transportation Plan must:

- Include operation and management of the transportation system in the general objectives of the planning process
- Address transportation planning area boundary relationship to non-attainment area boundaries; boundaries established on date of enactment remain as is, but future expansions of non-attainment area boundaries do not force expansion of transportation planning area unless agreed to by the Governor and Metro
- Coordinate with neighboring MPOs where a project crosses planning area boundaries
- Specifically identify freight shippers and users of public transit on the list of stakeholders to be given opportunity to comment on plans and TIPs
- Cooperate with ODOT and transit agencies in the development of financial estimates that support plan and TIP development
- Identify projects that will be implemented within a forecast of revenues that can be reasonably expected to be available over the life of the Regional Transportation Plan. The Regional Transportation Plan may also include additional projects that may be identified for illustrative purposes, and would be included in plans and TIPs if additional resources were available. Additional action by ODOT, Metro and the Secretary of Transportation is required to advance such projects

The RTP meets the TEA-21 provisions through its policies and project selection criteria. A summary of RTP compliance with these provisions is included in the RTP Background Document.

#### 6.1.2 Air Quality Conformity: Criteria that Constitutes a Conformed Plan

The 20202025 Preferred and Priority Systems both requires new revenue sources and go beyond federal requirements that long-range transportation plans be based upon "constrained resources." Air quality conformity of this plan will be based on a scaled-down 20202025 Priority-Preferred System that can likely be implemented within the federally defined fiscally constrained level of reasonably available resources. This system will be termed the 20202025 Fiscally Financially Constrained System. Air quality conformity entails:

- Making reasonable progress on Transportation Control Measures as identified in the SIP
- Staying within the carbon monoxide and ozone emissions budgets set for transportation with the SIP based upon a fiscally constrained transportation network

Portland is currently designated a maintenance area for the National Ambient Air Quality Standards (NAAQS) for ozone and carbon monoxide under the Clean Air Act Amendments of 1990.

# 6.1.3 Demonstration of Air Quality Conformity

The Financially Constrained System and the 2020 Priority System have been found to conform to federal air quality requirements. Appendix 4.0 provides detailed information to support this finding on the air quality conformity analysis to be completed on the 2025 Financially Constrained System.

# 6.2 Demonstration of Compliance with State Requirements

This section identifies the applicable state regulations for the regional transportation system plan and identifies the corresponding provisions contained in this RTP. Findings of Fact and Conclusions of Law explaining TPR compliance, which were will be adopted with the 2000 2004 RTP, are found and will be included in Appendix 5.0.

# 6.2.1 System Plan Required by Oregon Transportation Planning Rule

The Oregon Transportation Planning Rule (TPR) sets forth a number of requirements for Metro's Transportation System Plan (TSP). This RTP has a number of purposes. This Plan is adopted as the regional functional plan for transportation and the federal metropolitan transportation plan, as well as the regional TSP under state law. The RTP as regional TSP, must address provisions of Oregon Administrative Rule 660.012.000 applicable to regional TPSs.

The following TPR provisions are addressed in the portions of this multipurpose plan indicated under each applicable TPR requirement. Together, these portions of the <u>2000-2004</u> RTP comprise the regional TSP. Other portions of the RTP not indicated under the applicable TPR requirement address regional and federal planning issues beyond the regional TSP under this administrative rule.

• 660.012.0015(2) - MPOs shall prepare TSPs in compliance with TPR Metro is required to prepare a Transportation System Plan (TSP) for facilities of regional significance within Metro's jurisdiction. The portions of the 2000-2004 RTP which constitutes the regional transportation system plan are provisions of Chapters 1, 2, 5, 6 and the Appendix which address regional TSP issues, including the priority system of improvements.

• 660.012.0020 - TSP adequately serves regional transportation needs The RTP fully addresses this requirement by identifying the region's 20-year transportation needs in Chapter 2, including the future motor vehicle, public transportation, bicycle, pedestrian and freight system improvements, and complementary demand management, parking and financing programs in Chapter 5 adequate to respond to these identified needs.

• 660.012.0025 - Complying with Statewide Planning goals This is the first regional TSP adopted in the metro region. As such, the 2000–2004 RTP identifies transportation needs for regional facilities for the purpose of informing regional and local transportation and land-use planning. In some cases where a need has been established, decisions regarding function, general location and mode are deferred to a

refinement plan or local TSP. In these cases, the findings in Chapter 5 describe how these needs are met for the purpose of RTP analysis, and Sections 6.7.5 and 6.7.6 of this chapter establish the need for refinement planning, and base assumptions for specific refinement plans that are needed to ensure consistency with the RTP.

#### 660. 012.0025(3) - Refinement plans allowed

A number of refinement plans are proposed in the 2000 RTP, including 16 corridor plans and three area plans. Section 6.7 of this chapter describes the purpose and scope of refinement plans.

#### 660.012.0030 - Determination of transportation needs

The project development phase of the 2000–2004 RTP followed the congestion management requirements of Section 6.6.3 of this chapter, which incorporates the TPR requirements for determining transportation needs.

#### 660.012.0035 - Transportation system evaluation required

This 2000-2004 RTP represents a minor update to the 2000 RTP, which was is built on an extensive foundation of modeling and analysis. The Region 2040 project included five separate land use and transportation scenarios, including the alternative adopted and acknowledged in the 1995 Regional Urban Growth Goals and Objectives as the 2040 Growth Concept. A detailed transportation system was developed and modeled for each scenario, and the lessons learned from this effort were the starting point for the 2000 RTP update. Next, a level-of-service alternatives analysis was developed to further refine the region's system performance standards. Finally, the system development component of the 2000 RTP update included four separate rounds of modeling and analysis that combined the principles of the Region 2040 project and the level of service analysis.

For the purpose of complying with this requirement, the <u>Priority\_Preferred\_System</u> in Chapter 5-3 of the 2000-2004 RTP establishes a scale of the improvements that are adequate to meet state and regional travel needs in the Metro area, including the needs of the disadvantaged, the movement of goods and the protection of farm and forest resources within rural reserves.

#### 660.012.0035(4) - Reduction in vehicle miles traveled per capita

The 2000-2004 RTP addresses this requirement through the non-SOV modal targets set forth in Table 1.3 of this plan. The modal targets are linked to the 2040 Growth Concept, and if met, would result in satisfying the required 10 percent reduction in vehicle miles traveled per capita over the 20-year plan period. The non-SOV modal targets set the context for transportation improvements proposed in this plan. The analysis in Chapter 5 establishes that the region is making substantial progress toward meeting this TPR requirement, though the modal targets would not be met in all areas, due to the relative state of urbanization at the conclusion of the planning period. Areas with the greatest concentration of mixed-use development and quality transit service will easily meet the targets, while areas that are still developing are expected to meet the targets beyond the 20-year plan period.

These findings represent the good faith effort required to comply with this element of the TPR. An outstanding issue in Section 6.8.10 of this chapter directs future updates of the RTP to expand on alternative measures that both comply with the TPR, and improve on the plan's ability to identify appropriate transportation projects to meet identified needs.

660.012.0035(6) - Measures and objectives required for non-auto travel

The non-SOV modal targets in Table 1.3 of this plan provide the basic framework for compliance with this TPR provision, which requires a number of measures for demonstrating reduced reliance on the automobile. Other policies in Chapter 1 of this plan complement the non-SOV modal targets, and findings in Chapter 5-3 of this plan demonstrate a reduced reliance on the automobile based on the proposed system improvements.

660.012.0040 - Transportation funding program

The project descriptions in Appendix 1.1 and financial analysis in Chapter 4 of this plan satisfy the various TPR transportation funding requirements. Benchmarks in Section 6.5.3 of this chapter will address TPR requirements for implementation of the RTP through the MTIP.

660.012.0050 - Transportation project development Section 6.7 of this chapter establishes the regional project development requirements for improvements included in the RTP. These and other related requirements are consistent with TPR provisions for project development.

Metro's adoption of the 2000 2004 RTP provisions that address these applicable provisions of the TPR establishes the regional TSP for the Metro region. Through the consistency review process, local TSPs will be evaluated to ensure that local strategies needed to satisfy the above regional planning requirements are implemented. However, local TSPs are not required to make specific findings on these TPR provisions for the regional system, since the RTP establishes compliance for the Metro region. Appendix 5.0 <u>will</u> includes full findings of compliance with the TPR.

6.2.2 Regional TSP Provisions Addressed Through Local TSPs

The 2000-2004 RTP establishes compliance for regional TSP requirements with the policies, projects and financial analysis contained in this plan. Local consistency with the 2004 2000 RTP is described in Section 6.4.1. However, implementation of some regional TSP requirements will occur only through local implementation of RTP policies. These include adoption of the modal targets specified in Policy 19.0 of Chapter 1, and in parking management requirements contained in Title 2 of the Urban Growth Management Functional Plan. Local adoption of the Chapter 1 modal targets is necessary to demonstrate compliance with the VMT/Capita reduction findings described in Chapter 5.3 of the plan.

6.2.3 Special Designations in the Oregon Highway Plan (OHP)

The Oregon Highway Plan (OHP) establishes three special district designations for certain areas along state-owned facilities. The purpose of the designations is to respond to unique community access and circulation needs, while maintaining statewide travel function. Though these special districts are generally identified jointly between ODOT and local jurisdictions, the RTP establishes

6-7

a policy framework that supports these OHP designations through the 2040 Growth Concept and corresponding regional street design classifications contained in Section 1.3.5. The following is a summary of how RTP street design designations correspond to the OHP special district classifications:

• Special Transportation Area (STA): This designation is intended to provide access to community activities, businesses and residences along state facilities in a downtown, business district or community center. In these areas, the OHP acknowledges that local access issues outweigh highway mobility, except on certain freight routes, where mobility needs are more balanced with local access.

The RTP addresses this OHP designation through the boulevard design classifications, located in the 2040 central city, regional center, town center and main street land use components. In the Metro region, state routes designated as boulevards that also meet other standards as defined in the OHP, are eligible to be designated STAs. Further, the application of the boulevard design classifications also factors in major freight corridors, and this design classification is generally not applied to such routes.

• *Commercial Center:* This designation applies to relatively large (400,000 square feet) commercial centers located along state facilities. In these areas, the OHP allows for consolidate access roads or driveways that serve these areas, but such access is subject to meeting OHP mobility standards on the state highway serving the center. If the center has consolidated access roads and meets other OHP standards, the OHP mobility standard may be reduced.

The RTP supports this OHP designation with the throughway design classifications, which include freeway and highway design types. The throughway designs are mobility-oriented, and generally apply to routes that form major motor vehicle connections between the central city, regional centers and intermodal facilities. The throughway design classifications support the concept of limiting future access on a number of state facilities in the region that are designated as principal routes in the RTP.

• Urban Business Area (UBA): This designation recognizes existing commercial strips or centers along state facilities with the objective of balancing access need with the need to move through-traffic.

In the Metro region, these areas are generally designated as mixed-use corridors and neighborhoods in the 2040 Growth Concept, and a corresponding regional or community street design classification in the RTP which calls for a balance between motor vehicle mobility, and local access. These designs are multi-modal in nature, and include transit, bicycle and pedestrian design features, consistent with the OHP designation. The regional and community street classification can also be found in some regional and town centers, and where these are state routes, the facility is eligible for the OHP designation of Urban Business Area.

#### 6.2.4 Compliance with State Requirements

#### Compliance with Statewide Planning Goals

Together, the RTP and city and county TSPs that implement the RTP will constitute the land use decision about need, mode, and function and general location of planned transportation facilities and improvements shown in the RTP. As the regional transportation system plan, the RTP constitutes the land use decision about need, mode and function of planned transportation facilities and improvements. The RTP also identifies the general location of planned transportation facilities facilities and improvements.

The land use decision specifying the general location of planned regional transportation facilities and improvements will be made by cities and counties as they develop and adopt local TSPs that implement the RTP. While the specific alignment of a project may be incorporated into a TSP, such decisions are subject to the project development requirements in Section 6.7, and must include findings of consistency with applicable statewide planning goals, as described below.

In preparing and adopting local TSPs, cities and counties will prepare findings showing how specific alignment of planned regional facilities or general location or specific alignment of local facilities is consistent with provisions of the RTP, acknowledged comprehensive plans and applicable statewide planning goals, if any. If the actual alignment or configuration of a planned facility proposed by a city or county is inconsistent with the general location of a facility in the RTP, the process described in Section 6.4 to resolve such issues shall be used prior to a final land use decision by a city or county.

This section describes how cities and counties will address consistency with applicable local comprehensive plans and statewide planning goals.

#### General Location of Planned Transportation Facilities

Maps included in the RTP illustrate the general location of planned transportation facilities and improvements. For the purposes of this plan, the general location of transportation facilities and improvements is the location shown on maps adopted as part of this plan and as described in this section. Where more than one map in the RTP shows the location of a planned facility, the most detailed map included in the plan shall be the identified general location of that facility.

Except as otherwise described in the plan, the general location of planned transportation and facilities is as follows:

For new facilities, the general location includes a corridor within 200 feet of the location depicted on the maps included within the RTP. For interchanges, the general location corresponds to the general location of the crossing roadways. The general location of connecting ramps is not specified. For existing facilities that are planned for improvement the general location includes a corridor within fifty feet of the existing right-of-way. For realignments of existing facilities the general location includes a corridor within 200 feet of the segment to be realigned, measured from the existing right-of-way or as depicted on the plan map.

Local transportation system plans and project development are consistent with the RTP if a planned facility or improvement is sited within the general location shown on the RTP maps and described

above in this section. Cities and counties may refine or revise the general location of planned facilities as they prepare local transportation system plans to implement the RTP. Such revisions may be appropriate to lessen project impacts, or to comply with applicable requirements in local plans or statewide planning goals. A decision to authorize a planned facility or improvement outside of the general location shown and described in the RTP requires an amendment to the RTP to revise the proposed general location of the improvement.

# Transportation Facilities and Improvements authorized by existing acknowledged comprehensive plans

New decisions are required to authorize transportation facilities and improvements included in the RTP that are not authorized by the relevant jurisdiction's acknowledged comprehensive plan on August 10, 2000. Many of the facilities and improvements included in the RTP are currently authorized by the existing, acknowledged comprehensive plans. Additional findings demonstrating consistency with an acknowledged plan or the statewide planning goals are required only if the facility or improvement is not currently allowed by the jurisdiction's existing acknowledged comprehensive plan. Additional findings would be required if a local government changes the function, mode or general location of a facility from what is currently provided for in the acknowledged comprehensive plan.

#### Applicability of Statewide Planning Goals to decisions about General Location

Several statewide planning goals include "site specific" requirements that can affect decisions about the general location of planned transportation facilities. These include:

Goal 5	Open Spaces, Scenic, Historic and Natural Resources
Goal 7	Natural Hazards and Disasters
Goal 9	Economic Development, as it relates to protection of sites for specific uses (i.e. such as sites for large industrial uses)
Goal 10	Housing, as it relates to maintaining a sufficient inventory of buildable lands to meet specific housing needs (such as the need for multi-family housing)
Goal 15	Willamette River Greenway

Generally, compliance with the goals is achieved by demonstrating compliance with an acknowledged comprehensive plan. If City and county plans have been acknowledged to comply with the Goals and related rules, a planned improvement consistent with that plan is presumed to comply with the related goal requirement. Cities and counties may adopt the general location for needed transportation improvements, and defer findings of consistency with statewide planning goals to the project development phase. However, specific alignment decisions included in a local TSP must also include findings of consistency with applicable statewide planning goals.

In some situations, the Statewide Planning Goals and related rules may apply in addition to the acknowledged plan. This would occur, for example, if the jurisdiction is in periodic review, or an adopted statewide rule requirement otherwise requires direct application of the goal. Cities and

6-10

counties will assess whether there are applicable goal requirements, and adopt findings to comply with applicable goals, as they prepare local transportation system plans to implement the regional transportation plan.

If in preparing a local TSP, a city or county determines that the identified general location of a transportation facility or improvement is inconsistent with an applicable provision of its comprehensive plan or an applicable statewide planning goal requirement, it shall:

- propose a revision to the general location of the planned facility or improvement to accomplish compliance with the applicable plan or goal requirement. If the revised general location is outside the general location specified in the RTP, this would require an amendment to the RTP; or
- propose a revision to the comprehensive plan to authorize the planned improvement within the general location specified in the RTP. This may require additional goal findings, for example, if a goal-protected site is affected.

#### Effect of an Approved Local TSP on Subsequent Land Use Decisions

Once a local TSP is adopted and determined to comply with the RTP and applicable local plans and statewide planning goals, the actual alignment of the planned transportation facility or improvement is determined through the project development process. Subsequent actions to provide or construct a facility or improvement that are consistent with the local TSP may rely upon and need not reconsider the general location of the planned facility.

Additional land use approvals may be needed to authorize construction of a planned transportation improvement within the general location specified in an adopted local transportation system plan. This would occur if the local comprehensive plan and land use regulations require some additional review to authorize the improvement, such as a conditional use permits. Generally, the scope of review of such approvals should be limited to address siting, design or alignment of the planned improvement within the general location specified in the local TSP.

# 6.3 Demonstration of Compliance with Regional Requirements

In November 1992, the voters approved Metro's Charter. The Charter established regional planning as Metro's primary mission and required the agency to adopt a Regional Framework Plan (RFP). The plan was subsequently adopted in 1997, and now serves as the document that merges all of Metro's adopted land-use planning policies and requirements. Chapter 2 of the Regional Framework Plan describes the different 2040 Growth Concept land-use components, called "2040 Design Types," and their associated transportation policies. The Regional Framework Plan directs Metro to implement these 2040 Design Types through the RTP and Metropolitan Transportation Improvement Program (MTIP). These requirements are addressed as follows:

• Chapter 1 of the updated RTP has been revised to be completely consistent with applicable framework plan policies, and the policies contained in Chapter 1 of this plan incorporate all of the policies and system maps included in Chapter 2 of the framework plan. These policies served as a starting point for evaluating all of the system improvements proposed in this plan, and the findings in Chapter 3 and 5 of the

RTP demonstrate how the blend of proposed transportation projects and programs is consistent with the Regional Framework Plan and 2040 Growth Concept.

• The MTIP process has also been amended for consistency with the Regional Framework Plan. During the Priorities 2000 MTIP allocation process, project selection criteria were based on 2040 Growth Concept principles, and funding categories and criteria were revised to ensure that improvements critical to implementing the 2040 Growth Concept were adequately funded.

Prior to completion of this updated the 2000 RTP, several transportation planning requirements were included in the *Urban Growth Management Functional Plan* (UGMFP), which was enacted to address rapid growth issues in the region while the Regional Framework Plan and other long-range plans were under development. This The 2000 RTP now replaces replaced and expandeds the performance standards required for all city and county comprehensive plans in the region contained in Title 6 of the UGMFP. See Sections 6.4.4 through 6.4.7, 6.6, 6.6.3 and 6.7.3. In addition, parking policies contained in this plan were developed to complement Title 2 of the UGMFP, which regulates off-street parking in the region. See Section 1.3.6, Policy 19.1. Therefore, this RTP serves as a discrete functional plan that is both consistent with, and fully complementary of the UGMFP.

To ensure consistency between the 2000-2004 RTP and local transportation system plans (TSPs), Metro shall develop a process for tracking local TSP project and functional classification refinements that are consistent with the RTP, and require a future amendment to be incorporated into the RTP. Such changes should be categorized according to degrees of significance and impact, with major changes subject to policy-level review and minor changes tracked administratively. This process should build on the established process of formal comment on local plan amendments relevant to the RTP.

# 6.4 Local Implementation of the RTP

#### 6.4.1 Local Consistency with the RTP

The comprehensive plans adopted by the cities and counties within the Metro region are the mechanisms by which local jurisdictions plan for transportation facilities. These local plans identify future development patterns that must be served by the transportation system. Local comprehensive plans also define the shape of the future transportation system and identify needed investments. All local plans must demonstrate consistency with the RTP as part of their normal process of completing their plan or during the next periodic review. Metro will continue to work in partnership with local jurisdictions to ensure plan consistency.

The 2000-2004 RTP is Metro's regional functional plan for transportation. Functional plans by state law include "recommendations" and "requirements." The listed RTP elements below are all functional plan requirements. Where "consistency" is required with RTP elements, those elements must be included in local plans in a manner that substantially complies with that RTP element. Where "compliance" is required with RTP elements, the requirements in those elements must be included in local plans as they appear in the RTP.

For inconsistencies, cities and counties, special districts or Metro may initiate the dispute resolution process detailed in this chapter prior to action by Metro to require an amendment to a local comprehensive plan, transit service plan or other facilities plan. Specific elements in the 2000 RTP that require city, county and special district compliance or consistency are as follows:

- Chapter 1 Consistency with policies, objectives, motor vehicle level-of-service measure and modal targets, system maps and functional classifications including the following elements of Section 1.3:
  - regional transportation policies 1 through 20 and objectives under those policies
  - all system maps (Figures 1.1 through 1.19, including the street design, motor vehicle, public transportation, bicycle, pedestrian and freight systems)
  - motor vehicle performance measures (Table 1.2), or alternative performance measures as provided for in Section 6.4.7(1)
  - regional non-SOV modal targets (Table 1.3)

Chapter 2 Consistency with the 20202025 population and employment forecast contained in Section 2.1 and 2.3, or alternative forecast as provided for in Section 6.4.9 of this chapter, but only for the purpose of TSP development and analysis.

Chapter 6 Compliance with the following elements of the RTP implementation strategy:

- Local implementation requirements contained in Section 6.4
- Project development and refinement planning requirements and guidelines contained in Section 6.7

For the purpose of local planning, all remaining provisions in the RTP are recommendations unless clearly designated in this section as a requirement of local government comprehensive plans. All local comprehensive plans and future amendments to local plans are required by state law to be consistent with the adopted RTP. For the purpose of transit service planning, or improvements to regional transportation facilities by any special district, all of the provisions in the RTP are recommendations unless clearly designated as a requirement. Transit system plans are required by federal law to be consistent with adopted RTP policies and guidelines. Special district facility plans that affect regional facilities, such as port or passenger rail improvements, are also required to be consistent with the RTP.

The state Transportation Planning Rule (TPR) requires most cities and counties in the Metro region to adopt local Transportation System Plans (TSPs) in their comprehensive plans. These local TSPs are required by the TPR to be consistent with the RTP policies, projects and performance measures identified in this section.

#### 6.4.2 Local TSP Development

Local TSPs must identify transportation needs for a 20-year planning period, including needs for regional travel within the local jurisdiction, as identified in the RTP. Needs are generally identified either through a periodic review of a local TSP or a specific comprehensive plan amendment. Local TSPs that include planning for potential urban areas located outside the urban growth boundary shall also include project staging that links the development of urban infrastructure in these areas to future expansion of the urban growth boundary. In these areas, local plans shall also prohibit the construction of urban transportation improvements until the urban growth boundary has been expanded and urban land use designations have been adopted in local comprehensive plans.

Once a transportation need has been established, an appropriate transportation strategy or solution is identified through a two-phased process. The first phase is system-level planning, where a number of transportation alternatives are considered over a large geographic area such as a corridor or local planning area, or through a local or regional Transportation System Plan (TSP). The purpose of the system-level planning step is to:

- consider alternative modes, corridors, and strategies to address identified needs
- determine a recommended set of transportation projects, actions, or strategies and the appropriate modes and corridors to address identified needs in the system-level study area

The second phase is project-level planning (also referred to as project development), and is described separately in this chapter in Section 6.7.

Local TSP development is multi-modal in nature, resulting in blended transportation strategies that combine the best transportation improvements that address a need, and are consistent with overall local comprehensive plan objectives.

### 6.4.3 Process for Metro Review of Local Plan Amendments, Facility and Service Plans

Metro will review local plans and plan amendments, and facility plans that affect regional facilities for consistency with the RTP. Prior to adoption by ordinance, local TSPs shall be reviewed for consistency with these elements of the RTP. Metro will submit formal comment as part off the adoption process for local TSPs to identify areas where inconsistencies with the RTP exist, and suggest remedies.

Upon adoption of a local TSP, Metro will complete a final consistency review, and a finding of consistency with applicable elements of the RTP will be forwarded to the state Department of Land Conservation and Development (DLCD) for consideration as part of state review of local plan amendments or local periodic review. A finding of non-compliance for local TSPs that are found to be inconsistent with the RTP will be forwarded to DLCD if conflicting elements in local plans or the RTP cannot be resolved between Metro and the local jurisdiction.

6-14

The following procedures are required for local plan amendments:

- 1. When a local jurisdiction or special district is considering plan amendments or facility plans which are subject to RTP local plan compliance requirements, the jurisdiction shall forward the proposed amendments or plans to Metro prior to public hearings on the amendment.
- 2. Within four weeks of receipt of notice, the Transportation Director shall notify the local jurisdiction through formal written comment whether the proposed amendment is consistent with RTP requirements, and what, if any, modifications would be required to achieve consistency. The Director's finding may be appealed by both the local jurisdiction or the owner of an affected facility, first to JPACT and then to the Metro Council.
- 3. A jurisdiction shall notify Metro of its final action on a proposed plan amendment.
- 4. Following adoption of a local plan, Metro shall forward a finding of consistency to DLCD, or identify inconsistencies that were not remedied as part of the local adoption process.

### 6.4.4 Transportation Systems Analysis Required for Local Plan Amendments

This section applies to city and county comprehensive plan amendments or to any local studies that would recommend or require an amendment to the Regional Transportation Plan to add significant single occupancy vehicle (SOV) capacity to the regional motor vehicle system, as defined by Figure 1.12. This section does not apply to projects in local TSPs that are included in the 2000-2004 RTP. For the purpose of this section, significant SOV capacity is defined as any increase in general vehicle capacity designed to serve 700 or more additional vehicle trips in one direction in one hour over a length of more than one mile. This section does not apply to plans that incorporate the policies and projects contained in the RTP.

Consistent with Federal Congestion Management System requirements (23 CFR Part 500) and TPR system planning requirements (660-12), the following actions shall be considered when local transportation system plans (TSPs), multi-modal corridor and sub-area studies, mode specific plans or special studies (including land-use actions) are developed:

- 1. Transportation demand strategies that further refine or implement a regional strategy identified in the RTP
- 2. Transportation system management strategies, including intelligent Transportation Systems (ITS), that refine or implement a regional strategy identified in the RTP
- 3. Sub-area or local transit, bicycle and pedestrian system improvements to improve mode split
- 4. The effect of a comprehensive plan change on mode split targets and actions to ensure the overall mode split target for the local TSP is being achieved

- 5. Improvements to parallel arterials, collectors, or local streets, consistent with connectivity standards contained in Section 6.4.5, as appropriate, to address the transportation need and to keep through trips on arterial streets and provide local trips with alternative routes
- 6. Traffic calming techniques or changes to the motor vehicle functional classification, to maintain appropriate motor vehicle functional classification
- 7. If upon a demonstration that the above considerations do not adequately and costeffectively address the problem, a significant capacity improvement may be included in the comprehensive plan

Upon a demonstration that the above considerations do not adequately and cost-effectively address the problem and where accessibility is significantly hindered, Metro and the affected city or county shall consider:

- 1. Amendments to the boundaries of a 2040 Growth Concept design type
- 2. Amendments or exceptions to land-use functional plan requirements
- 3. Amendments to the 2040 Growth Concept
- 4. Designation of an Area of Special Concern, consistent with Section 6.7.7.

Demonstration of compliance will be included in the required congestion management system compliance report submitted to Metro by cities and counties as part of system-level planning and through findings consistent with the TPR in the case of amendments to applicable plans.

6.4.5 Design Standards for Street Connectivity

The design of local street systems, including "local" and "collector" functional classifications, is generally beyond the scope of the 2000 RTP. However, the aggregate effect of local street design impacts the effectiveness of the regional system when local travel is restricted by a lack of connecting routes, and local trips are forced onto the regional network. Therefore, streets should be designed to keep through trips on arterial streets and provide local trips with alternative routes. The following mapping requirements and design standards are intended to improve local circulation in a manner that protects the integrity of the regional transportation system.

Cities and counties within the Metro region are required to amend their comprehensive plans, implementing ordinances and administrative codes, if necessary, to comply with or exceed the following mapping requirements and design standards:

1. Cities and counties must identify all contiguous areas of vacant and redevelopable parcels of five or more acres planned or zoned for residential or mixed-use development and prepare a conceptual new streets plan map. The map shall be adopted as a part of the Transportation System Plan element of the local Comprehensive Plan. The purpose of this map is to provide guidance to land-owners and developers on desired street connections that will improve local access and preserve the integrity of the regional street system.

6-16

The conceptual street plan map should identify street connections to adjacent areas in a manner that promotes a logical, direct and connected street system. Specifically, the map should conceptually demonstrate opportunities to extend and connect to existing streets, provide direct public right-of-way routes, and limit the potential of cul-de-sac and other closed-end street designs.

- 2. In addition to preparing the above conceptual street plan map, cities and counties shall require new residential or mixed-use development involving construction of new street(s) to provide a site plan that reflects the following:
  - a. Street connections:
    - Responds to and expands on the conceptual street plan map as described in Section 6.4.5(1) for areas where a map has been completed.
    - Provides full street connections with spacing of no more than 530 feet between connections except where prevented by barriers such as topography, railroads, freeways, pre-existing development, or where lease provisions, easements, covenants or other restrictions existing prior to May 1, 1995 which preclude street connections.
    - Where streets must cross water features identified in Title 3 of the Urban Growth Management Functional Plan (UGMFP), provide crossings at an average spacing of 800 to 1,200 feet, unless habitat quality or length of crossing prevents a full street connection.
  - b. Accessways:
    - When full street connections are not possible provides bike and pedestrian accessways on public easements or rights-of-way in lieu of streets. Spacing of accessways between full street connections shall be no more than 330 feet except where prevented by barriers such as topography, railroads, freeways, pre-existing development, or where lease provisions, easements, covenants or other restrictions existing prior to May 1, 1995 which preclude accessway connections.
    - Bike and pedestrian accessways that cross water features identified in Title 3 of the UGMFP should have an average spacing no more than 530 feet, unless habitat quality or length of crossing prevents a connection.
  - c. Centers, main streets and station communities:
    - Where full street connections over water features identified in Title 3 of the UGMFP cannot be constructed in centers, main streets and station communities (including direct connections from adjacent neighborhoods), or spacing of full street crossings exceeds 1,200 feet, provide bicycle and pedestrian crossings at an average

spacing of 530 feet, unless exceptional habitat quality or length of crossing prevents a connection.

- d. Other considerations:
  - Limits the use of cul-de-sac designs and other closed-end street systems to situations where barriers prevent full street extensions.
  - Includes no closed-end street longer than 200 feet or with more than 25 dwelling units.
  - Includes street cross-sections demonstrating dimensions of right-of-way improvements, with streets designed for posted or expected speed limits.

For replacement or new construction of local street crossings on streams identified in Title 3 of the Urban Growth Management Functional Plan, Cities and Counties, TriMet, ODOT and the Port of Portland shall amend design codes, standards and plans to allow consideration of the stream crossing design guidelines contained in the Green Streets handbook.

Figure 6.1 demonstrates a site plan map that a developer would provide to meet code regulations for the subdivision of a single parcel. Figure 6.2 shows a street cross-section that could be submitted by a developer for approval during the permitting process.

Figure 6.1 Site Plan Map



Source: Metro

Figure 6.2 Street Cross Section – Local Street, mid-block



Source: Metro

- 3. Street design code language and guidelines must allow for:
  - a. Consideration of narrow street design alternatives. For local streets, no more than 46 feet of total right-of-way, including pavement widths of no more than 28 feet, curb-face to curb-face, sidewalk widths of at least 5 feet and landscaped pedestrian buffer strips that include street trees. Special traffic calming designs that use a narrow right-of-way, such as woonerfs and chicanes, may also be considered as narrow street designs.
  - b. Short and direct public right-of-way routes to connect residential uses with nearby commercial services, schools, parks and other neighborhood facilities.
  - c. Consideration of opportunities to incrementally extend streets from nearby areas.
  - d. Consideration of traffic calming devices to discourage traffic infiltration and excessive speeds on local streets.
- 4. For redevelopment of existing land-uses that require construction of new streets, cities and counties shall develop local approaches to encourage adequate street connectivity.

#### 6.4.6 Alternative Mode Analysis

Improvement in non-SOV mode share will be used as the key regional measure for assessing transportation system improvements in the central city, regional centers, town centers and station communities. For other 2040 Growth Concept design types, non-SOV mode share will be used as an important factor in assessing transportation system improvements. These modal targets will also be used to demonstrate compliance with per capita travel reductions required by the state TPR. This section requires that cities and counties establish non-SOV regional modal targets for all 2040 design types that will be used to guide transportation system improvements, in accordance with Table 1.3 in Chapter 1 of this plan:

6-19

- 1. Each jurisdiction shall establish an alternative mode share target (defined as non-single occupancy vehicle person-trips as a percentage of all person-trips for all modes of transportation) in local TSPs for trips into, out of and within all 2040 Growth Concept land-use design types within its boundaries. The alternative mode share target shall be no less than the regional modal targets for these 2040 Growth Concept land-use design types to be established in Table 1.3 in Chapter 1 of this plan.
- 2. Cities and counties, working with Tri-Met and other regional agencies, shall identify actions in local TSPs that will result in progress toward achieving the non-SOV modal targets. These actions should initially be based on RTP modeling assumptions, analysis and conclusions, and include consideration of the maximum parking ratios adopted as part of Title 2, section 3.07.220 of the *Urban Growth Management Functional Plan*; regional street design considerations in Section 6.7.3, Title 6, transportation demand management strategies and transit's role in serving the area. Local benchmarks for evaluating progress toward achieving modal targets may be based on future RTP updates and analysis, if local jurisdictions are unable to generate this information as part of TSP development.
- 3. Metro shall evaluate local progress toward achieving the non-SOV modal targets during the 20-year plan period of a local TSP using the Appendix 1.8 "TAZ Assumptions for Parking Transit and Connectivity Factors" chart as minimum performance requirements for local actions proposed to meet the non-SOV requirements.

# 6.4.7 Motor Vehicle Congestion Analysis

Motor Vehicle Level-Of-Service (LOS) is a measurement of congestion as a share of designed motor vehicle capacity of a road. Policy 13.0 and Table 1.2 of this plan establish motor vehicle level-of-service policy for regional facilities. These standards shall be incorporated into local comprehensive plans and implementing ordinances to replace current methods of determining motor vehicle congestion on regional facilities. Jurisdictions may adopt alternative standards that do not exceed the minimum LOS established in Table 1.2. However, the alternative standard must not:

- result in major motor vehicle capacity improvements that have the effect of shifting unacceptable levels of congestion into neighboring jurisdictions along shared regional facilities;
- result in motor vehicle capacity improvements to the principal arterial system (as defined in Figure 1.12) that are not recommended in, or are inconsistent with, the RTP.
- increase SOV travel to a measurable degree that affects local consistency with the modal targets contained in Table 1.3.

By definition, the RTP addresses congestion of regional significance through the projects identified in Chapter 5 or refinements plans contained in this chapter of the plan. Other, more localized congestion is more appropriately addressed through the local TSP process, and includes any locations on the regional Motor Vehicle System (Figure 1.12) that are not addressed by the RTP. Localized congestion occurs where short links within the transportation system are exceeding LOS standards, though the overall system in the vicinity of the congested link is performing acceptably.

6-20

In cases where these localized areas of congestion are located on Principal Arterial routes (as defined in Figure 1.12) or the Regional Freight System (Figure 1.17), they shall be evaluated as part of the local TSP process to determine whether an unmet transportation need exists that has not been addressed in the RTP. Should a local jurisdiction determine that an unmet need exists on such a facility, the jurisdiction shall identify the need in the local TSP, and propose one of the following actions to incorporate the need and recommended solution into the RTP:

- Identify the unmet need and proposed projects at the time of Metro review of local TSPs for consistency, but incorporate the project into the regional TSP during the next scheduled RTP update; or
- Propose an amendment to the RTP for unmet needs and resulting projects where a more immediate update of the regional TSP is appropriate or required.

Intersection analysis and improvements also generally fall outside of the RTP, and capacity improvements recommended in this plan generally apply to links in the regional system, not intersections.

For the purpose of demonstrating local compliance with Table 1.2 as part of a periodic review or plan amendment, the following procedure for conducting the motor vehicle congestion analysis shall be used:

1. Analysis – A transportation need is identified in a given location when analysis indicates that congestion has reached the level indicated in the "exceeds deficiency threshold" column of Table 1.2 and that this level of congestion will negatively impact accessibility, as determined through Section 6.4.7(2). The analysis should consider a mid-day hour appropriate for the study area and the appropriate two-hour peak-hour condition, either A.M. or P.M. or both, to address the problem. Other non-peak hours of the day, such as midday on Saturday, should also be considered to determine whether congestion is consistent with the acceptable or preferred operating standards identified in Table 1.2. The lead agency or jurisdictions will be responsible for determining the appropriate peak and non-peak analysis periods.

An appropriate solution to the need is determined through requirements contained in this chapter. For regional transportation planning purposes, the recommended solution should be consistent with the acceptable or preferred operating standards identified in Table 1.2. A city or county may choose a higher level-of-service operating standard where findings of consistency with section 6.4.4 have been developed as part of the local planning process. The requirements in Section 6.6.2 shall also be satisfied in order to add any projects to the RTP based on the higher level-of-service standard.

2. Accessibility – If a deficiency threshold is exceeded on the regional transportation system as identified in Table 1.2, cities and counties shall evaluate the impact of the congestion on regional accessibility using the best available quantitative or qualitative methods. If a determination is made by Metro that exceeding the deficiency threshold negatively impacts regional accessibility, cities and counties shall follow the transportation systems

analysis and transportation project analysis procedures identified in Sections 6.4.2 and 6.7.3.

3. Consistency – The identified function or the identified capacity of a road may be significantly affected by planning for 2040 Growth Concept design types. Cities and counties shall take actions described in Section 6.7 of this chapter, including amendment of their transportation plans and implementing ordinances, if necessary, to preserve the identified function and identified capacity of the road, and to retain consistency between allowed land-uses and planning for transportation facilities.

#### 6.4.8 Future RTP Refinements Identified through Local TSPs

The 2000 RTP represents the most extensive update to the plan since it was first adopted in 1982. It is the first RTP to reflect the 2040 Growth Concept, Regional Framework Plan and state Transportation Planning Rule. In the process of addressing these various planning mandates, the plan's policies and projects are dramatically different than the previous RTP. This update also represents the first time that the plan has considered growth in urban reserves located outside the urban growth boundary but expected to urbanize during the 20-year plan period. As a result, many of the proposed transportation solutions are conceptual in nature, and must be further refined.

In many cases, these proposed transportation solutions were initiated by local jurisdictions and special agencies through the collaborative process that Metro used to develop the updated RTP. However, the scope of the changes to the RTP will require most cities and counties and special agencies to make substantial changes to comprehensive, facility and service plans, as they bring local plans into compliance with the regional plan. In the process of making such changes, local jurisdictions and special agencies will further refine many of the solutions included in this plan.

Such refinements will be reviewed by Metro and, based on a finding of consistency with RTP policies, specifically proposed for inclusion in future updates to the RTP. Section 6.3 requires Metro to develop a process for to ensure consistency between the 2000 RTP and local TSPs by developing a process for tracking local project and functional classification refinements that are consistent with the RTP, but require a future amendment to be incorporated into the RTP. This process will occur concurrently with overall review of local plan amendments, facility plans and service plans, and is subject to the same appeal and dispute resolution process. While such proposed amendments to the RTP may not be effective until a formal amendment has been adopted, the purpose of endorsing such proposed changes is to allow cities and counties to retain the proposed transportation solutions in local plans, with a finding of consistency with the RTP, and to provide a mechanism for timely refinements to local and regional transportation plans.

#### 6.4.9 Local 20202025 Forecast - Options for Refinements

The 2000 RTP is a 20-year plan, with a 20202025 forecast developed from 1994-2000 base data. Metro produced an updated 20202025 forecast that accounts for urban reserveurban growth boundary actions, and estimates the amount of jobs and housing expected in urban reserves in 20202025. Local TSPs using the 20202025 forecast may experience different modeling outcomes in these areas than were observed during the development of the RTP. Therefore, Metro will accept local plans under the following four options:

- 1. Local plans in areas unaffected by urban reserve growth boundary actions may be developed using the RTP forecast for 20202025 (which is based on 1994-2000 data).
- 2. Local plans already under way at the time of RTP adoption, and which include areas affected by urban reserve growth boundary actions, may be developed using the RTP forecast for 20202025 (based on 1994-2000 data), with population and employment allocations adjusted by the local jurisdiction to reflect urban reserve actions. However, adjustments to population and employment allocations shall (a) remain within the holding capacity of a traffic zone or area, as defined by Metro's productivity analysis, and (b) not exceed traffic zone or area assumptions of the updated 20202025 forecast.
- 3. Local plans in areas affected by urban reserve actions may use the updated 20202025 forecast, and any subsequent differences in proposed transportation solutions will be reconciled during Metro's review of the local plan.
- 4. Local plans may be based on updated, locally developed population and employment data, conditions and 20202025 forecasts. However, population and employment data and forecasts, and the methodology for generating the data and forecasts shall be coordinated at the county level, and accepted by Metro technical staff and TPAC as statistically valid. Subsequent adjustments to the population and employment allocations for traffic zones may be made in the local planning to reflect updated population and employment data and 20202025 forecasts. Metro shall consider the updated locally developed data and forecasts in future RTP forecasts of population and employment. Subsequent differences in local TSP project recommendations that result from the differences in population and employment forecasts will be resolved in the next scheduled RTP update.

Metro will update the 20202025 population and employment allocations periodically to reflect local and regional land-use decisions. For example, changes to the 20202025 population and employment allocations could result if an urban reserve area is reduced in size or taken out altogether if the urban growth boundary is expanded or if local zoning capacity is amended to increase or decrease. The provisions in this section are for the purpose of TSP development and analysis, and do not necessarily apply to other planning activities.

#### 6.4.10 Transit Service Planning

Efficient and effective transit service is critical to meeting mode-split targets, and the regional transit functional classifications are tied to 2040 Growth Concept land-use components. Local transportation system plans shall include measures to improve transit access, passenger environments and transit service speed and reliability for:

- rail station areas, rapid bus and frequent bus corridors where service is existing or planned
- regional bus corridors where services exists at the time of TSP development

To ensure that these measures are uniformly implemented, cities and counties shall:

- 1. Adopt a transit system map, consistent with the transit functional classifications shown in Figure 1.16, as part of the local TSP.
- 2. Amend development code regulations to require new retail, office and institutional buildings on sites at major transit stops to:
  - 1. Locate buildings within 20 feet of or provide a pedestrian plaza at the major transit stops
  - 2. Provide reasonably direct pedestrian connections between the transit stop and building entrances on the site
  - 3. Provide a transit passenger landing pad accessible to disabled persons (if not already existing to transit agency standards)
  - 4. Provide an easement or dedication for a passenger shelter and underground utility connection from the new development to the transit amenity if requested by the public transit provider
  - 5. Provide lighting at a transit stop (if not already existing to transit agency standards).
- 3. Consider designating pedestrian districts in a comprehensive plan or other implementing land use regulations as a means of meeting or exceeding the requirements of OAR 660-012-0045 (4a-c) and this plan section 6.4.10(2) above. Pedestrian district designation shall address the following criteria:
  - (a) A connected street and pedestrian network, preferably through a local street and pedestrian network plan covering the affected area.
  - (b) Designated pedestrian districts should specifically consider, but are not limited to these elements: Transit/pedestrian/bicycle interconnection; parking and access management; sidewalk and accessway location and width; alleys; street tree location and spacing; street crossing and intersection design for pedestrians; street furniture and lighting at a pedestrian scale; and traffic speed. When local transportation system plans are adopted, designated pedestrian districts should be coordinated with the financing program required by the Transportation Planning Rule.
- 4. Provide for direct and logical pedestrian crossings at transit stops and marked crossings at major transit stops.
- 5. Consider street designs which anticipate planned transit stop spacing, location, and facilities (such as shelters, benches, signage, passenger waiting areas) and are consistent with the Creating Livable Streets design guidelines.

6-24

Public transit providers shall consider the needs and unique circumstances of special needs populations when planning for service. These populations include, but are not limited to, students, the elderly, the economically disadvantaged, the mobility impaired and others with special needs. Consideration shall be given to:

- 1. adequate transit facilities to provide service
- 2. hours of operation to provide transit service corresponding to hours of operation of institutions, employers and service providers to these communities
- 3. adequate levels of transit service to these populations relative to the rest of the community and their special needs

# 6.5 Metropolitan Transportation Improvement Program (MTIP)

#### 6.5.1 The Role of the MTIP in Regional Planning

An important tool for implementing the RTP is the Metropolitan Transportation Improvement Program (MTIP). The region's four-year funding document, the MTIP schedules and identifies funding sources for projects of regional significance to be built during a four-year period. Federal law requires that all projects using federal funds be included in the MTIP. In developing the MTIP, the region gives top priority to strategic transportation investments that leverage and reinforce the urban form outlined in Chapter 1, of this plan. The MTIP is adopted by Metro and the Oregon Transportation Commission for inclusion into a unified State TIP (STIP), that integrates regional and statewide improvement plans. The MTIP is updated every two years.

ISTEA and TEA-21 created important new fiscal requirements for the TIP. The TIP is fiscally constrained and includes only those projects for which federal resources are reasonably available. Projects are grouped by funding category, with project costs not to exceed expected revenue sources. The MTIP financial plan is not comprehensive; it covers only federal funds for capital improvements, and does not include operations, maintenance and preservation or local funds for capital costs.

It is the responsibility of the cities, counties, ODOT, Tri-Met and the Port of Portland to implement necessary improvements to the regional system, as well as those needed for local travel. These agencies are eligible to receive federal funds allocated through the MTIP process for projects included in the RTP. The TIP is prepared by Metro in consultation with these agencies. Interregional coordination throughout the planning and programming process will help to ensure that improvement projects are consistent with regional objectives and with each other.

Projects included in the MTIP must also be included in the RTP financially constrained system. For the purpose of this plan, the assumptions used to develop the financially constrained system are defined in Appendix 4.2. Projects included in the financially constrained system are identified by an asterisk (\*) in Figures 5.8 through 5.14 in Chapter 5. However, while the financially constrained system should provide the basis for most MTIP funding decisions, other projects from the RTP may

6-25

also be selected for funding. In the event that such projects are drawn from the plan for funding, the RTP financially constrained system will be amended to include the project or projects. In addition, when the financially constrained system is amended, continued financial constraint must be demonstrated by identifying additional revenues or removal of other projects from the financially constrained system. Except in the case of exempt projects (as defined by the federal and state conformity rules) such actions require an air quality conformity determination.

#### 6.5.2 How the MTIP is Developed

Though the MTIP development process is initiated by Metro, the work begins at the local level, with city and county elected officials receiving input from citizens through local planning efforts, and later sharing their transportation needs at the Joint Policy Advisory Committee on Transportation (JPACT). Additional public input is received at the regional level, as well, when JPACT and the Metro Council review the MTIP for final approval. Upon adoption by the Council, the MTIP is submitted to the Oregon Transportation Commission (OTC) for approval as part of the State Transportation Improvement Plan (STIP).

In 1999, more than \$75 million in regional funds were allocated to a wide variety of projects, ranging from safety improvements and system expansion to projects that leverage the 2040 Growth Concept. Priorities 2000 was the process for developing the fiscal year 2000 to 2003 MTIP. The first step in Priorities 2000 was developing criteria for ranking projects by transportation modes. The second step was a solicitation for project submittals. Local governments, Tri-Met and the Port of Portland submitted 150 transportation projects, with a cost of more than \$300 million, for funding consideration. In the third step, projects were ranked by technical and administrative criteria. Next, the Priorities 2000 projects were reviewed at a series of public workshops and hearings held throughout the region.

The final funding recommendation included 65 projects. The funding package broke new ground in Metro's objective of creating strong linkages between planned land-uses and the allocation of transportation funding. Based on the flow of federal transportation funding, the "Priorities" process for updating the MTIP and allocating revenues will occur every two years.

#### 6.5.3 RTP Implementation Benchmarks

The RTP establishes an general direction for implementation of needed improvements that reflects a wide variety of factors, including expected development trends, existing safety and operational deficiencies, and anticipated revenue. The project timing proposed in the RTP also reflects an effort to create a balanced, multi-modal transportation system. As such, the projects are organized according to those needed during the first five, second five and final ten years of the planning period. To ensure that incremental funding decisions that occur through the MTIP follow this general RTP direction, benchmarks shall be established for monitoring RTP implementation over time, and:

1. The benchmarks shall be tied to Chapter 1 objectives and shall address the relative performance of the system and the degree to which the various RTP projects are being implemented.

6-26

2. Findings for consistency with the benchmarks shall be developed as part of the biennial MTIP update, or as necessary in conjunction with other RTP monitoring activities.

In addition, benchmarks should be designed to track the following general information to the degree practicable for ongoing monitoring:

- progress on financing the strategic system
- progress in completing the modal systems described in Chapter 1
- relative change in system performance measures
- progress toward land use objectives related to the RTP
- relative comparisons with similar metropolitan regions on key measures

#### 6.5.4 Improvements in Urban Reserves

During the MTIP process, improvements that add capacity or urban design elements to rural facilities in urban reserves should:

- be coordinated with expansion of the urban growth boundary
- not encourage development outside of the urban growth boundary
- not disrupt the economic viability of nearby rural reserves
- be consistent with planned urban development or other transportation facilities

#### 6.6 **Process for Amending the RTP**

#### 6.6.1 RTP Policy, System Map and Compliance Criteria Amendments

When Metro amends policies or system maps in Chapter 1 of this plan or compliance criteria in this chapter, it will evaluate and adopt findings regarding consistency with the Regional Framework Plan. Decisions on amendments made at this level are land-use decisions for need, mode, corridor, general scope and function of a proposed project. Subsequent land-use decisions on final project design and impact mitigation will be needed prior to construction. Such analysis to evaluate impacts could lead to a "no-build" decision where a proposed project is not recommended for implementation, and would require reconsideration of the proposed project or system improvements. As such, amendments at this level shall be reviewed through the post-acknowledgement process. However, a decision on an amendment to the Regional Transportation Plan should not foreclose or appear to foreclose full and fair consideration of all relevant goal issues at such time that specific projects and programs are adopted by a local jurisdiction.

It is Metro's responsibility to adopt findings based on project need, mode, corridor, general scope and function of projects proposed in the Regional Transportation Plan. The affected jurisdiction is responsible for preparing the specific local plan amendments and findings related to specific location, project design and impact mitigation and for scheduling them for hearing before the governing body in time for action by that body by the time required.

#### 6.6.2 RTP Project Amendments

The RTP establishes a comprehensive policy direction for the regional transportation system and recommends a balanced program of transportation investments to implement that policy direction. However, the recommended investments do not solve all transportation problems and are not intended to be the definitive capital improvement program on the local transportation system for the next 20 years.

Rather, the RTP identifies the projects, programs or further refinement studies required to adequately meet regional transportation system needs during the 20-year planning period. Local conditions will be addressed through city and county TSPs, and will require additional analysis and improvements to provide an adequate transportation system. Section 6.7 of this chapter anticipates such refinements, particularly given the degree to which this RTP has been updated from previous plans. Similarly, refinements to the RTP may result from ongoing corridor plans or area studies. The following processes may be used to update the RTP to include such changes:

- 1. Amendments resulting from major studies: as the findings of such studies are produced, they will be recommended by a resolution of JPACT and the Metro Council. These amendments must be incorporated into the RTP through a quasi-judicial or legislative process, as needed.
- 2. Amendments resulting from local TSPs: new roadway, transit, bikeway, pedestrian, freight and demand management projects necessary to meet the objectives of the RTP shall be accompanied by an demonstration of consistency with the RTP based on the following criteria:
  - a. The objectives to be met by the proposed projects(s) are consistent with RTP goals, policies and objectives (Chapter 1).
  - b. The proposed action is consistent with the modal function of the facility as defined in Chapter 1.
  - c. The impact of the proposed projects(s) on the balance of the regional system is evaluated through a CMS analysis.
  - d. The proposed action is needed to achieve the motor vehicle level-of-service performance criteria identified in the RTP, or alternative performance criteria adopted in local TSPs under the provisions of Section 6.4.7, as follows:
    - A) principal, major and minor arterial capacity improvements are necessary to maintain compliance with Policy 13.0, Table 1.2, or alternative performance criteria adopted in local TSPs. Improvements that are designed to provide a higher level of service than

6-28

the minimum acceptable standard established in Policy 13.0 can be designed and/or provided at the option of the implementing jurisdiction. Such actions must be consistent with the RTP as outlined in this section and demonstrate that either:

- i) a long-range evaluation of travel demand indicates a probable need for right-ofway preservation beyond that necessary for the 20-year project design, or
- ii) the additional service provided by the higher level design is the result of a design characteristic necessary to achieve the minimum motor vehicle performance measure
- B) local transportation system improvements must be consistent with the following:
  - i) the local system must adequately serve the local travel demands expected from development of the land-use plan to the year 20202025 to ensure that the regional system is not overburdened with local traffic

- ii) local analysis shall incorporate required street connectivity plans
- iii) the local system provides continuity between neighboring jurisdictions, consistency between city and county plans for facilities within city boundaries and consistency between local jurisdictions and ODOT plans
- e. The need for the proposed action based on Metro's adopted population and employment projections, or refinements as noted in Section 6.4.8.
- f. The proposed action is consistent with the regional non-SOV modal targets specified in Table 1.3 of Chapter 1.
- g. The proposed action represents the lowest cost system alternative solution acceptable.
- h. The proposed action is not prohibited by unacceptable environmental impacts or other considerations.
- i. A goal, policy or system plan element in the federal RTP would likely change as the result of a "no-build" project decision later in the process.
- j. The project is in the local jurisdiction's TSP, or a final local land-use action occurred.
- k. The project is contained in or consistent with the RTP, adopted comprehensive plan, or implementation plan(s) of any other affected jurisdictions.
- 1. Sufficient public involvement activities have occurred regarding the proposed action.

The amount of information required to address these criteria shall be commensurate with the scope of the project. Such additions will be amended into the RTP as part of the project update process described in this section. Operations, maintenance and safety improvements are deemed

6-29

consistent with the policy intent of the RTP if (a) they are needed to serve the travel demand associated with Metro's adopted population and employment forecasts, and (b) they are consistent with affected jurisdictional plans.

3. Amendments resulting from updates to the Regional Framework Plan or related functional plans.

#### 6.6.3 Congestion Management Requirements

This section applies to any amendments to the Regional Transportation Plan to add significant single occupancy vehicle (SOV) capacity to multi-modal arterials and/or highways. Consistent with Federal Congestion Management System requirements (23 CFR Part 500) and TPR system planning requirements (OAR 660-12), the following actions shall be considered through the RTP when recommendations are made to revise the RTP to define the need, mode, corridor and function to address an identified transportation needs, and prior to recommendations to add significant SOV capacity:

- 1. Regional transportation demand strategies
- 2. Regional transportation system management strategies, including intelligent transportation systems (ITS)
- 3. High occupancy vehicle (HOV) strategies
- 4. Regional transit, bicycle and pedestrian system improvements to improve mode split
- 5. Unintended land-use and transportation effects resulting from a proposed SOV project or projects
- 6. Effects of latent demand from other modes, routes or time of day from a proposed SOV project or projects
- 7. If upon a demonstration that the considerations in 1 through 6 do not adequately and costeffectively address the problem, a significant capacity improvement may be included in the regional transportation plan

#### 6.6.4 Plan Maintenance

The RTP is updated every three to five years, and covers a minimum 20-year plan period. Periodic amendments to the plan will also occur, as needed, to reflect recommendations from corridor or subarea planning studies. As preparation for each scheduled update, development throughout the region will be monitored to determine whether growth (and the associated travel demand) occurs as forecast. Metro will review its population and employment forecasts annually and update them at least every five years for the following conditions:

- national or regional growth rates differ substantially from those previously assumed
- significant changes in growth rate or pattern develop within jurisdictions

6-30

- changes to the urban growth boundary are adopted
- a jurisdiction substantially changes its land-use plan

New information gathered during the course of the year on such issues as energy price and supply, population and employment growth, inflation and new state and federal laws may result in different conditions to be addressed by the plan. These modifications will be incorporated as needed during periodic updates to the plan. Each update will occur in cooperation with affected jurisdictions, state agencies and public transit providers.

# 6.7 Project Development and Refinement Planning

#### 6.7.1 Role of RTP and the Decision to Proceed with Project Development

Metro is the regional planning agency for the metropolitan area. Metro does not complete local transportation system plans, engineer or build transportation facilities or permit land uses or transportation projects. These activities occur at the local level. After a project has been incorporated in the RTP, it is the responsibility of the local sponsoring jurisdiction to determine the details of the project (design, operations, etc.). The local jurisdiction responsible for the applicable transportation system plan shall reach a decision on whether to build the improvement based upon detailed environmental impact analysis, adoption of actions to mitigate impacts and findings demonstrating consistency with applicable comprehensive plans and applicable statewide planning goals. If this process results in a decision not to build the project, the RTP will be amended to delete the recommended improvement and an alternative must be identified to address the original transportation need.

#### 6.7.2 New Solutions Re-submitted to RTP if No-Build Option is Selected

When a "no-build" alternative is selected at the conclusion of a project development process, a new transportation solution must be developed to meet the original need identified in the RTP, or a finding that the need has changed or been addressed by other system improvements. In these cases, the new solution or findings will be submitted as an amendment to the RTP, and would also be evaluated at the project development level.

#### 6.7.3 Project Development Requirements

Transportation improvements where need, mode, function and general location have already been identified in the RTP and local plans for a specific alignment must be evaluated on a detailed, project development level. This evaluation is generally completed at the local jurisdiction level, or jointly by affected or sponsoring agencies, in coordination with Metro. The purpose of project development planning is to consider project design details and select a project alignment, as necessary, after evaluating engineering and design alternatives, potential environmental impacts and consistency with applicable comprehensive plans and the RTP. The project need, mode, function and general location do not need to be addressed at the project level, since these findings have been previously established by the RTP.

The TPR and Metro's Interim 1996 Congestion Management System (CMS) document require that measures to improve operational efficiency be addressed at the project level, though system-wide considerations are addressed by the RTP. Therefore, demonstration of compliance for projects not included in the RTP shall be documented in a required Congestion Management System report that is part of the project-level planning and development (Appendix D of the Interim CMS document). In addition, the CMS requires that street design guidelines be considered as part of the project-level planning process. This CMS requirement does not apply to locally funded projects on local facilities. Unless otherwise stipulated in the MTIP process, these provisions are simply guidelines for locally funded projects.

Therefore, in addition to system-level congestion management requirements described in Section 6.6.3 in this chapter, cities, counties, TriMet, ODOT, and the Port of Portland shall consider the following project-level operational and design considerations during transportation project analysis as part of completing the CMS report:

- 1. Transportation system management (e.g., access management, signal inter-ties, lane channelization, etc.) to address or preserve existing street capacity.
- 2. Street design policies, classifications and design principles contained in Chapter 1 of this plan. See Section 1.3.5, Policy 11.0, Figure 1.4. Implementing guidelines are contained in *Creating Livable Streets: Street Design Guidelines for 2040* (2nd edition, 2002) or other similar resources consistent with regional street design policies.
- 3. Environmental design guidelines, as contained in *Green Streets*: *Innovative Solutions for Stormwater and Street Crossings* (2002), and *Trees for Green Streets*: *An Illustrated Guide* (2002), or other similar resources consistent with federal regulations for stream protection.

Transportation providers in the Metro region, including the cities and counties, TriMet, ODOT, and the Port of Portland are required to amend their comprehensive plans, implementing ordinances and administrative codes, if necessary, to consider the *Creating Livable Streets* design guidelines as part of project development. Transportation providers shall amend design codes, standards and plans to allow consideration of the guidelines contained in *Green Streets: Innovative Solutions for Stormwater and Street Crossings*.

#### 6.7.4 Refinement Planning Scope and Responsibilities

In some areas defined in this section, the need for refinement planning is warranted before specific projects or actions that meet and identified need can be adopted into the RTP. Refinement plans generally involve a combination of transportation and land use analysis, multiple local jurisdictions and facilities operated by multiple transportation providers. Therefore, unless otherwise specified in this section, Metro or ODOT will initiate and lead necessary refinement planning in coordination with other affected local, regional and state agencies. Refinement planning efforts will be multi-modal evaluations of possible transportation solutions in response to needs identified in the RTP, including land use alternatives and to address consistency with applicable statewide planning goals Refinement plans fall into two broad groups of scope and complexity:

6-32

- Type I Major corridor refinements are necessary where a transportation need exists, but mode, function and general location of a transportation improvement are not determined, and a range of actions must be considered prior to identifying specific projects.
- Type II Minor corridor refinements are necessary where both the need and mode for a transportation improvement are identified in the RTP, but a specific project has not been identified.

Appendix 3.1 describes the 2000 RTP prioritization for major corridor refinements and minor corridor refinements <u>defined by the Corridor Studies process in 2000</u>. Refinement plan and corridor study prioritization and specific scope for each corridor is subject to annual updates as part of the Unified Work Plan (UWP).

#### 6.7.5 Type I – Major Corridor Refinements

Type I, major corridor refinements will be conducted by state or regional agencies working in partnership with local governments in the following areas. In each case, a transportation need has been established by the RTP, and in some cases, mode, function or general location may be determined or the decision on these elements narrowed at the TSP level to focus the refinement planning work. A transportation need is identified when regional standards for safety, mobility, or congestion are exceeded. In many of these corridors, RTP analysis indicates several standards are exceeded.

The purpose of Type I major corridor refinements is to develop an appropriate transportation strategy or solution through the corridor planning process that determined mode, function and general location of a project or set of projects. For each corridor, a number of transportation alternatives will be examined over a broad geographic area or through a local TSP to determine a recommended set of projects, actions or strategies that meet the identified need. This section of the RTP also identifies a number of corridor planning issues that shall be addressed as part of the refinement planning process.

For refinement planning in corridors located outside the urban growth boundary, this work shall also address relevant statewide planning goal exception requirements pursuant to Section 660.012.0070 of the state transportation planning rule. These findings shall expand on exceptions findings made as part of the 2000 RTP adoption ordinance, but address more localized issues relevant to the refinement level of planning.

The specific project recommendations from Type I major corridor refinements are then incorporated into the RTP, as appropriate. This section contains the following specific considerations that must be incorporated into corridor studies as they occur:

#### Interstate-5 North (I-84 to Clark County)

This heavily traveled route is the main connection between Portland and Vancouver. In addition to a number of planned and proposed highway capacity improvements, light rail is proposed along Interstate Avenue to the Expo Center, and may eventually extend to Vancouver. As improvements are implemented in this corridor, the following design considerations should be addressed:
- consider HOV lanes and peak period pricing
- transit alternatives from Vancouver to the Portland Central City (including light rail transit and express bus)
- maintain an acceptable level of access to the central city from Portland neighborhoods and Clark County
- maintain off-peak freight mobility, especially to numerous marine, rail and truck terminals in the area
- consider adding reversible express lanes to I-5
- consider new arterial connections for freight access between Highway 30, port terminals in Portland and port facilities in Vancouver, Wa.
- maintain an acceptable level of access to freight intermodal facilities and to the Northeast Portland Highway
- construct interchange improvements at Columbia Boulevard to provide freight access to Northeast Portland Highway
- address freight rail network needs
- consider additional Interstate Bridge capacity sufficient to handle project needs
- develop actions to reduce through-traffic on MLK and Interstate to allow main street redevelopment

#### Interstate-5 South (Highway 217 to WilsonvilleWillamette River/Boones Bridge)

This facility serves as the major southern access to and from the central city. The route also serves as an important freight corridor, where Willamette Valley traffic enters the region at the Wilsonville "gateway," and provides access to Washington County via Highway 217. Projections for this facility indicate that growth in traffic between the Metro region and the Willamette Valley will account for as much as 80 percent of the traffic volume along the southern portion of I-5, in the Tualatin and Wilsonville area. <u>A joint ODOT and Wilsonville study<sup>1</sup> concludes that in 2030</u> widening of I-5 to eight lanes would be required to meet interstate freeway capacity standards set by Metro and ODOT and that freeway access capacity would not be adequate with an improved I-5/Wilsonville Road interchange. For this these reasons, the appropriate improvements in this corridor are unclear at this time. However, I-5 serves as a critical gateway for regional travel and commerce, and an acceptable transportation strategy in this corridor has statewide significance. A major corridor study is proposed to address the following issues:

6-34

<sup>&</sup>lt;sup>1</sup> I-5/Wilsonville Freeway Access Study, DKS Associates, November 2002

- the effects of widening I-205 on the I-5 South corridor
- the effects of the I-5 to 99W Connector on the Stafford Road interchange and the resultant need for increased freeway access
- •\_\_\_\_\_the effects of peak period congestion in this area on regional freight mobility and travel patterns
- the ability of inter-city transit service, to/from neighboring cities in the Willamette Valley, including commuter rail, to slow traffic growth in the I-5 corridor
- the ability to maintain off-peak freight mobility with capacity improvements
- the potential for better coordination between the Metro region and valley jurisdictions on land-use policies
- the effects of a planned long-term strategy for managing increased travel along I-5 in the Willamette Valley
- the effects of UGB expansion and Industrial Lands Evaluation studies on regional freight mobility
- <u>the effects to freight mobility and local circulation due to diminished freeway access</u> <u>capacity in the I-5/Wilsonville corridor</u>

In addition, the following design elements should be considered as part of the corridor study:

- peak period pricing and HOV lanes for expanded capacity
- provide rapid bus service on parallel Barbur route, connecting Wilsonville to the central city
- provide additional overcrossings in West Portland town center to improve local circulation and interchange access
- provide additional freeway access improvements in the I-5/Wilsonville corridor to improve freight mobility and local circulation, (e.g. a new Boeckman Road interchange)
- add capacity to parallel arterial routes, including 72nd Avenue, Boones Ferry, Lower Boones Ferry and Carmen Drive
- add overcrossings in vicinity of Tigard Triangle to improve local circulation
- extend commuter rail service from Salem to the central city, Tualatin transit center and Milwaukie, primarily along existing heavy rail tracks
- additional I-5 mainline capacity (2030 demand on I-5 would exceed capacity)

#### provision of auxiliary lanes between all I-5 freeway on- and off-ramps in Wilsonville

#### Interstate 205

Improvements are needed in this corridor to address existing deficiencies and expected growth in travel demand in Clark, Multnomah and Clackamas counties. Transportation solutions in this corridor should address the following needs and opportunities:

- provide for some peak period mobility for longer trips
- preserve freight mobility from I-5 to Clark County, with an emphasis on connections to Highway 213, Highway 224 and Sunrise Corridor
- maintain an acceptable level of access to the Oregon City, Clackamas and Gateway regional centers and Sunrise industrial area
- maintain acceptable levels of access to PDX, including air cargo access

Potential transportation solutions in this corridor should evaluate the potential of the following design concepts:

- auxiliary lanes added from Airport Way to I-84 East
- consider express, peak period pricing or HOV lanes as a strategy for expanding capacity
- relative value of specific ramp, overcrossing and parallel route improvements
- eastbound HOV lane from I-5 to the Oregon City Bridge
- truck climbing lane south of Oregon City
- potential for rapid bus service or light rail from Oregon City to Gateway
- potential for extension of rapid bus service or light rail north from Gateway into Clark County
- potential for refinements to 2040 land-use assumptions in this area to expand potential employment in the subarea and improve jobs/housing imbalance
- potential for re-evaluating the suitability of the Beavercreek area for urban growth boundary expansion, based on ability to serve the area with adequate regional transportation infrastructure

McLoughlin-Highway 224

6-36

Long-term improvements are needed in this corridor to preserve access to and from the Central City from the Clackamas County area, to provide access to the developing Clackamas regional center and to support downtown development in the Milwaukie town center. The recently completed South/North light rail study demonstrated a long-term need for high-capacity transit service in this corridor. The long-term transit need is critical, as demonstrated in the RTP analysis, where both highway and high-capacity transit service were needed over the 20-year plan period to keep pace with expected growth in this part of the region. The 2040 Growth Concept also calls for the regional centers and central city to be served with light rail. Transportation solutions in this corridor should address the following design considerations

- institute aggressive access management throughout corridor, including intersection grade separation along Highway 224 between Harrison Street and I-205
- design access points to McLoughlin and Highway 224 to discourage traffic spillover onto Lake Road, 34th Avenue, Johnson Creek boulevard, 17th Avenue and Tacoma Street
- monitor other local collector routes and mitigate spillover effect from congestion on McLoughlin and Highway 224
- consider an added reversible HOV or peak-period priced lane between Ross Island Bridge and Harold Street intersection
- expand highway capacity to a total of three general purpose lanes in each direction from Harold Street to I-205, with consideration of express, HOV lanes or peak period pricing for new capacity
- provide a more direct transition from McLoughlin to Highway 224 at Milwaukie to orient long trips and through traffic onto Highway 224 and northbound McLoughlin
- provide improved transit access to Milwaukie and Clackamas regional centers, including rapid bus in the short term, and light rail service from Clackamas regional center to Central City in the long term

#### Powell-Boulevard/Foster Road

The concentration potential urban growth boundary expansions in Clackamas County and southeast Multnomah County will place heavy demands on connecting routes that link these areas with employment centers in Portland and Multnomah County. Of these routes, the Foster/Powell corridor is most heavily affected, yet is also physically constrained by slopes and the Johnson Creek floodplain, making capacity improvements difficult. More urban parts of Foster and Powell Boulevard are equally constrained by existing development, and the capacity of the Ross Island Bridge.

As a result, a corridor study is needed to explore the potential for high capacity transit strategies that provide access from the developing Pleasant Valley and Damascus areas to employment areas

6-37

along the Foster/Powell corridor, Gresham regional center, Columbia South Shore industrial area and central city. Such a study should consider the following transportation solutions:

- aggressive transit improvements, including rapid bus service from Central City to Damascus town center via Powell and Foster roads, and primary bus on 172nd Avenue and to the Gresham regional center, Eastside MAX and Columbia South Shore
- capacity improvements that would expand Foster Road from two to three lanes from 122nd to 172nd avenues, and from two to five lanes from 172nd Avenue to Highway 212, phased in coordination with planned capacity improvements to Powell Boulevard between I-205 and Eastman Parkway
- extensive street network connection improvements in the Mount Scott and Pleasant Valley areas to reduce local travel demand on Foster Road and Powell Boulevard, and to improve access between these areas and adjacent East Multnomah and northeast Clackamas Counties
- ITS or other system management approaches to better accommodate expected traffic growth on the larger southeast Portland network, East Multnomah and northeast Clackamas County network

#### Powell Boulevard/Foster Road Phase 2

The Powell Boulevard/Foster Road Corridor represents both a key transportation challenge and an opportunity to meet 2040 regional land use goals. The Powell/Foster Corridor is a top priority among corridors requiring refinement plans. Despite policy changes to level-of-service standards that permit greater levels of congestion, significant multi-modal improvements will be needed in order to continue to serve transportation needs of the communities and industrial areas in southeast Portland and Gresham. The corridor is also critical to providing access to the planned growth areas in Pleasant Valley, along with Damascus and Springwater that have recently been added to the Urban Growth Boundary. In addition, the corridor is constrained by significant topographical and environmental features.

As a result of the findings from Phase 1 of the Powell Boulevard/Foster Road Corridor Plan, which was completed in 2003, specific multi-modal projects have been identified that address transportation needs on Powell Boulevard between inner SE Portland and Gresham, and on Foster Road west of Barbara Welch Road. System level decisions for transit service were also made for the corridor.

Several outstanding transportation problems in the Pleasant Valley, Damascus and south Gresham areas, require additional planning work before specific multi-modal projects can be developed and implemented. The Phase 2 plan should closely coordinated with concept plans for Damascus and the Springwater area, in order to incorporate the updated land use and transportation assumptions. It should examine the following transportation solutions and strategies:

6-38

- Determine the appropriate cross section on Foster Road between Barbara Welch Road and Jenne Road and the project timing, to meet roadway, transit, pedestrian and bike needs.
- Explore possibilities for potential new street connection improvements in the Mount Scott area that reduce local travel demand on Foster Road and improve access to the Pleasant Valley area.
- Develop conceptual designs and determine right-of-way for an improvement and extension of SE 174<sup>th</sup> Avenue between Powell Boulevard and Giese Road, or another new north-south roadway in the area, to accommodate travel demand and improve access to Pleasant Valley. The alignment should consider engineering feasibility, land use and environmental affects, safety, and overall costs.
- Further define the three-lane Highland Drive and Pleasant View Drive option that was recommended as part of Phase 1. This option needs to address design, operational, and safety-related issues.
- Work with local jurisdictions to provide for access management on arterials serving Pleasant Valley and Damascus.
- Address other regional north-south transportation needs identified by the Damascus Concept Plan and Springwater concept planning effort. Further evaluate alignment issues, engineering cost estimates, and right-of-way impacts of future roadway projects north of Damascus that are identified as part of the concept planning effort.

#### Highway 217

Improvements in this corridor are needed to accommodate expected travel demand, and maintain acceptable levels of access to the Beaverton and Washington Square regional centers. The following design and functional considerations should be included in the development of transportation solutions for this corridor:

- expand highway to include a new lane in each direction from I-5 to US 26
- address the competing needs of serving localized trips to the Washington Square and Beaverton regional centers and longer trips on Highway 217
- consider express, HOV lanes and peak period pricing when adding new capacity
- design capacity improvements to maintain some mobility for regional trips during peak travel periods
- design capacity improvements to preserve freight mobility during off-peak hours
- retain auxiliary lanes where they currently exist
- improve parallel routes to accommodate a greater share of local trips in this corridor

- consider improve light rail service or rapid bus service with substantially improved headways
- coordinate with planned commuter rail service from Wilsonville to Beaverton regional center

#### Tualatin Valley Highway

A number of improvements are needed in this corridor to address existing deficiencies and serve increased travel demand. One primary function of this route is to provide access to and between the Beaverton and Hillsboro regional centers. Tualatin Valley Highway also serves as an access route to Highway 217 from points west along the Tualatin Valley Highway corridor. As such, the corridor is defined as extending from Highway 217 on the east to First Avenue in Hillsboro to the west, and from Farmington Road on the south to Baseline Road to the north. The following design considerations should be addressed as part of a corridor study:

- develop an access management plan as part of a congestion management strategy
- implement TSM and other interim intersection improvements at various locations between Cedar Hills Boulevard and Brookwood Avenue
- the relative trade-offs of a variety of capacity and transit improvements, including:
  - a. improvements on parallel routes such as Farmington, Alexander, Baseline and Walker roads as an alternative to expanding Tualatin Valley Highway
  - b. seven-lane arterial improvements from Cedar Hills Boulevard or Murray Boulevard to Brookwood Avenue or Baseline Road in Hillsboro
  - c. a limited access, divided facility from Cedar Hills Boulevard or Murray Boulevard to Brookwood Avenue, with three lanes in each direction and some grade separation at major intersections
  - d. transit service that complements both the function of Tualatin Valley Highway and the existing light rail service in the corridor
  - evaluate impacts of the principal arterial designation, and subsequent operation effects on travel within the Beaverton regional center
- evaluate motor vehicle and street design designations as part of the study to determine the most appropriate classifications for this route

#### North Willamette Crossing

The RTP analysis shows a strong demand for travel between Northeast Portland Highway and the adjacent Rivergate industrial area and Highway 30 on the opposite side of the Willamette River. The St. Johns Bridge currently serves this demand. However, the St. Johns crossing has a number of limitations that must be considered in the long term in order to maintain adequate freight and

6-40

general access to the Rivergate industrial area and intermodal facilities. Currently, the St. Johns truck strategy is being developed (and should be completed in 2000) to balance freight mobility needs with the long-term health of the St. Johns town center. The truck strategy is an interim solution to demand in this corridor, and does not attempt to address long-term access to Rivergate and Northeast Portland Highway from Highway 30. Specifically, the following issues should be considered in a corridor plan:

- build on the St. Johns Truck Strategy recommendations to adequate freight and general access to Rivergate, while considering potentially negative impacts on the development of the St. Johns town center
- incorporate the planned development of a streamlined Northeast Portland Highway connection from I-205 to Rivergate to the crossing study
- include a long-term management plan for the St. John's Bridge, in the event that a new crossing is identified in the corridor plan recommendations

#### Barbur Boulevard/ I-5

This corridor provides access to the Central City and to neighborhoods and commercial areas in the inner southwest quadrant of the region. Barbur Boulevard is identified as a multi-modal facility with potential light rail or Rapid Bus as well as serving a regional role for motor vehicle, bicycle and pedestrian systems. I-5 in this corridor is a Main Roadway route for freight and a Principle Arterial for motor vehicles extending southward beyond the region.

Segments of both Barbur Boulevard and I-5 in this corridor experience significant congestion and poor service levels even with Priority System improvements, especially from the Terwilliger interchange northward. However, Rapid Bus service along Barbur and other expanded bus services are expected to experience promising ridership levels. Significant localized congestion occurs along the intersecting street segments of Bertha, Terwilliger and Capitol Highway/Taylors Ferry roads. Broad street cross-sections, angled intersections and limited signalized crossing opportunities along Barbur Boulevard creates traffic safety hazards and inhibits walking to local destinations and access to transit services.

Transportation solutions in the corridor should include the following considerations:

- Regional and local transit services and facilities needed to serve the Barbur corridor within the RTP planning horizon.
- Possible new locations or relocations for I-5 on-ramps and off-ramps and street connections across the freeway right-of-way.
- Opportunities for new or improved local street connections to Barbur Boulevard.
- Facilities to improve bicycle and pedestrian safety along Barbur and access to transit services and local destinations.

6-41

- Traffic management and intelligent transportation system improvements along the corridor.
- Potential mainline freeway improvements including possible southbound truck climbing lanes.

#### 6.7.6 Type II - Minor Corridor Refinements

Type II minor corridor refinements will be conducted by state or regional agencies working in partnership with local governments in the following areas. In each case, a transportation need has been established by the RTP, and in some cases, mode, function or general location may be determined or the decision on these elements narrowed at the TSP level to focus the refinement planning work. A transportation need is identified when regional standards for safety, mobility, or congestion are exceeded. In many of these corridors, RTP analysis indicates several standards are exceeded.

The purpose of the minor corridor refinement process is to identify specific projects consistent with the identified need, mode and general corridor. These proposed transportation projects must be developed to a more detailed level before construction can occur. This process is described in Section 6.7.3 of this chapter. For minor refinement planning in corridors located outside the UGB, this work shall also address relevant statewide planning goal exception requirements pursuant to Section 660.012.0070 of the state transportation planning rule. These findings shall expand on exceptions findings made as part of the 2000 RTP adoption ordinance, but address more localized issues relevant to the refinement level of planning. The specific project recommendations from major corridor studies are then incorporated into the RTP, as appropriate.

Because minor corridor refinements are more specific in location and mode, local TSPs shall consider measures to protect future right-of-way options within the affected corridors. Likewise, the refinement planning process shall make recommendations for corridor preservation or right-of-way acquisition strategies to ensure that final project recommendations are not precluded by land use decisions within the corridor.

The project development stage determines design details, and a project location or alignment, if necessary, after evaluating engineering and design details, and environmental impacts. While all projects in this plan must follow this process before construction can occur, the following projects must also consider the design elements described in this section:

#### Banfield (Interstate 84) Corridor

Despite the relatively heavy investments made in transit and highway capacity in this corridor in the 1980s, further improvements are needed to ensure an acceptable level of access to the central city from Eastside Portland neighborhoods and East Multnomah County. However, physical, environmental and social impacts make highway capacity improvements in this corridor unfeasible. Instead, local and special district plans should consider the following transportation solutions for this corridor:

- mitigate infiltration on adjacent corridors due to congestion along I-84 through a coordinated system of traffic management techniques (ITS)
- improve light rail headways substantially to keep pace with travel demand in the corridor
- improve bus service along adjacent corridors to keep pace with travel demand, including express and non-peak service
- consider additional feeder bus service and park-and-ride capacity along the eastern portion of the light rail corridor to address demand originating from East Multnomah and North Clackamas Counties
- develop TSM strategies for the Gateway regional center to mitigate expected spillover effects on the development of the regional center

#### Northeast Portland Highway

As radial urban highways such as the Banfield and Interstate-5 are increasingly burdened by peak period congestion, freight mobility will rely more heavily on circumferential routes, including I-205 and Northeast Portland Highway, for access to industrial areas and intermodal facilities. Northeast Portland Highway plays a particularly important role, as it links the Rivergate marine terminals and PDX air terminals to industry across the region (this route includes Killingsworth and Lombard streets from I-205 to MLK Jr. Boulevard, and Columbia Boulevard from MLK Jr. Boulevard to North Burgard). Though Northeast Portland Highway appears to have adequate capacity to serve expected <del>20202025</del> demand, a number of refinements in the corridor are needed. Local and special district plans should consider the following transportation solutions as improvements are made in this corridor:

- improve Northeast Portland Highway as a strategy for addressing Banfield corridor and east Marine Drive congestion
- develop a long-term strategy to serve freight movement between Highway 30 and Rivergate
- implement aggressive access management along Northeast Portland Highway
- implement and refine Columbia Corridor improvements to address full corridor needs of Northeast Portland Highway, from Rivergate to I-205
- consider future grade separation at major intersections
- streamline the Northeast Portland Highway connection from the Lombard/Killingsworth section to Columbia Boulevard with an improved transition point at MLK Jr. Boulevard

6-43

- improve the Columbia Boulevard interchange at I-5 to provide full access to Northeast Portland Highway
- construct capacity and intersection improvements between 82nd Avenue and I-205
- Implement the St. Johns Truck Strategy recommendations in order to direct truck traffic onto the designated freight system, as shown in Figure 1.17, and protect the Lombard main street and St. Johns town center from truck traffic impacts.

#### Interstate-84 to US 26 Connector

The long-term need to develop a highway link between I-84 and Highway 26 exists, but a series of interim improvements to Hogan Road are adequate to meet projected demand through 20202025. The RTP calls for a series of interim improvements that will better connect Hogan Road to both I-84 on the north, and Highway 26 to the south.

These improvements are needed to ensure continued development of the Gresham regional center and expected freight mobility demands of through traffic. They also benefit transit-oriented development along the MAX light rail corridor, as they would move freight traffic from its current route along Burnside, where it conflicts with development of the Rockwood town center and adjacent station communities. In addition to planned improvements to the Hogan Road corridor, local plans or a corridor study should address:

- more aggressive access management between Stark Street and Powell Boulevard on 181st, 207th and 257th avenues
- redesigned intersections improvements on Hogan at Stark, Burnside, Division and Powell to streamline through-flow
- the need for a long-term primary freight route in the corridor
- the potential for a new alignment south of Powell Boulevard to US 26.

#### Sunrise Corridor

The full Sunrise Corridor improvement from I-205 to Highway 26 is needed during the 20-year plan period, but should be implemented with a design and phasing that reinforces development of the Damascus town center, and protect rural reserves from urban traffic impacts. This corridor includes rural areas outside the Metro area urban growth boundary. Impacts on rural resources in these areas shall be addressed through statewide planning goal exception findings that expand on findings already adopted in the 2000 RTP, pursuant to Section 660.012.0070 of the state transportation planning rule. Though a draft environmental impact statement has been prepared for this corridor, the final environmental impact statement should be refined to consider the following elements:

6-44

- Construct the segment from I-205/Highway 224 interchange to existing Highway 212 at Rock Creek as funds become available
- preserve right-of-way (ROW) from Rock Creek to Highway 26 as funds become available
- consider phasing Sunrise construction as follows: (a) complete I-205 to Rock Creek segment first, followed by (b) ROW acquisition of remaining segments, then (c) construction of 222nd Avenue to Highway 26 segment and (d) lastly, construction of middle segment from Rock Creek to 222nd Avenue as Damascus town center develops
- consider express, peak period pricing and HOV lanes as phases of the Sunrise Corridor are constructed
- reflect planned network of streets in Damascus/Pleasant Valley area in refined interchange locations along the Sunrise Route, including a connection at 172nd Avenue, the proposed major north/south route in the area
- implement bus service in parallel corridor from Damascus to Clackamas regional center via Sunnyside Road
- avoid premature construction that could unintentionally increase urban pressures in rural reserves east of Damascus
- examine the potential for the highway to serve as a "hard edge" in the ultimate urban form of the Damascus area
- develop a concurrent plan to transition the function of the existing Highway 212 facility into a major arterial function, with appropriate access management and intersection treatments identified
- pursue a Green Corridor intergovernmental agreement (IGA) for the Sunrise Corridor from the Damascus town center to US 26, with the specific western terminus for the IGA flexible to future expansion of the urban growth boundary.

#### I-5 to 99W Connector

An improved regional connection between Highway 99W and I-5 is needed in the Tualatin area to accommodate regional traffic, and to move it away from the Tualatin, Sherwood and Tigard town centers. The RTP has narrowed the corridor to include two alternatives that depart from I-5 in the same general corridor, but split to form northern and southern alignments relative to the City of Sherwood. Impacts on rural resources in both alignments of this corridor shall be addressed through statewide planning goal exception findings that expand on findings already adopted in the 2000 RTP, pursuant to Section 660.012.0070 of the state transportation planning rule. This connection will also have significant effects on urban form in this rapidly growing area, and the following considerations should be addressed in a corridor plan:

6-45

- balance improvement plans with impacts on Tualatin and Sherwood town centers and adjacent rural reserves
- in addition to the northern alignment considered in the Western Bypass Study, examine the benefits of a southern alignment, located along the southern edge of Tualatin and Sherwood, including the accompanying improvements to 99W that would be required with either alignment
- identify parallel capacity improvements to Tualatin-Sherwood Road and 99W in Tigard from I-5 to Highway 217 that could be used to phase in, and eventually complement future highway improvements
- link urban growth boundary expansion in this area to the corridor plan and examine potential the proposed highway to serve as a "hard edge" in the ultimate urban form of the Sherwood area
- develop an access management and connectivity plan for 99W in the Tigard area that balances accessibility needs with physical and economic constraints that limit the ability to expand capacity in this area
- consider express, peak-period pricing and HOV lanes
- pursue a Green Corridor intergovernmental agreement (IGA) for the I-5/99W connector and Highway 99W south of the connector.

#### Sunset Highway

Improvements are needed in this corridor to preserve access to and from the central city and the Sunset Corridor employment area, and provide access to Hillsboro regional center. The following elements should be considered as improvements are implemented in this corridor:

- maintain off-peak freight mobility
- phase in capacity improvements from the Sylvan interchange to 185th Avenue, expanding to a total of three general purpose lanes in each direction
- improve light rail service, with substantially increased headways
- construct major interchange improvements at Sylvan, Cedar Hills Boulevard and Cornelius Pass Road
- identify and construction additional overcrossings in the vicinity of interchanges to improve connectivity and travel options for local traffic, thus improving interchange function
- consider express, peak period pricing or HOV lanes when adding highway capacity, especially west of Highway 217

6-46

#### Highway 213

Improvements to this highway link between I-205 and the Willamette Valley should be built in phases, and consider the following:

- continued development of the Oregon City regional center
- interim improvements identified in the 1999 Highway 213 Urban Corridor Study (and included in this plan)
- freight mobility demands
- access needs of Beavercreek urban area, including a re-evaluation of the suitability of Oregon City urban growth boundary expansion in light of transportation constraints
- transit service to areas south of Oregon City.

6-47

#### Macadam/Highway 43

Though heavy travel demand existing along Macadam/Highway 43, between Lake Oswego and the central city, physical and environmental constraints preclude major roadway expansion. Instead, a long-term strategy for high-capacity transit that links the central city to southwest neighborhoods and Lake Oswego town center is needed. As this service is implemented, the following options should be considered in local and special district plans:

- interim repairs to maintain Willamette Shores Trolley excursion service
- implement frequent bus service from Lake Oswego town center to Portland central city in the Macadam corridor
- phasing of future streetcar commuter service or commuter rail in this corridor to provide a high-capacity travel option during congested commute periods, using either the Willamette Shore Line right-of-way, the Macadam Corridor Design Guidelines (1985) rail alignment or other right-of-way as appropriate.
- implement bicycle safety improvements where appropriate south of the Sellwood Bridge

#### 6.7.7 Areas of Special Concern

Section 660.012.0060 of the state Transportation Planning Rule (TPR) allows local plans to "modify planned function, capacity and performance standards, as needed, to accept greater motor vehicle congestion to promote mixed-use, pedestrian friendly development where multi-modal choices are provided." Facilities in the areas or corridors described in this section are expected to exceed the motor vehicle level of service policy set forth in this plan, and fall under this designation, as they are planned mixed use areas that will have a wide range of transportation alternatives.

However, in each case, the range of transportation solutions needed to address an RTP motor vehicle deficiency represents an unacceptable social, financial or environmental impact, and would be inconsistent with other local, regional and statewide planning goals. Further, each of these areas or corridors represents a relatively localized impact on the overall regional system, and other, alternative travel routes that would continue to conveniently serve regional travel needs. Strategies for managing traffic impacts and providing adequate transportation performance in these areas could include bicycle, pedestrian and transit improvements, demand management programs or changes to land-use plans.

In these areas where motor vehicle performance measures will be exceeded, local TSPs shall adopt one of the following approaches for establishing other transportation performance standards for Areas of Special Concern:

- 1. Adopt the following performance measures, and provide an analysis that demonstrates progress toward meeting these measures in the local TSP:
  - a. Non-SOV modal targets consistent with Table 1.3 in Chapter 1 of this plan

6-48

- b. parking ratios consistent with Title 2 of the Urban Growth Management Functional Plan (UGMFP)
- c. a street connectivity plan for the Area of Special Concern that meets the connectivity requirements set forth in Section 6.4.5 of this chapter
- d. a plan for mixed-use development
- 2. Establish an Area of Special Concern action plan that:
  - a. anticipates the growth and subsequent impacts of motor vehicle traffic on multi-modal travel in these areas
  - b. establishes an action plan for mitigating the growth and subsequent impacts of motor vehicle traffic
  - c. establishes performance standards for monitoring and implementing the action plan

The action plan shall consider land-use strategies, as well as transportation solutions for managing the effects of continued traffic growth.

For either strategy, the adopted approach and performance measures shall be incorporated into Appendix 3.6 of the RTP during the next scheduled update. For an Area of Special Concern, adopted performance measures consistent with this section are required at the time of a plan amendment that significantly affects a regional facility, consistent with OAR 660.012.0060.

The following Areas of Special Concern where refinement planning to establish performance measures shall occur as part of the local TSP process, in accordance with this section:

Highway 99W



The Highway 99W corridor between Highway 217 and Durham Road is designated as a mixed-used corridor in the 2040 Growth Concept, and connects the Tigard and King City town centers. This route also experiences heavy travel demand. The City of Tigard has already examined a wide range of improvements that would address the strong travel demand in this corridor. The RTP establishes the proposed I-5 to 99W connector as the principal route connecting the Metro region to the 99W corridor outside the region. This emphasis is intended to change in the long term the function of 99W, north of Sherwood, to a major arterial classification, with less need to accommodate longer, through trips.

However, for much of Washington County, Highway 99W will still be a major connection, linking Sherwood and Tigard to the rest of the County and linking the rest of the County to the Highway 99W corridor outside of the region. A number of alternatives for relieving congestion have been tested as part of the RTP update, and by the City of Tigard in earlier planning efforts. These efforts led to the common conclusion the latent travel demand in the Highway 99W corridor is too great to be reasonably offset solely by capacity projects. While the RTP proposed new capacity on 99W between I-5 and Greenburg Road, no specific capacity projects are proposed south of Greenburg Road, due to latent demand and the impacts that a major road expansion would have on existing development. As a result, this section of Highway 99W is not expected to meet the region's motor vehicle level of service policies during mid-day and peak demand periods in the future, and an alternative approach to managing and accommodating traffic in the corridor is needed.

Since statewide, regional and local travel will still need to be accommodated and managed for sometime ODOT, Metro, Washington County and Tigard should cooperatively address the means for transitioning to the future role of the facility to emphasize serving circulation within the local community. This will include factoring in the social, environmental and economic impacts that congestion along this facility will bring. Additionally the analysis should specifically document the schedule for providing the alternatives for accommodating the regional and statewide travel. Similarly the local TSPs should include the agreed upon action plans and benchmarks to ensure the local traffic and access to Highway 99W is managed in a way that is consistent with broader community goals. Additional alternative mode choices should be ensured for Tigard and King City town centers. Tri-Met should be a major participant in the alternative mode analysis. The results of this cooperative approach should be reflected in the local TSPs and the RTP.

In addition, other possible solutions, such as ODOT's new program for local street improvements along highway corridors, may provide alternatives for managing traffic growth on 99W. Finally, the local TSPs should also consider changes to planned land use that would minimize the effects of growing congestion.

#### Gateway Regional Center



Gateway is at a major transportation crossroads, and suffers and benefits from the level of access that results. The Preferred System analysis shows that from the perspective of employers looking at labor markets, the Gateway area is the most accessible place in the Metro region. At the same time, spillover traffic from the Banfield Freeway corridor exceeds the LOS policy established in Table 1.2 on a number of east/west corridors in the Gateway area, including Halsey, Glisan, Burnside, Stark and Division streets.

The local TSP should examine the ability of local streets in these areas to absorb travel demand to a degree that cannot be measured in the regional model. A traffic management plan for

these streets should be integrated with the overall TSP strategy, but should establish specific action plans and benchmarks for facilities determined to exceed the LOS policy in the local analysis. Alternative mode choices should be identified to further reduce travel demand. The local

6-50

TSP should also consider strategies for providing better access to LRT, including park and ride facilities at station areas.

Tualatin Town Center



Tualatin town center is adjacent to an important industrial area and employment center. New street connections and capacity improvements to streets parallel to 99W and I-5 help improve local circulation and maintain adequate access to the industrial and employment area in Tualatin. However, the analysis of travel demand on regional streets shows that several streets continue to exceed the LOS policy established in Table 1.2, including Hall Boulevard and Boones Ferry Road.

The Tualatin transportation system plan should further evaluate ITS or other system management strategies to further address travel demands and peak-hour expected congestion

along Hall Boulevard and Boones Ferry Road entering the town center. In addition, the local TSP should examine the ability of local streets in these areas to absorb travel demand to a degree that cannot be measured in the regional model. A traffic management plan for these streets should be integrated with the overall TSP strategy, but should establish specific action plans and benchmarks for facilities determined to exceed the LOS policy in the local analysis. Alternative mode choices should be identified to further reduce travel demand in addition to placing an emphasis on connectivity, including new development, retrofits and interconnected parking lots in commercial/employment areas. Overall, commuter rail is expected to be an important part of the modal mix of improvements for this part of the region because it offers separate right-of-way for transit service in a corridor that is expected to experience congestion during the morning and evening two-hour peak period. The local TSP should also consider strategies for providing better access to commuter rail.

#### 6.8 Outstanding Issues

The section describes a number of outstanding issues that could not be addressed at the time of adoption of this plan, but should be addressed in future updates to the RTP.

#### 6.8.2 Damascus/Boring-Pleasant-Valley TCSP-Concept Planning

Metro was recently awarded a special federal TCSP grant from the US Department of Transportation to complete an urban reserve plan for the Damascus-Pleasant-Valley-area of Clackamas County. The work scope for the project is broad, encompassing land-use, transportation, and environmental planning. The project is scheduled to begin in early 2000. The objective of the study is to prepare concept plans for this large urban reserve area in anticipation of future urbanization. Metro will work with a number of local partners to complete the project, including the cities of Portland, Gresham and Happy Valley, and Multnomah and Clackamas counties. A citizen policy advisory committee that includes residents and key stakeholders will guide the project.

6-51

The Damascus-Pleasant Valley planning effort will include conceptual transportation planning for regional facilities in the area, and more detailed street planning for northern portions of the area that are already included in the urban area. Transportation and land use scenarios will be developed to reflect a variety of land use alternatives for the area, and will be analyzed with the regional transportation model.

The preferred alternative will likely include refinements to the Damaseus Pleasant Valley street functional classifications and transportation improvements included in this plan.

Metro received federal grant money for the purpose of completing a concept plan for a new urban area in the Damascus/Boring area. Clackamas County and Metro will jointly develop the concept plan, with the assistance of a Contractor and the participation of area citizens, key organizations, service providers and cities. ODOT will also participate in the process. The concept planning is aniticpated to start in winter of 2003, will take approximately two years to complete. There will be extensive public involvement during this process.

The Damascus/Boring Concept Plan will be a cooperative planning effort to create plan and implementation strategies for development of approximately 12,000 acres located south of Gresham and east of Happy Valley in Clackamas County. The concept plan is a follow-up to a December 2002 decision by Metro to bring the area inside the Urban Growth Boundary. The Damascus/Boring Concept plan will be closely coordinated with the environmental analysis of the Sunrise Corridor Unit 1 effort and will address the general need, modes, function, and location of the proposed Sunrise Corridor Unit 2. Important components of the concept plan are expected to include:

- <u>A land-use element that locates a combination of uses and densities that support local and regional housing and employment needs, provides a diverse range of housing, and identifies commercial and industrial employment opportunities that allow residents to work near their home</u>
- <u>A multi-modal transportation system element that serves interstate, regional and</u> community travel needs and informs the Sunrise Corridor Unit 2 planning process
- <u>A natural resources element that identifies natural resource areas and protection strategies</u>
- A public infrastructure and facilities element for water, sewer, storm water, parks, schools, fire and police

The concept plan will provide the basis for future comprehensive plan amendments and development code regulations that must be adopted before development can take place. The Damascus/Boring Concept Plan will identify and evaluate multi-modal transportation system alternatives to serve regional and community needs in the area. The alternatives will include combinations of highway, arterial, boulevard and transit improvements that are complemented by a network of local streets, multi-use trails and bicycle and pedestrian connections. If the Damascus/Boring Concept Plan reaffirms that Sunrise Corridor Unit 2 improvements are needed, the concept plan will identify transportation alternatives to be evaluated through a future DEIS process similar to that already initiated for the Unit 1 portion of the Sunrise Corridor.

Proposed amendments to the RTP would be considered upon completion of the study, which is scheduled to conclude in Fall 2002. The preferred alternative will also include future street plans for some local streets that may be incorporated into local TSPs.

6-52

#### 6.8.3 Regional Transportation Model Enhancements

#### Multi-modal Performance Measure Development

Section 660.012.0060 of the state Transportation Planning Rule allows for the development of alternative measures for evaluating transportation function and efficiency. Though the principal measure in this plan measures motor vehicle performance, future updates to the plan should uses a multi-modal measure that better reflects transportation needs and potential solutions. Such measures are already used for Areas of Special Concern identified in Chapter 1 of this plan, but should also be considered in other areas to better evaluate both the need and relative effectiveness of multi-modal transportation solutions.

#### Tour-Based Modeling and TRO Enhancements

Tour-based modeling represents a departure from the current trip-based model used to develop the RTP. In contrast to the current model, tour-based modeling allows for a much more detailed analysis, since it does not rely on the somewhat generalized assumptions that accompany the current model. In the current system, land-use and transportation assumptions are created for each of 1,260 traffic zones that form the smallest building block for analysis. Tour-based modeling will allow data to be evaluated to the tax lot or parcel level, which will result in a much more detailed and flexible system for testing proposed transportation improvements.

The recently completed Traffic Relief Options (TRO) project was the first Metro effort to use tourbased modeling. This study tested the effects of congestion pricing on travel in the region, and allows relative pricing costs to be evaluated in terms of the ability to redistribute travel and manage congestion. The tour-based model with TRO enhancements could offer a unique new tool for future RTP updates, as the concepts of congestion pricing and tolling are likely to be considered as major transportation strategies.

#### Bicycle and Pedestrian Modeling

The existing regional transportation model probably underestimates bicycle and pedestrian trips, and does not predict bicycle travel according to the transportation network. Instead, the current model predicts bicycle and pedestrian trips as part of the "mode choice" step of the modeling process, but does not assign these trips to a network to predict how they might be distributed. While pedestrian trips are generally short enough to make a network assignment impractical, bicycle trips are of sufficient length to be assigned to a network and evaluated at this level. As part of a future update to the RTP or the Regional Bicycle Plan, Metro will develop a bicycle network modeling process that will improve the region's ability to plan for bicycle travel.

#### The ODOT Willamette Valley Model

ODOT has developed a more detailed set of travel zones for the Willamette Valley, which will allow Metro to better predict travel demand at "gateway" points where Willamette Valley traffic enters the region. Currently, the regional model simply projects historic traffic volumes on such routes, but is unable to evaluate how congestion, parallel routes, and distribution of employment in and outside the region affects travel demand at these "gateway" locations. The ODOT Valley

6-53

Model has been used in other Metro transportation projects, and should be considered for the next RTP update.

#### 6.8.4 Connectivity Research

In1996, Metro completed the Regional Street Design study, a project that resulted in new regional street design classifications in the RTP and connectivity provisions in the UGMFP. The connectivity provisions were based on a series of five case studies of subareas within the Metro region. These areas averaged two square miles in area, and ranged from a very urbanized neighborhood in Portland, to developing areas in Clackamas and Washington counties. For each subarea, conceptual street systems were used to evaluate the benefits of varying levels of street connectivity. The results of this analysis are published in Metro's technical report Street Connectivity Analysis (1997).

The connectivity analysis in the 1996 study was limited to motor vehicles, and while the findings from the study are conclusive, the consultant for the project recommended an expanded analysis of one or two of the subareas to confirm the sensitivity analysis included in the original study.

A follow-up study is proposed to confirm the motor vehicle findings of the 1996 study, and expand the analysis to examine the effects of varying levels of connectivity on pedestrian, transit and bicycle travel. This follow-up study could result in proposed changes to existing UGMFP connectivity requirements. This follow-up study is scheduled to be conducted by Metro upon completion of the 2000 RTP update, and recommendations from the study could be considered for adoption in 2001.

#### 6.8.5 Ramp Metering Policy and Implications

During the 1990s, ODOT has increasingly managed access to the principal arterial system (freeways and highways) with ramp metering. This system of signaled ramp controls allows ODOT to remotely manage traffic flows onto the system to streamline merges and prevent bottlenecks during peak travel periods. Ramp meters provide a low-cost alternative for adding system capacity and enhancing safety. However, as traffic volumes continue to increase on the principal arterial system as well as connecting major and minor arterial routes, the practice of ramp metering will become more complex. Already, local concerns about ramp "storage" capacity forcing backups onto local routes have required ramp expansions in some locations where metering is used.

As part of the next update of the RTP, the policy considerations raised by ramp metering should be addressed. The fundamental principle behind ramp metering is to maintain traffic flows on principal routes as a priority over local arterial routes. However, this assumption should be carefully evaluated on the basis of the performance and reliability requirements of the freeway system in the context of the new land use patterns and street classifications and configurations evolving out of the Region 2040 growth concept.

#### 6.8.6 Green Corridor Implementation

Green corridors were adopted as part of the 2040 Growth Concept. They are designated in rural areas where state-owned highways connect neighbor cities to the metro area. The purpose of green corridors is to prevent unintended urban development along these often heavily traveled routes, and

6-54

maintain the sense of separation that exists between neighbor cities and the Metro region. The green corridor concept calls for a combination of access management and physical improvements to limit the effects of urban travel on the routes on adjacent rural activities.

In several corridors, Metro has already developed inter-governmental agreements (IGAs) with local governments to address access management issues. However, IGAs are not in place in most corridors, and physical improvements, such as street and driveway closures, landscaping and public signage have not been implemented in any green corridors. During the next several years, Metro will continue to work with ODOT and affected local jurisdictions to complete IGAs for the remaining green corridors, and develop plans for necessary improvements. Such improvements should be incorporated into future updates of the RTP.

#### 6.8.7 2040 Land-use and Transportation Evaluation

Though the RTP contains a number of land-use recommendations, more work is needed to further evaluate RTP and 2040 Growth Concept to determine potential land-use changes that would be beneficial to the transportation system. This evaluation would consider directing growth away from areas that do not have adequate transportation systems, and focusing growth in areas with surplus transportation capacity, as well as improving the balance of jobs and housing to reduce long-distance commuting on the principal arterial system. The evaluation would also include an analysis of the effect of relative wages on the mix of jobs and housing needed to realize transportation benefits.

- Damascus & Pleasant Valley Urban Reserves: The overall jobs/housing imbalance in Clackamas County results in heavy travel demand on routes like I-205 and Highway 224 that link Clackamas County to employment areas. A review of the Damascus and Pleasant Valley Urban Reserves should consider the potential for improving jobs/housing balance in these areas. This review should include areas in the Pleasant Valley areas that have been recently incorporated into the urban area, but are largely undeveloped.
- Beavercreek Urban Reserves: Urbanization of these reserves would require major improvements to Highway 213 and connecting arterial streets that may be inappropriate in scale and cost, and could negatively impact adjacent areas in Oregon City.

6-55

#### 6.8.8 Industrial Lands Evaluation

Additional work is needed in Tier 2, 3 and 4 urban reserve lands to determine where strategic transportation improvements could be implemented to make industrial land more viable for development. This evaluation would identify key areas for industrial development where non-transportation actions would enable industrial development that complements the planned transportation system.

#### 6.8.9 TDM Program Enhancements

The TDM Subcommittee is in the process of developing a 3-5 year strategic plan that clearly articulates a new vision and proposed direction for the Regional Travel Options program. The strategic direction is to develop a more collaborative marketing program that eliminates duplication of marketing effort and that delivers a clear message to all of our customers (students, commuters, aging population, shoppers, etc). The regional evaluation program will also become more collaborative as we work to develop performance measure and evaluate progress toward non-SOV modal targets for regional centers and industrial areas. The strategic plan will update TDM policies resulting in RTP Amendments that reflect new strategies for promoting travel options to the region.

In addition, t<sup>T</sup>he TDM program should be continually updated to include new strategies for regional demand management. One such strategy that should be considered is the Location Efficient Mortgage (LEM). The LEM is a mortgage product that increases the borrowing power of potential homebuyers in "location efficient" neighborhoods. Location efficient neighborhoods are pedestrian friendly areas with easy access to public transit, shopping, employment and schools. The LEM recognizes that families can save money by living in location efficient neighborhoods because the need to travel by car is reduced. Instead of owning two cars, a family living in a location efficient neighborhood could get by with one - or none. The LEM requires bankers to look at the average monthly amount of money that applicants would be spending on transportation if they had to use a car for day-to-day transport and applies it to the servicing of a larger mortgage. This increases the purchasing power of borrowers when buying a home in location efficient neighborhoods, stimulating home purchases in existing urban areas.

#### 6.8.10 Transportation Performance Measures

The 2000 RTP marks marked the first time in the 18-year evolution of the plan that a performance measure other than congestion is adopted as regional policy. The newly incorporated Area of Special Concern designation allows for a broader definition of performance in mixed use centers and corridors, where transportation solutions solely aimed at relieving congestion are inappropriate for functional, physical, financial or environmental reasons.

However, the Area of Special Concern designation is only a first step toward a more broadly defined set of performance measures. Future updates of the RTP should continue to expand the definition of performance to encompass all modes of travel as they relate to planned land uses. While congestion should be factored into a more diverse set of measures, it should be evaluated in a more comprehensive fashion to ensure that transportation solutions identified in future RTP updates represent the best possible approaches to serving the region's travel demand.

6-56

#### Section 6.8.11 Transit Stop Planning

Tri-Met, in cooperation with regional partners, defined most of the major transit stops as a part of the Primary Transit Network planning process in 1997. Planning for the location of transit station continues as Tri-Met and other transit providers participate in specific corridor planning or implements elements of their strategic plan. Amendments to Figure 1.16 will be necessary as these planning efforts continue. As these planning efforts will include participation from the affected local jurisdictions, amendments to their transportation system plans should be made as planning is completed.

As a part of these planning efforts, transit providers may consider policy standards for station spacing for particular types of service lines, amenities to be provided at transit stops and design standards for those amenities. Jurisdictions are also encouraged to undertake transit stop area plans at major transit stops on rapid bus lines, similar to previous planning efforts for light rail stations.

#### 6.8.12 Job Access and Reverse Commute

The Transportation Efficiency Act (TEA-21) of 1998 included the Job Access and Reverse Commute Program to address the mobility challenges facing welfare recipients and low-income persons. This grant program requires States to develop solutions collaboratively with Metropolitan Planning Organizations (MPOs), local and regional transportation agencies and social service providers. The federal Job Access and Reverse Commute Program provides grants to help States and localities develop a coordinated, regional approach to new or expanded transportation services that connect welfare recipients and other low-income persons to jobs and other employment services. Job Access projects support developing new or expanded transportation services such as shuttles, vanpools, new bus routes, guaranteed ride home programs and other transit service expansion for welfare recipients and low-income persons. Reverse Commute projects provide transportation services to suburban employment centers from urban, rural and other suburban locations for all persons.

In response to the federal legislation, the purpose of the Portland Job Access Plan is to connect lowincome persons and those receiving Temporary Assistance to Needy Families (TANF) with employment areas and related services in the Portland metropolitan region. The community to be served includes approximately 220,000 people with incomes 150 percent below the poverty level. In 1999, Phase I funding for Portland's Job Access Plan matched existing local resources with federal funds to provide over 87,000 new transit rides for low-income and welfare recipients in Washington, Clackamas and Multnomah counties. The new services improved connections and services to both urban and rural areas of the tri-county area using a combination of public, non-profit and private providers. This has allowed individuals with limited resources to enhance their access to the regional transit network and reduce their transportation burdens. The Regional Job Access Committee represents more than 20 organizations, including Metro, transit providers, social service agencies, child care providers and employers.

Many of today's entry-level positions do not work traditional work hours and the public transportation system is less efficient or non-existent during off-peak shift times. More than 75 employers, representing more than 25,000 employees, have new transportation options for these "hard to serve" shifts from the first year federal Job Access funds. New transportation options range

6-57

from carpool incentives to evening or early morning shuttle services which allow low-income job seekers access to otherwise unattainable employment locations.

While job training is a key to job placement, the Portland Job Access Plan recognizes that travel training is a key to job retention. Knowing how to use the available transportation services can ease the commute and provide options for childcare. The plan stresses regional coordination and information access as a key to preparing welfare recipients for their commute.

#### 6.8.13 Financial Implementation

JPACT will convene a committee to address transportation funding issues. This committee will consider the information and concepts addressed in Section 5.4 and report back to JPACT with a funding implementation strategy and an analysis of how the strategy addresses the principles identified in Section 5.4.1. JPACT and its transportation funding committee will work with other government agencies, private sector and non-profit agency efforts to address transportation funding in the state and region as it considers its implementation strategy. This effort will lead to proposals for new sources of transportation revenue to build, operate and maintain the RTP Priority system.

#### 6.8.14 RTP Modal Targets Implementation

Metro was recently awarded state Transportation/Growth Management funds to identify best practices and further clarify what constitutes a minimum requirements for local transportation system plans to meet the RTP modal targets. Metro's primary goal is to ensure that the planning programs be adopted, and that on-the-ground progress be demonstrated over time. However, progress toward the non-SOV modal targets is an output of the regional travel demand model, but cannot be generated by local jurisdictions. Progress would be periodically evaluated as part of RTP updates. The project will:

- Identify best practices and minimum requirements for local governments to demonstrate that local TSPs can meet non-SOV mode split targets in the RTP. Meeting this objective will allow Metro to ensure RTP compliance with Section 660-012-0035(5) of the Transportation Planning Rule.
- Ensure that minimum requirements identified are reasonably sufficient to enable loca l jurisdictions to achieve the Non SOV Modal Targets of Table 1.3 and the Alternative Mode Analysis of section 6.4.6 of the RTP.
- Ensure that minimum requirements identified can be carried out by Metro and/or local jurisdictions without a significant commitment of staff time or other resources.
- Provide education on the benefits of reducing non-SOV mode trips.

This effort could result in amendments to the RTP.

#### 6.8.15 Defining System Adequacy

Section 660.012.0060 of the Oregon Transportation Planning Rule (TPR) requires local governments to evaluate amendments to acknowledged plans and regulations to ensure that the changes are consistent with planned transportation improvements. For the Metro region, the RTP defines the "preferred" system of improvements for major transportation facilities as the basis for evaluating such amendments.

6-58

However, given that a XX percent funding shortfall between the preferred system and existing revenue projections exists, this methodology can result in plan amendments being justified by transportation improvements that are unlikely to occur in a timely period, due to the current funding shortfall. Under this scenario, a more realistic basis for evaluating the system might be the "financially constrained" system, which represents just XX percent of the larger "preferred" system, and is based on recent funding history. Conversely, using the much more conservative financially constrained system for this analysis risks turning away unanticipated economic development that is consistent with the general intent of a local plan, but requiring greater transportation infrastructure than is provided in the constrained scenario.

Prior to the next update to the 2004 RTP, the issue of defining an adequate system of improvements for the purpose of evaluating local plan amendments should be addressed in detail to ensure a balance between allowing desired development and preventing land use actions that outstrip the public ability to provide transportation infrastructure. This effort should include a cross-section of local and regional interests and state agency officials, and could lead to recommended RTP amendments that implement a new strategy for considering such proposals. The effort should be led jointly by Metro and the Oregon Department of Transportation.

#### 6.8.16 Wilsonville I-5 South Corridor

Based on the results of the I-5/Wilsonville Freeway Access Study (DKS Associates, November 2002, prepared for ODOT and the City of Wilsonville, with Metro's participation), there will be a future deficiency for freeway access capacity in Wilsonville based on year 2020 PM peak forecasts. Improvements were identified in the City of Wilsonville's 2003 Transportation Systems Plan to address this deficiency, but did not include the effects of the planned southern alignment for the I-5 to 99W Connector to the Stafford Road Interchange, the plans for which were outside of the scope of the TSP. The improvements include an improved local street system in Wilsonville, freeway access improvements and I-5 operational improvements. Improvements to the local roadway system are not adequate by themselves to mitigate the future 2020 interchange access needs without interchange improvements. In evaluating two freeway access improvement alternatives (an enhanced Wilsonville Road diamond interchange and a new Boeckman Road interchange to I-5) it was found that improvements to the Wilsonville Road interchange would be necessary with either interchange alternative. Based upon the findings of study, an enhanced Wilsonville Road diamond interchange, currently in preliminary engineering, is needed to meet future 2020 capacity demands. Implementation of the enhanced Wilsonville Road diamond interchange project depends upon funding availability.

The analysis of future freeway access needs was conducted with a wide range of travel forecasts, assessing the sensitivity of the findings in the 2020 PM peak period with various travel demand assumptions. In each case, the findings noted above were found to be consistent in terms of the required first step being the enhanced Wilsonville Road diamond interchange. However, utilizing an approximation technique to extend 2020 forecasts to 2030, it was found that in 2030 widening of I-5 to eight lanes would be required to meet interstate freeway capacity standards set by Metro and ODOT and that freeway access capacity would not be adequate with the improved I-5/Wilsonville Road interchange and further access improvements would be necessary. Thus, other freeway access improvements (e.g. a new Boeckman Road interchange) must be considered in future regional capacity studies, including the Regional Transportation Plan update, I-5 South Corridor Study, I-5

to 99W Connector and / or a Stafford / I-205 Study in conjunction with possible urban growth boundary expansions and industrial land evaluations.

#### 6.8.17 National Highway System (NHS) Routes Update

A component of the federal requirements that warrants special effort is a needed update to the National Highway System (NHS) designations in the RTP. These routes were originally designated in the early 1990s, and are due for an update that considers 2040 land use and transportation considerations that have since been adopted into regional and local plans. This effort will occur prior to the next RTP update.

# How to Comment on the update to the 2004 Regional Transportation Plan

The public comment period for the 2004 Regional Transportation Plan (RTP) begins on October 31, 2003 and concludes with a public hearing on December 4, 2003. You may submit comments online at Metro's website:

#### www.metro-region.org/rtp

Comments and questions may also be mailed using the form below, or left on Metro's Transportation hotline at (503) 797-1900, Option 2.

### **Comments:**

	 <u>.</u>	
	 · · · · · · · · · · · · · · · · · · ·	
·	 	·······

## Submitted by:

Name	
Street Address	City/Zip
Phone	E-Mail
Send me more info:	
2000 RTP Document CD	Other RTP Info:
Please add me to the RTP i	nterested citizens mailing/e-mail lists

## **Regional Transportation Plan Update Calendar**

- October 31 Public comment period begins; staff recommendation on draft 2004 RTP released for 30-day public comment period; draft RTP and conformity determination submitted to FHWA and FTA to begin review
- November 3 Air quality conformity analysis begins
- November 5 MTAC comments on draft 2004 RTP
- November 12 MPAC comments on draft 2004 RTP
- November 13 JPACT tentative action on draft 2004 RTP
- November 13 Metro Council first reading of Ordinance on draft 2004 RTP
- **November 26** TPAC review and discussion of draft 2004 RTP and air quality conformity analysis
- **December 4** Public hearing on draft 2004 RTP; public comment period ends at 5 p.m.
- December 5 TPAC special meeting to comment on draft 2004 RTP
- **December 10** Tentative final MPAC action on 2004 RTP
- December 11 Tentative final JPACT action on 2004 RTP
- **December 11** Metro Council second reading of Ordinance and consideration of adoption of 2004 Regional Transportation Plan

#### FOLD HERE



Place first class postage here.

Metro 600 NE Grand Avenue Portland, Oregon 97232 Attention: Marilyn Matteson





## 2004 Regional Transportation Plan **Air Quality Conformity**

## October 31, 2003



PEOPLE PLACES OPEN SPACES

#### Metro People places • open spaces

Metro serves 1.3 million people who live in Clackamas, Multnomah and Washington counties and the 24 cities in the Portland metropolitan area. The regional government provides transportation and land-use planning services and oversees regional garbage disposal and recycling and waste reduction programs.

Metro manages regional parks and greenspaces and owns the Oregon Zoo. It also oversees operation of the Oregon Convention Center, the Portland Center for the Performing Arts and the Portland Metropolitan Exposition (Expo) Center, all managed by the Metropolitan Exposition Recreation Commission.

#### Your Metro representatives

Metro Council President – David Bragdon Metro Councilors – Rod Park, District 1; Brian Newman, District 2; Carl Hosticka, District 3; Susan McLain, District 4; Rex Burkholder, District 5; Rod Monroe, District 6. Auditor – Alexis Dow, CPA

Metro's web site: www.metro-region.org

Metro 600 NE Grand Ave. Portland, OR 97232-2736 (503) 797-1700 TDD (503) 797-1804

Printed on 100 percent recycled paper, 30 percent post-consumer fiber



## 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

## Conformity Determination Report Table of Contents

А.	Introduction	1
	1. Background	1
	2. Reason for Determination	5
В.	Overview of 2004 RTP and Major Changes in Network Assumptions	7
	1. Unchanged Network Assumptions Carried Over from 2000 RTP	7
	2. New 2004 RTP Network Assumptions	8
C.	Summary of Relevant Conformity Requirements & Findings of Compliance	9
	1. Consistency with the latest planning assumptions (OAR 340-252-0110)	9
	2. Latest emissions model (OAR 340-252-0120)	26
	3. Consultation (OAR 340-252-0130)	26
	4. Timely implementation of TCMs (OAR 340-252-0140)	31
	5. Support achievement of NAAQS	31
	6. Quantitative analysis (OAR 340-252-0190)	32

#### Appendices

Appendix 1: Financially Constrained System Project List (Not included in this draft - See RTP Packet 2 - Project Amendments)

Appendix 2: 2004 RTP Public Involvement

Appendix 3: 2004 RTP and 2004-07 Metropolitan Transportation Improvement Program Conformity Analysis Protocol

Appendix 4: Transportation Analysis Zone (TAZ) Assumptions

Appendix 5: 2004-07 MTIP Public Involvement

Appendix 6: Published notice, other notices

Appendix 7: Evaluation of emissions analysis for transportation activities which cross borders of MPOs

Appendix 8: Evidence of Compliance with Metro Interim Land Use Measures

Appendix 9: Interagency Coordination - October 2, 2003 Meeting Summary

Appendix 10: Portland Area Motor Vehicle Fleet Assumptions

Page 1



## 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program Conformity Determination

### A. Introduction

#### Background

The federal Clean Air Act provides the main framework for national, state and local efforts to protect air quality. Under the Clean Air Act, the Environmental Protection Agency (EPA) is responsible for setting standards, known as national ambient air quality standards (NAAQS), for pollutants considered harmful to people and the environment. These standards are set at levels that are meant to protect the health of the most sensitive population groups, including the elderly, children and people with respiratory diseases. Air quality planning in this region is focused on meeting the NAAQS and deadlines set by the federal Environmental Protection Agency and state Department of Environmental Quality for meeting the standards. Further, the United States Department of Transportation has established regulations which make failure to meet these standards result in a loss of transportation funding from state and federal sources and increased health risks to the region.

The 2004 Regional Transportation Plan (RTP) and 2004-07 Metropolitan Transportation Improvement Program are subject to an air quality conformity determination under federal regulation (40 CFR Parts 51 and 93) and state rule (OAR 340 Division 252). Metro, as the federally designated Metropolitan Planning Organization (MPO) for the Oregon portion of the Portland-Vancouver air shed, is the lead agency for the conformity determination. In addition, the Transportation Policy Alternatives Committee (TPAC) is called out under the state rule as the standing committee designated for "interagency consultation" as required by the rule. In order to demonstrate that the 2004 Regional Transportation Plan (RTP) and the 2004-07 MTIP meet federal and state air quality planning requirements, Metro must complete a technical analysis that is known as air quality conformity. The need for this analysis came from the integration of requirements in the Clean Air Act Amendments of 1990 and the Intermodal Surface Transportation Equity Act for the 21st Century (TEA21) in 1998. Conformity is a regulation requiring that all transportation plans and programs in air quality non-attainment or maintenance areas conform to the State's air quality plan, known as the State Implementation Plan (SIP). Transportation plans and programs such as the 2004 RTP and the 2004-07 MTIP must not result in air quality violations.

The Portland/Vancouver area has one interconnected airshed. However, given the State boundary along the Columbia River and the differing jurisdictions and state laws, the Federal government approved

Page 2

each side of the airshed taking responsibility for its area. For the Oregon side a Portland Area Airshed was established. However, as there are several types of pollutants of concern in the Portland Area, several geographic areas were established for differing air pollutants.

For Carbon monoxide, the Metro jurisdictional boundary was established as the geographic extent of concern for which emission budgets (maximum pollutant levels) were created. Within that area, their were sub-areas established with their own emission budgets. These sub-areas were the Portland Central City sub-area and the 82nd Avenue subarea.

For precusors of ozone, commonly called smog, geographic boundaries were set that pertained to the level of hydrocarbons (also known as volatile organic compounds) and nitrogen oxide. The Portland Air Quality Maintenance Area was established for addressing ozone and the emission budgets for this area.

The following map shows these boundaries.

Page 3



Plot time: Oct 28, 2003 J:\holsted\03283\_Mark\_Turpel\_AQ\_bdy\metro\_air\_quality\_boundaries.mxd

Please recycle with mixed paper

#### **Reason for Determination**

Metro is the Portland area's designated Metropolitan Planning Organization (MPO). As the MPO, Metro is the lead agency for development of regional transportation plans and the scheduling of federal transportation funds in the Portland urban area. Regulations of the United States Department of Transportation (USDOT) require the MPO to develop a 20-year Regional Transportation Plan (RTP). The Plan must identify revenue that can be reasonably anticipated over a 20-year period for transportation purposes. It must also state the region's transportation goals and policies and identify the range of multi-modal transportation projects that are needed to implement them. Just as Metro is required to develop an RTP, it is also mandated to develop a Metropolitan Transportation Improvement Program (MTIP) for the Portland urban area. The MTIP "program" process is used to determine which projects included in the Plan will be given funding priority year by year.

The U.S. DOT and the U.S. Environmental Protection Agency (EPA) approved and acknowledged the 2000 RTP air quality conformity determination on January 26, 2001. Under federal regulations, the RTP must be updated every three years to ensure that the plan adequately addresses future travel needs and is consistent with the federal Clean Air Act. As a result, an update to the 2000 RTP began in September 2003.

On June 19, 2003, the Joint Policy Advisory Committee on Transportation (JPACT) and the Metro Council approved Resolution No. 03-3335, approving a regional allocation of federal funds for the years 2006 and 2007, pending an air quality conformity analysis for the 2004-07 MTIP. The 2004-07 Metropolitan Transportation Improvement Program (MTIP) schedules spending of federal transportation funds in coordination with significant state and local funds in the Portland metropolitan region for the federal fiscal years 2004 through 2007. It also demonstrates how these projects relate to federal regulations regarding project eligibility, air quality impacts, environmental justice and public involvement.

On August 11, 2003 the U.S. DOT recommended that the 2004 RTP air quality conformity analysis and determination be completed jointly with the conformity analysis for the 2004-07 Metropolitan Transportation Improvement Program (MTIP).

On December 11, 2003, the Metro Council is scheduled to take action on the 2004 Regional Transportation Plan (RTP), the 2004-07 MTIP and the conformity determination for both plans. In order to ensure that the 2004 RTP is in compliance with air quality requirements, this Conformity Determination has been prepared for the financially constrained system of the 2004 Regional Transportation Plan (RTP) which also includes projects identified in the 2004-07 MTIP.<sup>1</sup> It has been prepared because the RTP and

Page 5

<sup>&</sup>lt;sup>1</sup> Defined in Chapter 5 of the 2004 Regional Transportation Plan and in Appendix 1 to this document, the financially constrained system responds to federal planning requirements. This system of projects and programs is limited to current funding sources, and those new sources that can be reasonably expected to be available during the 20-year plan period. As the federally recognized system, the financially constrained system is also the source of transportation projects that may be funded through the Metropolitan Transportation Improvement Program (MTIP). The MTIP allocates federal funds in the region. The 2004 RTP not only provides an updated set of financially constrained projects and programs for future MTIP allocations, but also establishes more formal procedures and
the MTIP must be conformed every three years, as described in OAR Chapter 340, Division 252, section 50. A new plan and MTIP demonstrating conformity with the Clean Air Act must approved and acknowledged by US DOT and US EPA in a formal conformity determination by January 26, 2004, when the current US DOT/US EPA conformity determination for the 2000 RTP expires.

Section B of this conformity determination provides an overview of the 2004 RTP and major changes to road and transit network assumptions. The State Transportation Conformity Rule requires that the air quality conformity determination comply with several subsections of OAR Chapter 340, Division 252, including:

- 1. OAR 340-252-0110 Use of the Latest Planning Assumptions
- 2. OAR 340-252-0120 Use of Latest Emissions Model
- 3. OAR 340-252-0130 Consultation
- 4. OAR 340-252-0140 Timely Implementation of Transportation Control Measures (TCMs)
- 5. OAR 340-252-0190 Motor Vehicle Emissions Budget

Section C discusses the relevant conformity determination requirements and demonstrates that this Determination complies with each requirement. Metro's technical analysis indicates that regional emissions will remain within established budgets in all analysis and budget years (i.e., 2006, 2007, 2010, 2015, 2020 and 2025). The following analysis demonstrates how the conformity determination for the 2004 Regional Transportation Plan complies with applicable requirements of OAR Chapter 340, Division 252. Inapplicable subsections of Division 252 are not cited in this conformity determination.

This October 31, 2003 draft document contains the assumptions, methodology and budgets (maximum pollutant levels) for determining air quality conformity. However, the calculations to determine whether the proposed financially constrained 2004 RTP and the MTIP meet air quality conformity standards have not yet been completed. Accordingly, reviewers may comment on the assumptions and methodology. Where calculation results are being completed, there is text indicating "Results Pending". Conformity determination results will be made available at a later date for technical and public review. As the financially constrained system of the 2004 RTP is very similar to the 2000 RTP as amended in 2002 and 2003, it is assumed that the 2004 RTP and 2004-07 MTIP will meet conformity standards. Should the calculations result in findings that the 2004 RTP or 2004 MTIP not conform to air quality standards, the technical and public review schedule will be revised to allow for revisions to the RTP and MTIP, revision of air quality calculations and public and technical comment prior to MPO consideration and adoption.

objectives for implementing long-range regional transportation policies through incremental funding decisions. These new MTIP provisions are set forth in Chapter 6 of the 2004 RTP.

Page 6

# B. OVERVIEW OF THE 2004 RTP AND MAJOR CHANGES IN NETWORK ASSUMPTIONS

The 2004 RTP Update represents a minor update to the 2000 RTP that focuses on meeting state and federal requirements, and incorporated new policy direction set by JPACT and the Metro Council as part of various corridor and special studies conducted since 2000. The update will also incorporate a number of "friendly amendments" proposed as part of local transportation plans being adopted over the past three years This update builds on the extensive planning work and analysis that was completed for the 2000 RTP. The 2004 RTP continues to implement the 2040 Growth Concept, the region's long-range plan for addressing expected growth while preserving the region's livability. The 2004 RTP represents a nearly 20-year evolution from a mostly road-oriented plan to a more balanced multi-modal plan that is closely tied to land use and the 2040 Growth Concept. The 2004 plan remains relatively unchanged in terms of the mix of projects, and continues to rely on greater emphasis on a multi-modal transportation system that enhances opportunities for walking, bicycling and use of transit, transportation demand management, street connectivity, and a 2040-based level of service policy that tolerates some congestion, particularly during two-hour peak period in select locations based on availability of other modes of travel such as walking, biking and transit.

The total reasonably expected revenue base assumed in the 2004 RTP for the road system is about...

#### **Results Pending**

The following section summarizes some of the more important similarities and distinctions between the two networks.

- 1. Network Assumptions Carried Over the from 2000 RTP:
- Annual average transit service increase of 1.5 percent through 2006;
- LRT extended from Milwaukie to Vancouver, Washington by 2020, including a first phase Interstate Avenue LRT alignment from the Rose Quarter to the Expo Center (though the opening day for Interstate MAX has changed from September 2004 to May 2004);
- LRT extended from Gateway Regional Center to Clackamas Regional Center and LRT extended along the Portland Transit Mall from the Steel Bridge to PSU along 5th and 6th Avenues.
- Early implementation of an interim "Rapid Bus" system in the 99E corridor on McLoughlin from downtown Portland to Milwaukie.
- Wilsonville/Beaverton Commuter Rail;
- Added freeway lanes:
  - I-5 from Greeley to Interstate Bridge;

Page 7

- US 26 from Highway 217 to Murray Boulevard;
- Highway 217 from Tualatin Valley Highway to 72nd Avenue Interchange.
- Signal system interconnection on significant regional arterial streets.
- Implementation of the central city streetcar from NW Portland to the Macadam district in two phases.
- Improved bus headways and occupancy on numerous priority routes due to implementation of amenities and structural improvements (e.g., "coach-style" buses, dedicated transit lanes, queue jump lanes, signal priority systems, "real-time" on-street bus arrival information displays, etc.)
- Slightly reduced geographic coverage of bus service to emphasize service on the most productive routes;
- Phase 1 construction of the Sunrise Highway from I-205 to Rock Creek;
- Hogan Interchange construction at I-84 to Stark Street.
- The 2000 RTP plans for construction of 34 additional arterial lane miles and 108 more freeway lane miles than assumed in the 1995 RTP (which froze road construction at 2015 levels).
- 2. New 2004 RTP Network Assumptions:
- Base year of 2000.

#### Results Pending

The 2004 RTP builds on the policy direction established in the 2000 RTP, which was to use transportation investment as a means to implement and reinforce the region's land use goals, and more fully defines the methods and projects that will effect this purpose. Extensive interagency consultation was conducted to develop and refine the current financially constrained system project list. The resultant network continues to rely extensively on auto trip making (**Results Pending**) percent of daily trips are single-occupant auto trips in 2025) and therefore continues to reflect significant investment in maintenance and expansion of the region's freeway and street facilities.

However, a more refined multi-modal approach is also exhibited in the 2004 RTP's specification of precise pedestrian and bike system improvements, and the identification of "boulevard-design" locations where the intent is to retrofit designated streets for walking, biking and transit. The retrofits of major streets include wider sidewalks, safer street crossings, bike lanes and improved bus stops and shelters along streets that serve the central city, regional centers, town centers and other areas. The 2004 RTP congestion level of service standards reflect a policy that the associated impacts of wider, faster streets and freeways needed to achieve the traditional service level are too often accompanied by unacceptable impacts on costs, surrounding neighborhoods and alternative travel modes. Some funds previously dedicated to attempts to meet the traditional level of service standard have been freed up to pursue more balanced system investment that is more reliant on system and demand management, walking, bicycling and transit to meet regional trip demand. And as the comparative data above, and in Section C.1(b), below, suggest, this approach yields meaningful reductions of auto trip dependency.

Page 8

### C. Relevant Conformity Requirements and Findings of Compliance

#### 1. Consistency with the Latest Planning Assumptions (OAR 340-252-0110).

a. **Requirement:** The State Rule requires that Conformity Determinations be based "on the most recent planning assumptions" derived from Metro's approved "estimates of current and future population, employment, travel and congestion."

**Finding of compliance:** The *quantitative* analysis (see Section C.6) employs the transportation system planning assumptions completed for the 2004 RTP, and population, employment and development assumptions that reflect Metro adoption of the Regional Framework Plan and its implementing ordinances. The 2000 base year reflects Metro's official estimates of population and employment calibrated to 2000 Census data. Metro has completed a population/employment projection for 2025. The 2025 population/employment projection is the foundation for all analysis years used in this Conformity Determination.

Travel and congestion forecasts in the analysis years of 2000, 2010 and 2025 are derived from the population/employment data using Metro's regional travel demand model and the EMME/2 transportation planning software. Within subroutines of the regional travel demand model, Metro calculates the transit/bike/walk mode split for calculated travel demand based on a variety of factors, including trip distance, car per worker relationship, transit headways, total employment within one mile, intersection density and a zone-based mixed-use index of the ratio of total employment to total population (see Appendix 4). Both the population and employment estimates and the methodology employed by the EMME/2 model have been the subject of extensive interagency consultation and agreement (discussed further in Section C.3).

The resulting estimates of future year travel and motor vehicle congestion are then used with the outputs of the EPA approved MOBILE 5a-h emissions model to determine regional emissions. In all respects, the model outputs reflect input of the latest approved planning assumptions and estimates of population, employment, travel and congestion.

b. **Requirement:** The State Rule requires that changes in transit policies and ridership estimates assumed in the previous conformity determination must be discussed.

**Finding of compliance:** Changes in transit policies and ridership estimates are discussed below for each type of transit service assumed in the 2004 RTP transit network: light rail, commuter rail, rapid bus, frequent bus, regional bus and community bus.

**LRT Extension.** The *transit policies* which guide modeled implementation of light rail transit (LRT) service in the South/North corridor are consistent with previous Conformity

Page 9

modeling of the Westside and Hillsboro LRT service starts. Bus resources providing downtown radial service are replaced with LRT service. Previous short-haul service between former radial trunk routes is reconfigured to support new LRT stations and surrounding neighborhoods. This represents continuation of *existing transit policy* and its extension to the expanded LRT system. The same principles are further extended to implementation of planned commuter rail in South Washington County.

Previous conformity determinations have reflected policy changes that call for the construction of the South Corridor LRT Project in two phases. The first phase to include I-205 LRT from Gateway Regional Center to Clackamas Regional Center and LRT on the downtown Portland Transit Mall by 2008. A second phase is assumed that would include LRT from downtown Portland to Milwaukie town center. A new assumption is more rapid implementation of the Interstate MAX from downtown Portland to the Expo Center to the Expo Center. LRT service extension from Expo Center to Vancouver, Washington continues to be assumed to be part of the Preferred System, but is now not included in the Financially Constrained RTP.

**Commuter Rail.** A previous Determination has assessed introduction of commuter rail into the regional transit service strategy. The 2004 RTP makes no changes to the assumptions previously modeled. Only one alignment and service parameter is identified: Wilsonville to Beaverton in Washington County during the a.m. and p.m. peak periods with supporting park and ride facilities and a slight increase and realignment of supporting feeder bus service. If other alignments should be determined to be feasible, amendment of the regionally defined system would be needed.

**Bus Transit**. The 2004 RTP carries forward a hierarchy of regional bus transit service described in the 2000 RTP. From a modeling perspective, one of the most significant factors effecting transit ridership is transit service headways. The 2000 RTP identified four gradations of bus service: Rapid bus, Frequent bus, Regional bus and Community bus which are continued in the 2004 RTP. Rapid bus service would most closely emulate LRT in speed, frequency and comfort serving major transit routes with limited stops. Rapid bus service is characterized by some dedicated rights-of-way, signal preemption capability, 15-minute headways and high quality station and passenger amenities. Passenger amenities are concentrated at transit centers such as schedule information, ticket machines, bicycle parking and covered shelters. The 2004 RTP continues with an approach of deploying a limited number of Rapid bus lines in high demand commuter corridors.

Frequent bus service is characterized by 10-minute headways, wider geographic coverage, utilization of some dedicated right-of-way (e.g., queue jumps, dedicated turn lanes, etc.), signal preemption capabilities, and enhanced passenger amenities that include covered bus shelters, special lighting. Some overlap of Rapid and Frequent bus service is conceivable. However, bus stops (rather than stations) would characterize the frequent bus system and much more frequent stops would occur. The vehicles would be typical transit buses.

Page 10

Regional bus service would represent the majority of planned regional bus service. Radial trunk service would be provided on major arterials. Stops would be located every two to three blocks, and amenities would be prioritized to high ridership locations. Headways would not be more than 15-minutes during regular operating hours. The 2004 RTP continues the 2000 RTP approach which assumed expansion of the system to provide not only central city radial service but also to interconnect emerging regional and town centers, main streets and corridors with the central city and with one another.

The Community transit network is an innovation of the 2000 RTP that grew from Tri-Met's Transit Choices for Livability program. In addition to local bus service to neighborhoods and employment areas, community bus service includes decentralization of some transit services to a multitude of community-based transit providers dedicated to providing localized, "shuttle-like" service to destinations within a very limited geography. Vehicle types are expected to vary from traditional buses to van-type shuttles and taxi and carshare programs. The service is focused on more accessibility, frequency along the route and coverage to a wide range of land use options rather than on speed between two points. Community bus service generally is designed to serve travel with one trip end occurring within the 2040 Growth Concept town centers, main streets, station communities and corridors.

**Transit Ridership**. The broadest measure of ridership assumptions is revenue hours. The previous network, used to conform the 2000 RTP, as amended, reflected changes to the South/North alignment and timing. Also, it included introduction of Commuter Rail in Washington County.

The following data points highlight the practical effect of changed system configuration and funding assumed in the 2004 RTP relative to previous assumptions used in the 2000 RTP:

✤ Total projected revenue hours projected for the 2004 RTP is Results Pending

The 2004 RTP projects Average Weekday (AWD) transit trips in 2025 ... Results Pending

- The 2004 RTP projects that the percent of regional daily trips that are transit is ...Results Pending)
- The 2004 RTP projects that, the percent of households and employment within 1/4-mile of transit service in 2025 to be Results Pending
- AWD originating riders per revenue hour are Results Pending
- c. **Requirement:** The State Conformity Regulations require that reasonable assumptions be used regarding transit service, and increases in fares and road and bridge tolls over time.

Finding of compliance: There are no road or bridge tolls in place in the Portland

Page 11

metropolitan area, and none are assumed in the 2004 RTP or proposed in the MTIP. No decision to deploy such a project has been made and this Determination does not model evaluation of such a program. However, in the future some of the projects included in the Financially Constrained System Project List may include value pricing considered during individual project evaluation and alternative selection.

Auto operating costs are factored into the mode choice subroutines of the regional travel model. These costs are held constant to 1985 dollars. Parking costs for the Central City and for Tier 1 regional centers are based on the South/North DEIS parking costs developed from survey data to reflect parking control strategies. Parking factors for the remaining regional centers, station communities, town centers and mainstreets are scaled back by 50 percent from these costs. No parking factors are assumed for corridors, neighborhoods, employment areas, industrial areas, greenspaces and areas outside the urban growth boundary. The three-zone transit fare structure adopted in 1992 is held constant through 2025. User costs (for both automobile and transit) are assumed to keep pace with inflation and are calculated in 1985 dollars. Free transit areas are assumed for the central business and Lloyd districts and Tier 1 regional centers and within Wilsonville town center.

Service assumptions (i.e., transit vehicle headways) also affect trip assignment to transit. The South Corridor LRT Project Locally Preferred Alternative has selected the I-205 LRT segment and the downtown Portland Transit Mall LRT segment as a first phase recommended for completion by 2007 and a downtown Portland to Milwaukie LRT segment as a second phase.

LRT along Interstate Avenue from the Rose Quarter to the Expo Center is ahead of schedule with startup now planned for May 2004. These service assumptions were previously modeled in the FY 02-05 Metropolitan Transportation Improvement Program (MTIP) Conformity Determination, approved January 20, 2000 and as amended August 14, 2003.

The 2000 RTP assumed a 1.5 percent annual service hour increase for regional bus service through 2006. The bulk of the increase was allocated to building a service base along the Interstate Avenue corridor. At 2007, these bus resources were assumed to be reallocated throughout the region and feeder service within the LRT Corridor was reinforced. Service increases were assumed to Results Pending

The 2004 RTP continues these early program assumptions. However, with added regional support in the FY 2002 – 2005 MTIP, earlier attention has been focused on building service in two of four newly identified priority rapid bus corridors: the Barbur/99W and McLoughlin corridors, which link downtown with southeast Washington County and west Clackamas County, respectively. Rather than general reallocation of the Interstate LRT service hours, service in these corridors will be expanded. In addition, the 2004 RTP (as did the 2000 RTP) extends the 1.5 percent increase through 2025. Finally, rapid bus

Page 12

service is extended to the McLoughlin Boulevard/Highway 224 corridor and on Division Street to Gresham regional center in east Multhomah County.

d. **Requirement:** The State Conformity Regulations require that the latest existing information be used regarding the effectiveness of TCMs that have already been implemented. It must also be demonstrated that the Plan does not delay or impede the implementation of TCMs

The the Portland area maintenance plans for ozone and carbon monoxide include TCMS that are identical, except for section 2 of the non-funding based TCMs. Following are the TCM quoted verbatim (shown in italics) from the air quality maintenance plans and unless noted, are the same in each maintenance plan. The maintenance plan TCMs are followed by a description of actions taken by the region to comply:

#### "Non-funding based Transportation Control Measures

#### 1. Metro 2040 Growth Concept

Metro's 2040 Growth Concept is included because it changes typical growth patterns to be less reliant on motor vehicle travel, thereby reducing motor vehicle emissions. Two elements of the land use plan (the Interim Measures and the Urban Growth Boundary) provide appropriate implementation mechanisms to meet FCAA enforceability requirements for control strategies.

#### a. Metro Interim Land Use Measures relating to:

- Requirements for Accommodation of Growth;
- Regional Parking Policy; and
- Retail in Employment and Industrial Areas.

The text of the interim land-use measures is included in Appendix D1-17 (for Ozone, Appendix D2-10 for CO).

#### b Urban Growth Boundary.

The Urban Growth Boundary (UGB) as currently adopted or amended before EPA approval of the maintenance plan, assuming an amendment does not significantly affect the air quality plan's transportation emission projections.

#### 2. Central City Parking Requirements (Carbon Monoxide)

The Portland City Council adopted the <u>Central City Transportation Management Plan, Plan and</u> <u>Policy</u>, and other supporting documents on December 6, 1995. The Central City Transportation Management Plan (CCTMP) was adopted by Ordinance No. 169535, Resolution 35472. The Ordinance became effective January 8, 1996. A key supporting document was the Zoning Code Amendments, containing the maximum parking ratios for new development, the requirements for providing structured parking to serve older historic buildings and other regulations on parking.

Page 13

Key elements of the Zoning Code Amendments related to CO air quality projections are incorporated into this document as given below.

The CCTMP replaced the former Downtown Parking and Circulation Policy, first adopted in 1975 and updated in 1980 and 1985. The 1980 update of the parking policy served as a foundation for the 1982 Portland area CO attainment plan. The CCTMP is designed to minimize new vehicle traffic in the Central City and encourage alternative travel modes by extending the downtown maximum parking ratio concept to the entire Central City area. The CCTMP provided for the lifting of the downtown parking lid upon EPA approval of the maintenance plan and the request" for attainment redesignation. However, until EPA approval, the CCTMP retains the parking lid.

The parking offset program (OAR 340-020-0400 through OAR 340-020-0430), designed to allow the city to increase the parking lid by up to a maximum of 1,370 spaces, was also retained until after EPA approval of the maintenance plan. The DEQ's emission projection figures for the CCTMP emissions inventory area include an estimate for the emissions associated with 827 parking spaces, as documented in Appendix D2-4-4. These are the parking spaces yet to be developed, but which were authorized by the parking offset program.

The following is a list of zoning code amendments that were incorporated directly into the Portland Carbon Monoxide Maintenance Plan. The text of critical code provisions (such as maximum parking ratios for new development and parking provisions for existing buildings) is contained in Appendix D2-8. A list of other zoning code amendments used as supporting documents for the maintenance plan is contained in Appendix D2-13 of Volume 3 of the Oregon State Implementation Plan.

Items in Volume 3 of the SIP are federally enforceable. With regard to Volume 3 items, EPA has allowed DEQ to make changes which are merely administrative, without requiring public process. DEQ and EPA make a determination as to whether a proposed change by the City of Portland is merely administrative rather than substantive.

Section 1: Incorporated Amendments to Chapter 33.510, Central City Plan District

<u>Code Number</u> 33.510.261 -33.510.261.E

(33.510.261.E.1.a(1)-(2),b,E.2.a(1)-(2),b)

33,510.263 -33.510.263.A (33.510.263.A.1.a-c(1)-(4),A.2-4.a-b(1)-(3),A.5-7.a-d)

33.510.263.B -(33.510.263.B.1.a-c(1)-(2),B.2-4.a)

33.510.263.E -(33.510.263.E.1.a-b,E.3.a-c) <u>Code Title</u> Parking Site split by subdistrict or parking sector boundaries

Parking in the Core Area Growth Parking

**Preservation Parking** 

Residential/Hotel Parking

All Parking

Page 14

2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program draft Air Quality Conformity Determination October 31, 2003

33.510.263.G -

33.510.263.G.4 -(33.510.263.G.4.a. (1)-(2), G.4.d(1)-(3»)

33.510.264

33.510.264.A (33.510.264.A.1.a-c(1)-(4),A.2.a,A.4.a)

33.510.264.B 33.510.264.B.1.a-c(I)-(2),B.2.a-c,B.4.a-c)

33.510.264.F

33.510.264.F.4 (33.510.264.F.4.e.(1)-(3)

33.510.265

33.510.265.A (33.510.265.A.1.a-c,A.2.a,A.4.a)

33.510.265.B (33.510.265.B.1.a-c(1)-(4),B.2.a,b) (33.510.265.B.4.a-c) Surface parking lots.

Parking in Lloyd District

Growth Parking

Preservation Parking

All Parking

Surface parking lots

Parking in the Goose Hollow Subdistrict and Central Eastside Sectors 2 and 3

Growth Parking

<u>Code Title</u>

Review Status

Preservation Parking

Loss of Central City Parking

General Approval Criteria for

Central City Parking Review

If the site is in the Core Area:

Section 2: Incorporated Portion of New Chapter 33.808, Central City Parking Review

<u>Code Number</u>

33.808.050

33.808.100

33.808.100.G

33.808.100.J 33.808.100.J.2.a

33.808.100.M

Section 3: Incorporated Maps

<u>Map Number</u> 510-8 <u>Map Title</u> Core and Parking Sectors - EPA

Page 15

Section 4:

Incorporated Portion of CCTMP Administration Section

VI.D.1.a.(1)-(5)

Administration Section: Preservation Parking

Unless it is a substitution of a Transportation Control Measure producing equivalent emission reduction, any change in the Portland Metro Area CO Maintenance Plan language will require adoption of a formal amendment by the EQC and approval by EP A. The City of Portland may make changes to City policies and regulations which are included in the Portland Metro Area CO Maintenance Plan provided they do not relax the stringency of the air quality control strategies. DEQ will work with the City to notify EPA of such changes. These changes will be incorporated into the Portland Metro Area CO Maintenance Plan detro Area CO Maintenance Plan ta a future convenient time.

Changes to documents supporting the Portland Metro Area CO Maintenance Plan' (zoning code amendments not directly incorporated into the Portland Metro Area CO Maintenance Plan, but listed in Appendix D2-13 of Volume 3 of the Oregon State Implementation Plan) which do not affect the stringency of the air quality control strategies will not require adoption of a formal amendment by the EQC and approval by EP A. DEQ and the City of Portland will review potential changes to the supporting documents to determine whether they affect the stringency of the air quality strategies. If it is determined that stringency will not be affected, DEQ will submit those changes to EPA for concurrence and administrative incorporation into the Portland Metro Area CO Maintenance Plan.

#### 2. DEQ Employee Commute Options Program (ozone)

A 10% trip reduction target is required for employers who employ more than 50 employees at the same work site. See discussion above and Appendix D1-13.

#### 3. DEQ Voluntary Parking Ratio Program (ozone)

Implement a voluntary parking ratio program providing incentives to solicit participation, including exemption from the Employee Commute Options program. See discussion above and Appendix D1-14.

Page 16

#### Funding based Transportation Control Measures

- 1. Increased Transit Service
  - a. Regional increase in transit service hours averaging 1.5% annually.

This commitment includes an average annual capacity increase in the Central City area equal to the regional capacity increase. The level of transit capacity increase is based on the regional employment growth projections adopted by Metro Council on Dec. 21, 1995. These projections assume that the Central City will maintain its current share of the regional employment. Should less employment growth occur in the Region and/or the Central City, transit service increase may be reduced proportionately.

- b. Completion of the Westside Light Rail Transit facility.
- c. Completion of Light Rail Transit (LRT) in the South/North corridor by the year 2007.
- 2. Bicycle and Pedestrian Facilities
  - a. Multimodal facilities.

Consistent with ORS 366.514<sup>2</sup>, all major roadway expansion or reconstruction projects on an arterial or major collector shall include pedestrian and bicycle improvements where such facilities do not currently exist. Pedestrian improvements are defined as sidewalks on both sides of the street. Bicycle improvements are defined as bikeways within the Metro boundary and shoulders outside the Metro boundary but within the Air Quality Maintenance Area.

b. RTP Constrained Bicycle System.

In addition to the multimodal facilities commitment, the region will add at least a total of 28 miles of bicycle lanes, shoulder bikeways or multi-use trails to the Regional Bicycle System as defined in the Financially Constrained Network of Metro's Interim Federal RTP (adopted July 1995) by the year 2006. Reasonable progress toward implementation means a minimum of five miles of new bike lanes, shoulder bikeways or multi-use trails shall be funded in each two-year Transportation Improvement Program (TIP) funding cycle.

Bike lanes are striped lanes dedicated for bicycle travel on curbed streets, a width of five to six feet is preferred; four feet is acceptable in rare circumstances. Use by autos is prohibited. Shoulder bikeways are five to six foot shoulders for bicycle travel and

<sup>2</sup> This provides for the following exceptions:

- absence of any need;
- contrary to public safety; and
- excessively disproportionate cost.

Page 17

emergency parking. Multi-use trails are eight to 12 foot paths separate from the roadway and open to non-motorized users.

#### c. Pedestrian facilities.

In addition to the multimodal facilities commitment, the region will add at least a total of nine miles of major pedestrian upgrades in the following areas, as defined by Metro's Region 2040 Growth Concept: Central City/Regional Centers, Town Centers, Corridors & Station Communities, and Main Streets. Reasonable progress toward implementation means a minimum of one and a half miles of major pedestrian upgrades in these areas shall be funded in each two-year TIP funding cycle."

**Finding of compliance:** All non funding and funding based TCMs are fully supported by local, regional and State actions as well as the 2004 RTP and MTIP. This includes:

#### Metro 2040 Growth Concept

Since its adoption in 1995, the Metro Growth Concept has continued to serve as a means of coordinating land use and transportation, emphasizing a compact urban form, mixed uses where high quality transit service is provided or planned, a balanced transportation system that serves the Growth Concept and providing for transportation choices. Both the Metro 2000 RTP and 2004 RTP use the transportation system to implement the 2040 Growth Concept. This includes using a 2040 land use hierarchy to guide transportation plans and MTIP criteria that direct transportation investment decisions with 2040 Growth Concept implementation in mind. The MTIP includes incentives for serving 2040 centers (mixed use areas) and reducing vehicle miles traveled. As a result, during the period 1990 to 2000, while total vehicle miles increased by 35 percent, TriMet ridership increased 49 percent. Further, from the local adoption of the air quality maintenance plan requirements (1996) to the year 2000 (the latest data available), vehicle miles per capita decreased from 21.7 vehicle miles traveled per capita (vmt/c) to 20 vmt/c - an eight percent decrease.

#### **Metro Interim Land Use Measures**

In 1996, the Metro Council adopted the Urban Growth Management Functional Plan, which was a set of recommendations and requirements for the twenty-four cities and the urban portions of three counties for implementing the 2040 Growth Concept. These regulations are not interim measures, rather, they provide lasting measures to address land use/transportation coordination. The Functional Plan set targets for cities and counties within the region for new jobs and housing as a means of encouraging land use patterns that are supportive of transit, walking and biking as well as setting standards for street connectivity and reducing the amount of land devoted to surface parking. As of January 2003, the Metro Council concluded (See appendix 8, which includes Metro Resolution No. 03-3299, compliance tables and the Functional Plan recommendations and requirements) that 25 of the 27 jurisdictions complied with the minimum density standards, all jurisdictions complied with land partitioning standards, all but one

Page 18

complied with accessory dwelling unit standards. The total residential capacity demonstrated by the local jurisdictions was 94 percent of the total envisioned by the targets, without counting the capacity of the City of Wilsonville or unincorporated Multnomah County. The regional total for accommodating jobs was 107percent of the regional targets.

With regard to parking, all but one jurisdiction, as of January 2003, had complied with reviewing parking space sizes and ratios and lowering the total amount of land devoted to surface parking.

Finally, for Title 4, Retail in Employment and Industrial Areas, every city or county with employment or industrially zoned lands complied. In addition, Metro is currently looking at further protection of encroachment on employment and industrial lands with additonal regulations now being discussed by the Metro Council.

In addition, Metro adopted a Title 6, which pertained to transportation accessibility and connectively. While not included as a land use measure in the air quality maintenance plans, these regional requirements for local government implementation encouraged street systems that connected more frequently which, in turn, encourages walking, biking and transit use - all contributing to better air quality. All 27 jurisdictions complied with connectivity standards.

#### **Urban Growth Boundary**

As noted above, the 2040 Growth Concept was envisioned to encourage a more compact urban form and to provide for land use patterns that encourage transportation choice. The urban growth boundary was not intended to be static. Since the late 1970s, the boundary has been moved about three dozen times. Most of those moves were small - 20 acres or less. There were two times that Metro authorized more substantial additions:

- in 1998 about 3,500 acres were added to make room for approximately 23,000 housing units and 14,000 jobs. Acreage included areas around the Dammasch state hospital site near Wilsonville, the Pleasant Valley area in east Multnomah, the Sunnyside Road area in Clackamas County, and a parcel of land south of Tualatin.
- in 1999 another 380 acres were added based on the concept of "subregional need." An example of "subregional need" would occur when a community needed land to balance the number of homes with the number of jobs available in that area.
  These expansions represented an increase of only about 2 percent, even though the Metro region's population has increased by about 17 percent since 1990.

In early 2002, the voters of the region approved ballot measure 26-29, which prohibits Metro from requiring higher densities within existing neighborhoods. Metro's goal is to locate higher density housing, such as townhouses and apartments, within "centers" such as the downtowns of Portland, Beaverton and Gresham, or along transportation corridors, particularly where there is a light-rail line.

Page 19

Further, in 2002, the Metro Council completed a two-year process reviewing the region's capacity for housing and jobs by expanding the UGB by an additional 18,638 acres, with 2,851 acres dedicated to employment purposes.

As part of the 2002 UGB decision, the Metro Council adopted new policies that address the protection of existing neighborhoods and additional job land, and the improvement of downtown commercial centers and main streets. Accordingly, transportation and air quality modeling have assumed urban land use consistent with population, housing and job forecasts. In turn, transportation system improvements have also been assumed to serve the area. The air quality conformity determination, once modeling has been completed, will demonstrate the estimated future air quality results.

#### **Central City Parking Requirements**

Central City Parking Requirements were enacted as cited in the *Portland Area Carbon Monoxide Maintenance Plan* as a means of addressing concerns about concentrations of this pollutant in the Portland downtown area. A monitoring station located at 4th and Alder Streets in downtown Portland has provided actual measurements of carbon monoxide. The 1-hour and 8 hours averages for the years 1996 through 2001 expressed in parts per million (ppm) are as follows:

#### Table 1

Central City (4th and Alder) Carbon Monoxide Measurements

	1 Hour	1 Hour	8 Hour
Year	Oct-April Average	Maximum	Maximum
1996	1.36	8.6	6.4
1997	1.37	7.8	4.8
1998	1.13	8.4	4.6
1999	1.23	11.6	7.5
2000	1.14	9.3	5.4
2001	1.04	6.3	3.6

The 1 hour standard is 35 ppm and the 8 hour standard is 9 ppm. Because the actual carbon monoxide concentrations were so far below the standards, in 2002, the Oregon Department of Environmental Quality removed the air quality monitoring station.

Accordingly, it is concluded that carbon monoxide pollution in the Central City is no longer a significant problem, in part because of the array of transportation control measures that have been implemented.

#### **DEQ Employee Commute Options Program**

The ECO rule (OAR 340-242-0100 through 0290), applies to employers in the Portland area with *more than 50 employees* reporting to a single work site. Affected employers must provide incentives for employee use of alternative commute options. The incentives must have the potential to reduce commute trips to the work site by ten percent within three years. Annual employee surveys measure progress toward this goal.

Page 20

Popular programs include transit subsidies, carpool matching and preferential parking for carpools, compressed work weeks (4/10's for example), telecommuting, and bike/walk programs. Most companies offer a guaranteed ride home for personal emergencies for commuters.

Failure to comply with the ECO rule is a Class II environmental violation and carries penalties that typically range from \$500 - \$2,000 for each day of violation.

Ongoing ECO rule implementation is the basis for concluding that this TCM has been fully implemented.

#### **DEQ Voluntary Parking Ratio Program**

The Metro Functional Plan adopted in 1996, provide a more rigorous parking ratio approach. See Metro Interim Land Use Measures, above. Accordingly, in 1999, the DEQ eliminated this program.

Because of the Metro Functional Plan requirements, this TCM has been fully implemented.

#### Transit Service

Table 2 below displays the total region-wide annual service hours for light rail and bus vehicles by year since the adoption of the region's transportation control measures (1996).

#### Table 2

#### Region-wide Annual Transit Service Hours

	Service Hours			Percent Change		
Fiscal Year	Rail	Bus	Total	cumulative from 1996	year-to-year	
1996	59,544	1,821,120	1,880,664	0.0%		
1997	59,748	1,819,320	1,879,068	-0.1%	-0.0%	
1998	66,708	1,869,324	1,936,032	2.9%	3.0%	
1999	130,236	1,938,048	2,068,284	9.9%	6.8%	
2000	143,100	2,009,148	2,152,248	14.4%	4.0%	
· 2001	144,672	2,032,944	2,177,616	<u> </u>	1.1%	
2002	183,648	2,048,484	2,232,132	18.6%	2.5%	
2003	192,500	2,049,100	2,241,600	19.1%	0.4%	
Average					2.6%	

Page 21

TriMet has actually increased transit service by an average of 2.6 percent since adoption of this transportation control measure. This is more than 1 percent greater than the 1.5 percent average transit service increase required annually. Furthermore, a large percentage of the increase in vehicle service hours have been provided on light rail vehicles which have three to six times the passenger carrying capacity of a bus, depending on whether a one or two car train is operating.

This level of transit service increase was made possible by large increases in payroll tax revenues within the TriMet district due to a favorable economic climate. It is unlikely TriMet will be able to sustain this level of growth over a long period of time. Service and financial planners at TriMet have forecast modest growth in service hours through the MTIP years, however, that will easily exceed the commitment to averaging 1.5 percent annual growth. Recently acquired authority from the 2003 State Legislature to increase the payroll tax rate once the recession has ended will further enable TriMet to meet this goal.

The corresponding change in transit service in the Portland Central City also showed that the annual capacity increase in the Central City increased by an average annual rate of 3.9 percent for seated capacity and by 5.7 percent for total capacity during the years 1996 and 2003, each well above the TCM mandate of 1.5 percent average annual increase. This is illustrated in Table 3, below.

#### Table 3

**Transit Service in the Portland Central City** 

Mode	Seated Capacity			Total Capacity (seated and standing)		
	Fall 1996	Fall 2003	Annual Average % Increase	Fall 1996	Fall 2003	Annual Average % Increase
Bus	1,172,354	1,214,256		1,830,016	1,895,494	
Rail	163,328	486,524		423,632	1,261,922	
Total	1,335,682	1,700,780	3.9%	2,253,648	3,157,346	5.7%

#### Pedestrian System TCMs

New pedestrian projects awarded funding in the most recent Transportation Priorities process focused on improving the safety of pedestrian crossings at intersections. This includes the Central Eastside bridge heads project (which also includes access from Water Avenue to the Morrison Bridge) and the St. John's town center pedestrian improvements. The length of the improvements across intersections and the new Morrison Bridge access are approximately .4 miles in length. The Forest Grove town center pedestrian improvement project will be providing approximately 1.2 miles of new sidewalks in the 2006-07 biennium. A data base and

Page 22

map to illustrate these improvements is not currently available. However, Metro should complete such a database and map for future conformity determinations.

#### Bicycle System TCMs

A data base of constructed bike lanes and related facilities should be completed for future conformity determinations. As a surrogate, a map comparing the bike system in 1999 and 2002 was prepared from the Metro *Bike There!* maps. The below map shows the 103 miles of new bike lanes and multi-purpose paths added during the period 1999 to 2002. That is, from a 1999 total of 519 miles, 103 miles of bikeway were added for a 2002 total of 622 miles. Of the current 622 miles of bikeways, 512 are bike lanes, defined as "striped portions of the roadway designated as a bicycle travel lane". The balance, 110 miles are regional multi-use paths defined as "physically separated from motor vehicle traffic, used by bicyclists, pedestrians, skaters and other non-motorized travelers." Further review is in order and if the analysis is confirmed, the region will have achieved this TCM adopted in 1996 that "... the region will add at least a total of 28 miles of bicycle lanes, shoulder bikeways or multi-use paths to the Regional Bicycle System as defined in the Financially Constrained Network of Metro Interim Federal RTP (adopted July 1995) by the year 2006."

In addition to bike lanes constructed as part of associated road improvements, this Transportation Priorities process allocated funding for approximately 3.8 miles of new offstreet multi-use paths for bicycle and pedestrian use in the 2006-07 biennium. Funding for the design of an additional 4.5 miles of multi-use path was also provided as a part of these projects.

#### Page 23



Bease registe withmixed paper

**Other TCMs.** Effectiveness of implemented and planned TCMs is also reflected in emission credits approved by DEQ for use in this Determination's calculation of daily regional emissions. Credits were assumed for compact land form called for in the Region 2040 Growth Concept, expansion of the I/M Boundary; implementation of enhanced I/M; and implementation of the Employee Commute Option (ECO) program. Credit for the region's Voluntary Parking Ratio program was eliminated in 1999 because very few businesses chose to participate in the program. All of these programs are founded in enforceable regulations.

In addition, the 2004 MTIP includes \$125,000, which in conjunction with State of Washington contributions, would explore TDM/TSM policies for the I-5 Corridor. Metro has also initiated a Strategic Plan for TDM in the Metro area as a means of establishing a comprehensive approach throughout the Metro region.

#### 2. Latest Emissions Model (OAR 340-252-0120)

a. **Requirement:** The State Conformity Regulations require that the conformity determination must be based on the most current emission estimation model available.

**Finding of compliance:** Metro employed EPA's recommended Mobile 5a-h emissions estimation model in preparation of this conformity determination. Additionally, Metro uses EPA's recommended EMME/2 transportation planning software to estimate vehicle flows of individual roadway segments. These model elements are fully consistent with the methodologies specified in OAR 340-252-0120.

In addition, Metro has begun running the MOBILE6 model in order to begin familiarization with this new model in anticipation of its use in future conformity determinations.

#### 3. Consultation (OAR 340-252-0130)

a. **Requirement:** The State Conformity Regulations require the MPO to consult with the state air quality agency, local transportation agencies, DOT and EPA regarding enumerated items. TPAC is specifically identified as the standing consultative body in OAR 340-225-0060(1)(b).

**Finding of compliance:** Specific topics are identified in the Regulations that require consultation. TPAC is identified as the Standing Committee for Interagency Consultation. Most of the agencies defined as eligible to participate during interagency consultation for the Determination were participants in development of the 2004 RTP and the MTIP, (EPA and the Federal Transit Administration, whose closest offices are located in Seattle have not been able to participate at TPAC) including development of the financially constrained system, at both the region's technical and policy committee levels (TPAC and JPACT) during the development of the 2004 RTP. However, a special interagency meeting was

Page 26

convened with all eligible participants in order to review an early draft of this document and discuss the conformity determination approach, schedule and assumptions (see Appendix 9)

Further, an independent analysis of the air quality conformity process throughout the nation (*Exhausting Options: Assessing SIP-Conformity Interations*, Resources for the Future, 2003) was completed and which included six case studies, including the Portland area. On page 88 regarding the Portland area, the Report states:

"DEQ has been aggressive in its role in conformity since the rule was first released. For example, it was DEQ that pushed through an interagency consultation agreement. DEQ also devised out-year motor vehicle emission budgets. To avoid the planning horizon mismatch, the MVEBs were allowed to increase in the out-years to allow for growth in vehicle emissions. DEQ has played a very active role in transportation planning in general and conformity in particular. Its staff has a good understanding of the analytical elements of the conformity process and especially how modeling assumptions can affect conformity determinations."

#### It further states:

i.

"...the air quality authority participates fully in transportation planning, and the interagency consultation process works well."

Determination of which Minor Arterial and other transportation projects should be deemed "regionally significant."

Metro models virtually all proposed enhancements of the regional transportation network proposed in the MTIP, the 2004 RTP and by local and state transportation agencies. This level of detail far exceeds the minimum criteria specified in both the State Rule and the Metropolitan Planning Regulations for determination of a regionally significant facility. This detail is provided to ensure the greatest possible accuracy of the region's transportation system predictive capability. The model captures improvements to all principal, major and minor arterial and most major collectors. Left turn pocket and continuous protection projects are also represented. Professional judgment is used to identify and exclude from the model those proposed intersection and signal modifications, and other miscellaneous proposed system modifications, (including bicycle system improvements) whose effects cannot be meaningfully represented in the model. The results of this consultation were used to construct the analysis year networks identified in Appendix 1 of this Determination.

*ii.* Determine which projects have undergone significant changes in design concept and scope since the regional emissions analysis was performed.

Page 27

All agencies defined as eligible to participate during interagency consultation for the Determination were participants in development of the 2004 RTP and 2004-07 MTIP and commented extensively on the Plan's preparation, including development of the 2004 RTP financially constrained system, at both the region's technical and policy committee levels (TPAC and JPACT).

iii. Analysis of projects otherwise exempt from regional analysis.

All projects capable of being modeled have been included in the Conformity Analysis quantitative networks, regardless of funding source or "degree of significance".

*iv.* Advancement of TCMs.

All past and present TCMs have been implemented on schedule. There exist no obstacles to implementation to overcome. See 1(d) in this section., above.

v. PM10 Issues.

The region is in attainment status for PM10 pollutants.

vi. forecasting vehicle miles traveled and any amendments thereto.

The forecast of vehicle miles is the product of the modeled road and transit network defined in the financially constrained system, which was approved during extensive consultation with all concerned agencies including DEQ as part of TPAC and JPACT.

vii. determining whether projects not strictly "included" in the TIP have been included in the regional emission analysis and that their design concept and scope remain unchanged.

All projects capable of being modeled have been included in the Conformity Analysis quantitative networks, regardless of funding source or "degree of significance".

viii. project sponsor satisfaction of CO and PM10 "hot-spot" analyses.

The MPO defers to ODOT staff expertise regarding project-level compliance with localized CO conformity requirements and potential mitigation measures which are considered on a project-by-project basis as a part of the environmental assessment. There exist no known  $PM_{10}$  hot spot locations of concern.

ix. evaluation of events that will trigger new conformity determinations other than those specifically enumerated in the rule.

This section is not applicable to the 2004 RTP or MTIP conformity determination.

Page 28

evaluation of emissions analysis for transportation activities which cross borders of MPOs or nonattainment or maintenance areas or basins.

The Portland-Vancouver Interstate Maintenance Area (ozone) boundaries are geographically isolated from all other MPO and nonattainment and maintenance areas and basins. Emissions assumed to originate within the Portland-area (versus the Washington State) component of the Maintenance Area are independently calculated by Metro. The Clark County Regional Transportation Commission (RTC) is the designated MPO for the Washington State portion of the Maintenance area. Metro and RTC coordinate in development of the population, employment and VMT assumptions prepared by Metro for the entire Maintenance Area. RTC then performs an independent Conformity Determination for projects originating in the Washington State portion of the Maintenance Area.

Conformity of projects occurring outside the Metro boundary but within the Portland-area portion of the Interstate Maintenance Area were assessed by Metro as provided in State regulations. A request was made of each county to forward projects within the Maintenance Area boundary. While several projects were forwarded to Metro from Multnomah County for analysis, none of these projects was considered a regionally significant project. (see Appendix 12) No regionally significant projects outside the urban boundary have been declared to Metro for analysis.

xi. disclosure to the MPO of regionally significant projects, or changes to design scope and concept of such projects that are not FHWA/FTA projects.

In the process of updating the 2000 RTP and the 2004 RTP, local jurisdictions and regional and state agencies made changes to the projects. These changes will be reflected in the air quality modeling and considered in the conformity determination.

xii. the design schedule and funding of research and data collection efforts and regional transportation model development by the MPO.

This consultation occurs in the course of MPO development and adoption of the annual Unified Planning Work Program.

#### xiii. development of the TIP.

Х.

Development of the MTIP included review by TPAC, which is the designated body for interagency consultation.

#### xiv. development of RTPs.

Development of the 2004 RTP was directly reviewed by TPAC, which is the standing body for interagency consultation.

Page 29

xv. establishing appropriate public participation opportunities for project level conformity determinations.

In line with other project-level aspects of conformity determinations, it is most appropriate that project management staff of the state and local operating agencies be responsible for any public involvement activities that may be deemed necessary in making project-level conformity determinations.

b. **Requirement:** The State Conformity Regulations require a proactive public involvement process that provides opportunity for public review and comment by providing reasonable public access to technical and policy information considered by the agency at the beginning of the public comment period and prior to taking formal action on the conformity determination for all transportation plans.

**Finding:** The 2004 RTP and 2004-07 MTIP had public outreach during November 2003, during a 30-day comment period. The 2004 RTP is, by and large, extending plans and approaches that were concluding during development of the 2000 RTP which was crafted during five years (1995-2000). Design of the 2000 RTP was also guided by input from a 21-member citizen advisory committee, local officials and staff from the region's cities and counties, residents, community groups and businesses throughout the region. Numerous opportunities for public comment were provided during the five-year process, which concluded with a 45-day public comment period prior to adoption by ordinance. Appendix 2 contains a timeline that describes key products and opportunities for public comment as part of the 2004 RTP. In addition, development of the MTIP included extensive public review and comment opportunities.

On September 29, 2003 a notice of Metro's intent to update the 2000 RTP and conduct an air quality conformity analysis of the 2004 RTP and 2004-07 MTIP was sent to affected governments and interested residents, businesses and community groups. This notice summarized the public process and a timeline for adoption of the 2004 RTP, the 2004-07 MTIP and a conformity determination for both plans. On October 31, 2003, a 30-day public comment period began on the draft 2004 RTP air quality conformity analysis procedures and methodologies. Metro's website and transportation hotline also supplied information on the plan update and conformity determination process, including opportunities for public comment. Appendix 2 contains copies of the 45-day kickoff notice and Oregonian notice. In addition, a post card was mailed to approximately 2,500 persons who had asked to be placed on either the RTP or MTIP interested persons mailing list. The post cards were also mailed to representatives of neighborhood organizations and community planning organizations. Finally, a email newsletter was also sent out to elected officials and representatives of local, regional and state officials. Table 4 describes the 2004 RTP and 2004-07 MTIP conformity process.

Page 30

Table 4 2004 Regional T	ransportation Plan /2004-07 MTIP Conformity Analysis Timeline
September 29, 2003	Notification of 2004 RTP and joint 2004 RTP/2004-07 MTIP air quality conformity process to affected governments, interested citizens, community groups
October 31, 2003	Begin 30-day public comment period on draft 2004 RTP and draft conformity determination document for the 2004 RTP and 2004-07 MTIP
December 4, 2003	Metro Council Public hearing on 2004 RTP, 2004-07 MTIP and draft conformity determination; close of public comment period
December 5, 2003	Review of 2004 RTP and air quality conformity analysis results and tentative action by TPAC
December 11, 2003	Tentative action on 2004 RTP and joint 2004 RTP/2004-07 MTIP air quality conformity findings by JPACT and Metro Council

#### 4. Timely Implementation of TCMs (OAR 340-252-0140).

a. Requirement: The State Conformity Regulations require MPO assurance that "the transportation plan, [and] TIP... must provide for the timely implementation of TCMs from the applicable implementation plan."

Finding: See C.1(d), above.

#### 5. Support Achievement of NAAQS

a. **Requirement:** The State Implementation Plan (SIP) requires the 2004 RTP and 2004-07 MTIP to support achievement of NAAQS.

**Finding:** The 2004 RTP and 2004-07 MTIP were prepared by Metro. SIP provisions are integrated into the RTP and MTIP as described below, and by extension into subsequent TIPs, which implement the 2004 RTP. In addition, the 2004-07 MTIP is consistent with the 2004 RTP, and accordingly, both the 2004 RTP and MTIP are consistent with this requirement.

The scope of the 2004 RTP requires that it possess a guiding vision which recognizes the inter-relationship among (a) encouraging and facilitating economic growth through improved accessibility to services and markets; (b) ensuring that the allocation of increasingly limited fiscal resources is driven by both land use and transportation benefits; and (c) protecting the region's natural environment in all aspects of transportation planning process.

Chapter 1 of the 2004 RTP describes this guiding vision:

Page 31

- balance transportation and land use plans to protect livability in the region
- reduce reliance on any single mode of travel by expanding transportation choices
- · sustain economic health by providing access to jobs and industry
- target transportation investments to leverage the 2040 Growth Concept
- maintain access to the natural areas around the region
- protecting the region's natural environment in all aspects of transportation planning process

In addition, several policies and objectives in Section 1.3.4 of the 2004 RTP directly support achievement of National Ambient Air Quality Standards (NAAQS). These objectives are achieved through a variety of measures affecting transportation system design and operation, also described in Chapter 1 of the 2004 RTP. The plan sets forth goals and objectives for road, transit, freight, bicycle, and pedestrian improvements as well as for implementation of system and demand management strategies.

The highway system is functionally classified to ensure a consistent, integrated, regional highway system of principal routes, arterial and collectors. Acceptable level-of-service standards are set for maintaining an efficient flow of traffic. The RTP also identifies regional bicycle and pedestrian systems for accommodation and encouragement of non-vehicular travel. System performance is emphasized in the RTP and priority is established for implementation of transportation system management (TSM) measures.

The transit system is similarly designed in a hierarchical form of regional transitways, radial trunk routes and feeder bus lines. Standards for service accessibility and system performance are set. Park-and-ride lots are emphasized to increase transit use in suburban areas. The RTP also sets forth an aggressive demand management program to reduce the number of automobile and person trips being made during peak travel periods and to help achieve the region's goals of reducing air pollution and conserving energy.

In conclusion, 2004 RTP and the 2004-07 MTIP is in conformance with the SIP in its support for achieving the NAAQS. Moreover, the RTP provides adequate statements of guiding policies and goals with which to determine whether projects not specifically included in the RTP at this time may be found consistent with the RTP in the future. Section 1.3.7 in Chapter 1 of the 2004 RTP identifies key policies that guide the selection of projects and programs to implement the RTP. Conformity of such projects with the SIP would require interagency consultation.

#### 6. Quantitative Analysis (OAR 340-252-0190)

#### 1. Conduct a Quantitative Analysis

**Requirement:** OAR 340-252-0190 requires that a quantitative analysis be conducted as part of the 2004 RTP conformity determination. The analysis must demonstrate that emissions resulting from the entire transportation system, including all regionally significant projects expected within the time frame of the plan, must fall within budgets established in the

Page 32

maintenance plan for criteria pollutants. In the Portland-Vancouver Air Quality Maintenance Area these include ozone precursors (HC and NOx) and carbon monoxide (CO). A specified methodology must be used to calculate travel demand, distribution and consequent emissions as required by OAR 340-20-1010. The Portland metropolitan area has the capability to perform such a quantitative analysis.

**Finding:** For the Oregon portion of the Portland-Vancouver airshed, emission budgets have been set for various sources of pollutants (mobile, point, area) and are included in the SIP and in the region's Ozone and Carbon Monoxide Maintenance Plans. The 2004 RTP and 2004-07 MTIP must conform to the SIP mandated mobile emission budgets. Mobile emission budgets are set for winter carbon monoxide (CO) and for two summer ozone precursors: nitrogen oxides (NOx), and hydrocarbons (HC).

The region's approved Maintenance Plans identify two sets of analysis years, one set for winter CO and one set for summer ozone precursors (NOx and HC). The CO budget years are 2007, 2010, 2015 and 2020. The ozone analysis years are 2006, 2010, 2015 and 2020. In addition, a plan horizon year must also be evaluated. For the 2004 RTP, the horizon year is 2025. Table 5 shows the budget years and associated emissions budgets. The 2004-07 MTIP is a subset of the financially constrained system described in the 2004 RTP.

	2004 RTP/2004-07 MT	IP Mobile Emiss	sions Budgets <sup>1</sup>
	Winter CO (thousand pounds/day)	Summer HC (tons/day)	Summer NOx (tons/day)
2006	n/a	41	51
2007	775	n/a	n/a
2010	760	40	52
2015	788	40	55
2020	842	40	59
2025	842	40	59

<sup>1</sup> Budgets are from the Maintenance Plan adopted in 1996 except as noted. Year 2025 budget based on Ozone Maintenance Plan emission budget "for years 2020 and beyond".

Source: Metro

The network that was analyzed is summarized in Appendix 1. The protocol for definition of the Determination's analysis and budget years is summarized in Appendix 3, including discussion of why each analysis year was selected. Appendix 4 contains a summary of the principle model assumptions, including a discussion of assumed transit costs, parking factors, and intersection density and the impact of these factors on travel mode selection by 2040 design type (e.g., central city, regional centers, town centers, station communities, mainstreets, employment areas, corridors, etc.). A detailed description of the network assumptions coded into Metro's regional model is contained in a 2004 RTP Financially Constrained System Atlas, available for review at Metro located at 600 NE Grand Avenue, Portland, OR 97232. The Atlas

Page 33

includes information about system and individual link capacities in the 2000 base year and capacities assumed after planned improvements as well as the year of expected operation of each planned improvement. The results of the quantitative analysis are shown in Table 3 and Figures 1, 2 and 3. In summary, Metro's analysis indicates that, with regard to the established budgets in all budget years (i.e., 2006, 2007, 2010, 2015, 2020 and 2025), that regional emissions will. Results Pending.

#### 2. Determine Analysis Years.

a. **Requirement:** The State Conformity Regulations) require the first analysis year to be no later than 10 years from the base year used to validate the transportation demand planning model (340-252-0070), that subsequent analysis years be no greater than 10 years apart and that the last year of the 2004 RTP must be an analysis year (340-252-0070).

**Finding:** See Appendix 3 regarding selection of analysis and budget years, including discussion of why each analysis year was selected.

#### 3. **Perform the Emissions Impact Analysis.**

a. **Requirement:** The State Conformity Regulations) require Metro to conduct the emissions impact analysis.

**Finding:** Calculations were prepared, pursuant to the methods specified at OAR 340-20-1010, of CO and Ozone precursor pollutant emissions assuming travel in each analysis year on networks that have been previously described. A technical summary of the regional travel demand model, the EMME/2 planning software and the Mobile 5a-h methodologies is available from Metro upon request. The methodologies were reviewed by TPAC.

#### 4. Determine Conformity.

a. **Requirement:** Emissions in each analysis year must be consistent with (i.e., must not exceed) the budgets established in the maintenance plan for the appropriate criteria pollutants (OAR 340-252-0190).

**Finding:** Metro's analysis indicates that regional emissions will remain within established budgets in all budget years (i.e., 2006, 2007, 2010, 2015, 2020 and 2025). Table 6 provides a summary of these emissions and shows that the 2004 RTP and 2004-07 MTIP, conform with the SIP.

Page 34

	2004 RTP/2004-07 MTIP Conformity Results					
	Wir	nter CO	Su	Immer HC	Sur	nmer NOx
	(thousand pounds/day)		(tons/day)		(tons/day)	
	Budget	Model Result	Budget	Model Result	Budget	Model Result
2006	n/a	Results Pending	41	Results Pending	51	Results Pending
2007	775	Results Pending	n/a	Results Pending	n/a	Results Pending
2010	760	Results Pending	40	Results Pending	52	Results Pending
2015	788	Results Pending	40	Results Pending	55	Results Pending
2020	842	Results Pending	40	Results Pending	59	Results Pending
2025	842	Results Pending	40	Results Pending	59	Results Pending

<sup>1</sup> Budgets are from the Maintenance Plan adopted in 1996. Year 2025 budget should be adjusted based on emission budget input factors.

#### Source: Metro

Table 6

Figures 1, 2 and 3 show graphs of the conformity results that compare the emissions budgets with the modeled results for each analysis year for winter carbon monoxide (CO) and for two summer ozone precursors: nitrogen oxides (NOx), and hydrocarbons (HC) respectively. Figures 4 and 5 show graphs of the conformity results that compare the emissions budgets with the modeled results for each analysis year for winter carbon monoxide (CO) in the Portland central city subarea and 82nd Avenue subarea.

Page 35



Based on RTP Financially Constrained System and 2004-07 MTIP Source: Metro

Figure 2 Add forecast emissions including 2025 numbers



#### Summer HC Emissions Air Quality Maintenance Area Boundary

Figure 3 Add forecast emissions including 2025 numbers

Page 36









Page 37

Figure 5 Add forecast emissions including 2025 numbers



Based on RTP Financially Constrained System and 2004-07 MTIP. Source: Metro

Page 38



Appendix 1

### 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

### Financially Constrained System Project List

(Note: because RTP Packet 2 - Project Amendments contains the identical information and is being distributed with this draft conformity determination, please see RTP Packet 2. The final conformity determination will include this list.)



## 2004 RTP UPDATE Calendar of Activities

September 5	TPAC review and discussion on RTP Work Program
September 9	Metro meeting with TriMet on RTP finance and project assumptions
September 16	Council Work Session review of RTP Work Program
September 18	JPACT review of RTP Work Program
September 18	Metro meeting with City of Portland and Port of Portland on RTP finance and project assumptions
September 23	Metro meeting with Clackamas County Coordinating Committee TAC on RTP finance and project assumptions
September 24	Metro meeting with East Multnomah County Transportation Committee on RTP finance and project assumptions
September 25	Metro meeting with Washington County Coordinating Committee TAC on RTP finance and project assumptions
September 25	Metro meeting ODOT and other MPOS on State finance assumptions
September 26	TPAC discussion on defining the preferred system and financial constraint analysis
October 2	FTA/FHWA/DEQ/EPA and TPAC interagency consultation on air quality conformity
Early October	Preferred system analysis begins
October 7	TPAC Workshop – Finalize Preferred RTP System and continue discussion on Financially Constrained RTP System
October 14	TPAC Workshop – Finalize Financially Constrained RTP System 9:30-noon, Cooper Mountain Room (Rm 370 A)
Mid-October	Financially constrained system analysis begins
October 22	TPAC Workshop – General amendments to the RTP 9:30-noon, Cooper Mountain Room (Rm 370 A)

Updated October 8, 2003

October 28 Metro Council work session on draft 2004 RTP

- October 31 Staff recommendation on "technical" draft 2004 RTP released at TPAC to kick-off public comment period; draft RTP and conformity determination (not including emissions results) documents submitted to FHWA and FTA to begin review
- **November 3** Air quality conformity analysis begins
- November 5 MTAC discussion on draft 2004 RTP
- November 12 MPAC discussion on draft 2004 RTP
- November 13 JPACT discussion on draft 2004 RTP
- November 13 First Metro Council reading of Ordinance on draft 2004 RTP
- **November 19** MTAC comments on draft 2004 RTP (*tentative*)
- **November 26** TPAC discussion on draft 2004 RTP; review and discussion of air quality conformity analysis
- **December 4** Public hearing on draft 2004 RTP and air quality conformity procedures; public comment period ends at 5 p.m.
- December 5 TPAC Special Meeting comments on draft 2004 RTP
- December 10 MPAC consideration of 2004 RTP
- December 11 JPACT consideration of 2004 RTP
- **December 11** Second Council reading of Ordinance and Resolution, and consideration of adoption of 2004 RTP
- **December 12** RTP and final conformity determination submitted to FHWA and FTA for Federal review, pending approval by Metro Council

January 262000 RTP expires; deadline for federal conformity finding on 2004 RTP<br/>and conformity analysis to prevent lapse of RTP

**Appendix 3** 



### 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program Air Quality Conformity Analysis Protocols

#### **Transportation Emissions Budget Years**

For the Oregon portion of the Portland-Vancouver airshed, emission budgets (maximum air pollutant levels) have been set for various sources of pollutants (mobile, point, and area) and are included in the State Implementation Plan (SIP) and in the region's Ozone and Carbon Monoxide Maintenance Plans. The 2004 Regional Transportation Plan (RTP) and 2004-07 Metropolitan Transportation Improvement Program (MTIP) must conform to the SIP mandated transportation emissions budgets. Transportation emissions budgets are set for winter carbon monoxide (CO) and for two summer ozone precursors: nitrogen oxides (NOx), and hydrocarbons (HC). The geographic extent of the carbon monoxide transportation emission budget is the Metro jurisdictional boundary. For the carbon monoxide transportation emission budget, the geographic extent is the Air Quality Maintenance Area (AQMA) . However, emission budgets for carbon monoxide have also been established for the Central City Transportation Management Plan area (the central city of Portland) as well as an area along SE 82nd Avenue area from SE Division Street to SE Woodstock Avenue in southeast Portland. These areas are shown in the following map.


Plot time: Oct 28, 2003 J:\holsted\03283\_Mark\_Turpel\_AQ\_bdy\metro\_air\_quality\_boundaries.mxd

Please recycle with mixed paper

The region's approved Maintenance Plans identify two sets of budget years, one set for winter CO and one set for summer ozone precursors (NOx and HC). The CO budget years are 2007, 2010, 2015 and 2020. The ozone budget years are 2006, 2010, 2015 and 2020. In addition, a plan horizon year must also be evaluated. For the 2004 RTP, the horizon year is 2025. Table 1 shows the budget years and associated emissions budgets.

	2004 RTP Tra	Table 1 Insportation Emi	ssions Budgets <sup>1</sup>	l	
		Winter CO (thousand pounds/da	ıy)	Summer HC (tons/day)	Summer NOx (tons/day)
	Region	PDX Central	82nd	Region	Region
	(Metro boundary)	City Sub-area	Ave Sub-area	(AQMA)	(AQMA)
2006	n/a	n/a	n/a	41	51
2007	775	70	4	n/a	n/a
2010	772	68	4	40	52
2015	801	71	4	40	55
2020	856	76	4	40	59
2025	856	76	4	40	59

#### **Relationship of Budget Years to Analysis Years**

On October 2, 2003, Metro, DEQ, EPA, FHWA and FTA staff met and reviewed the conformity requirements. The process is technically complex, requires extensive staff and computer time and is, therefore, expensive. Metro fully models as few analysis years as possible to the degree the rules allow. As permitted by the conformity rule, Metro identifies and models key analysis years and interpolates between them to establish that regional mobile emissions meet all established emissions budgets. As noted in the table below, full transportation model runs, include forecasts of trip characteristics such as trip origin and destinations, time, length and duration. These full transportation model runs are completed for years 2000, 2010 and 2025. These transportation models are based on assumptions about future transportation improvements, the location and amount of future population and job growth and transportation facility characteristics (propensity to drive, use transit, etc). Future air quality conditions using air quality software (MOBILES5a-h) are then estimated using the output of the transportation model runs is used. This approach uses the trip tables from the 2010 and 2020 full model runs and assesses the results of these trips on a transportation network with improvements assumed to be made by 2015. Then the air quality model is run to estimate the air quality conditions in the year 2015.

This approach is acceptable under the federal rule and is called out in its preamble as follows: "A full regional emissions analysis must be performed for each pollutant and precursor for the last year of the transportation plan's forecast period (i.e., 2025)..." as well as for intervening years, not to exceed 10 years between analyses. For the other years for which the *budget test* is required to be demonstrated, the

<sup>&</sup>lt;sup>1</sup> Budgets are from the Maintenance Plan adopted in 1996. The maintenance plans include no specific year emission budget after year 2020, but other transportation planning requirements mandate that the planning forecast year also be conformed. The planning forecast year is 2025. The year 2025 budget uses the same budget as year 2020, as both the ozone and carbon monoxide maintenance plans call for the same budget "For Years 2020 and Beyond".

estimate of regional emissions does not necessarily need to be based on a full regional emissions analysis performed for the specific year; the estimate of regional emissions may be based on an interpolation between the years for which the full regional emissions analysis was performed.

Table 2 identifies the years for which a full conformity analysis was performed and the years for which interpolation was performed for both summer ozone precursors and winter carbon monoxide. Sub-area analyses are derived from the regional results.

	2004 Regional Tra	ansportation Plan (	Conformity Analysi	s Years
			Winter CO	Ozone
				(HC and NOx)
Year	Budget	Modeling	Emission	Emission
	Established	_	Calculation	Calculation
2006	Ozone		None - not	Emission
			required	Interpolation*
2007	Winter CO		Emission	None - not
			Interpolation*	required
2010	Both	Full Model run	MOBILE5a-h	MOBILE5a-h
2015	Both	Trip Assignment (Partial Model run)	MOBILE5a-h	MOBILE5a-h
2020	Both		Emission Interpolation	Emission Interpolation
2025	All years after 2020 to use 2020 budget	Full Model run	MOBILE5a-h	MOBILE5a-h

 Table 2

 2004 Regional Transportation Plan Conformity Analysis Year

\* A full model run was performed for year 2000. Emissions for 2006 and 2007 were interpolated using the 2000 and 2010 model runs.

### **Regional Travel Demand Model Inputs, Assumptions and Methodology**

For a full analysis, air quality conformity requires demand model outputs such as vehicle miles traveled, trip ends, and network speeds. Emissions calculations are performed on a link-by-link and matrix basis for stabilized emissions and trip end emissions, respectively. Metro's model requires the following inputs to be assembled or created, if not already available (for a given year):

- Population and employment forecasts
- Transit fare and parking cost data
- Transit network assumptions (PM peak, Midday; including bus routes and park & ride sheds)
- Highway network definitions (PM peak, Midday)
- Vehicle emission factors

The model run consists of the following steps:

- Trip generation (e.g., how many total trips are expected in the region)
- Destination choice (e.g., determination of where each of the approximately 5 million daily trips are coming from and going to)
- Mode choice
- Time of day identifications (AM peak, PM peak, midday, rest of the day)

Page 4

Assignment of trips to the network (path choice)

In addition, air quality conformity model runs require stratification of the trips by inspection maintenance area (Oregon I/M, Washington State I/M, and Non-inspected). Once the data are assembled and the demand model steps are completed, the results are used for the calculation of emissions. Ozone and CO gases are computed, and then reported in various geographies depending on the project requirements.

To summarize, a full model analysis was performed for year 2000, 2010 and the 2004 RTP horizon year of 2025. New trip assignments were prepared for 2015. Data for all other budget years were interpolated between these four analysis years. The interpolated results were then compared to actual emission budgets to establish that the 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program conform to the emissions budgets in all years for which they are established in the region's CO and Ozone maintenance plans.

#### MOBILE5a-h Air Quality Model Assumptions

The MOBILE5a-h air quality computer model is used to estimate the future air quality conditions for the Portland area should the 2004 RTP and 2004-2007 MTIP be implemented. More specifically, on-road motor vehicle emissions of carbon monoxide and precursors of ozone and will be determined using EPA's Mobile5a\_h Emissions Factor Model and the following parameters:

Fleet Data: Vehicle registration distribution and vehicle age distribution for Light Duty Gas Vehicles (LDGV) and Light Duty Diesel Vehicles (LDDV) will be derived from Oregon Dept. of Motor Vehicles registration records for Clackamas, Multnomah and Washington Counties 2002. Vehicle type and age distributions for other vehicle groups will be determined by national averages. Vehicles originating in Clark County, Washington will be characterized the same way if possible. If 2002 registration data are not available, national averages will be used to describe that portion of the fleet.

I/M Program: Vehicles registered in the Portland Metropolitan area are subject to Oregon DEQ's Inspection/Maintenance (Emissions Testing) Program. Details of the I/M program reflected in the Mobile5a\_h model are:

OBD Test: 1996 and newer vehicles are subject to On Board Diagnostics testing.

Enhanced Test: 1981 through 1995 model year vehicles are subject to BAR 31 "enhanced" emissions testing (modeled as EPA's I/M 240 enhanced test).

Basic Test: 1975 through 1980 model year vehicles are subject to the 2500 two speed idle emissions test.

Exemption: Most vehicles are not subject to emissions testing until they become four years old.

Waiver Rate: There is no repair cost threshold at which a vehicle does no have to meet the emissions test requirement.

I/M Program Start Year: 1975

Program Type: Centralized

Compliance Rate: 90%

Inspection Frequency: Biennial

Tampering Rates: Mobile5 rates.

Speed: One average speed used for all vehicle types.

Basic Emission Rates: derived from Mobile5 Basic Emission Rates.

Refueling Emissions: None calculated. (Accounted for under "Area Sources")

Summer Temperatures: Min: 61 deg. F; Max: 98 deg. F

Winter Temperature: Ambient = 39.8 deg. F

Summer Reid Vapor Pressure: 7.8 psi

Winter Reid Vapor Pressure: 13.6 psi

Winter Fuel Type: 2.7% Oxygen



Appendix 4

Metro

## 2004 Regional Transportation Plan Transportation Analysis Zone Assumptions

		2025	2025	2025	2025
	2040 Group Characteristics	Intersection	Parking	Transit	Fareless
2040 Grouping		Density	Factors	Pass	Areas
		(connections per	(indexed to	Factor	(for internal
		mile)	CBD	(% of Full	trips)
			in '94 dollars)	Fare)	
		FC	FC	FC	FC
Central City 1 Downtown Business District	Highest planned employment and housing density in the region, with				
	highest level of access by all modes. LRT exists and current land uses reflect planned mix and densities.	20	6.08	60%	<b>x</b> .
Central City 2 Lloyd District	Highest planned employment and housing density in the region, with highest level of access by all modes. LRT exists and current land uses reflect planned mix and densities.	20	3.94	60%	x
Central City 3 River District and Northwest	Planned high employment and housing density, with highest level of access by all modes. LRT exists and current land uses				
	approach planned mix and densities.	20	3.94	65%	
Central City 4 Central Eastside Industrial District	Planned high employment and housing density, with highest level of access by all modes. LRT exists and current land uses do not reflect planned mix and densitie	20	2.96	65%	
Control City 5	Densities.				
South Waterfront District	housing density, with highest level of access by all modes. LRT exists and current land uses do not reflect planned mix and densities.	18	3.04	65%	
Regional Centers -	Planned high employment and				
Tier 1	housing density, with highest level				
Gresham	of access by all modes. LRT				
Gateway	exists and current land uses				
Beaverton	approach planned mix and	>14	0.80	80%	х
Clockamoo	aensities.				
Decianal Contains	Dianned high employment and			<u> </u>	
Tion 2	Fianned nigh employment and		ļ		
Mashington Square	of access by all medees planned				
Oregon City	I PT Current land uses do not				
Cregon Oily	reflect planned mix and densities	>10	0.60	95%	
•		210	0.00	00,0	

	1	0000	00.07	0.000	
2040 Grouping	Group Characteristics	2025 Intersection Density (connections per mile)	2025 Parking Factors (indexed to CBD in '94 dollars)	2025 Transit Pass Factor (% of Full Fare)	2025 Fareless Areas (for internal trips)
		FC	FC	FC	FC
Station Communities Tier 1 Banfield Corridor Westside Corridor	High housing density mixed with commercial services; highest level of access for transit, bike and walk: existing LRT.	>12	0.80	80%	
Station Communities	Planned high housing density				
Tier 2 South/North Corridor	mixed with commercial services, with high level of transit, bike and walk; planned LRT. Current land uses do not reflect planned mix and densities.	>10	0.60	95%	
Town Centers - Tier 1 St. Johns Hollywood Lents Fairview/Wood Village Troutdale Rockwood	Moderate housing and employment density planned, with high level of access by all modes. Currently has good mix of uses, well connected street system and good transit.	>16	0.45	85%	
Lake Oswego Tualatin Forest Grove Milwaukie Sherwood Wilsonville					
Town Centers - Tier 2 West Portland Raleigh Hills Hillsdale Gladstone West Linn Sunset Cornelius Orenco	Moderate housing and employment density planned, with high level of access by all modes. Currently has some mix of uses, moderately connected street system and some transit. Existing topography or physical barriers may limit bike and pedestrian travel.	>10	0.36	100%	
Town Centers - Tier 3 Happy Valley Lake Grove Cedar Mill Tannasbourne	Moderate housing and employment density planned, with high level of access by all modes. Currently has modest mix of uses, poorly connected street system and poor transit. Existing topography or physical barriers may limit bike and pedestrian travel.	>8	0.28	100%	
Town Centers - Tier 4 Pleasant Valley Damascus Bethany Murrayhili	Moderate housing and employment density planned, with high level of access by all modes. Currently undeveloped or developing urban uses, with skeletal street system and poor transit. Existing topography or physical barriers may limit bike and pedestrian travel.	>8	0.18	100%	
Mainstreets - Tier 1 Eastside Portland to 60th	Moderate housing and employment density planned, with high level of access by all modes. Currently has good mix of uses, well connected street system and good transit.	>14	0.45	100%	

		2025	2025	2025	2025
	Group Characteristics	Intersection	Parking	Traneit	Farelace
2040 Grouping	aroup endiactoristics	Density	Factors	Paee	Areae
2040 Grouping		(connections	lindexed to	Factor	(for internal
		per mile)	CBD	(% of Full	trins)
			in '94 dollars)	Fare)	
		FC	FC	FC	FC
Mainstreets - Tior 2	Moderate housing and	· · · ·	·	· · ·	<b>`</b>
Remaining Region	employment density planned				
riemanning riegion	with high level of access by all				-
	modes Currently has some mix				
	of uses moderate connectivity	~8	0.36	100%	
	and some transit	~	0.00	10070	
Corridors	Moderate housing and			· · · ·	
Full Begion	employment density planned				
i un riegion	with high level of access by all				
	modes Currently has modest	>10	None	100%	
	mix of uses, moderate			10070	
	connectivity and some transit.				
Inner Neighborhoods	Low density housing planned.				I
Full Region	with moderate level of access				
	by all modes. Currently has	>10	None	100%	
	moderate connectivity and				
	some transit.				
Outer Neighborhoods -	Low density housing planned.				
Tier 1	with moderate level of access				
Current Urban Areas	by all modes. Currently has	>8	None	100%	
	poorly connected street system				
	and little transit.				
Outer Neighborhoods -	Low density housing planned,				
Tier 2	with moderate level of access				
Urban Reserve Areas	by all modes. Currently has	>6.	None	100%	
	skeletal street system and no				
	transit.				
Employment Areas	Low density employment				
Full Region	planned, with moderate level of				
	access by all modes. Currently				
	has poorly connected street	>8	None	100%	
	system and limited transit.				
Industrial Areas - Tier 1	Low density employment				
Rivergate	planned, with high level of				
Swan Island	access by rail and truck freight,				
Airport	and moderate access by other		(		
	modes. Currently has	>10	None	100%	
	somewhat connected street				
Industrial Areas Tier C	System and some transit.			·······	
South Shore	planned with high level of				
Clackamas	access by rail and truck freight				
Tualatin	and moderate access by other				
Beaverton	modes. Currently has		None	100%	
Sunset	developing street system and				
	poor transit.				
Greenspaces	Recreational uses are planned.				
Same as Tier 2 Outer	with moderate level of access	:			
Neighborhoods.	by all modes	>6	None	100%	
-					
Rural Reserves	Urban uses are not planned in				
Same as Tier 2 Outer	the foreseeable future.				l l
Neighborhoods.	Currently has skeletal street	>6	None	100%	
	system and no transit.				
Special Area 1					
Portland International Airport		*	6.14	60%	

Page 3

Special Area 2 Oregon Health Sciences University	These places are relatively small geographic areas with special characteristics.	•	1.86	60%	
Special Area 3 Oregon Zoo		*	1.86	100%	
Special Area 4 SMART (Wilsonville)		*	*	*	x

\* Use parent zone values. 10/29/03

à

Appen	dix 5
-------	-------



## 2004-07 Metropolitan Transportation Improvement Program (MTIP) Calendar of Activities

June 19 Council action on final Transportation Priorities program, pending air quality analysis.

September 26 TPAC review of draft MTIP report.

October 2 MTIP/RTP Air Quality interagency consultation meeting. 10-11:30a.m., Cooper Mountain Room (Rm 370 A)

**October 9** JPACT Review of draft MTIP report.

**October** Draft RTP financially constrained system defined and analyzed.

October 31 Draft conformity determination (not including emissions results) submitted to FHWA/FTA to begin review. Public comment period begins on 2004-07 MTIP and draft conformity determination.

**November 3** Joint RTP/MTIP air quality conformity analysis begins.

- **November 14** Public comment period on draft conformity determination (RTP and MTIP) begins
- **November 26** TPAC review and discussion of air quality conformity analysis.
- **December 4** Public hearing on 2004 RTP and 2004-07 MTIP air quality conformity determination at Metro Council. Public comment period closes at 5:00 pm.

**December 11** Final JPACT action on 2004-07 MTIP and air quality conformity

- **December 11** Metro Council action on 2004-07 MTIP and air quality conformity determination (by Resolution).
- December 122004 RTP and 2004-07 MTIP final conformity determinations<br/>submitted to FHWA and FTA for Federal review, pending<br/>approval by Metro Council.

January 26 Anticipated federal approval of 2003 RTP and 2004-07 MTIP air quality conformity determinations.

September 10, 2003



**Appendix 6** 

### METRO 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

## **Published Notice**



Copy of Post Card sent via US Mail to about 2,500 people (RTP & MTIP Interested Persons mailing list and neighborhood and community planning organizations within the region)



## **Transportation plan update begins**

Public comment will be taken Oct. 31 to Dec. 4

Metro is starting a periodic update of the Regional Transportation Plan (RTP) in order to maintain continued compliance with the Federal Clear Air Act and state guidelines. The update will include an air quality analysis of the 2004 RTP and 2004-07 Metropolitan Transportation Improvement Program.

The plan, updated every three years to ensure that it addresses future travel needs, will focus on projects for roads and freight movement, bicycling, transit and walking. These projects already have been adopted in local and regional plans and corridor studies through a public process.

Public comment will be taken Oct. 31 through Dec. 4. The staff recommendation on the technical draft of the plan will be available for public review on Oct. 31.

#### Public hearing will be held Dec. 4

A public hearing will be held during the Thursday, Dec. 4, Metro Council meeting. The meeting begins at 2 p.m. at Metro Regional Center, 600 NE Grand Ave., Portland.

The council will take action on the update on Dec. 11 (tentative). For more information, visit **www.metro-region.org** or call (503) 797-1839.

#### Other ways to comment

Phone	(503) 797-1900 option 2
Fax	(503) 797-1911
E-mail	trans@metro.dst.or.us
Mail	Kim Ellis, Metro
	600 NE Grand Ave.
	Portland, OR 97232

#### October 2003

#### METRO'S REGIONAL PLANNING E-NEWSLETTER

Welcome to Metro's Regional Planning e-newsletter. It is e-mailed periodically to interested persons. Check the end of the newsletter for "subscription" information.

#### FEATURED IN THIS ISSUE:

- · Periodic update of Regional Transportation Plan
- Downtown Mall revitalization comments solicited
- Powell/Foster Corridor Study recommendation due
- TGM grant received for Centers and Corridors Study
- Fish and wildlife habitat protection events

#### UPDATE BEGINS ON REGIONAL TRANSPORTATION PLAN

Metro is starting a periodic update of the Regional Transportation Plan (RTP), in order to maintain continued compliance with the Federal Clean Air Act and state guidelines. The update will include both a 2004 RTP and 2004-07 Metropolitan Transportation Improvement Program (MTIP) air quality analysis.

The RTP is updated every three years to ensure the plan addresses future travel needs. For this update, the plan will focus on projects for roads and freight movement, bicycling, transit and walking that have already been adopted in local and regional plans and corridor studies through a public process.

Public comment will be taken Oct. 31 through Dec. 4, 2003. The staff recommendation on the technical draft of the plan and the air quality analysis will be available for public review on Friday, Oct. 31.

Comments will be taken at a public meeting at 2 p.m. Thursday, Dec. 4 at Metro, 600 NE Grand Avenue in Portland.

The Metro Council is scheduled to take action on the RTP update on Thursday, Dec. 11 (tentative). For more information, visit www.metro-region.org or call (503) 797-1839.

#### PUBLIC COMMENT SOLICITED ON THE DOWNTOWN MALL REVITALIZATION PROJECT

Metro, TriMet and the City of Portland are considering adding light rail to the Portland Mall as part of an effort to revitalize Fifth and Sixth avenues.

Appendix 7



### 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

### Evaluation of emissions analysis for transportation activities which cross borders of MPOs or nonattainment or maintenance areas or basins

The following maps and data were forwarded from Multnomah County. They constitute all projects within Multnomah County and inside the Air Quality Maintenance Area and outside the Metro boundary.

These projects were not considered regionally significant as they do not appear to add to the road capacity and therefore should have no air quality impact.

	· .	• • •	· ·	
Fish Passage Culvert Project - Field Form         Culvert model Name, Culvert #, Mile Point, Size         Culvert ID No.       Road Name, Culvert #, Mile Point, Size       Stream         Milepoint       Priority       Owner       USGS Quad MapName				
)3-06 282ND Av, SE - # 2 - MP: 2.046 84 549250 5034300 Johnso	x 40 IRIS: 493 n Creek 3.5 High	Multnomah County	Sandy	Preliminary Assessment
IRIS Material Type: CP inlet T Stats Coating Type: C Outlet	reatment. BH Offset Distance: Treatment BH Cover Depth:	16 Slope: 0 2 Skew: 45	Rise Height 84 Draihag Span Width: 84 Conditio	e Adequacy: A n: G Road MP 2.046
Coho Salmon: Verified Cutth	roat Trout: Verified Steelhe	ead: None	Vinter Steelhead: Verified	Rainbow Trout: None
			· · · · · · · · · · · · · · · · · · ·	
			· · · · · · · · · · · · · · · · · · ·	
		Measure	ments - Outfall Drop:	Depth of Pool:

· · ·

Project Nam	282nd /	Ave/Stone Rd				
Project #:	705	Category: Sign	al/Intersection	Fu	nctional Class: Rural Arte	rial
Project Description:	Widen offset o	282nd Ave to creat of east and west leg	te left turn poo js.	kets to S	Stone Rd. Widen Stone Re	d to reduce
RTP No:		IRIS #: 493	Mile Point:	2.09	ROW Cost:	\$20,000
TIF					Construction Cost:	\$150,000

Score:

5

Construction Cost: \$150,000 \$170,000 Total Cost:



Map not to Scale		-
·	Existing	New
Travel Lanes:	2	3
Sidewalks:	No	No
Bike Lanes:	No ·	
Drainage:	Ditch	Ditch
Illumination:	No	No
Turn Lanes:	No	Yes
Intersection:	No	Yes
	1	

10/8/2003 02:36 PM

Project Nam Beaver Creek Bridge on Historic Columbia River Hwy

Project #:	724	Category: Bridge	Functional Class: Major Co	ollector
Project Description:	Replac	e Bridge	· ·	
RTP No:		IRIS #: 490	ROW Cost:	\$60,C

TIF Score:

□ 30 IRIS

ROW Cost:	\$60,000
Construction Cost:	\$987,000
Total Cost:	\$1,047,000



Map not to Scale					
	Existing	New			
Travel Lanes:	2	2			
Sidewalks:		Yes			
Bike Lanes:	No	Yes			
Drainage:	Storm	Storm			
Illumination:	No	No			
Turn Lanes:	No	No			
Intersection:	Yes	No			

10/8/2003 02:33 PM

Project Nam	238th I	Dr: Glisan St-Arata Rd Safet	/ Improvements	
Project #:	722	Category: Arterial	Functional Class: Minor Arte	ərial
Project Description:	Widen ahead	existing pavement near entra sign with beacons.	nce to Tree Hill Condominiums, and i	nstall signal
RTP No:		IRIS #: 403	ROW Cost:	\$0
TIF	П	From Mile Point: 0.000	Construction Cost:	\$125,000

0.641

\$125,000

Total Cost:



Existing	New
3	3
	No
No	No
Storm	Storm
Yes	Yes
Yes	Yes
Yes	Yes
	Existing 3 No Storm Yes Yes Yes

10/8/2003 02:37 PM

20

Score:

To Mile Point:

Project Nam	Stark St	Viaduct		
Project #:	736	Category: Bridge	Functional Class: Rural Arter	ial
Project Description:	Reconst	ruct Stark St Viaduct		

RTP No:		IRIS #: 404	Mile Point:	2.64	ROW Cost:	\$0
TIF					Construction Cost:	\$679,000
Score:	10				Total Cost:	\$679,000



Map not to Scale	T	
	Existing	New
Travel Lanes:	2	
Sidewalks:		
Bike Lanes:	No	No
Drainage:	Ditch	Ditch
Illumination:	No	No
Turn Lanes:	No	No
Intersection:	No	No
		1

10/8/2003 02:34 PM

Project Nam Orient Dr/Dodge Park Blvd

Project #:	703	Category: Signal/Intersection	Functional Class: Rural Arterial
Project Description:	Widen Or	ient Dr to create eastbound left turr	ו lane.

RTP No:		IRIS #: 434	Mile Point:	2.06	ROW Cost:	\$10,000
TIF					Construction Cost:	\$90,000
Score:	5				Total Cost:	\$100,000



Map not to Scale	· · · · · · · · · · · · · · · · · · ·	
	Existing	New
Travel Lanes:	2	3
Sidewalks:	No	No
Bike Lanes:	No	No
Drainage:	Ditch	Ditch
Illumination:	No	No
Turn Lanes:	No	Yes
Intersection:	No	Yes
		-

10/8/2003 02:38 PM

**Appendix 8** 



## 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

### **Evidence of Compliance with Metro Interim Land Use Measures**

Attached is Metro Resolution No. 03-3299, which documents the results of the Metro Urban Growth Management Functional Plan. The air quality maintenance plans for the Portland area call for "Metro Interim Land Use Measures relating to: Requirements for Accommodation of Growth; Regional Parking Policy; and Retail in Employment and Industrial Areas."

The relevant portions of the The Urban Growth Management Functional Plan (titles 1, 2 and 4) are also attached to document the recommendations and requirements of the Functional Plan and how they concern the cities and counties of the region.

#### BEFORE THE METRO COUNCIL

#### FOR THE PURPOSE OF ENTERING AN ORDER RELATING TO COMPLIANCE WITH THE URBAN GROWTH MANAGEMENT FUNCTIONAL PLAN

#### **RESOLUTION NO. 03-3299**

Introduced by Councilor Rod Park

WHEREAS, Title 8 of the Urban Growth Management Functional Plan ("UGMFP") requires the Metro staff to submit to the Metro Council a report on the status of compliance of each local government with each requirement of the UGMFP, and to provide public notice of the report; and

)

WHEREAS, the Executive Officer submitted two reports jointly entitled "2002 Urban Growth Management Functional Plan Compliance Reports", one part on the status of compliance with UGMFP Titles 1 through 6 and a second part on the status of compliance with Title 7, to the Council on December 2, 2002, and provided public notice of the reports; and

WHEREAS, Title 8 requires the Council to hold a public hearing for the purpose of taking testimony on the question whether cities and counties have complied with the UGMFP; and

WHEREAS, the Council held a hearing for that purpose on January 30, 2003, and heard testimony from interested persons, and from the staff on actions to comply with the UGMFP taken by local governments after the December 2, 2002, reports; and

WHEREAS, Title 8 requires the Council to enter an order that determines the status of each city's and county's compliance with the requirements of the UGMFP, and to send a copy of the order to all cities and counties and all persons who participated at the hearing; now, therefore,

BE IT RESOLVED:

- 1. That the Council adopt Order No. 03-001, with its attachments, as the Council's determination of the status of city and county compliance with the UGMFP, pursuant to subsection 3.07.880C.
- 2. That the Council direct the Metro staff to send a copy of Order No. 03-001 to all cities and counties and all persons who participated at the hearing, pursuant to subsection 3.07.880C.

ADOPTED by the Metro Council this day of <u>Apri</u> 2003.

David Bragdon, Council President

Approved as to Form:

Daniel B. Coop etro Attorney

Page 1 - Resolution 03-3299 m: httom//onifidential/1.4.3.7.3/03-3299.004 OMARPB/kww (03/28/03)



#### Order No. 03-001

#### RELATING TO COMPLIANCE WITH THE URBAN GROWTH MANAGEMENT FUNCTIONAL PLAN

#### IT IS ORDERED THAT:

1. The Council accepts the December 2, 2002, combined reports from the Executive Officer entitled "2002 Urban Growth Management Functional Plan Compliance Reports" and the January 24, 2003, hearing report presented by staff at the January 30, 2003, public hearing as fulfilling the requirement of Urban Growth Management Functional Plan (UGMFP) Title 8, section 3.07.880A. The reports are attached and incorporated into this order as Exhibits A and B, respectively.

2. Based upon the staff reports described in section 1 of this order and testimony received at the public hearing, the Council adopts Exhibit C, entitled "Status of Compliance by Jurisdiction – 2002", attached and incorporated into this order, as its determination of the status of city and county compliance with UGMFP requirements of Titles 1 through 7, as required by Title 8, section 3.07.880C.

3 Based upon the determinations in Exhibit C, the Council concludes that the cities of Beaverton, Durham, Johnson City, King City, Lake Oswego, Maywood Park, Milwaukie, Troutdale and Wilsonville and Clackamas and Washington Counties have not achieved the target housing capacities required by Title 1 (Requirements for Housing and Employment Accommodation). The Council further concludes that the cities of Beaverton, Happy Valley, Johnson City, Maywood Park, Milwaukie, Oregon City, Rivergrove and Wilsonville and Clackamas County have not achieved the target employment capacities required by Title 1. However, in 1998 and 1999, the Council expanded the urban growth boundary (UGB) to add housing and employment capacity, in part because it was not possible for some cities to achieve their targets. As a result of UGB expansion and actions taken by local governments after the expansion, the region as a whole has achieved and exceeded the housing and employment targets set in Title 1. Given this achievement, on December 5, 2002, the Council adopted Ordinance No. 02-969B, amending Title 1 to replace the housing and employment targets of Table 3.07-1 with zoned capacity. Revised Table 3.07-1 displays actual zoned capacities for housing and employment achieved by city and county actions taken to comply with Title 1. Revised Title 1 accepts these capacities and prohibits net reductions. Having considered these past actions by the Council, the Council concludes that no further action need be taken by cities or counties or the Council to achieve the housing or employment targets specified in the now-repealed version of Table 3.07-1.

4. The staff reports do not indicate whether cities and counties have complied with the requirement in Title 1, section 3.07.140A, to report on density of residential development between 1990 and 1995, and to take action if actual density fell below 80 percent of maximum zoned density. The Council assumes, therefore, that cities and counties have not complied with the reporting requirement. However, all cities and counties except the cities of Durham and Oregon City have now adopted minimum densities that prevent development below 80 percent of maximum zoned density (both Durham and Oregon City reported to Metro that residential development in their cities is taking place at least at 80 percent of maximum zoned densities).

Page 1 of 2 Order No. 03-001 to Resolution No. 03-3299 m:/sttorney/confidential/7.4.3.7.3/03-3299.Order 03-001.cln 003 OMA/RPE/Lyw (03/28/03)

These minimum densities are the basis for the zoned capacity for each city and county displayed on Table 3.07-1. Accordingly, Ordinance No. 02-969B amended Title 1 to revise the requirements of section 3.07.140A. Hence, the Council concludes that no further action need be taken by cities or counties or the Council to achieve compliance with the reporting requirement of section 3.07.140 as it read prior to revision by Ordinance No. 02-969B.

5. The staff reports do not indicate whether cities and counties reported on actions to achieve the target housing or employment capacities in mixed-use areas, or whether they achieved the target capacities, as required by Title 1, section 3.07.160B. The Council assumes, therefore, that cities and counties have not complied with the reporting requirement. The Council notes, however, that the target capacities for mixed-use areas are subsumed by each city's and county's overall targets for housing and employment. Ordinance No. 02-969B amended Title 1 to replace the housing and employment targets of Table 3.07-1 with zoned capacity and to remove from that table separate targets or capacities for mixed-use areas. In place of targets or capacities for mixed-use areas, the Council adopted a new Title 6 for Centers (Central City, Regional and Town Centers, Station Communities) and a program to facilitate increased housing and employment capacities in Centers. For these reasons, the Council concludes that no further action need be taken by cities or counties or the Council to achieve compliance with the requirements of section 3.07.160B as it read prior to revision by Ordinance No. 02-969B.

6. The staff reports ask the Council to interpret language in subsection 3.07.730B of Title 7 that requires cities and counties to consider amendment of their comprehensive plans to adopt affordable housing strategies. The Council interprets the subsection to mean that the governing body of the city or county must consider each strategy listed in the subsection and either amend its land use regulations to adopt the strategy or explain why it has decided not to adopt the strategy.

ENTERED this 10 day of April, 2003 avid Bragdon, Council President Approved as to Form: PPROUVE

METRO

Daniel B. Cooper Metro Attorney

Page 2 of 2 Order No. 03-001 to Resolution No. 03-3299 m: wittorney/confidential/7.4.3.7.3/03-3299.Order 03-001.ela 003 OMA/RPB/Lyw (03/28/03)

### Status of Compliance by Jurisdiction - January 2003

Title 1: Housing and Employment Accommodation					
	2.A minimum density	2.B partitioning	2.C accessory	3.A map of design	5.A capacity analysis
		standards	dwelling units	types	
Beaverton	in compliance	in compliance	in compliance	in compliance	housing, employment low
Cornelius	in compliance	in compliance	in compliance	in compliance	in compliance
Durham	exception requested	in compliance	in compliance	in compliance	housing low
Fairview	in compliance	in compliance	in compliance	in compliance	in compliance
Forest Grove	in compliance	in compliance	in compliance	in compliance	in compliance
Gladstone	in compliance	in compliance	in compliance	in compliance	in compliance
Gresham	in compliance	in compliance	in compliance	in compliance	in compliance
Happy Valley	in compliance	in compliance	in compliance	in compliance	employment low
Hillsboro	in compliance	in compliance	in compliance	in compliance	in compliance
Johnson City	in compliance	in compliance	in compliance	in compliance	housing low employment low
King City	in compliance	in compliance	in compliance	in compliance	housing low
Lake Oswego	in compliance	in compliance	in compliance	in compliance	in compliance
Maywood Park	in compliance	in compliance	in compliance	in compliance	housing low, employment low
Milwaukie	in compliance	in compliance	in compliance	in compliance	housing low, employment low
Oregon City	extension to 12/02	in compliance	extension to 12/02	in compliance	employment low
Portland	in compliance	in compliance	in compliance	in compliance	in compliance
Rivergrove	in compliance	in compliance	in compliance	in compliance	employment low
Sherwood	in compliance	in compliance	in compliance	in compliance	in compliance
Tigard	in compliance	in compliance	in compliance	in compliance	in compliance
Troutdale	in compliance	in compliance	in compliance	in compliance	housing low
Tualatin	in compliance	in compliance	in compliance	in compliance	in compliance
West Linn	in compliance	in compliance	in compliance	in compliance	in compliance
Wilsonville	in compliance	in compliance	in compliance	extension to 09/02	extension to 09/02
Wood Village	in compliance	in compliance	in compliance	in compliance	in compliance
Clackamas C.	in compliance	in compliance	in compliance	in compliance	housing low, employment low
Multnomah C.	in compliance	in compliance	in compliance	in compliance	targets to Portland Gresham, Troutdale
Washington C.	in compliance	in compliance	in compliance	in compliance	housing low

Exhibit C to Resolution 03-3299

Page 1 of 6

	Title 2: Regional Parking Policy				
	2.A.1&2 Minimum/Maximum standards	2.A.3 Variance Process	2.B Blended Ratios		
Beaverton	in compliance	in compliance	in compliance		
Cornelius	in compliance	in compliance	in compliance		
Durham	scheduled for February 2003 adoption	scheduled for February 2003 adoption	scheduled for February 2003 adoption		
Fairview	in compliance	in compliance	in compliance		
Forest Grove	in compliance	in compliance	in compliance		
Gladstone	in compliance	in compliance	in compliance		
Gresham	in compliance	in compliance	in compliance		
Happy Valley	in compliance	in compliance	in compliance		
Hillsboro	in compliance	in compliance	in compliance		
Johnson City	in compliance	in compliance	in compliance		
King City	in compliance	in compliance	in compliance		
Lake Oswego	in compliance	in compliance	in compliance		
Maywood Park	in compliance	in compliance	in compliance		
Milwaukie	in compliance	in compliance	in compliance		
Oregon City	in compliance	in compliance	in compliance		
Portland	in compliance	in compliance	in compliance		
Rivergrove	in compliance	in compliance	in compliance		
Sherwood	in compliance	in compliance	in compliance		
Tigard	in compliance	in compliance	in compliance		
Troutdale	in compliance	in compliance	in compliance		
Tualatin	in compliance	in compliance	in compliance		
West Linn	in compliance	in compliance	in compliance		
Wilsonville	in compliance	in compliance	in compliance		
Wood Village	in compliance	in compliance	in compliance		
Clackamas County	in compliance	in compliance	in compliance		
Multhomah County	In compliance	in compliance	in compliance		
Washington County	in compliance	in compliance	in compliance		

	Title 3: Water Quality, Flood Mgmt and Fish and Wildlife Conservation							
	4.A Flood Mgmt Performance Standards	4.B Water Quality Performance	4.C Erosion and Sediment Control					
Beaverton	in compliance	in compliance	in compliance					
Comelius	in compliance	in compliance	in compliance					
Durham	in compliance	in compliance	in compliance					
Fairview	in compliance	In compliance	in compliance					
Forest Grove	in compliance	in compliance	in compliance					
Gladstone	in compliance	in compliance	in compliance					
Gresham	in compliance	in compliance	in compliance					
Happy Valley	in compliance	in compliance	in compliance					
Hillsboro	in compliance	in compliance	in compliance					
Johnson City	in compliance	in compliance	in compliance					
King City	in compliance	in compliance	in compliance					
Lake Oswego	In compliance	extension to 12/02	in compliance					
Maywood Park	N/A	N/A	in compliance					
Milwaukie	In compliance	in compliance	in compliance					
Oregon City	in compliance	In compliance	in compliance					
Portland	in compliance	in compliance	in compliance					
Rivergrove	in compliance	in compliance	in compliance					
Sherwood	in compliance	in compliance	in compliance					
Tigard	in compliance	in compliance	in compliance					
Troutdale	in compliance	in compliance	in compliance					
Tualatin	in compliance	in compliance	in compliance					
West Linn	in compliance	extension to 12/02	in compliance					
Wilsonville	in compliance	in compliance	in compliance					
Wood Village	N/A	in compliance	in compliance					
Clackamas County	in compliance	extension to 12/02	in compliance					
Mutthomah County	in compliance	in compliance	in compliance					
Washington County	in compliance	in compliance	in compliance					

•

	Title 4: Retail in Employ	Title 5: Neighbor Cities and Rural Reserves		
	2.A Retail Restrictions - Industrial Areas	2.B Retall Restrictions – Employment Areas	2. Rural Reserves	2. Green Corridors
Beaverton	in compliance	in compliance	N/A	N/A
Cornelius	in compliance	in compliance	N/A	N/A
Durham	in compliance	in.compliance	N/A	N/A
Fairview	in compliance	in compliance	N/A	N/A
Forest Grove	in compliance	in compliance	N/A	N/A
Gladstone	N/A	in compliance	N/A	N/A
Gresham	in compliance	in compliance	N/A	in compliance
Happy Valley	N/A	N/A	N/A	N/A
Hillsboro	in compliance	in compliance	N/A	in compliance
Johnson City	N/A	N/A	N/A	N/A
King City	N/A	N/A	N/A	N/A
Lake Oswego	in compliance	in compliance	N/A	N/A
Maywood Park	N/A	N/A	N/A	N/A
Milwaukie	in compliance	in compliance	N/A	N/A
Oregon City	in compliance	in compliance	N/A	extension to 12/02
Portland	in compliance	in compliance	N/A	N/A
Rivergrove	N/A	N/A	N/A	N/A
Sherwood	in compliance	in compliance	N/A	in compliance
Tigard	in compliance	in compliance	N/A	N/A
Troutdale	in compliance	in compliance	N/A	N/A
Tualatin	in compliance	in compliance	N/A	in compliance
West Linn	N/A	in compliance	N/A	in compliance
Wilsonville	in compliance	in compliance	N/A	in compliance
Wood Village	in compliance	in compliance	N/A	N/A
Clackamas County	in compliance	in compliance	in compliance	in compliance
Multnomah County	in compliance	in compliance	N/A	in compliance
Washington County	in compliance	in compliance	in compliance	in compliance

Page 4 of 6 `

.

	Title 6: Regional Accessibility							
·	2. Regional Street Designs	3. Design Standards for Connectivity						
Beaverton	in compliance	In compliance						
Comelius	in compliance	in compliance						
Durham	in compliance	In compliance						
Fairview	in compliance	in compliance						
Forest Grove	in compliance	in compliance						
Gladstone	in compliance	in compliance						
Gresham	in compliance	in compliance						
Happy Valley	in compliance	in compliance						
Hillsboro	in compliance	In compliance						
Johnson City	in compliance	in compliance						
King City	in compliance	In compliance						
Lake Oswego	in compliance	in compliance						
Maywood Park	in compliance	in compliance						
Milwaukie	in compliance	in compliance						
Oregon City	in compliance	in compliance						
Portland	in compliance	in compliance						
Rivergrove	in compliance	in compliance						
Sherwood	in compliance	in compliance						
Tigard	in compliance	in compliance						
Troutdale	in compliance	in compliance						
Tualatin	in compliance	in compliance						
West Linn	in compliance	in compliance						
Wilsonville	extension to 09/02	in compliance						
Wood Village	in compliance	in compliance						
Clackamas County	in compliance	in compliance						
Multnomah County	in compliance	in compliance						
Washington County	in compliance	in compliance						

. •

Page 5 of 6

4

.

Exhibit C to Resolution 03-3299

Title 7: Affordable Housing										
	Progress	Voluntary	Comprehensive Plan and Implementing Ordinances					Other stro	tegies	
Jurisdiction	Reports	Goals	Diversity Strategy	Maintain Supply and Increase	Supply for All Income Levels	Land Use Strategies (Seven)				
	(Title 7: 3.07.740)	(îitle 7: 3.07.720)	(Title 7: 3.07.730.A.1)	(Title 7: 3.07.730.A.2)	(Title 7: 3.07.730.A.3)	(Title 7: 3.07.730.B)			(Title 7: 3.07.760)	
			·			Existing	Discussed	Considered	Metro list (five)	Local initiative
Beaverton	Yes	Discussed	NAR	NAR	NAR	NAR	NAR	NAR	2	1
Comelius				[					1	
Durham	Yes	NAR	NAR	NAR	NAR	NAR	NAR	NAR	NAR	NAR
Fairview					· · ·	·				•
Forest Grove						· ·			1	
Gladstone										
Gresham	Yes	Discussed	NAR	NAR	NAR	2	7	6	2	NAR
Happy Valley								· · · · · · · · · · · · · · · · · · ·		
Hillsboro	Yes	NAR	NAR	NAR	NAR	1	NAR	NAR	1	NAR
Johnson City									1	1
King City	Yes									1
Lake Oswego						1.	1			
Maywood Park						· ·				
Milwaukie	Requested Extension									
Oregon City		·								
Portland	Yes	NAR	NAR	NAR	NAR	6	7	NAR	5	16
Rivergrove						1				1
Sherwood			1	-						
Tigard	Yes	Discussed	NAR	NAR	NAR	2	2	1	2	5
Troutdale			1	1			•		1	
Tualatin	Yes	NAR	NAR	NAR	NAR	2	NAR	NAR	NAR	
West Linn										
Wilsonville										1
Wood Village	Yes	NAR	NAR	NAR	NAR	NAR	NAR	NAR	NAR	1
Clackamas	Yes	Consider in	NAR	NAR	NAR	5	NAR	NAR	3	3
Multhomab			l	· · · · · · · · · · · · · · · · · · ·						+
County	[							<u> </u> .		1
Washington County	Yes		NAR	NAR	NAR	2	0	NAR	1	NAR
Definitions:	Discussed =	Discussed after	January 2001		······································					

Discussed = Discussed after January 2001 Existing = Adopted prior to January 2001. Considered = Discussed at a local elected officials public meeting after January 2001, and adoption of an ordinance which amends the comprehensive plan and implementing ordinances to include new tools and strategies or tools and strategies which were considered but not adopted and the revision(s) not adopted. NAR = No action reported

Exhibit C to Resolution 03-3299

Page 6 of 6 '

#### Title 1, Table 1 Compliance -- May 29, 2002 - Dwelling Unit Capacity

	Table 1 Tar	get	Capacity	Difference	% of	% Short of	Notes
		· ·	Analysis		Jurisdiction	Total Region	
					Target	Target	
Beaverton	1:	5,021	13,635	(1,386)	91%	0.6%	
Cornelius		1,019	1,285	266	126%		
Durham		262	243	(19)	93%	0.008%	
Fairview		2,921	2,929	8	100%		·
Forest Grove		2,873	3,054	181	106%		
Gladstone		600	880	280	146%	•	
Gresham	1	6,817	16,920	103	101%		
Happy Valley		2,030	2,558	528	126%	·	does not include newly annexed areas
Hillsboro	1,	4,812	14,896	. 84	101%		·
Johnson City		168	38	(130)	23%	0.05%	allocation process did not account for existing mobile homes
King City		182	100	(82)	55%	0.03%	
Lake Oswego	3,353 4	,212'	4,049	(163)	96%	0.07%	859 units from Clackamas County
Maywood Park		27	12	(15)	44%	0.006%	
Milwaukie		3,514	3,188	(326)	90%	0.1%	
Oregon City	6,157 10	),630 <sup>1</sup>	7,994	(2,836)	75%	1.2%	City's preliminary estimate – will submit a revised capacity analysis – 4,473 units from the County
Portland	7	0,704	71,036	332	100%		mid point between zoned capacity of 66,994 and comp. plan capacity of 75,078.
Rivergrove		(15)	20	35	233%		
Sherwood		5,010	5,216	206	104%		
Tigard		6,073	6,308	235	104%		
Troutdale		3,789	3,260	(529)	86%	0.2%	
Tualatin		3,635	4,009	374	110%		
West Linn	2,577 3	3,226 <sup>1</sup>	3,732	506	116%		649 units from Clackamas County
Wilsonville		4,425	N/A	(4,425)	N/A	1.8%	capacity analysis not available
Wood Village		423	458	35	108%		
Clackamas C.	19,530 13	3,549 <sup>1</sup>	12,540	(1,007)	93%	0.4%	5,983 to be included in LO OC and WL
Multnomah C.		3,089	N/A	(3,089)	N/A	1.3%	need to coordinate with cities
Washington C.	5	4,999	51,649	(3,350)	94%	1.4%	
Regional Total	24	3,995	230,009	(13,986)	94%	6.0%	Wilsonville, Multnomah to report; Oregon City to submit revised capacity analysis

<sup>1</sup>Clackamas County allocated a portion of its targets for the areas where Lake Oswego, Oregon City and West Linn have planning jurisdiction over unincorporated areas.

1

and a start of the cost of the second start of the

#### Title 1, Table 1 Compliance – May 29, 2002 - Employment Capacity

	Table 1	1 Target	Capacity	Difference	% of	% Short of	Notes
			Analysis		Jurisdiction	Total Region	
					Target	Target	
Beaverton		25,122	21,368	(3,754)	85%	0.8%	
Cornelius		2,812	3054	242	109%		
Durham		498	522	24	105%		
Fairview		5,689	7,063	1,374	124%		
Forest Grove		5,488	5,943	455	108%		
Gladstone		1,530	1,569	39	103%		
Gresham		23,753	24,579	826	103%		
Happy Valley		1,767	510	(1,257)	29%	0.3%	includes 304 jobs from newly annexed areas
Hillsboro		58,247	59,082	835	101%		
Johnson City		180	82	(98)	45%	0.02%	allocation process did not account for existing mobile homes
King City		241	350	109	145%		
Lake Oswego	8,179	10,587 <sup>1</sup>	13,268	2,681	125%		2,408 jobs from County
Maywood Park		5	5	0	100%		
Milwaukie		7,478	3,650	(3,828)	49%	0.8%	
Oregon City	8,185	11,1721	7,665	(3,507)	68%	0.8%%	City's preliminary estimate – will submit a revised capacity analysis – 2,987 jobs from County
Portland		158,503	208,115	49,612	131%		mid point between zoned capacity of 191,913 and comp. plan capacity of 224,318.
Rivergrove		41	0	(41)	0%	0.009%	· · · · · · · · · · · · · · · · · · ·
Sherwood		8,156	9,518	1,362	117%		
Tigard		14,901	17,801	2,900	119%		
Troutdale		5,570	7,222	1,652	130%		
Tualatin		9,794	12,286	2,492	125%		
West Linn	2,114	2459 <sup>1</sup>	2,935	476	119%		345 jobs from County
Wilsonville		15,030	N/A	(15,030)	N/A	3.3%	
Wood Village		736	1,074	338	145%	·	
Clackamas C.	42,685	36,945'	31,101	(5,844)	84%	1.2%	5,670 jobs to LO, OC and WL
Multnomah C.		2,381	N/A	(2,381)	N/A	0.5%	
Washington C.		52,578	55,921	3,343	106%		
Regional Total		461,663	494,683	33,020	107%		Wilsonville, Multnomah to report; Oregon City to submit revised capacity analysis

<sup>1</sup> Clackamas County allocated a portion of its targets for the areas where Lake Oswego, Oregon City and West Linn have planning jurisdiction over unincorporated areas.

I:\gm\community\_development\projects\COMPLIANCE\Compliance Status\Table 1 summary of compliance .doc

#### CHAPTER 3.07

#### URBAN GROWTH MANAGEMENT FUNCTIONAL PLAN

SECTIONS TITLE

3.07.010 Purpose

- 3.07.020 Regional Policy Basis
- 3.07.030 Structure of Requirements
- TITLE 1: REQUIREMENTS FOR HOUSING AND EMPLOYMENT ACCOMMODATION
- 3.07.110 Purpose and Intent
- 3.07.130 Design Type Boundaries Requirement

3.07.160 Local Plan Accommodation of Expected Growth Capacity for Housing and Employment-Performance Standard

3.07.170 Design Type Density Recommendations

#### • TITLE 2: REGIONAL PARKING POLICY

3.07.210 Intent

3.07.220 Performance Standard

Table 3.07-2 - Regional Parking Ratios

#### TITLE 3: WATER QUALITY, FLOOD MANAGEMENT AND FISH AND WILDLIFE CONSERVATION

- 3.07.310 Intent
- 3.07.320 Applicability
- 3.07.330 Implementation Alternatives for Cities and Counties
- 3.07.340 Performance Standards
- 3.07.350 Fish and Wildlife Habitat Conservation Area
- 3.07.360 Metro Model Ordinance Required
- 3.07.370 Variances

Table 3.07-3 - Protected Water Features

TITLE 4: INDUSTRIAL AND OTHER EMPLOYMENT AREAS

- 3.07.410 Purpose and Intent
- 3.07.420 Protection of Regionally Significant Industrial Areas
- 3.07.430 Protection of Industrial Areas

3.07.440 Protection of Employment Areas

- Table 3.07-4
- TITLE 5: NEIGHBOR CITIES AND RURAL RESERVES
- 3.07.510 Intent

3.07.520 Rural Reserves and Green Corridors

- 3.07.530 Invitations for Intergovernmental Agreements
- 3.07.540 Metro Intent with Regard to Green Corridors

TITLE 6: CENTRAL CITY, REGIONAL CENTERS, TOWN CENTERS AND STATION COMMUNITIES

3.07.610 Purpose and Intent

3.07.620 Local Strategy to Improve Centers

(Effective 9/24/03) 3.07 - 1

- 3.07.630 Special Transportation Areas
- 3.07.640 Government Offices
- 3.07.650 Reporting on Center Progress

#### TITLE 7: AFFORDABLE HOUSING

3.07.710 Intent

- 3.07.720 Voluntary Affordable Housing Production Goals
- 3.07.730 Requirements for Comprehensive Plan and Implementing Ordinance Changes

3.07.740 Requirements for Progress Report

- 3.07.750 Metro Assessment of Progress
- Recommendations to Implement Other Affordable Housing 3.07.760 Strategies

Table 3.07-7

Five-Year Voluntary Affordable Housing Production Goals

#### TITLE 8: COMPLIANCE PROCEDURES

3.07.810 Compliance With the Functional Plan

- 3.07.820 Compliance Review by the Chief Operating Officer
- 3.07.830 Review of Compliance by Metropolitan Policy Advisory Committee
- 3.07.840 Review by Metro Council
- 3.07.850 Extension of Compliance Deadline

3.07.860 Exception from Compliance

3.07.870 Enforcement of Functional Plan 3.07.880 Compliance Report and Order 3.07.890 Citizen Involvement in Compliance Review

#### TITLE 9: PERFORMANCE MEASURES

3.07.910 Intent

3.07.920 Performance Measures Adoption

#### TITLE 10: FUNCTIONAL PLAN DEFINITIONS

3.07.1010 Definitions

#### TITLE 11: PLANNING FOR NEW URBAN AREAS

3.07.1105 Purpose and Intent

3.07.1110 Interim Protection of Areas Brought into the Urban Growth Boundary

3.07.1120 Urban Growth Boundary Amendment Urban Reserve Plan Requirements

3.07.1130 Implementation of Urban Growth Boundary Amendment Urban Reserve Plan Requirements

3.07.1140 Effective Date and Notification Requirements

#### TITLE 12: PROTECTION OF RESIDENTIAL NEIGHBORHOODS

3.07.1210 Purpose and Intent

3.07.1220 Residential Density

3.07.1230 Access to Commercial Services

3.07.1240 Access to Parks and Schools

(Effective 9/24/03)

3.07 - 2

NOTE: The Urban Growth Management Functional Plan was adopted by the Metro Council by Ordinance No. 96-647C, and amended by Ordinance No. 97-691C, prior to being codified as Metro Code Chapter 3.07 by Ordinance No. 97-715B.

(Effective 9/24/03)

3.07 - 3

#### 3.07.010 Purpose

The regional policies which are adopted by this Urban Growth Management Functional Plan recommend and require changes to city and county comprehensive plans and implementing ordinances. The purpose of this functional plan is to implement regional goals and objectives adopted by the Metro Council as the Regional Urban Growth Goals and Objectives (RUGGO), including the Metro 2040 Growth Concept and the Regional Framework Plan. The comprehensive plan changes and related actions, including implementing regulations, required by this functional plan as a component of the Regional Framework Plan, shall be complied with by cities and counties as required by Section 5(e)(2) of the Metro Charter.

Any city or county determination not to incorporate all required functional plan policies into comprehensive plans shall be subject to the conflict resolution and mediation processes included within the RUGGO, Goal I provisions, prior to the final adoption of inconsistent policies or actions.

(Ordinance No. 97-715B, Sec. 1.)

#### 3.07.020 Regional Policy Basis

The regional policies adopted in this Urban Growth Management Functional Plan are formulated from, and are consistent with, the RUGGOS, including the Metro 2040 Growth Concept. The overall principles of the Greenspaces Master Plan are also incorporated within this functional plan. In addition, the updated Regional Transportation Plan (RTP)<sup>1</sup>, when adopted, will serve as the primary transportation policy implementation of the 2040 Growth Concept. However, early implementation land use policies in this functional plan are integrated with early implementation transportation policies derived from preparation of the 1996 Regional Transportation Plan, and consistent with the Metro 2040 Growth Concept.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-972A, Sec. 1.)

#### 3.07.030 Structure of Requirements

The Urban Growth Management Functional Plan is a regional functional plan which contains "requirements" that are binding on cities and counties of the region as well as recommendations that are not binding. "Shall" or other directive words are used with requirements. The words "should" or "may" are used with recom-

<sup>1</sup> Metro has an adopted Regional Transportation Plan. However, because of changing local and regional conditions, as well as state and federal requirements, the RTP is scheduled to be amended in 1997.

(Effective 9/24/03) 3.07 - 4
mendations. In general, the plan is structured so that local jurisdictions may choose either performance standard requirements or prescriptive requirements. The intent of the requirements is to assure that cities and counties have a significant amount of flexibility as to how they meet requirements. Performance standards are included in most titles. If local jurisdictions demonstrate to Metro that they meet the performance standard, they have met that requirement of the title. Standard methods of compliance are also included in the plan to establish one very specific way that jurisdictions may meet a title requirement, but these standard methods are not the only way a city or county may show compliance. In addition, certain mandatory requirements that apply to all cities and counties are established by this functional plan.

(Ordinance No. 97-715B, Sec. 1.)

## (Effective 9/24/03)

## REGIONAL FUNCTIONAL PLAN REQUIREMENTS

## TITLE 1: REQUIREMENTS FOR HOUSING AND EMPLOYMENT ACCOMMODATION

## 3.07.110 Purpose and Intent

One goal of the Framework Plan is the efficient use of land. Title 1 intends to use land within the UGB efficiently by increasing its capacity to accommodate housing and employment. Title 1 directs each city and county in the region to consider actions to increase its capacity and to take action if necessary to accommodate its share of regional growth as specified in this title.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance 02-969B, Sec. 1.)

## 3.07.120 Housing and Employment Capacity

- A. Each city and county shall determine its capacity for housing and employment in order to ensure that it provides and continues to provide at least the capacity for the city or county specified in Table 3.01-7. Local governments shall use data provided by Metro unless the Metro Council or the Chief Operating Officer determines that data preferred by a city or county is more accurate.
- B. A city or county shall determine its capacity for dwelling units by cumulating the minimum number of dwelling units authorized in each zoning district in which dwelling units are authorized. A city or county may use a higher number of dwellings than the minimum density for a zoning district if development in the five years prior to the determination has actually occurred at the higher number.
- C. If a city annexes county territory, the city shall ensure that there is no net loss in regional housing or employment capacity, as shown on Table 3.07-1, as a result of amendments of comprehensive plan or land use regulations that apply to the annexed territory.
- D. After completion of its initial determination of capacity, each city or county shall report changes in its capacity by April 15 of the first calendar year following completion of its initial determination and by April 15 of every following year.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-972A, Sec. 1; Ordinance No. 02-969B, Sec. 1.)

(Effective 9/24/03) 3.07 - 6

#### 3.07.130 Design Type Boundaries Requirement

For each of the following 2040 Growth Concept design types, city and county comprehensive plans shall be amended to include the boundaries of each area, determined by the city or county consistent with the general locations shown on the 2040 Growth Concept Map:

<u>Central City</u>--Downtown Portland is the Central City which serves as the major regional center, an employment and cultural center for the metropolitan area.

<u>Regional Centers</u>--Seven regional centers will become the focus of compact development, redevelopment and high-quality transit service and multimodal street networks.

Station Communities--Nodes of development centered approximately one-half mile around a light rail or high capacity transit station that feature a high-quality pedestrian environment.

Town Centers--Local retail and services will be provided in town centers with compact development and transit service.

<u>Main Streets</u>--Neighborhoods will be served by main streets with retail and service developments served by transit.

<u>Corridors</u>-Along good quality transit lines, corridors feature a high-quality pedestrian environment, convenient access to transit, and somewhat higher than current densities.

Employment Areas--Various types of employment and some residential development are encouraged in employment areas with limited commercial uses.

<u>Industrial Areas</u>--Industrial area are set aside primarily for industrial activities with limited supporting uses.

<u>Regionally Significant Industrial Areas</u>--Industrial areas with site characteristics that are relatively rare in the region that render them especially suitable for industrial use.

<u>Inner Neighborhoods</u>--Residential areas accessible to jobs and neighborhood businesses with smaller lot sizes are inner neighborhoods.

<u>Outer Neighborhoods</u>--Residential neighborhoods farther away from large employment centers with larger lot sizes and lower densities are outer neighborhoods.

(Effective 9/24/03)

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 1.)

#### 3.07.140 Measures to Increase Development Capacity

- A. Each city and county shall adopt a minimum dwelling unit density, as prescribed in this subsection, for each zoning district in which dwelling units are authorized inside the UGB:
  - 1. Any city or county minimum density standard deemed to comply with the Urban Growth Management Functional Plan pursuant to Section 3.07.810 prior to January 1, 2003, shall be deemed to comply with this subsection.
  - 2. A city or county shall not approve a subdivision or development application that will result in a density below the minimum density for the zoning district.
  - 3. A city or county may change the dwelling unit density of any zoning district so long as the zoning district continues to comply with this subsection and so long as the city or county continues to provide at least the overall capacity for housing for the city or county specified in Table 3.07-1.
- B. A city or county shall not prohibit the partition or subdivision of a lot or parcel that is at least twice the size of the minimum size for new lots or parcels in any zoning district in which dwelling units are authorized.
- C. A city or county shall authorize the establishment of at least one accessory dwelling unit for each detached singlefamily dwelling unit in a zoning district and for each detached or attached single-family dwelling unit in a Regional Center or Station Community. The authorization may be subject to reasonable regulation for siting and design purposes.
- D. In order to assist Metro to evaluate the effectiveness of Title 1 in aid of accomplishment of the 2040 Growth Concept, and to comply with state progress reporting requirements in ORS 197.301, by April 15 of each even-numbered year beginning 2004, each city and county shall report to Metro the actual density of new residential development per net developed acre authorized in those zoning districts that allow residential development in the preceding 24 months.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 1.)

(Effective 9/24/03)

## 3.07.150 Transfer of Capacity

- A. A city or county may amend its comprehensive plan and land use regulations to transfer capacity for housing or employment shown on Table 3.07-1 to another city or county inside the UGB upon a demonstration that:
  - The transfer complies with the policies of the Regional Framework Plan;
  - The transfer will not reduce the capacity of the region for housing or employment specified on Table 3.07-1;
  - 3. The housing or employment capacity to be transferred is reasonably likely to occur at the receiving site within the 20-year planning period of Metro's last UGB capacity review under ORS 197.299; and
  - 4. The transfer does not move capacity from a designated Center to an Inner or Outer Neighborhood, or from a Regional Center to a Town Center.
- B. A city or county may seek a transfer of capacity as authorized in subsection A by filing an application on a form provided for that purpose by Metro. After receipt of a complete application, Metro shall set the matter for a public hearing before the Metro Council and shall notify MPAC and those persons who request notification of requests for transfers of capacity.
- C. The Metro Council shall hold a public hearing to consider the request for a transfer of capacity. Any person may participate in the hearing. The Metro Council may set terms and conditions upon approval of a transfer so long as they relate to the criteria in subsection A and are incorporated into the Metro Council's order.
- D. The Metro Council shall issue an order with its conclusions and analysis and send a copy to the local governments involved in the transfer and any person who participated in the hearing before the Metro Council. Any person who participated in the hearing may seek review of the Metro Council's order as a land use decision under ORS 197.015(10)(a)(A).

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 01-925E, Sec. 4; Ordinance No. 02-972A, Sec. 1; Ordinance No. 02-969B, Sec. 1.)

(Effective 9/24/03)

3.07.160 Local Plan Accommodation of Expected Growth Capacity for Housing and Employment-Performance Standard

All cities and counties within Metro shall demonstrate that:

- A. The provisions required in Section 3.07.140 of this title have been included in comprehensive plans and implementing ordinances; and
- B. Using the computation method in Section 3.07.120, calculated capacities will achieve the target capacities for dwelling units and full-time and part-time jobs contained in Table 3.07-1; and
- C. Effective measures have been taken to reasonably assure that the calculated capacities will be built for dwelling units and jobs; and
- D. Expected development has been permitted at locations and densities likely to be achieved during the 20-year planning period by the private market or assisted housing programs, once all new regulations are in effect.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 1.)

#### 3.07.170 Design Type Density Recommendations

A. For the area of each of the 2040 Growth Concept design types, the following average densities for housing and employment are recommended to cities and counties:

Central City - 250 persons per acre Regional Centers - 60 persons per acre Station Communities - 45 persons per acre Town Centers - 40 persons per acre Main Streets - 39 persons per acre Corridor - 25 persons per acre Employment Areas - 20 persons per acre Industrial Areas - 9 employees per acre Regionally Significant Industrial Area - 9 employees per acre Inner Neighborhoods - 14 persons per acre Outer Neighborhoods - 13 persons per acre

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 1.)

(Effective 9/24/03)

Table 3.07-1				
Zoned Capacity for Housing and Employment Units – Year 1994 to 2017				
Section 3.07.120(A)(1)(b)				
City or County	Dwelling Unit Capacity	Job Capacity		
Beaverton	13, 635	21,368		
Cornelius	1,285	3,054		
Durham	243	522		
Fairview	2,929	7,063		
Forest Grove	3,054	5,943		
Gladstone	880	1,569		
Gresham <sup>3</sup>	20,020	27,679		
Happy Valley <sup>4</sup>	5,705	1,418		
Hillsboro <sup>5</sup>	16,106	59,566		
Johnson City	38	82		
King City <sup>6</sup>	461	470		
Lake Oswego	4,049	13,268		
Maywood Park	12	5		
Milwaukie	3,188	3,650		
Oregon City	9,750	8,298		
Portland <sup>3</sup>	72,136	209,215		
Rivergrove	20	0		
Sherwood	5,216	9,518		
Tigard	6,308	17,801		
Troutdale	3,260	7,222		
Tualatin <sup>7</sup>	4,054	12,301		
West Linn	3,732	1,935		
Wilsonville <sup>2</sup>	4,425	15,030		
Wood Village	458	1,074		
Clackamas County <sup>1,3</sup>	13,340	31,901		
Multnomah County <sup>8</sup>	0	0		
Washington County <sup>1</sup>	51,649	55,921		
Regional Total	246,053	516,873		

<sup>1</sup>Standards apply to the urban unincorporated portion of the county only.

<sup>2</sup> Wilsonville has not completed its capacity analysis (as of October 2002), 1996 Title 1 data used.
 <sup>3</sup>Includes capacity for Pleasant Valley Concept Plan, former Urban Reserve Nos. 4 and 5.
 <sup>4</sup>Includes capacity for former Urban Reserve Nos. 14 and 15.

<sup>5</sup>Includes capacity for former Urban Reserve No. 55.

<sup>6</sup>Includes capacity for former Urban Reserve No. 47.

<sup>7</sup>Includes capacity for former Urban Reserve No. 43. <sup>8</sup>Capacity for unincorporated Multnomah County is included in the capacities of the Cities of Gresham, Portland and Troutdale.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 1.)

(Effective 9/24/03)

## TITLE 2: REGIONAL PARKING POLICY

#### 3.07.210 Intent

The State's Transportation Planning Rule calls for reductions in vehicle miles traveled per capita and restrictions on construction of new parking spaces as a means of responding to transportation and land use impacts of growth. The Metro 2040 Growth Concept calls for more compact development as a means to encourage more efficient use of land, promote non-auto trips and protect air quality. In addition, the federally mandated air quality plan adopted by the state relies on the 2040 Growth Concept fully achieving its transportation objectives. Notably, the air quality plan relies upon reducing vehicle trips per capita and related parking spaces through minimum and maximum parking ratios. This title addresses these state and federal requirements and preserves the quality of life of the region.

A compact urban form requires that each use of land is carefully considered and that more efficient forms are favored over less efficient ones. Parking, especially that provided in new developments, can result in a less efficient land usage and lower floor to area ratios. Parking also has implications for transportation. In areas where transit is provided or other non-auto modes (walking, biking) are convenient, less parking can be provided and still allow accessibility and mobility for all modes, including autos. Reductions in auto trips when substituted by non-auto modes can reduce congestion and increase air quality.

(Ordinance No. 97-715B, Sec. 1.)

## 3.07.220 Performance Standard

- A. Cities and counties are hereby required to amend their comprehensive plans and implementing regulations, if necessary, to meet or exceed the following minimum standards:
  - 1. Cities and counties shall require no more parking than the minimum as shown on Table 3.07-2, Regional Parking Ratios, attached hereto; and
  - 2. Cities and counties shall establish parking maximums at ratios no greater than those listed in the Regional Parking Ratios Table and as illustrated in the Parking Maximum Map. The designation of A and B zones on the Parking Maximum Map should be reviewed after the completion of the Regional Transportation Plan and every three years thereafter. If 20-minute peak hour transit service has become available to an area within a

(Effective 9/24/03) 3.07 - 12

one-quarter mile walking distance for bus transit or one-half mile walking distance for light rail transit, that area shall be added to Zone A. If 20-minute peak hour transit service is no longer available to an area within a one-quarter mile walking distance for bus transit or one-half mile walking distance for light rail transit, that area shall be removed from Zone A. Cities and counties should designate Zone A parking ratios in areas with good pedestrian access to commercial or employment areas (within 1/3 mile walk) from adjacent residential areas.

3. Cities and counties shall establish an administrative or public hearing process for considering ratios for individual or joint developments to allow a variance for parking when a development application is received which may result in approval of construction of parking spaces either in excess of the maximum parking ratios; or less than the minimum parking ratios.

Cities and counties may grant a variance from any maximum parking ratios through a variance process.

- в. Free surface parking spaces shall be subject to the regional parking maximums provided for Zone A and Zone B. Parking spaces in parking structures, fleet parking, parking for vehicles that are for sale, lease, or rent, employee car pool parking spaces, dedicated valet parking spaces, spaces that are user paid, market rate parking or other high-efficiency parking management alternatives may be exempted from maximum parking standards by cities and counties. Sites that are proposed for redevelopment may be allowed to phase in reductions as a local option. Where mixed land uses are proposed, cities and counties shall provide for blended parking rates. It is recommended that cities and counties count adjacent on-street parking spaces, nearby public parking and shared parking toward required parking minimum standards.
- C. Cities and counties may use categories or measurement standards other than those in the Regional Parking Ratios Table, but must provide findings that the effect of the local regulations will be substantially the same as the application of the Regional Parking Ratios.
- D. Cities and counties shall monitor and provide the following data to Metro on an annual basis:
  - The number and location of newly developed parking spaces; and

(Effective 9/24/03)

2. Demonstration of compliance with the minimum and maximum parking standards, including the application of any variances to the regional standards in this title. Coordination with Metro collection of other building data should be encouraged.

(Ordinance No. 97-715B, Sec. 1.)

(Effective 9/24/03)

Table 3.07-2 - Regional Parking Ratios				
	(Section 3.07.220(A)(1))			
(parking ratios are based on spaces per 1,000 sq. ft of gross leasable area unless otherwise stated)				
Land Use	Minimum Parking	Maximum	Maximum	
	Requirements	Permitted Parking	Permitted Parking	
	(See Central City	- Zone A:	Ratios	
	Transportation		- Zone B:	
	Management Plan for			
	downtown Portland stds)			
	Requirements May Not	Transit and	Rest of Region	
	Exceed	Pedestrian		
		Accessible		
		Areas		
General Office (includes Office Park, "Flex-	2.7	3.4	4.1	
Space", Government Office & misc.				
Services) (gsf)				
Light Industrial	1.6	None	None	
Industrial Park				
Manufacturing (gsf)				
Warehouse (gross square feet; parking	0.3	0.4	0.5	
ratios apply to warehouses 150,000 gsf or				
greater)				
Schools: College/	0.2	0.3	0.3	
University & High School				
(spaces/# of students and staff)				
Tennis Racquetball Court	1.0	1.3	1.5	
Sports Club/Recreation Facilities	4.3	5.4	6.5	
Retail/Commercial, including shopping	4.1	5.1	6.2	
centers				
Bank with Drive-In	4.3	5.4	6.5	
Movie Theater	0.3	0.4	0.5	
(spaces/number of seats)				
Fast Food with Drive Thru	9.9	12.4	14.9	
Other Restaurants	15.3	19.1	23	
Place of Worship	0.5	0.6	0.8	
(spaces/seats)				
Medical/Dental Clinic	3.9	4.9	5.9	
Residential Uses				
Hotel/Motel	1	none	none	
Single Family Detached	1	none	none	
Residential unit, less than 500 square feet	1	none	none	
per unit, one bedroom				
Multi-family, townhouse, one bedroom	1.25	none	none	
Multi-family, townhouse, two bedroom	1.5	none	none	
Multi-family, townhouse, three bedroom	1.75	none	none	

<sup>1</sup> Ratios for uses not included in this table would be determined by cities and counties. In the event that a local government proposes a different measure, for example, spaces per seating area for a restaurant instead of gross leasable area, Metro may grant approval upon a demonstration by the local government that the parking space requirement is substantially similar to the regional standard.

(Ordinance No. 97-715B, Sec. 1.)

(Effective 9/24/03)

## TITLE 4: INDUSTRIAL AND OTHER EMPLOYMENT AREAS

#### 3.07.410 Purpose and Intent

The Regional Framework Plan calls for a strong economic climate. To improve the region's economic climate, the plan seeks to protect the supply of sites for employment by limiting incompatible uses within Industrial and Employment Areas. To protect the capacity and efficiency of the region's transportation system for movement of goods and services and to promote the creation of jobs in centers, the plan encourages efficient patterns and mixes of uses within designated Centers and discourages certain kinds of commercial retail development outside Centers. It is the purpose of Title 4 to achieve these policies. Metro will consider amendments to this title in order to make the title consistent with new policies on economic development adopted as part of periodic review.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance 02-969B, Sec. 5.)

## 3.07.420 Protection of Regionally Significant Industrial Areas

- A. Regionally Significant Industrial Areas are those areas that offer the best opportunities for family-wage industrial jobs. Each city and county with land use planning authority over areas shown on the Generalized Map of Regionally Significant Industrial Areas adopted in Ordinance No. 02-969 shall derive specific plan designation and zoning district boundaries of the areas from the Map, taking into account the location of existing uses that would not conform to the limitations on non-industrial uses in subsections C, D and E of this section and the need of individual cities and counties to achieve a mix of types of employment uses.
- B. Each city and county with land use planning authority over an area designated by Metro on the 2040 Growth Concept Map, as amended by Ordinance No. 02-969, as a Regional Significant Industrial Area shall, as part of compliance with Section 3.07.1120 of the Urban Growth Management Functional Plan, derive plan designation and zoning district boundaries of the areas from the Growth Concept Map.
- C. After determining boundaries of Regionally Significant Industrial Areas pursuant to subsections A and B, the city or county shall adopt implementing ordinances that limit development in the areas to industrial uses, uses accessory to industrial uses, offices for industrial research and development and large corporate headquarters in compliance with subsection E of this section, utilities, and those

(Effective 9/24/03)

non-industrial uses necessary to serve the needs of businesses and employees of the areas. Ordinances shall not allow financial, insurance, real estate or other professional office uses unless they are accessory to an industrial or other permitted use.

- D. Notwithstanding subsection C, a city or county shall not approve:
  - A commercial retail use with more than 20,000 square feet of retail sales area in a single building or in multiple buildings that are part of the same development project; or
  - Commercial retail uses that would occupy more than five percent of the net developable portion of all contiguous Regionally Significant Industrial Areas.
- E. As provided in subsection C of this section, a city or county may approve an office for industrial research and development or a large corporate headquarters if:
  - 1. The office is served by public or private transit; and
  - 2. If the office is for a corporate headquarters, it will accommodate for the initial occupant at least 1,000 employees.
- F. A city or county may allow division of lots or parcels into smaller lots or parcels as follows:
  - Lots or parcels less than 50 acres may be divided into any number of smaller lots or parcels;
  - Lots or parcels 50 acres or larger may be divided into smaller lots and parcels so long as the resulting division yields the maximum number of lots or parcels of at least 50 acres;
  - 3. Notwithstanding paragraphs 2, 3 and of this subsection, any lot or parcel may be divided into smaller lots or parcels or made subject to rights-of-way for the following purposes:
    - a. To provide public facilities and services;
    - b. To separate a portion of a lot or parcel in order to protect a natural resource, to provide a public amenity, or to implement a remediation plan for a

(Effective 9/24/03)

site identified by the Oregon Department of Environmental Quality pursuant to ORS 465.225;

- c. To separate a portion of a lot or parcel containing a nonconforming use from the remainder of the lot or parcel in order to render the remainder more practical for a permitted use;
- d. To reconfigure the pattern of lots and parcels pursuant to subsection G of this section; or
- e. To allow the creation of a lot for financing purposes when the created lot is part of a master planned development.
- G. A city or county may allow reconfiguration of lots or parcels less than 50 acres in area if the reconfiguration would be more conducive to a permitted use and would result in no net increase in the total number of lots and parcels. Lots or parcels 50 acres or greater in area may also be reconfigured so long as the resulting area of any such lot or parcel would not be less than 50 acres.
- H. Notwithstanding subsections C and D of this section, a city or county may allow the lawful use of any building, structure or land at the time of enactment of an ordinance adopted pursuant to this section to continue and to expand to add up to 20 percent more floor area and 10 percent more land area. Notwithstanding subsection F of this section, a city or county may allow division of lots or parcels pursuant to a master plan approved by the city or county prior to December 31, 2003.
- By December 31, 2003, Metro shall, following consultation Ι. with cities and counties, adopt a map of Regionally Significant Industrial Areas with specific boundaries derived from the Generalized Map of Regionally Significant Industrial Areas adopted in Ordinance No. 02-969, taking into account the location of existing uses that would not conform to the limitations of non-industrial uses in subsections C, D and E of this section and the need of individual cities and counties to achieve a mix of types of employment uses. Each city and county with land use planning authority over the area shall use the map in the application of the provisions of this section until the city or county adopts plan designations and zoning district boundaries of the area as provided by subsection A of this section.

(Effective 9/24/03)

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 5.)

## 3.07.430 Protection of Industrial Areas

- A. In Industrial Areas mapped pursuant to Metro Code section 3.07.130 that are not Regionally Significant Industrial Areas, cities and counties shall limit new and expanded retail commercial uses to those appropriate in type and size to serve the needs of businesses, employees and residents of the Industrial Areas.
- B. In an Industrial Area, a city or county shall not approve:
  - A commercial retail use with more than 20,000 square feet of retail sales area in a single building or in multiple buildings that are part of the same development project; or
  - 2. Commercial retail uses that would occupy more than ten percent of the net developable portion of the area or any adjacent Industrial Area.
- C. Notwithstanding subsection B of this section, a city or county may allow the lawful use of any building, structure or land at the time of enactment of an ordinance adopted pursuant to this section to continue and to expand to add up to 20 percent more floorspace and 10 percent more land area.

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 5.)

## 3.07.440 Protection of Employment Areas

- A. Except as provided in subsections C, D and E, in Employment Areas mapped pursuant to Metro Code Section 3.07.130, cities and counties shall limit new and expanded commercial retail uses to those appropriate in type and size to serve the needs of businesses, employees and residents of the Employment Areas.
- B. Except as provided in subsections C, D and E, a city or county shall not approve a commercial retail use in an Employment Area with more than 60,000 square feet of gross leasable area in a single building, or commercial retail uses with a total of more than 60,000 square feet of retail sales area on a single lot or parcel, or on contiguous lots or parcels, including those separated only by transportation right-of-way.

(Effective 9/24/03) 3.0

- C. A city or county whose zoning ordinance applies to an Employment Area and is listed on Table 3.07-4 may continue to authorize commercial retail uses with more than 60,000 square feet of gross leasable area in that zone if the ordinance authorized those uses on January 1, 2003.
- D. A city or county whose zoning ordinance applies to an Employment Area and is not listed on Table 3.07-4 may continue to authorize commercial retail uses with more than 60,000 square feet of gross leasable area in that zone if:
  - 1. The ordinance authorized those uses on January 1, 2003;
  - 2. Transportation facilities adequate to serve the commercial retail uses will be in place at the time the uses begin operation; and
  - 3. The comprehensive plan provides for transportation facilities adequate to serve other uses planned for the Employment Area over the planning period.
- E. A city or county may authorize new commercial retail uses with more than 60,000 square feet of gross leasable area in Employment Areas if the uses:
  - 1. Generate no more than a 25 percent increase in sitegenerated vehicle trips above permitted non-industrial uses; and
  - Meet the Maximum Permitted Parking Zone A requirements set forth in Table 3.07-2 of Title 2 of the Urban Growth Management Functional Plan.

**Table 3.07-4** (Section 3.07.420(B))

Clackamas County unincorporated Commercial Commercial Industrial

Lake Oswego

General Commercial Highway Commercial

Troutdale General Commercial

Hillsboro General Commercial

Sherwood

General Commercial

Tigard

General Commercial Commercial Professional

Tualatin Commercial General

•

Wilsonville Planned Development Commercial

(Ordinance No. 97-715B, Sec. 1. Amended by Ordinance No. 02-969B, Sec. 5.)

## (Effective 9/24/03)

**Appendix 9** 



# METRO 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

# Interagency Coordination - October 2, 2003 Meeting Summary

# **Meeting Summary**

Interagency Consultation Meeting Air Quality Conformity & the 2004 RTP/2004-2007 MTIP October 2, 2003

Subcommittee Participation. The meeting commenced at approximately 10:08am and began with completing teleconferencing connections with Wayne Elson, US Environmental Protection Agency, and Rebecca Reyes-Alicea and Jennifer Bowman, Federal Transit Administration. Those in attendance in room 370 A at Metro included: Fred Patron and Michelle Eraut, Federal Highways Administration; Dave Nordberg and Marianne Fitzgerald, Oregon Department of Environmental Quality; Chris Smith, TPAC citizen member; Robin McArthur, Vince Carrow and Thomas Picco, Oregon Department of Transportation; Phil Sellinger, TriMet; and Andy Cotugno, Tom Kloster, Dick Walker, Kim Ellis, Ted Leybold, Jean Alleman, John Mermin and Mark Turpel, Metro. These individuals representing their respective agencies constituted the Interagency Consultation subcommittee (Subcommittee).

**Reference Documents**. Several documents were discussed and made available at the meeting including: *Interagency Consultation Draft Air Quality Conformity Determination*, (Determination) dated September 25, 2003, *Interagency Consultation Agenda* dated October 2, 2003, *Interagency Consultation Meeting Summary of Responses to Agenda Items* dated October 2, 2003 (Summary), and a one-page excerpt from page 2, Appendix 3 of the *Interagency Consultation Draft Air Quality Conformity Determination* (Excerpt).

Agenda. Discussion began with a query as to whether there were other items that should be discussed beyond the 13 items included in the Summary. The status of the Vancouver, Washington airshed and technical comments on the Determination were added.

Air Quality Model to be Used. Discussion of the Summary commenced with no disagreement on the responses in the Summary for item 1, MOBILE (air quality software) model to be used. However, Wayne Elson noted that MOBILE5b could also be used in addition to the MOBILE5a-h Metro has been using.

Subcommittee Conclusions:

- MOBILE5a-h is suitable for use in the air quality conformity determination for the 2004 RTP/2004-07 MTIP, MOBILE5b could also be used;
- Metro staff have begun testing MOBILE6 in order to transition to its use in the future.

Analysis Years. The second agenda item, analysis years, was discussed and the subcommittee agreed that changes were need for both the Determination and Excerpt documents. The Subcommittee discussed Table 2 of Appendix 3 in detail.

Subcommittee Conclusions:

- the year 2000 should be clarified that there is no CO or Ozone Budget established for this year;
- the type of budget (CO or Ozone) should be specified;
- no analysis would be completed for CO for the year 2006 as there is no emission budget for this pollutant for that year;
- an explanation about the difference between a full analysis and an analysis based on trip assignments was suggested;

These changes are reflected in a revised Table 2 below, and which also will be reflected in other tables and references in the Determination.

			Winter CO	Ozone
Year	Budget Established	Modeling	Emission Calculation	Emission Calculation
2006	Ozone		None - not required	Emission Interpolation*
2007	Winter CO		Emission Interpolation*	None - not required
2010	Both	Full Model run	MOBILE5a-h	MOBILE5a-h
2015	Both	Trip Assignment (Partial Model run)	MOBILE5a-h	MOBILE5a-h
2020	Both		Emission Interpolation	Emission Interpolation
2025	All years after 2020 to use 2020 budget	Full Model run	MOBILE5a-h	MOBILE5a-h

Table 2		
2004 Regional Transportation Plan Conformity	Analysis	Year

\* A full model run was performed for year 2000. Emissions for 2006 and 2007 were interpolated using the 2000 and 2010 model runs.

**Motor Vehicle Emission Budgets** Agenda item 3 addresses the State Implementation Plan (SIP) and MVEB (motor vehicle emission budget). There was no disagreement with the statements in the Summary. However, there was discussion of the subregional budget included in the Winter CO Maintenance Plan.

Subcommittee Conclusions:

- the Determination should clearly state the source of the emission budgets;
- sub- area CO budgets should not be ignored, rather, some response was needed;
- CO pollution levels in the Portland Central City area and 82nd Avenue areas (the sub-areas specifically included in the CO Maintenance Plan with their own emission budgets) have not been a problem. In fact, for the Central City area, actual Winter CO rates were only about ½ the allowed maximum and DEQ has removed the monitoring station because of the relatively low levels of actual CO.
- Marianne Fitzgerald, DEQ, agreed to investigate the SIP and Federal regulations to see whether separate sub-area budget analyses were absolutely required;
- Metro would likely prepare, unless the DEQ investigation showed no sub-area analysis was needed, an analysis of the sub-areas that addressed sub-area budgets, but the sub-area analysis may be less rigorous that the region-wide analysis based on subcommittee review.

Geographic Area Analysis The subcommittee discussed the statements in the Summary.

Subcommitee Conclusions:

- The statements about the geographic analysis area in the Summary are correct;
- Maps of these areas and sub-areas should be provided and included in the Determination.

**Transportation Control Measures** Agenda item 5, listed transportation control measures in EPA approved State Implementation Plans and their status was discussed.

Subcommittee Conclusions:

- This section of the Determination should be substantially improved by quoting each maintenance plan's TCMs and then documenting what has been done, noting those TCMs that may have been completed and future planned actions to implement those that have not yet been completed.
- Phil Sellinger, TriMet, noted that Table 1, page 11 of the Determination did not include street car service and that he would provide this data. He further noted that the Ozone Maintenance Plan included a TCM for transit service levels in the Portland Central City and that he would also provide this data.
- The first bullet on page 10 under the heading of "increased transit" should be revised to note that the annual service increase is on average and the last phrase beginning with question marks referring to a time period after the year 2020 should be deleted.
- It was noted that the first sentence below Table 1 on page 11 should be revised to clarify that the TCM is for transit service to increase by an average of 1.5 percent

per year, that TriMet had increased it by 2.6 percent per year, the result being actual transit service levels 1 percent more than the required TCM.

Latest Planning Assumptions This item, number 6 on the Agenda, was briefly discussed by the Subcommittee.

## Subcommittee Conclusions:

• The Subcommittee concurred with the responses in the Summary.

Motor Vehicle Fleet Information. The Subcommittee discussed this item.

## Subcommittee Conclusions:

• The Subcommittee concluded that specific fleet assumptions, especially the date of the data, must be included in Determination.

Public Comment Period. There was very substantial discussion of this item by the Subcommittee. FHWA and FTA representatives expressed concern about the schedule and the fact that while the October 31 Draft Determination would have descriptions of assumptions and methodology, it would not have the resulting air quality modeling output. Specifically the data that would show whether the region would meet emission budgets would not be available during most of the public comment period. Metro staff noted that the schedule was designed, in part, to be responsive to a letter from FHWA and FTA asking that the conformity information be provided 60 days or more before the lapse date, January 26, 2004. In addition, Metro recognized USDOT concerns expressed in the letter about the risks involved with a conformity lapse. Metro further stated that the schedule would only be implemented if the modeling, based on the stated assumptions and methodology available for public review and comment, met emission budgets. If the emission budgets were not met, then Metro would have to make revisions to the RTP and MTIP, rerun the analysis and revise the schedule accordingly. Discussion of preparation of an interim RTP, showing those projects that could proceed in the event of an air quality conformity lapse was suggested by FHWA representatives.

Subcommittee Conclusions:

- A draft interim RTP project list by analysis year assumptions should be prepared to illustrate the consequences of a conformity lapse and meet Federal reporting requirements. This task will be completed in a coordinated effort among Metro, ODOT and USDOT representatives. Ideally, this list should be circulated to the Subcommittee prior to its inclusion in the October 31 revised Determination.
- the 2004 RTP Update Calendar of Activities, dated September 26, 2003 which includes public outreach and comment period, will not be changed at this time.
- Should emission modeling show that the 2004 RTP and/or 2004-07 MTIP do not meet emission budgets, the Calendar will be revised after consideration of possible RTP/MTIP revision issues, modeling time, interagency consultation and other relevant factors. This explanation should be added to the Determination.

**Emission Reduction Credits.** The responses included in the Summary were discussed, with emphasis on describing which credits were applied after running the emissions model.

## Subcommittee Conclusion:

• The emission credits cited in the Summary (item 9) should be added to the Determination with an explanation of how they have been applied.

**Exempt Projects.** This item was discussed at the same time as items 11, list of projects by analysis year. (The criteria for projects which are eligible for exemption are located at 40CFR Part 93.126 which may be found at:

http://www.access.gpo.gov/nara/cfr/cfrhtml\_00/Title\_40/40cfr93\_00.html Generally, projects eligible for exemption include roadway safety projects; transit projects which involve service changes, but not new construction; air quality improvement programs like vanpooling, bicycle projects; and other activities that do not directly lead to construction.)

Subcommittee Conclusion:

• As noted under the Public Comment item, above, a draft interim RTP project list by analysis year, transit service levels and level of service assumptions should be prepared to illustrate the consequences of a conformity lapse and meet Federal reporting requirements.

Project list by Analysis Year. See conclusions under Exempt Projects, above.

**Transit System and Level of Service Assumptions**. The Subcommittee discussed this item, recognizing that these assumptions had not yet been completed.

Subcommittee Conclusion:

• When Metro has a draft of transit system and level of service assumptions, these should be circulated to the Subcommittee, ideally prior to publication of a revised Determination on October 31.

**Contingency Measures in Case of Violation**. The Subcommittee concluded that the conformity determination should discuss what happens in the event of a contingency lapse, not NAAQS violations.

Subcommittee Conclusion:

• The Subcommittee agreed that should the air quality analysis not demonstrate conformity, then Metro would make revisions to the RRTP and /or MTIP, or take other actions that would bring the region into conformity. An explanation of this approach should be included in the revised Determination.

Affect of Possible Metro Area Conformity Lapse on Clark County. This question was raised to clarify the impact on Clark County should a lapse occur in the Metro area.

Subcommittee Conclusion.

• The Subcommittee deferred to the EPA representative, who stated that a conformity lapse in the Metro area would not adversely impact Clark County Washington air quality conformity.

**Determination Document Comments**. FHWA representatives included several comments including:

- A reference to assessment of environmental justice on page 2, third paragraph of the Determination was questioned. Metro staff responded that included in the MTIP was an environmental justice assessment and that the statement in the Determination was accurate. Accordingly, no revision to the Determination on this point is planned.
- The Determination should be revised on page 13, to note that TPAC and JPACT do not include all relevant agencies (ie, FTA and EPA) that should be included in the development of the RTP and MTIP. Further, the revised Determination should reference the fact that the Subcommittee has met, reviewed the Determination and commented. The revised Determination should note that the Subcommittee meeting and coordination, along with TPAC and JPACT meetings, does result in a full review and coordination with all necessary and relevant agencies.
- The Determination should be revised in the last paragraph on page 15 under item x. and the response to item xi, to reflect changes. Specifically, the MOU cited under section x has been superceded by an amendment to the OAR. The OAR should be referenced, explained and the region's response should be described as a replacement to the existing paragraph. For the section under xi, the process that Metro is completing should replace the existing language.
- The RTP Work Plan, page 5, should be revised to clarify what changes are going to be made to the timeline and to reflect the need to update the planning boundary. Metro staff agreed to revising the work plan and completing the tasks.
- The MTIP is required to include estimates of the air quality benefits of each CMAQ project. While some of the projects are carried over from previous years and do not require new estimates, newer CMAQ projects do.

The Subcommittee, having no further comments or recommendations, adjourned at approximately 12:05.

**Appendix 10** 



# METRO 2004 Regional Transportation Plan and 2004-07 Metropolitan Transportation Improvement Program

# Portland Area Motor Vehicle Fleet Assumptions

On-road motor vehicle emissions of carbon monoxide and precursors of ozone and will be determined using EPA's Mobile5a\_h Emissions Factor Model. The inputs for these computer analyses will reflect the following parameters:

Fleet Data: Vehicle registration distribution and vehicle age distribution for Light Duty Gas Vehicles (LDGV) and Light Duty Diesel Vehicles (LDDV) will be derived from Oregon Dept. of Motor Vehicles registration records for Clackamas, Multnomah and Washington Counties 2002. Vehicle type and age distributions for other vehicle groups will be determined by national averages.

Vehicles originating in Clark County, Washington will be characterized the same way if possible. If 2002 registration data are not available, national averages will be used to describe that portion of the fleet.

I/M Program: Vehicles registered in the Portland Metropolitan area are subject to Oregon DEQ's Inspection/Maintenance (Emissions Testing) Program. Details of the I/M program reflected in the Mobile5a\_h model are:

OBD Test: 1996 and newer vehicles are subject to On Board Diagnostics testing.

Enhanced Test: 1981 through 1995 model year vehicles are subject to BAR 31 "enhanced" emissions testing (modeled as EPA's I/M 240 enhanced test).

Basic Test: 1975 through 1980 model year vehicles are subject to the 2500 two speed idle emissions test.

Exemption: Most vehicles are not subject to emissions testing until they become four years old.

Waiver Rate: There is no repair cost threshold at which a vehicle does no have to meet the emissions test requirement.

I/M Program Start Year: 1975

Program Type: Centralized

Compliance Rate: 90%

Inspection Frequency: Biennial

Tampering Rates: Mobile5 rates.

Speed: One average speed used for all vehicle types.

BERs: Mobile5 Basic Emission Rates.

Refueling Emissions: None calculated. (Accounted for under "Area Sources")

Summer Temperatures: Min: 61 deg. F; Max: 98 deg. F

Winter Temperature: Ambient = 39.8 deg. F

Summer Reid Vapor Pressure: 7.8 psi

Winter Reid Vapor Pressure: 13.6 psi

Winter Fuel Type: 2.7% Oxygen

# How to Comment on the update to the **2004 Regional Transportation Plan**

The public comment period for the 2004 Regional Transportation Plan (RTP) begins on October 31, 2003 and concludes with a public hearing on December 4, 2003. You may submit comments online at Metro's website:

# www.metro-region.org/rtp

Comments and questions may also be mailed using the form below, or left on Metro's Transportation hotline at (503) 797-1900, Option 2.

# **Comments:**

# Submitted by:

Name	
Street Address	City/Zip
Phone	E-Mail
Send me more info:	
2000 RTP Document CD	Other RTP Info:
Please add me to the RTP inte	erested citizens mailing/e-mail lists

# **Regional Transportation Plan Update Calendar**

- October 31 Public comment period begins; staff recommendation on draft 2004 RTP released for 30-day public comment period; draft RTP and conformity determination submitted to FHWA and FTA to begin review
- **November 3** Air quality conformity analysis begins
- November 5 MTAC comments on draft 2004 RTP
- November 12 MPAC comments on draft 2004 RTP
- November 13 JPACT tentative action on draft 2004 RTP
- November 13 Metro Council first reading of Ordinance on draft 2004 RTP
- **November 26** TPAC review and discussion of draft 2004 RTP and air quality conformity analysis
- **December 4** Public hearing on draft 2004 RTP; public comment period ends at 5 p.m.
- December 5 TPAC special meeting to comment on draft 2004 RTP
- December 10 Tentative final MPAC action on 2004 RTP
- December 11 Tentative final JPACT action on 2004 RTP
- December 11
   Metro Council second reading of Ordinance and consideration of adoption of 2004 Regional Transportation Plan

#### FOLD HERE



Place first class postage here.

Metro 600 NE Grand Avenue Portland, Oregon 97232 Attention: Marilyn Matteson

# STAFF REPORT

# CONSIDERATION OF ORDINANCE NO. 03-1024 FOR THE PURPOSE OF ADOPTING THE 2004 REGIONAL TRANSPORTATION PLAN AS THE REGIONAL TRANSPORTATION SYSTEM PLAN AND THE REGIONAL FUNCTIONAL PLAN FOR TRANPORTATION TO MEET STATE PLANNING REQUIREMENTS

Date: November 4, 2003

Presented by: Andrew C. Cotugno

# PROPOSED ACTION

This ordinance would adopt the 2004 Regional Transportation Plan (RTP) as the regional transportation system plan (TSP) and the regional functional plan for transportation, as required by ORS 268.390, and would bring the RTP into compliance with the state Transportation Planning Rule (TPR). The 2004 RTP includes:

- <u>RTP Policies</u> Chapter 1 of the RTP includes the policy component of plan. It has been updated to incorporate functional map amendments recommended in local transportation plans adopted since 2000 and endorsed by Metro as "friendly amendments" as part of the local review process. This action will also amend Ordinance No. 97-715B, updating Chapter 2 of the Regional Framework Plan with the updated Chapter 1 of the RTP.
- <u>RTP Projects and Systems Analysis</u> Chapters 2 through 5 of the RTP identify the 20-year transportation needs for the region, detail the scope and nature of proposed improvements that address the 20-year needs and a financial plan for implementing the recommended projects. The chapters have been updated to incorporate project amendments recommended in local transportation plans adopted since 2000 and endorsed by Metro as "friendly amendments" as part of the local review process and technical or factual updates to the plan text that reflect updated population, employment and other empirical data needed to establish a new planning horizon year of 2025. Chapter 3 includes a description of the preferred system, which is intended to satisfy the state TPR requirements for an "adequate" system, as well as procedures and criteria in Chapter 6 for amending the projects.
- <u>RTP Implementation</u> Chapter 6 of the RTP establishes regional compliance with state and federal planning requirements, and sets requirements for city and county compliance with the RTP. This chapter also establishes criteria for amending the RTP project lists, and the relationship between the RTP and the Metro Transportation Improvement Program (MTIP). Chapter 6 also identifies future studies needed to refine the RTP as part of future updates. These future studies are consistent with state TPR provisions that require refinement planning in areas where a transportation need exists, but further analysis is required to define specific solutions.

# EXISTING LAW

Metro is required to complete a periodic update of the Regional Transportation Plan (RTP) in order to maintain continued compliance with the federal Clean Air Act. The U.S. Department of Transportation (DOT) and the U.S. Environmental Protection Agency (EPA) approved and acknowledged the 2000 RTP air quality conformity determination on January 26, 2001. Under federal regulations, the RTP must be updated every three years to ensure that the plan adequately addresses future travel needs and is consistent with the federal Clean Air Act. As a result, a new plan demonstrating conformity with the Clean Air Act must approved and acknowledged by US DOT and US EPA in a formal conformity determination by January 26, 2004, when the current US DOT/US EPA conformity determination for the 2000 RTP expires. If the conformity determination expires, the plan is considered to "lapse," meaning that federally-funded transportation improvements could not be obligated during the lapse period. This consequence would apply to engineering, right-of-way acquisition or construction of any federally funded or permitted transportation project, except those defined as exempt because they do not have the possibility of increasing vehicle emissions.

Because the 2000 RTP was the result of a major update and was completed relatively recently, the 2004 update represents a minor effort that was limited to meeting state and federal requirements, and incorporating new policy direction set by Joint Policy Advisory Committee on Transportation (JPACT) and the Metro Council as part of various corridor and special studies adopted since 2000. The update also incorporated a number of "friendly amendments" proposed as part of local transportation plans adopted since 2000.

The next RTP update (which will be required by 2007) is proposed to be a more expansive effort that involves broader public discussion of plan policies and projects. By limiting this update to previously adopted local plans and corridor studies, projects that are included have been subject to past public involvement. This approach would establish a cycle of every other update being a "major" effort that reopens discussion of the RTP on a more fundamental level at six-year intervals.

# FACTUAL BACKGROUND AND ANALYSIS

# **Background on the RTP**

The 2000 RTP was the culmination of a major, five-year effort to completely overhaul the plan to reflect new federal and state regulations and the (then) newly adopted 2040 Growth Concept. It was the first RTP to be acknowledged by the Land Conservation and Development Commission as consistent with statewide planning goals.

The 2000 Regional Transportation Plan was developed to include separate layers of planned projects and programs that respond to differing federal, state and regional planning mandates. These layers are:

• the financially constrained system, which responds to federal planning requirements, and is based on a financial forecast of limited funding over the 20-year plan period

- the **priority system**, which responds to state planning requirements, and assumes that significant new revenue must be identified in order to provide an adequate transportation system over the 20-year plan period
- the **preferred system**, which responds to regional planning policies adopted as part of the 2040 Growth Concept and Regional Framework Plan, including specific system performance measures.

The federal "metropolitan transportation plan" is contained in applicable provisions of Chapter 1, 2, 3, 4 and 6 of the 2000 RTP. The policies and financial analysis in Chapters 3 and 4 for the preferred system of policies and facility improvements are for regional, not state, transportation planning requirements. The priority system described in Chapter 5 of this plan serves as the statement of adequacy for the purpose of compliance with the state TPR. The priority system includes a broad set of needed transportation projects and programs that generally keep pace with growth in the region, while implementing key elements of the 2040 Growth Concept.

The 2000 RTP was adopted in three stages: (1) an interim, federal element in 1995 that ensured continued certification under federal regulations, (2) a greatly expanded policy document approved in 1996 that established a new direction for the RTP that mirrored the 2040 Growth Concept and (3) a system component approved in 1999 that updated and expanded the planned projects called for in the region during the 20-year plan period. These components were assembled and jointly adopted by the Metro Council and JPACT in August 2000 as a complete plan addressing all federal, state and regional requirements.

The August 2000 adoption triggered a state requirement that local transportation plans be updated for consistency with the RTP within one year of the August 10, 2000 adoption date. As of today, all local plans have been updated for consistency, and have either been adopted or are in the final stages of adoption. To this extent, the elements of the RTP that are implemented through local plans, including design considerations for boulevards, local street connectivity requirements and a new "congestion management" process for developing transportation projects that requires thorough review of alternatives to road expansion before new road projects are identified.

The August 2000 action also included an update to the Title 2 Parking requirements, including the provision to design large parking lots with street-like features and layouts that encourage infill development and support walking and bicycling. These new parking requirements have also largely been incorporated into local plans.

# Major Tasks for the 2004 RTP Update

## Federal Regulations and Air Quality Conformity

The most pressing need for this update to the RTP is continued compliance with the federal Clean Air Act. The U.S. Department of Transportation last made a conformity determination on the 2000 RTP on January 26, 2001, and a new plan demonstrating conformity with the Clean Air

Act must be in place on January 26, 2004, when the 2000 RTP conformity determination expires. The conformity determination is made jointly by the Federal Highway Administration (FHWA) and Federal Transit Administration (FTA). Failing to adopt an updated RTP within the three year federal timeline means that federal-funded transportation improvements could not be obligated during the lapse period.

Most of the federal requirements only required minor revisions to the RTP in order to maintain compliance. The more involved efforts involve the requirement for a "financially constrained" plan and demonstration of conformity with the federal Clean Air Act. The conformity finding is based on the projects that make up the "financially constrained" plan. The financial constraint exercise consists of developing a projection of reasonably expected transportation funding over the 20-year plan period, and selecting a subset of projects from the plan that fit within this "constraint".

As the federally recognized system, the financially constrained system is also the source of transportation projects that may be funded through the Metropolitan Transportation Improvement Program. The MTIP allocates federal funds in the region, and is updated every two years, and includes a rolling, four-year program of transportation improvements.

Given that the larger set of "priority" RTP projects is nearly four times the project revenue in the existing 2000 RTP, was a difficult task to accomplish. The function of the "financially constrained" set of projects is further elevated by the fact that this list defines which projects in the plan are eligible for federal funding. The 2004 Regional Transportation Plan provides an updated set of financially constrained projects and programs for future MTIP allocations.

# **Previous Post-Acknowledgement Amendments**

In June 2002, the Metro Council and JPACT adopted a series of three "post-acknowledgement" amendments. These changes to the RTP reflected recently completed studies that had been anticipated in the original RTP adoption action, and were approved as a resolution that directed staff to bring the amendment to the next regular update to the RTP.

The "post-acknowledgement" amendments included changes resulting from the *Green Streets Study*, the *Elderly and Disabled Transit Study* and the *Corridor Priorities Project*, both completed in late 2001. These studies addressed specific, outstanding needs identified in the 2000 RTP. A third "post-acknowledgement" amendment was comprised of a number of minor text changes that were generated by the LCDC order that acknowledged the plan in June 2001.

Because the "post-acknowledgement" amendments were reviewed in detail as part of resolutions approved by JPACT and the Metro Council, they will simply be forwarded as part of the overall RTP update ordinance, with no further changes proposed.

# Local Transportation Plan "Friendly Amendments"

Under state rules, local governments in the Metro region were required to update local transportation plans for consistency with the 2000 RTP. Metro was involved in these local updates at a detailed level, with project staff assigned to each jurisdiction. As each local plan was completed, any proposed amendments to the RTP were called out and identified as "friendly amendments" in Metro's formal comments on the local plans. These "friendly amendments" represent refinements to RTP maps and project descriptions and have been incorporated into the 2004 RTP.

## Transportation Planning Rule and State Planning Goals

In 1991, the Land Conservation and Development Commission adopted the Oregon Transportation Planning Rule (TPR). The TPR implements State Land Use Planning Goal 12, Transportation, which was adopted by the Oregon Legislature in 1974. The TPR requires most cities and counties and the state's four MPOs to adopt transportation system plans that consider all modes of transportation, energy conservation and avoid principal reliance on any one mode to meet transportation needs. By state law, local plans in MPO areas must be consistent with the regional transportation system plan (TSP). Likewise, the regional TSP must be consistent with the Oregon Transportation Plan, adopted in 1992 by the Oregon Transportation Commission.

The state TPR requires that transportation system plans provide an adequate system of improvements that meet adopted performance measures. The 2004 RTP consolidates the preferred and priority systems from the 2000 RTP into a single "preferred" system that will serve as the regional TSP. This analysis of this system will then be used to make a determination of adequacy for the purpose of compliance with the state TPR.

However, projects identified in this new system cannot be funded through the MTIP process unless they are also included in the smaller financially constrained system. Instead, these projects and programs are intended to guide local transportation plans and land use actions, and serve as the source of future projects in the financially constrained system, either through amendments to the Regional Transportation Plan, or through the regular updates that occur every three to five years.

Two major highway corridors will continue to remain "outside the plan" until exception findings on rural and resource goals for the portions of the corridors located outside of the urban growth boundary are completed and approved by LCDC. These include the Sunrise Corridor Unit 2 and I-5 to 99W connector.

The Sunrise corridor work will begin shortly, as part of the parallel Sunrise Corridor Unit 1 DEIS and Damascus/Boring Concept Plan projects, but the recommendations from these studies will not be available before the RTP update is scheduled to conclude in early 2004. Likewise, a proposed corridor study for the I-5 to 99W connector was allocated funding through the MTIP process, and could be completed in the next few years, but would remain "outside" the RTP until then. Both corridors will continue to be portrayed on the RTP system maps, which set the longrange vision for the region's key transportation corridors, but those portions of the corridors located outside the urban growth boundary will not be included as projects in the plan until the respective corridor studies are complete and exceptions findings are approved by LCDC.

# Thresholds for Changes to the RTP

Given time and resource constraints, the Metro Council directed staff in May 2003 to complete a "housekeeping" update to the RTP, with the understanding that the next update (which will be required by 2007) will be a more expansive effort that involves broader public discussion of plan policies and projects. This approach eastablished a cycle of every other update being a "major" effort that reopens discussion of the RTP on a more fundamental level at six-year intervals. The 2004 RTP update was limited to regulatory and other mandated changes needed to keep the plan current, and following guidelines listed below:

- 1. Revisions required by federal statute or regulation.
- 2. Revisions required by state statute or administrative rule.
- 3. RTP amendments approved by Council Ordinance since August 2000, such as the South Corridor map and project amendments.
- 4. RTP amendments forwarded by Council Resolution to this scheduled update, such as the I-5 Trade Corridor and Green Streets amendments.
- 5. Amendments to the Regional Street Design map resulting from ODOT's effort to create a comprehensive map of Special Transportation Area (STA) designations.
- 6. Local functional map and project amendments recommended in local transportation plans adopted since August 2000, and endorsed by Metro as part of the local plan review process as "friendly amendments".
- 7. Technical or factual updates to the plan text that reflect updated population, employment and other empirical data needed to establish a new planning horizon year of 2025.
- 8. Limited transportation analysis updates based on the limited modeling proposed to meeting air quality conformity requirements.
- 9. Identification of new topics warranting further study as "outstanding issues" in Chapter 6 of the updated RTP.

As the final point suggests, these guidelines deferred major topics not already described in this staff report to be addressed as discrete RTP amendments, or deferred to a subsequent RTP update.

p. 6 of 7

# **Technical Considerations**

Because of the inherent time and resource constraints, a single round of modeling and analysis was utilized for this update. The principal purpose for this approach was to complete the federal air quality conformity analysis required to demonstrate that the updated plan is consistent with the region's air quality maintenance plan.

To achieve this, the 2004 RTP update combined the preferred and priority systems contained in the 2000 RTP as a single preferred system that established the universe of projects eligible for inclusion in the financially constrained system that is eligible for federal funding. Exceptions to this guideline were local and regional projects identified in corridor refinements and local transportation plans since the 2000 RTP was adopted. This approach focused TPAC's activities on defining the financially constrained system, and was based on the assumption that the combination of preferred system projects from the existing plan, and new projects from subsequent studies, will be adequate to meet travel demand in the new 2025 horizon year.

As part of documenting findings from this limited RTP modeling exercise, staff will review and update system performance conclusions from the 2000 RTP, as appropriate, to reflect the new systems. The 2004 RTP Update did not include an iterative process of multiple rounds of modeling to test new projects against the congestion management system and other RTP performance measures, since the new preferred system of improvements is expected to perform adequately. Any outstanding issues that are identified will be referenced for future corridor or area studies.

## **2004 RTP Update Products**

The results of the 2004 RTP update work tasks are included in the 2004 Regional Transportation Plan Public Comment document, which is included as Exhibit "A." A 30-day public comment period was held from October 31, 2003 through December 4, 2003.

## **BUDGET IMPACT**

None.

KE:acc

p. 7 of 7